Clemson University TigerPrints

All Dissertations

**Dissertations** 

December 2020

# Human Factors Considerations in a Telemedicine-Integrated Ambulance-Based Caregiving Environment for Stroke Care

Hunter Rogers Clemson University, hkroger@g.clemson.edu

Follow this and additional works at: https://tigerprints.clemson.edu/all\_dissertations

## **Recommended Citation**

Rogers, Hunter, "Human Factors Considerations in a Telemedicine-Integrated Ambulance-Based Caregiving Environment for Stroke Care" (2020). *All Dissertations*. 2713. https://tigerprints.clemson.edu/all\_dissertations/2713

This Dissertation is brought to you for free and open access by the Dissertations at TigerPrints. It has been accepted for inclusion in All Dissertations by an authorized administrator of TigerPrints. For more information, please contact kokeefe@clemson.edu.

# HUMAN FACTORS CONSIDERATIONS IN A TELEMEDICINE-INTEGRATED AMBULANCE-BASED CAREGIVING ENVIRONMENT FOR STROKE CARE

A Dissertation Presented to the Graduate School of Clemson University

In Partial Fulfillment of the Requirements for the Degree Doctor of Philosophy Industrial Engineering

> by Hunter Rogers December 2020

Accepted by: Dr. Kapil Chalil Madathil, Committee Chair Dr. Anand Gramopadhye, Committee Co-Chair Dr. David Neyens, Committee Member Dr. Anjali Joseph, Committee Member

#### ABSTRACT

Telemedicine, the use of communications technology to connect patients to medical professionals remotely, can be applied to a variety of settings, for example connecting older adults with their physicians from home, specialists to rural county hospitals or patients to physicians for emergency care. This dissertation focuses on the use of telemedicine for ambulance-based care for stroke patients, including how the design of this system impacts caregivers.

The initial study investigated both the usability of a telemedicine system implemented in ambulances for stroke care as well as the possibility of human error when using it. The heuristic evaluation of usability violations found several issues that needed to be addressed, including the lack of clarity in the tab structure and the lack of suggestions for correct data inputs. Similarly, the analysis of possible errors also determined several issues with this system, with the two most common being miscommunication and difficulty in locating data input or selecting an incorrect option. Several remediations strategies were recommended based on this study: improvement of the labelling of the tab structure, consistent formatting, rigid or suggested formatting for data input, automation of task structure and camera movement, and audio/visual improvements to support communication.

The second study investigated the experience of caregivers with the ambulancebased stroke telemedicine system, focusing on the support of the distributed cognition of the caregiving teams. Teams comprised of a neurologist, nurse, and paramedic were observed conducting 13 simulated stroke consults, after which each caregiver completed a survey on the perceived workload, usability, and teamwork during the session and an interview about their experience with the telemedicine system. In total, thirty-nine caregivers were interviewed, and the data collected were analyzed for themes. The themes that emerged identified such barriers to and facilitators for using telemedicine for ambulance-based stroke caregiving as training and experience, technical difficulty barriers, and patient care and efficiency improvement facilitators. The findings from this study resulted in design recommendations for supporting healthcare professionals during caregiving, especially ones that support their distributed cognition when using ambulance-based telemedicine for stroke care.

The final study evaluated the effect of design recommendations implemented in a new telemedicine system on the neurologist's workload, situation awareness, and task performance in addition to evaluating the perceived usability of this new design and its support of distributed cognition. For this study based on a within-subjects experimental design, 20 neurologists completed simulated stroke assessments using both the new design and the design investigated in the two previous studies and evaluated each system. Overall, the results found that the neurologists experienced a lower workload, performed better in their task, exhibited higher situation awareness, and rated usability highly in the new design. In addition, most participants thought that the new design better supported distributed cognition principles and preferred the new system for ambulance-based stroke consults.

# DEDICATION

This work is dedicated to my family, my mom Stephanie, my dad Steve, and my sisters Kaitlyn and Meaghan who never gave up on me, never let me give up on myself and saw the light at the end of the tunnel even before I did. Your love and support kept me through and I am forever grateful.

#### ACKNOWLEDGEMENTS

I would like to thank Dr. Kapil Chalil Madathil for offering me the opportunity to pursue a Ph.D. and for supporting me throughout the process; your knowledge and encouragement have made this possible. I would also like to thank the other members of my doctoral committee, Dr. Anand Gramopadhye, Dr. David Neyens, and Dr. Anjali Joseph, for their valuable feedback and support in this research effort.

I would also like to thank Dr. Jeffrey Bertrand for his help in developing the simulations I used in my final study and Barbara Ramirez for all her help editing my journal and conference papers during the course of my Ph.D.

In addition, I want to thank all the members of the Human Systems Integration Lab, both current and previous, Dr. Sruthy Agnisarman, Dr. Shraddhaa Narasimha, Dr. Amro Khasawneh, Amal Ponathil, Aasish Bhanu, Snowil Lopes, Emma Scharett, and Tara Morris. Thank you for your support and encouragement; I know the experiences we shared and my memories of them will be with me forever. I would also like to thank Jennifer Iacono, Shannon Devlin, Kylie Gomes, Katie Jurewicz, and the many other friends I made during my time at Clemson for your encouragement and friendship.

I also want to thank my support system, my family--my parents, my sisters, and my grandmother--for their constant efforts to keep me sane while I was working on my Ph.D. Without you I could not have gotten to where I am now. Finally, I would like to thank my Lord and Savior Jesus Christ, without Whom none of this would be possible.

# TABLE OF CONTENTS

TITLE PAGEi
ABSTRACTii
DEDICATIONiv
ACKNOWLEDGEMENTSv
LIST OF TABLESxi
LIST OF FIGURESxii
CHAPTER
I. INTRODUCTION1
Problem Statement2Theoretical Framework4SEIPS 2.04Distributed Cognition6Research Objectives7Research Questions7Dissertation Organization8
II. A SYSTEMATIC REVIEW OF THE IMPLEMENTATION CHALLENGES OF TELEMEDICINE SYSTEMS IN AMBULANCES
Introduction9Materials and Methods10Institutional Review and Human Subject Determination10Data Sources10Search Terms10Inclusion and Exclusion Criteria11Article Selection11Data Abstraction12Results12Application Overview14

# Table of Contents (Continued)

14
14
17
18
24
25
27
27
29
29

#### 

Introduction	
Emergency Medical Services	
Human Error Taxonomies	
Human Factors Approach to Human Error	
Research Purpose and Ouestions	
Methodology	
Search Strategy	
Data Selection	
Quality Assessment	41
Data Extraction	
Results	
Location	
Method	
Population and Condition	
Possible Causes	
Discussion	
Types of Error	
Occurrence of Error	
Possible Causes	
Systematic Review Limitations	
Directions and Future Work	
Conclusion	

IV.	ANALYSIS OF AN AMBULANCE-BASED TELEMEDICINE-	
	INTEGRATED SETTING FOR STROKE CARE	71
	Introduction	71
	Telemedicine in Stroke Care	71
	Background	75
	Methodology	79
	Participants	79
	Procedure	
	Analysis Techniques	
	Results	
	НТА	
	Heuristic Evaluation	
	SHERPA	
	Discussion	
	Remediations	
	Limitations	
	Conclusion	111
	TELEMEDICINE-INTEGRATED AMBULANCE-BASED SE FOR STROKE CARE	ETTING 112
	Introduction	112
	Telemedicine Use for Strokes	112
	Theoretical Background	114
	Research Questions	
	Method	
	Participants	
	Equipment	
	Procedure	
	Measures	
	Analysis	
	Results	
	Workload	
	Team Effectiveness	
	Usability	
	Interview Analysis	
	Discussion	
	Conclusion	149

Table of Contents (Continued)

VI.	PRIORITIZED INFORMATION DISPLAY FOR ENHANCING	
	DISTRIBUTED COGNITION IN A TELEMEDICINE-INTE	GRATED
	AMBULANCE-BASED SYSTEM FOR STROKE CARE	152
	Introduction	
	Distributed Cognition	
	Proposed Interface Design	
	Research Questions	166
	Methodology	167
	Participants	167
	Experimental Setup	169
	Experimental Design	169
	Procedure	171
	Measures	174
	Analysis	175
	Results	176
	Workload	176
	Usability	179
	Situation Awareness	181
	Performance	185
	Preference	187
	Distributed Cognition	188
	Benefits of Telemedicine	193
	Discussion	194
	Limitations	211
	Conclusion	
VII.	CONCLUSION	
DEFEDE	NGEG	222
REFERE	NCES	
APPEND	DICES	254
A:	Search Strategy for Chapter 3	
B:	Results of Search.	
<u>C</u> :	Adapted STROBE Checklist	
D:	Quality Assessment	
E:	Role Specific HTA Diagrams	
F:	I eam H I A Diagrams	
G:	I eam HIA I able	

H:	Heuristic Evaluation Form	
I:	Heuristic Violations	
J:	SHERPA Table	
K:	Sample Interview Guide: Nurse and Neurologist	
L:	Sample Interview Guide: Paramedic	
M:	Study Protocol	
N:	SAGAT Questions	
O:	Interview Protocol	

# LIST OF TABLES

Table	Page
2.1	Application of Selected Studies13
2.2	Technology and Network of Selected Studies16
2.3	Key Findings of Selected Studies
3.1	Two Term Search List
3.2	Procedure Errors
3.3	Protocol Deviations
3.4	Medication Errors
3.5	Diagnosis Accuracy
3.6	Adverse Safety Events
4.1	Nielsen's 10 Heuristics adapted from (Nielsen 1994b)
4.2	SHERPA Error Mode Taxonomy
4.3	Summary of Heuristic Violations
4.4	Sample of SHERPA Table
5.1	Description Codes for Themes and Quotations
6.1	DiCoT Principles and Barriers or Facilitators Used for Design Features 159
6.2	Participant Demographics168

# LIST OF FIGURES

Figure	Page
1.1	The SEIPS 2.0 Model
2.1	Selection Flowchart
3.1	HFACS model Adapted from (Shappell & Wiegmann, 2000)
3.2	Selection Flowchart
4.1	REACH interface (Participants' faces masked for confidentiality)73
4.2	Process Flow Diagram of the Sessions
4.3	HTA Excerpt (See Appendices A and B for Detailed Task Analyses)92
4.4	Heuristic Evaluation Severity Summary
4.5	History Tab in REACH (Participants' faces masked for confidentiality).106
5.1	Emergency Stroke Care Process Flow115
5.2	from top down left to right a) Neurologist Set-Up (Participant's face masked for confidentiality), b) Nurse REACH Cart, c) Telemedicine Setup Diagram, d) Ambulance Overhead Speaker, e) Ambulance Laptop, f) Ambulance Camera
5.3	NASA-TLX Average Participant Response by Caregiving Role123
5.4	Average Team Effectiveness by Caregiving Role124
5.5	Average Usability Ratings by Caregiving Role125
6.1	REACH interface (Participants' faces masked for confidentiality)153
6.2	Proposed Interface Design
6.3	REACH Patient History and Medication Input161
6.4	History, Medication, and Allergy Input for Proposed Design

# Figure

6.5	Study Procedure Task Flow	173
6.6	Average Perceived Workload	179
6.7	Average Usability	181
6.8	Average Accuracy of SAGAT Freeze 1	
6.9	Average Accuracy of SAGAT Freeze 2	
6.10	Average Errors by Type	187
7.1	New Iteration of Prioritized Information Display	218

## CHAPTER ONE

# INTRODUCTION

Emergency Medical Services (EMS) stabilize and care for a variety of people and sometimes complex conditions. The nature of the work creates a fast-paced, high-stress environment with shifts longer than 8 hours for most Emergency Medical Technicians (EMTs). These factors, in addition to the types of conditions seen and their work environment, can create a high incidence of errors in both equipment and organization, but more importantly in human action or inaction. Errors in these cases are likely more frequent than in a normal health care setting because of high decision density, high severity conditions, and a small work area (Hobgood, Xie, Weiner, & Hooker, 2004; Patterson et al., 2012). These cases are also, by nature of being in an ambulance, emergencies which mean errors could quickly turn serious or fatal. It is for these reasons that patient safety in the pre-hospital setting is so vital to understand through research, specifically how humans make errors in this setting. However, the use of ambulances as not just transport, but vital stabilization and treatment facilities improves patient outcomes and time to treatment (Band et al., 2011; Bernard et al., 2010; Gardtman, Waagstein, Karlsson, & Herlitz, 2000; Jurriën et al., 2010). Specifically in the transport of cardiac and stroke patients, ambulance care can improve triage, decrease time to treatment, and increase use of time-limited lifesaving treatments such as tissue plasminogen activators (t-PA) (Kunisawa et al., 2014; Kwan, Hand, & Sandercock, 2004; Mosley et al., 2007; Quain et al., 2008; Wendt et al., 2015).

#### **PROBLEM STATEMENT**

Telemedicine, the use of telecommunications technology to provide access to healthcare when geographical barriers make face-to-face consultation impractical (Board on Health Care Services & Institute of Medicine, 2012; Kobb, Hoffman, Lodge, & Kline, 2003; National Academy of Sciences, 2014), is being integrated into such wide-ranging aspects of patient care as home health monitoring (Koch, 2006), out-of-office hospital consultation (Agnisarman et al., 2017), and ambulances (LaMonte et al., 2004). This growing field began in the 1960s when telephone communications and microwave signals were first introduced for transporting data (Field, 1996), followed by the second generation in the 1990s with the creation of the Internet and the World Wide Web, both of which relaxed the constraints of time and geography (Institute of Medicine, 2012). The third and current generation focuses on the use of mobile networks to transfer data, including video and voice communication, across vast distances (Yperzeele et al., 2014). Feasibility, pilot, and clinical studies using this third generation have explored both the benefits and limitations of this growing field (Barrett et al., 2016; Bergrath et al., 2011; Charash et al., 2011; Cho, Kwon, & Jeong, 2015; Czaplik et al., 2014; Felzen et al., 2016; Hadeed & Hadeed, 2011; Kwak et al., 2009; LaMonte et al., 2004; Liman et al., 2012; Lippman et al., 2016; Mandellos et al., 2004; Sibert et al., 2008; Takeuchi et al., 2015; Terkelsen et al., 2002; Valenzuela Espinoza et al., 2016; Walter et al., 2012; Xiao et al., 2000; Yperzeele et al., 2014), finding that these systems require low interference and a high level of functionality and usability to facilitate seamless interaction.

Emergency care, specifically that given in ambulances, is one area where telemedicine has the potential to play a critical role. Past research (LaMonte et al., 2004; Mandellos et al., 2004; Papai et al., 2014; Terkelsen et al., 2002; Valenzuela Espinoza et al., 2016; Walter et al., 2012; Yperzeele et al., 2014), found that its use can decrease treatment time by completing diagnosis steps in pre-hospital transport with a high level of diagnosis accuracy. For example, several studies (Barrett et al., 2016; Liman et al., 2012; Rörtgen et al., 2013; Walter et al., 2012) have found that stroke evaluation completed by teleconsultation in conjunction with EMS personnel can improve the level of care given before admittance to the hospital; this is especially important because according to a recent review (Schwamm et al., 2009), 74% of stroke cases died outside of the hospital (Norris, 1998). A recombinant tissue plasminogen activator (rt-PA) treatment, the gold standard for stroke care, can dissolve a clot that causes an ischemic stroke intravenously. However, 99.1% of hospitals in the U.S. alone have a rt-PA treatment rate of 10% or less (Kleindorfer et al., 2009). In most instances this is caused by the time to treatment being longer than the 3 hour window from symptom onset (LaMonte et al., 2004; Valenzuela Espinoza et al., 2016). Reducing transport time using telemedicine holds a large potential for increasing stroke survival rate (Saler, Switzer, & Hess, 2011). Integrating telemedicine systems in ambulances may reduce fatalities from not only strokes, but also other cardiovascular diseases.

With an abundance of research on the benefits and costs of using telemedicine systems in ambulances, there is a lack of research found in the literature that provides an understanding of how these systems influence EMTs or paramedics workflow or the demands placed on them in this work system. Limited research is also conducted on the demands placed on the nurses and emergency physicians that support the EMTs and paramedics through the telemedicine system.

#### **THEORETICAL FRAMEWORK**

#### SEIPS 2.0

The Systems Engineering for Patient Safety or SEIPS 2.0 model will be used in this research to understand the ambulance work system and how the addition of a telemedicine system changes that work system. This model takes a system approach to looking at health care and how work systems ultimately shape outcomes (Holden et al., 2013). This model has sought to improve healthcare using human factors principles and a focus on human system integration. The model is graphically represented in Figure 1.1. Here the work system is depicted as containing 6 interacting components: Tools & Technology, Organization, Tasks, Internal Environment, External Environment and Person(s). These components create processes that can be specified as being Professional work, Collaborative Professional - Patient work, and Patient work. These processes are also referred to as workflow in healthcare. These processes create outcomes that can be classified as Patient, Professional or Organizational and can be desirable or undesirable and proximal, the direct result of a work process, or distal, a result of causal actions and observed over time. Through all stages of this model, feedback or adaptation is present, outcomes are observed, and changes are made to work processes or the work system.



Figure 1.1: The SEIPS 2.0 model adapted from (Holden et al., 2013)

This research will investigate how telemedicine in an internal environment of an ambulance will interact with organization, task, external environment, and the people (EMTs or paramedics, nurses, and emergency physicians) components of a work system. This will be observed through the workflow or work processes identified in this model. Assessing patient condition, communication of symptoms, and development of diagnosis and treatment plan is classified by this model as Collaborative Professional - Patient work, with the assumption that the patient is conscious or is accompanied by a family member or caretaker. Interviews will improve understanding of the interaction of the components in this work system and how the interaction creates the current workflow as well as

perceptions of how that workflow impacts outcomes. This research will also consider how changes to the work system affect professional outcomes through system testing and provide feedback for future development of these work systems.

#### **Distributed Cognition**

By its nature, the placement of people and components in a telemedicine consultation are distributed by physical distance. There is also a separation of the mental models of all providers in a telemedicine consultation that needs to be considered. Specialists, emergency physicians, or nurses have a knowledge of diagnoses and treatments that they must relay to the patient or pre-hospital provider and likewise the patient or prehospital provider must communicate vital signs and symptoms. Distributed Cognition considers the information processing in a system when people are physically or mentally separated (Hollan, Hutchins, & Kirsh, 2000). Previous studies have approached areas such as medical informatics (Hazlehurst, Gorman, & McMullen, 2008), patient records (Bång & Timpka, 2003), and clinical research forms for data collection (Nahm, Nguyen, Razzouk, Zhu, & Zhang, 2010), using distributed cognition as a framework. Furniss and Blandford developed a method of modeling distributed cognition in teamwork through understanding the information flow, physical layout, and artifacts of the system called Distributed Cognition for Teamwork or DiCoT (Furniss & Blandford, 2006). This method of modeling has been used to improve mobile healthcare (McKnight & Doherty, 2008) and infusion administration in the Intensive Care Unit (ICU) (Rajkomar & Blandford, 2012). Understanding the information flow and the physical layout of the stroke telemedicine consultation, specifically completing communication analysis, is paramount to supporting distributed cognition and ultimately effective teleconsultation. This research will use the principles outlined in the DiCoT model to understand the telemedicine system and to develop interventions to enhance distributed cognition and improve workflow and usability of a telemedicine system for ambulatory stroke care.

#### **RESEARCH OBJECTIVES**

The primary objective of this dissertation is to understand how telemedicine systems can be used to support general patient care and specifically stroke care in ambulances. More specifically this dissertation explores the following research topics:

- 1. Understanding how telemedicine systems have been implemented in ambulances worldwide.
  - a) To understand the improvements and challenges found in implementation
  - b) To understand what human factors considerations have been made in implementation
- 2. Understanding the errors made by EMTs or paramedics in pre-hospital or ambulatory care of patients and how technology could mitigate the causes of errors.
- 3. Evaluate a telemedicine integrated ambulance-based system for stroke care and how that system affects workflow, information flow, and patient and professional outcomes.
- 4. Develop interventions for a telemedicine integrated ambulance-based system for stroke care to enhance distributed cognition among geographically dispersed caregivers.

## **Research Questions**

The overall research questions this dissertation aims to answer are listed below:

- 1. How are telemedicine systems being used or evaluated currently and what are the barriers to their use?
- 2. What human error occurrences impact ambulance-based emergency care?
- 3. What are the issues concerning usability and human error in a current ambulance-based telemedicine for stroke caregiving?
- 4. What are the cognitive demands and workload of geographically distributed caregivers in a current ambulance-based telemedicine for stroke caregiving?
- 5. What design interventions can be used to improve the work system, workflow, outcomes, and distributed cognition when using a telemedicine-equipped ambulance?

## **DISSERTATION ORGANIZATION**

This dissertation is organized as follows: Chapter Two details a systematic review of the literature on the challenges and benefits of implementing telemedicine systems in ambulances. Chapter Three summarizes the types and rates of human error of paramedics and EMTs in ambulances found in the literature. Chapter Four evaluates the usability and human error possibilities of the current telemedicine system integrated in ambulances in a South Carolina county used for stroke care. Chapter Five evaluates demands placed on EMTs, nurses, and physicians when stroke care is provided through telemedicine-equipped ambulances. Chapter Six explores the generation and testing of interface design interventions to enhance distributed cognition among caregivers for ambulatory stroke care. Chapter Seven summarizes findings of this dissertation, discusses impact of the findings, and provides future research recommendations.

#### CHAPTER TWO

# A SYSTEMATIC REVIEW OF THE IMPLEMENTATION CHALLENGES OF TELEMEDICINE SYSTEMS IN AMBULANCES

### **INTRODUCTION**

The use of telemedicine represents one of the benefits to society from advancements in information technology. The ability to send packets of data and communicate lifesaving information from almost anywhere represents possibilities to vastly improve patient care and quality of life. Whether for use in the home (Agnisarman et al., 2017; Newlin, McCall, Ottmar, Welch, & Khairat, 2018) or in emergency transport (Curry & Harrop, 1998; Kyriacou et al., 2003), telemedicine is being tested for benefits and barriers. The costs of the technology are apparent, but other obstacles such as usability or time investment need to be assessed as well. Improvements to patient care and health outcomes need to be understood in tandem.

This systematic review of the literature focuses on answering the following research questions:

- 1. What are the benefits of implementing telemedicine in ambulances?
- 2. What are the challenges or barriers associated with the implementation of telemedicine systems in ambulances?
- 3. On a larger scale, what are the implications of using telemedicine in ambulances?

Furthermore, this review summarizes the results from the research reviewed and identifies the challenges, best practices, and opportunities for further research.

### MATERIALS AND METHODS

#### Institutional Review and Human Subject Determination

This study was exempted from approval by Clemson University's Institutional Review Board, as it did not involve active human subject research. No individual patient participated in this study.

# **Data Sources**

Web of Science and PubMed were searched for articles published between the years 2000 and 2016. Duplicates were identified using Endnote and searching manually. An additional search was made through citations of articles selected for review for relevant studies published within the same time frame, but none were found to be useful for review.

# Search Terms

The databases were searched using keywords "Telemedicine AND Ambulance" (retrieved 134 articles from PubMed; retrieved 140 articles from Web of Science), "Telemedicine AND emergency healthcare" (retrieved 8 articles from PubMed; retrieved 14 articles from Web of Science), "Emergency Telemedicine" (retrieved 57 articles from PubMed; retrieved 55 articles from Web of Science), and "Telemedicine AND Emergency Medical Services" (retrieved 396 articles from PubMed; retrieved 72 articles from Web of Science), and selected titles and abstracts that incorporated telemedicine systems into ambulances, testing the system's feasibility in any patient category.

# **Inclusion and Exclusion Criteria**

## Domain

Articles were included from any domain as long as they focused on the implementation of telemedicine systems in ambulances.

### Article Format

Extended abstracts and posters were not included in this review, and studies were excluded if they did not contain at least a pilot study with simulated or real patients.

# Language

Only articles published in English were included in this review. *Publication Year* 

Articles were only chosen if they were published between 2000 and 2016. This time-period was chosen not only to keep studies relevant but also to document the beginning and development of the third generation of telemedicine, which is defined by the revolution of communication over mobile networks.

# **Article Selection**

The research team reviewed titles and abstracts of all articles found in initial search (864) to determine which could be included for full-text screening. Abstracts were not considered for full text review if they did not include any experimental results. With this screening process, 102 articles were chosen for full text analysis. Articles were excluded if they did not test telemedicine system capabilities in some way to show feasibility or implementation benefits and challenges. As a result, 23 studies were selected for review;

any conflicts in selection decisions were resolved through discussion between readers. Figure 2.1 describes the article search, screening, and selection in more detail.



Figure 2.1: Selection flowchart

# **Data Abstraction**

Data were taken from each article and placed into a table to organize results and details about each article. Method, sample size, objectives, technology, application, key findings, and limitations were identified in each article.

# RESULTS

Analysis of these 23 studies suggested three major domains of research: General Care, Stroke Care, and ST- Elevated Myocardial Infarction (STEMI) Care, with each study including the application, the technology explored, the data collection methods used, and an analysis of the data collected and the results found.

	Application			
Author	General Care	Stroke	STEMI	
Tang et al., 2006				
Yperzeele et al., 2014				
Bergrath et al., 2013				
Cho et al., 2015				
Xiao et al., 2000	•			
Espinoza et al., 2016		•		
Mandellos et al., 2004	•			
Takeuchi et al., 2015			•	
Papai et al., 2014			•	
Terkelsen et al., 2002			•	
Sibert et al., 2008				
Bergrath et al., 2011	•			
Barrett et al., 2016		•		
Felzen et al., 2016	•			
La Monte et al., 2004		•		

Table 2.1: Application of selected studies

Rörtgen et al., 2013

Charash et al., 2011

Liman et al., 2012

# **Application Overview**

An analysis of the 23 articles found they represented three application categories. Most studies (61%) involved general emergency care, with the remaining 39% focusing on one type of emergency patient. More specifically, six studies focused on stroke pre-hospital care and three on STEMI care. Table 2.1 categorizes the studies based on application domain.

.

.

.

# **Origin Overview**

Thirty-five percent of studies were conducted in the United States, another 30% were conducted in Germany, and the remaining countries of origin have no more than two studies. Specifically, two studies were conducted in each of the following countries: Greece, South Korea, and Belgium, and one study was conducted in each of the following countries: Japan and Denmark.

# Technology

The technology used most frequently in the studies reviewed was for the transmission of vitals or testing results over a network, a use case seen in all but six of the studies (Bergrath et al., 2013, 2011; Charash et al., 2011; Cho et al., 2015; Czaplik et al., 2014; Felzen et al., 2016; Hadeed & Hadeed, 2011; Lippman et al., 2016; Papai et al., 2014;

Rörtgen et al., 2013; Sibert et al., 2008; Takeuchi et al., 2015; Tang, Johnson, Tindall, & Zhang, 2006; Terkelsen et al., 2002; Valenzuela Espinoza et al., 2016; Walter et al., 2012; Xiao et al., 2000; Yperzeele et al., 2014). Such vital signs as blood pressure, blood sugar, body temperature, and electrocardiogram (ECG) were transmitted over various network types, with six studies using a Global System for Mobile communications (GSM) (Bergrath et al., 2011; Czaplik et al., 2014; Felzen et al., 2016; Mandellos et al., 2004; Terkelsen et al., 2002; Xiao et al., 2000), nine using 4th Generation Long Term Evolution (4G LTE)/3rd Generation (3G) networks (Barrett et al., 2016; Charash et al., 2011; Czaplik et al., 2014; LaMonte et al., 2004; Lippman et al., 2016; Sibert et al., 2008; Takeuchi et al., 2015; Valenzuela Espinoza et al., 2016; Yperzeele et al., 2014), and two using a High-Speed Downlink Packet Access (HSDPA) technology (Cho et al., 2015; Liman et al., 2012). The remaining studies either used radio links (Bergrath et al., 2013; Papai et al., 2014), Wireless Broadband Internet (Hadeed & Hadeed, 2011; Kwak et al., 2009), or did not specify the network (Rörtgen et al., 2013; Tang et al., 2006; Walter et al., 2012). In addition to the transmission of vital data, all but 2 (9%) used some form of communication from the ambulances to either a hospital or an out-of-office location (Mandellos et al., 2004; Tang et al., 2006). Sixty-seven percent of the studies used a cell phone system with a video feed or still images for communication with the hospital unit (Bergrath et al., 2013, 2011; Charash et al., 2011; Czaplik et al., 2014; Felzen et al., 2016; Kwak et al., 2009; LaMonte et al., 2004; Papai et al., 2014; Rörtgen et al., 2013; Sibert et al., 2008; Takeuchi et al., 2015; Terkelsen et al., 2002; Walter et al., 2012; Xiao et al., 2000), while the remaining 33% used a bi-directional video and audio system (Barrett et al., 2016; Cho et al., 2015; Hadeed & Hadeed, 2011; Liman et al., 2012; Lippman et al., 2016; Valenzuela Espinoza et al., 2016; Yperzeele et al., 2014). Table 2.2 shows the technologies and networks used in each study.

	Transmission Content				Transmission Method			
Author	Vitals	Voice	Bi-directional Video	GSM	HSDPA	4G/3G	Radio	Not specified
Tang et al., 2006	•							•
Yperzeele et al., 2014	•		•			•		
Bergrath et al., 2013	•	•					•	
Cho et al., 2015	•		•		•			
Xiao et al., 2000	•	•		•				
Espinoza et al., 2016	•		•			•		
Mandellos et al., 2004				•				
Takeuchi et al., 2015	•	•				•		
Papai et al., 2014	•	•					•	
Terkelsen et al., 2002	•	•		•				

Table 2.2: Technology and Network of Selected Studies

Sibert et al., 2008		•			•	
Bergrath et al., 2011	•	•		•		
Barrett et al., 2016			•		•	
Felzen et al., 2016	•	•		•		
La Monte et al., 2004		•			•	
Rörtgen et al., 2013	•	•				•
Charash et al., 2011	•	•			•	
Liman et al., 2012			•		•	

## **Data Collection and Analysis Methods**

Of the studies reviewed, 61% included some form of user evaluation (Barrett et al., 2016; Bergrath et al., 2013, 2011; Cho et al., 2015; Czaplik et al., 2014; Felzen et al., 2016; Hadeed & Hadeed, 2011; Kwak et al., 2009; Liman et al., 2012; Lippman et al., 2016; Sibert et al., 2008; Terkelsen et al., 2002; Xiao et al., 2000; Yperzeele et al., 2014) to collect the data from the participants. This evaluation focused on usability of the telemedicine system (Barrett et al., 2016; Kwak et al., 2009; Sibert et al., 2008; Xiao et al., 2000; Yperzeele et al., 2000; Yperzeele et al., 2016; Kwak et al., 2009; Sibert et al., 2008; Xiao et al., 2000; Yperzeele et al., 2000; Yperzeele et al., 2016; Kwak et al., 2009; Sibert et al., 2008; Xiao et al., 2000; Yperzeele et al., 2016; Kwak et al., 2009; Sibert et al., 2008; Xiao et al., 2000; Yperzeele et al., 2016; Kwak et al., 2009), the acceptance rate of the telemedicine system (Felzen et al., 2016; Kwak et al., 2009), the acceptance rate of the telemedicine system

(Bergrath et al., 2013; Yperzeele et al., 2014), and/or the technical quality of the telemedicine transmissions (Bergrath et al., 2011; Cho et al., 2015; Czaplik et al., 2014; Felzen et al., 2016; Hadeed & Hadeed, 2011; Liman et al., 2012; Lippman et al., 2016; Terkelsen et al., 2002). In addition, network signal strength and data transfer speeds were used for data collection, with 4 of the 23 studies using this evaluation method (Czaplik et al., 2014; Valenzuela Espinoza et al., 2016; Xiao et al., 2000; Yperzeele et al., 2014). Another two studies focused on comparing the tasks completed in a regular ambulance to those completed in a telemedicine-equipped ambulance (Charash et al., 2011; Rörtgen et al., 2013). Stroke care studies reviewed involved evaluating the possibility and reliability of stroke assessment (Barrett et al., 2016; LaMonte et al., 2004; Liman et al., 2012; Walter et al., 2012) and measured the time to evaluate a patient with National Institutes of Health Stroke Scale (NIHSS) to validate the system's feasibility (Barrett et al., 2016; Lippman et al., 2016). Time-to-treatment was a method used largely in Stroke and STEMI care studies (LaMonte et al., 2004; Papai et al., 2014; Takeuchi et al., 2015; Terkelsen et al., 2002; Walter et al., 2012), with one general care study using this method (Mandellos et al., 2004). Furthermore, diagnosis accuracy, the comparison of pre-hospital and final diagnosis, was used in three studies (Terkelsen et al., 2002; Valenzuela Espinoza et al., 2016; Yperzeele et al., 2014), while one study used a heuristic methodology to evaluate a telemedicine system, with evaluators and EMS users analyzing the prototype iterations based on their usability characteristics (Tang et al., 2006).

# **Reported Results**

The system user evaluations varied, with four studies measuring system usability (Kwak et al., 2009; Sibert et al., 2008; Xiao et al., 2000; Yperzeele et al., 2014) and five measuring the technical quality of the system (Bergrath et al., 2011; Cho et al., 2015; Czaplik et al., 2014; Felzen et al., 2016; Hadeed & Hadeed, 2011). A study measuring usability through a survey of users, asking them to rate their level of agreement with certain statements found high agreement with the statements, "the telemedicine system was easy to operate," "the telemedicine system was easy to learn," and "delays in sending videos posed few problems."(Xiao et al., 2000) Another study found that a new system has low usability and maintenance ratings, but had high clinical value (Kwak et al., 2009). In two studies, the quality of still images was rated as excellent to good in 73% (Bergrath et al., 2011) and 93% (Felzen et al., 2016) of cases and video streaming was rated as excellent to good in 89% (Bergrath et al., 2011) and 92% (Felzen et al., 2016) of the cases. Acceptance of the telemedicine system was assessed in two studies through user evaluation, with one study concluding high acceptance based on a 0% patient refusal rate (Bergrath et al., 2013). In addition, three studies (Felzen et al., 2016; Kwak et al., 2009; Yperzeele et al., 2014) measured the clinical value of telemedicine with the clinical value of images and video rated very helpful and helpful in routine use by 94% and 87% of the users, respectively, in one study (Felzen et al., 2016).

Because of the limitations in bandwidth and signal at various locations, the user evaluation of the audio and/or visual quality of transmissions varied, with three studies (Czaplik et al., 2014; Xiao et al., 2000; Yperzeele et al., 2014) measuring signal strength and speed. One study determined median upload and download speeds of out-of-office hours at ~200 kilobits per second (kbps) and 100 kbps and of in-office hours at ~750 and 75 kbps with a transmission success rate of 73.2% (Yperzeele et al., 2014).

Comparing traditional ambulance care to telemedicine ambulance care and survival rates, the studies reviewed here found that not only did EMTs in the latter complete more positive tasks in a simulation study but were also able to complete procedures, for which they had no previous training, successfully with the instruction of doctors through telemedicine using simulated patients; in addition, the telemedicine-equipped ambulance (Charash et al., 2011; Rörtgen et al., 2013) had a higher survival rate. Pre-hospital diagnosis and final diagnosis agreement rates in studies comparing the two were high, with a K statistic of 0.98 and 0.92 for neurological and non-neurological disease, respectively, and 100% for stroke cases using the ambulance (Yperzeele et al., 2014).

Heuristic evaluation decreased usability and heuristic violations by 42% and 50%, respectively, after one round of evaluation in one study. However, heuristic evaluation did not find all issues in the system during the four iterations of the design tested. An ethnographic test in this study revealed 48 usability problems, 21 being potentially catastrophic, with 6 of these 21 being verified in the heuristic evaluation (Tang et al., 2006).

For the NIHSS evaluation, agreement rates were measured by comparing different trained evaluator's ratings of a patient completed through teleconsultation and/or by comparing standard training video ratings, or in ambulance ratings to telemedicine system ratings; both of these studies revealed similar results (LaMonte et al., 2004; Liman et al., 2012). In a study with three raters, the overall interrater reliability K statistic in the 12 scenarios tested were 0.78 in the telestroke group and 0.69 in the control (Liman et al.,

2012). In a different study evaluating the success rate in telemedicine NIHSS evaluations, the assessments were successfully completed in 10 of 11 real patient cases and 5 of 5 simulated patients, with the mean time taken to complete the assessment being 7.6 min (Barrett et al., 2016).

Authors	Data Collection Method	Reported Results
Tang et al., 2006	• Heuristic Evaluation	• Heuristic evaluation decreased usability and heuristic violations by 42% and 50% respectively after one round of evaluation.
		• Ethnographic study revealed 48 usability problems, 21 being potentially catastrophic problems, only 6 of the 21 problems were verified in heuristic evaluation
Yperzeele et al., 2014	• Diagnosis Accuracy	• A K statistic of .98 and .92 for neurological and non-neurological disease respectively and 100% of cases with stroke were diagnosed in the ambulance
	• User Evaluation	• 75.4% system activation (high acceptability of the system) and usability was rated as 'Good" (median=4 from Likert scale)
	• Signal Strength and Speed	• Median upload and download speeds out of office hours at about 200 Kb/s and 100 Kb/s. Median upload and download speeds in office hours at about 750 Kb/s and 75 Kb/s. Transmission success rate of 73.2%
Bergrath et al., 2013	• User Evaluation	• 0% patient refusal rate
Cho et al., 2015	• User Evaluation	• Video transmission and Biosignal transmission quality received a satisfaction score of 4 by EMS personnel as well as physicians
Xiao et al., 2000	• User Evaluation	• Users agreed, in some cases strongly, that the telemedicine system was easy to operate, easy to learn and that delays in sending videos posed few problems

Table 2.3: Key Findings of Selected Studies

	• Signal Strength and Speed	• over 70% of pictures transferring at speeds from 1 to 10 frames per second
Espinoza et al., 2016	• Diagnosis Accuracy	• 100% of stroke cases were diagnosed and 83% of stroke pre- hospital diagnoses were confirmed
	• Signal Strength and Speed	• The average upload speeds (from the teleconsultant to the ambulance) was 40 kB/s The average download speeds (from the ambulance to the teleconsultant) was 127 kB/s
Mandellos et al., 2004	• Treatment and Transport Time	• Pre-hospital diagnosis could be made 4 minutes and 20 seconds before patient arrival to the hospital
Takeuchi et al., 2015	• Treatment and Transport Time	• The door-to-balloon time in the telemedicine group was $56.1 \pm 13.7$ minutes and $74.0 \pm 14.1$ minutes in the control group
Papai et al., 2014	• Treatment and Transport Time	• Door-to-balloon time (min) of the telemedicine group was $60.31 \pm 19.50$ and $63.73 \pm 21.13$ for the control group
		• Transport time (min) of telemedicine group was $53.75 \pm 32.97$ and $40.78 \pm 21.30$ in the control group
Terkelsen et al., 2002	• User Evaluation	• Technical quality of the ECGs transmitted were classified as 'good' in 78% of cases and technical quality of communications were classified as 'good' in 73% of cases
	• Diagnosis Accuracy	• Positive Predictive value of 56%
	• Treatment and Transport Time	• Transport time to local hospital in control group was 22 minutes, in telemedicine ambulances where diagnosis was not attempted 10 minutes, in telemedicine ambulances where diagnosis was attempted 14 minutes. Door-to-needle time in control group was 81 minutes, and in telemedicine ambulances 38 minutes
Sibert et al., 2008	• User Evaluation	• Ratings from 1-9 in usability and technical quality characteristics: Ultrasound = $5.1 \pm 1.6$ , Laryngoscope = 7.2.
Bergrath et al., 2011	• User Evaluation	• Quality of still images: 73% 'excellent'-'good', quality of video streaming 89% 'excellent'-'good'
Barrett et	User Evaluation	• Satisfaction rated at "satisfied" to "very satisfied" in 83% of cases
al., 2016	• Stroke Evaluation	• NIHSS evaluations successfully completed in 10 of 11 real patient cases and 5 of 5 simulated patients and mean time to complete the assessment was 7.6 minutes
--------------------------	-------------------------------------	---
Felzen et al., 2016	• User Evaluation	• Clinical value of images and video were rated very helpful to helpful in routine use in 2014 by 94% of users and 87% of users respectively, Quality was rated at Excellent to good for 93% of users for pictures and 92% of users for video
La Monte et al., 2004	• Stroke Evaluation	• Data showed no variability between patients on scores for NIHSS questions including level of consciousness, level of consciousness commands, best gaze, visual fields, and right motor arm testing
	• Treatment and Transport Time	• The mean time to treatment for historic control patients was $33 \pm 17$ minutes and that for patients with TeleBAT transport was $17.4 \pm 3.6$ minutes.
Rörtgen et al., 2013	• Task Evaluation and Comparison	• Based on the predefined checklist of sample care activities the telemedicine group's care was comparable to the control with some positive tasks being performed at a higher percentage than the control (obtaining allergies, and medications)
Charash et al., 2011	• Task Evaluation and Comparison	• 24 tests with the telemedicine groups only 2 patients expired, compared to the control where 16 of 16 simulated patients died. Other measures such as the interventions, signs and patients, the telemedicine groups % correct identification was much higher (92%, 96%, 98%) than the control (49%, 79%, 75%). and in the simulations that were outside of the EMTs scope of practice only the telemedicine group was able to complete procedures.
Liman et al., 2012	• User Evaluation	• mean audio signal quality ratings at 2 out of 6 and video signal quality ratings between 2 and 4 out of 6
	• Stroke Evaluation	• Interrater reliability K statistic in the 12 scenarios tested and the 3 raters was .78 in the telestroke group and .69 in the control

When evaluating time to treatment in stroke application, one study found the mean time to be 33 min in the control group and 17 min in the telemedicine group (LaMonte et al., 2004). The results from STEMI studies vary in the improvement in door-to-balloon time, but one study (Terkelsen et al., 2002) found the mean door-to-needle time was 38

min faster than the control group. This study also found a mean transport time that was faster in the telemedicine group by 8 to 12 min depending on if a diagnosis was attempted (Table 2.3).

#### DISCUSSION

This review was executed to understand the current state of research and detail benefits and challenges faced in studies implementing or testing telemedicine systems in ambulances. Limitations of these studies were documented and are discussed in this section to gain insight on what further study needs to be completed to eliminate these limitations.

Growing concerns in healthcare information technology include the feasibility of using technology to assist in patient care (Chalil Madathil et al., 2013; Madathil, Rivera-Rodriguez, Greenstein, & Gramopadhye, 2015), especially in emergency situations. Of particular interest is how to ensure that this technology works seamlessly with healthcare professionals to limit interference with patient care (Board on Health Care Services & Institute of Medicine, 2012; Koikkara, Greenstein, & Madathil, 2015). Usability testing and user evaluations were represented as a methodology used in more than half of the studies reviewed in this article, with one study devoted entirely to heuristic evaluation and ethnography (Tang et al., 2006); however, four studies reported low usability (Felzen et al., 2016; Kwak et al., 2009; Sibert et al., 2008) or low transmission quality (Liman et al., 2012; Sibert et al., 2008), and one study concluded that the transportation of the telemedicine unit was not feasible in situations where transport on scene to treat a patient outside of the ambulance was needed given the size and weight of the unit (Cho et al., 2015). Given the importance of implementing a highly functional system in a timedependent process, stakeholders such as EMTs, patients, and doctors should be at the heart of the design process. The timing, efficiency, and level of care are paramount in EMS, and design of the telemedicine systems has to be designed with them in mind to be integrated successfully.

One concern in human-centered design is the cognitive and physical stressors put on the stakeholders by the system (Narasimha, Agnisarman, et al., 2016; Narasimha, Chalil Madathil, et al., 2016; Sanderson et al., 2013; Valdez et al., 2016). Research developing possible stressors, how they affect work in ambulances and with a teleconsultant, and how to eliminate these stressors is needed specifically with telemedicine to be a viable tool for use in ambulances. Another consideration in design is the patient's level of perceived privacy, which has been addressed in one study included in this review. The study cited that patients did not feel that telecommunications intruded on their privacy and felt comfortable in the use of the system, but more studies should include an emphasis on this viewpoint in the patient experience (Xiao et al., 2000). Another aspect necessary to creating effective care when using telemedicine systems is designing those systems in an integrated environment (Madathil, Greenstein, Juang, Neyens, & Gramopadhye, 2013; Narasimha, Chalil Madathil, et al., 2016; Valdez et al., 2016). This method limits training needed to effectively use the system and reduces probability of errors given that the system is similar to other technologies and systems in the environment that users are familiar with (Valdez et al., 2016).

## **Simulated Studies**

Six of the 23 articles reviewed used a simulation study as their sole method of system testing (Charash et al., 2011; LaMonte et al., 2004; Liman et al., 2012; Lippman et al., 2016; Rörtgen et al., 2013; Sibert et al., 2008), a situation that limited the procedures tested because they used medical dummies. For example, the study evaluating the feasibility of completing ultrasounds and intubation procedures in an ambulance through teleconsultation, used a medical dummy to test intubation with a video laryngoscopy and 1 patient to test the ultrasound process, simplifying the investigation to a still, nonresponsive patient or a healthy participant (Sibert et al., 2008). A second study used an METI Human Patient Simulator; however, it simulated patient responses using only a limited number of noninvasive procedures to test the system, thus limiting the testing capacity of the study (Charash et al., 2011).

Another 9% of the studies used actors to simulate patients, specifically in the NIHSS evaluation. One study generated 12 scenario scripts that actors and clinical care transport staff completed, while three evaluators observed either by TV/VCR recording or through a TeleBAT system on a desktop computer, comparing the results to historic patients (LaMonte et al., 2004), while another study used actors trained in simulating right and left middle cerebral artery stroke syndromes with differing severity (Liman et al., 2012). These studies may introduce a bias in the simple NIHSS evaluation because of the previous experience of the actors in simulating strokes as well as limitations in evaluating clinical value of vital transmission as these studies used healthy patients. Two studies reviewed completed an onboard trial with a simulated patient to test the feasibility of the system and orient the medical staff with the equipment before a live trial to both test system

capabilities in real situations and validate the feasibility results (Barrett et al., 2016; Xiao et al., 2000).

#### Sample Size and System Testing

A common limitation cited in the studies reviewed was the sample size of the patient trials or simulation runs. Studies cited small sample sizes as they tested the system, using 4 to 12 participants. Similarly, the sample sizes of simulation tests were also small (n = 7) as were simulation scenarios which ranged from a sample size of 3 to 12; however, the number of runs of these scenarios ranged from 24 to 30. Four of the eight studies citing small sample sizes were feasibility or pilot studies, although further studies with larger sample sizes have been conducted in other studies included in this review. Limitations in system use were cited in 4 of the 23 studies reviewed; however, this issue was of particular importance in simulated patient studies. Two such studies only tested specific functions of a system (LaMonte et al., 2004; Sibert et al., 2008) rather than running tests in the field. An additional four studies investigated system functionality using simulated patients (Charash et al., 2011; Liman et al., 2012; Lippman et al., 2016; Rörtgen et al., 2013). Field test studies addressed this limitation by running tests for long periods of time with no dispatch routing rules and large sample sizes of patients, thus requiring a range of system functions to be used.

## **Bandwidth and Connection Limitations**

A common challenge discussed in these studies was the technical issues faced when using public networks. More specifically, 13 studies which cited this technical challenge or limitation used this method because of its widespread application and low cost. However, this method created a trade-off in data transfer size and speed. In addition, using public networks in rural areas resulted in signal loss. In one study, the system addressed this issue by allowing users to select the frame rate or picture quality for the telemedicine transmissions (Sibert et al., 2008).

#### **EMT Considerations**

Interactions among the stakeholders, including patient and EMT personnel, patient and doctor (in a remote facility), and EMT and doctor (remote facility), should be considered when designing a telemedicine-mediated healthcare environment. When designing telemedicine systems, from a human-system integration perspective, consideration should be given for the current tasks completed by EMTs and doctors in a technology-mediated environment. Integrating technology such as a telemedicinemediated environment into an ambulance setting can change the job requirements of EMTs and could change the methods that have proven to be successful for maintaining their situation awareness. It could create more work because now the telemedicine equipment, as well as the system itself, must be monitored.

In this development of communication systems, EMTs can do much more than just stabilize a patient; they could begin testing, create a tentative diagnosis, or even administer treatment, if authorized. This situation is tested in studies in this review, specifically those measuring diagnosis accuracy (Terkelsen et al., 2002; Valenzuela Espinoza et al., 2016; Yperzeele et al., 2014). Additional studies mentioned a pre-hospital diagnosis being performed, but did not measure the accuracy of those diagnoses to the final hospital diagnosis (Bergrath et al., 2011; Felzen et al., 2016). In the studies measuring diagnosis accuracy, the accuracy was found to be high, reflecting clinical value of EMTs diagnosing patients in ambulance or on the scene not just for simplicity of treatment post-transport, but also because this ability to diagnose pre-hospitally could shorten transport time and treatment time.

#### **Financial Implications**

Thirteen studies mentioned the limitations of using public networks for costeffectiveness and common, inexpensive materials to outfit ambulances as teleambulances when available; however, the cost of this initiative remains a concern. One study identified in the literature screening process for this review investigated the cost of using telemedicine to increase the use of rt-PA treatments in eligible patients. The study also evaluated the benefit-to-cost ratio of the system using different configurations of personnel at varying operating distances, and even the highest cost configuration of personnel achieved a benefit ratio greater than 1 at an operating distance of 20 km (Dietrich et al., 2014).

Another study investigated the benefit to cost comparison of congenital heart disease patients using an ambulance equipped with an ECG, a camera, an iPod touch and a laptop connected to a central server. The results of this study found cost savings between  $33,586 \in$  and  $35,740 \in$  within 1 year (Frexia et al., 2014). While these studies only represent care for a specific condition rather than general care, they show that it is feasible and cost-effective to implement a telemedicine system. Future study could include the possibility of using private networks with higher bandwidth capabilities and a cost analysis of applying a telemedicine system for general emergency care.

## Limitations

This review has limitations. Only peer-reviewed journal articles written in English were included in this review. Defining the search keywords was difficult, and it was learned that not all studies that should have been identified were found in the search.

### **Future Research Questions**

This review suggests a paucity of published studies describing scientifically valid and reproducible evaluations at various stages of telemedicine implementation in ambulances. The common limitations and unique testing challenges found in the studies reviewed here led to the formulation of several recommendations for future studies or for generating a focus for a study to further test the feasibility of telemedicine systems in improving patient care.

A system of systems is where a collection of different systems, originally designed for a specific purpose, are combined and/or coordinated to produce a more capable system. Inherently, such a system poses new challenges (National Research Council, Division of Behavioral and Social Sciences and Education, Committee on Human Factors, & Committee on Human-System Design Support for Changing Technology, 2007; Sanderson et al., 2013). A telemedicine-integrated ambulance is an example of a system of systems. Human capabilities and limitations must be considered while implementing such a system. A technology-mediated caregiving environment is operated by a team consisting of EMTs and doctors whose interactions must be considered while implementing a telemedicine system in an ambulance. System designers need to ensure that in such a remote caregiving process, considering the operational aspects of just the telemedicine system may not be enough. The key stakeholders involved in this process are required to operate multiple systems, including Electronic Health Records (EHRs) telemedicine systems, adding to the complexity of the caregiving process. Extreme cognitive, physical and temporal demands are placed on the caregivers when operating such a system of systems. Further studies need to be conducted to understand the sensemaking process and situation awareness of caregivers while providing care in such technology-mediated, high-stress environment.

Four studies reviewed discussed the impact of developing studies in a variety of areas, as the location determines such variables as response and transport times, signal availability, patient acceptance, and population statistics. One simulation study in this review determined that transportation time in rural areas could be as long as 40 to 50 min, thus creating simulations in which a patient would go critical within 40 min to test the capacity of EMTs to perform lifesaving measures over long transport times. The range of transport times, among other factors, suggests the need for future research exploring the implications in patient care and survival rates of implementing a telemedicine system in an urban area versus a rural area.

One future application of telemedicine that could improve emergency department function would be to use teleconsultation to triage emergency patients. The American College of Surgeons suggests an acceptable percentage of over-triaging is 25% to 35% (Sasser et al., 2012); however, even in that range, there is a cost associated with this practice. The Center for Disease Control and Prevention (CDC) cites a study comparing triaging guidelines from 1991 to 2006, identifying a potential \$568 million cost savings (Sasser et al., 2012). Telemedicine may be a useful tool for addressing this situation, if prehospital diagnosis could be utilized for triaging patients and preparing the hospital for their arrival. Future research needs to validate using remote triaging through teleconsultation to reduce over-triage rates and improve patient transfer.

#### CONCLUSION

This article reviewed the literature conducted on telemedicine systems in ambulances for emergency care. According to these studies, the usability ratings of the systems tested are high, and the effects of implementation are significant and positive. However, limitations in usability testing, simulation studies, public network bandwidth, sample size, and cost suggest the need for additional research. Thus, future work should focus on enhanced human-system integration, developing a high fidelity, usable system at a low cost and testing this system using live patients suffering from a variety of conditions and injuries to fully explore the use of telemedicine in ambulances. Such research studies could further support the use of telemedicine in ambulances as a viable way to treat emergency patients quickly and efficiently, thus improving the healthcare delivery system for this population.

#### CHAPTER THREE

# A SYSTEMATIC REVIEW OF HUMAN ERROR IN EMERGENCY AMBULANCE-BASED CARE

#### **INTRODUCTION**

#### **Emergency Medical Services**

Emergency medical services (EMS) stabilize and care for a variety of people, frequently involving complex health conditions, in a fast-paced, high-stress environment, with these prehospital providers working in shifts longer than 8 hours at a time. EMS responders, Emergency Medical Technicians (EMTs) or paramedics respond to 911 calls to assess the patient, stabilize their condition, conduct what tests they can, and in many cases transport the patient to a hospital. Depending on the severity and location of the patient emergency responders will transport the patient via an ambulance or helicopter. These emergency responders can be called out for a variety of symptoms, most commonly found in this review were difficulty breathing, anaphylaxis, stroke symptoms, or cardiac arrest symptoms. Some cases can be straightforward, clear and present symptoms of a heart attack for which EMS has protocols in place to treat and transport the patient. Other more complex situations such as unresponsive patients, or patients with multiple competing trauma injuries require more effort to stabilize or difficult procedures such as intubation. Protocols and treatments can be standardized across countries, states, or nations; however each EMS is different. Some EMS use volunteers to operate, or others use employed staffing, the number and training of individuals in the ambulance can also vary. When more individuals (EMTs or paramedics) are available to treat and transport the patient, tasks may be more manageable as they are distributed across the caregivers (collecting blood pressure, establishing an IV line, gathering blood samples). However, having multiple caregivers in this environment creates the need for coordination of tasks and communication which in addition to the fast pace and urgency in these tasks creates a complex sociotechnical system. These factors can create a high incidence of error in both equipment and organization, but more importantly, in human action or inaction. In fact, errors in this environment potentially occur more frequently than in other health care settings because of the high decision density, the severity of the health conditions being treated, and the small work area (Hobgood et al., 2004; Patterson et al., 2012). These cases are also, by nature of being in an ambulance, emergencies, meaning errors could quickly become serious or fatal. A common practice in healthcare is to measure adverse events, injury caused by medical management rather than disease or condition of the patient (Brennan et al., 1991). Research abounds in healthcare on the topic of adverse events, their occurrence in different fields or geographic locations (Davis et al., 2001; Kable et al., 2002; Rigby et al., 1999; E. J. Thomas & Brennan, 2000), methods of measurement (Leape, 2002; Thomas & Petersen, 2003; Walshe, 2000; Woloshynowych et al., 2005), effects on patient safety (Mardon et al., 2010; Rivard et al., 2008; X. Wang et al., 2014), and mitigation strategies (Pettker et al., 2009; Rafter et al., 2015). However, an important distinction is that not all adverse events are errors, preventable adverse events are events created by error. For these reasons, research on patient safety in the prehospital setting is critical, especially that of human errors in this setting.

## **Human Error Taxonomies**

James Reason in his seminal book, Human Error, describes the process of identifying error as a series of 3 yes or no questions: were the actions directed by some prior intention, did the actions proceed as planned, and did they achieve their desired end (Reason, 1990). Reason describes human error in this way to identify where in the mental process errors are made in a similar fashion to the skill-rule-knowledge (SRK) framework developed by Rassmusen 10 years prior (Rasmussen, 1983). James Reason furthermore created the Swiss Cheese model, which describes how hazards can travel through layers in a system and when holes in these layers align it allows for "a trajectory of accident opportunity" (Reason, 1990). This model identifies four layers of the cheese model, which allow for accidents to occur: organizational influences, unsafe supervision, preconditions for unsafe acts, and unsafe acts. This model was the foundation for the error taxonomy called Human Factors Analysis and Classification System (HFACS) created by Doug Wiegmann and Scott Shappell shown in Figure 3.1 (Shappell & Wiegmann 2000). This error taxonomy focuses on errors as a result of system failures rather than only the result of the actions of an individual. Unsafe acts are the actions which are made in the system that can be further classified as errors or violations. Errors in this case are acts that are not purposefully incorrect. Errors in this model are described by 3 categories: Decision Errors, Skill-based Errors and Perceptual Errors. These three categories are modeled after the SRK framework. Decision Errors are made by applying an incorrect rule or choice to a situation which is similar to rule-based errors in SRK. Skill-based Errors are typically due to misattention or failure of memory or technique, which is the same in SRK. Perceptual Errors

are made when a situation is misjudged or not perceived correctly, this type of error correlates to the knowledge-based errors in SRK. It is the classification level of unsafe acts that adds to the concept of violations which are unsafe acts or errors, in terms of how the system was designed, that are done purposefully. These violations occur in two ways; a routine violation would occur for example, when an assembly worker installs items based on experience, rather than the policy, to speed up the installations, also known as a workaround. Violations can also occur as an exception, an isolated departure from authority.



Figure 3.1: HFACS model Adapted from (Shappell & Wiegmann, 2000)

James Reason states that there is no "universally agreed classification of human error ... and no single scheme is likely to satisfy all needs." (Reason, 1990). However, these classifications support understanding how errors are made and provide avenues for interventions that minimize occurrences of error. The knowledge of what errors are or what mental processes cause them can identify solutions for minimizing errors in a systematic fashion.

## Human Factors Approach to Human Error

To Err Is Human asserts that the problem is not bad people in healthcare--it is that good people are working in bad systems that need to be made safer (Institute of Medicine (US) Committee on Quality of Health Care in America, 2014). Human Factors concepts provide frameworks or theory to understand error as a function of cognition and work systems. One such framework is the Systems Engineering Initiative for Patient Safety (SEIPS) model. The SEIPS model offers more specific human factors concepts to organize human error within healthcare than the generalized error frameworks described above. The updated SEIPS 2.0 model describes the healthcare socio-technical system and how that system affects the work and outcomes of work of professionals and non-professionals. The factors of the system are described in the figure, these factors impact the work processes which in turn creates outcomes of different kinds. Outcomes can further be described as desirable or undesirable and proximal (immediate result) or distal (result further removed from work process). Adaptation is also considered in this model, feedback from outcomes impact how work processes are conducted and the work system as a whole. SEIPS can provide insight and interventions for error by organizing the root cause of error by work system factors. By understanding the work system factors that impacted the processes creating errors, or undesirable outcomes, interventions can be focused on improving those work system factors.

## **Research Purpose and Questions**

Past studies have focused on understanding the prehospital setting (Brice et al., 2000; Callaham, 1997; Cohen & Patel, 2014). A prior review has also systematically reviewed patient safety in emergency medicine, however their inclusion for review was based generally on events in EMS that cause patient harm with no further qualification of how that event was committed (human, equipment, management) (Bigham et al., 2012). Currently, no attempts have been made to systematically review the types and occurrences of human error in ambulance based care. An understanding of human errors in this setting is critical for developing interventions to mediate error and improve patient safety. As a first step, this review summarizes the limited literature found to quantify the occurrences and causes of human error in prehospital care. Specifically, it addresses the following research questions:

- 1. What types of errors are made by prehospital providers during emergency care?
- 2. How often does human error occur in the prehospital setting?
- 3. What are some of the causes of the errors made by prehospital providers in emergency care?

### METHODOLOGY

## Search Strategy

38

Eleven databases: Web of Science, PubMed, Science Direct, AgeLine, and Alt Health Watch, Applied Science and Technology, PsychINFO, ACM Digital Library, Health Reference Academic, BioOne, and Academic OneFile, were searched between January 18-23, 2019 for articles to better understand the errors made by emergency response personnel in ambulances. These databases were searched for articles published between 1988 and 2019 with titles or abstracts containing terms emergency medical services and errors along with two term combinations of synonyms for each term. Table 3.1 lists all terms used. An example of the search strategy used for the BioOne database is ((ABSTRACT:(prehospital)) OR ABSTRACT: (Emergency Medical) OR ABSTRACT: (Emergency OR ABSTRACT: (Ambulance) Care) OR ABSTRACT:(paramedic)) AND (ABSTRACT:(error) OR ABSTRACT:(incident) OR ABSTRACT:(mistake) OR ABSTRACT:(failure) OR ABSTRACT:(adverse event)). The search strategy for each database varied based on the selection of terms available or the coding structure required. The complete search strategy for each database, and search results can be found in Appendices A and B. Additional articles were found by hand searching the references in the studies selected after full text review to exhaust all options of search. The number of citations and the filtering process are outlined in Figure 3.2

## **Data Selection**

After completing the database search the articles were reviewed following the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guidelines.

Emergency Medicine Terms	Error Terms
emergency medical services or EMS	error
prehospital, pre hospital, or pre-hospital	incident reporting or incident
emergency medical technician or EMT	mistake
ambulance	adverse event
emergency care	safety event
paramedic	mistake
	failure

#### Table 3.1: Two Term Search List

Articles were first reviewed by title to eliminate those that may have contained the search term *error* in the title but did not focus on the errors made in the prehospital setting. Then abstracts were reviewed first by finding the context of the study, followed by the methods for recording or determining errors, and finally identifying the type of errors. The inclusion criteria for this review were that the study measured the number of errors or error occurrences in some way, the errors were defined (for example as medication error, deviations from regulation procedure, failure of a procedure, diagnosis accuracy, or generally as adverse or safety events), and the study measured these errors only in an emergency setting and within the prehospital arena of patient care (ambulance treatment and transport). Non- English, abstract only, and commentary papers were excluded. Then the full text of the articles were reviewed using the same inclusion criteria.

Each of these stages of review were completed by two independent reviewers (Hunter Rogers and Mackenzie Wilson), then consensus coded after assessing Inter-Rater Reliability (IRR), which was calculated using prevalence-adjusted bias-adjusted kappa to adjust for the large number of papers removed at each stage (Chaturvedi & Shweta, 2015). IRR for the title review was 0.98, for the abstract review 0.73, and for the full text review was 0.72. Additionally, the quality of each study was assessed before inclusion in the review.

## **Quality Assessment**

Determining the quality of studies selected in a systematic review is generally good practice, but it is important to limit the studies reported to only high quality research to ensure that standards of reporting and measuring these errors were maintained in the studies. It is also important to assess researcher biases in selection by using a rigid method of quantifying quality. For these reasons a quality checklist was used to review all studies that remained after reviewing full text articles for inclusion criteria. Given the variety of the studies found and the number studies lacking an intervention, the Strengthening The Reporting of OBservational Studies in Epidemiology (STROBE) Checklist was used (von Elm et al., 2007). The checklist is structured by sections of an article selected for review, certain items are required in the introduction, methods, results, or discussion and conclusion. Checklist items were elements of the study or analysis that must be reported in the article. For example, describing limitations of the study, reporting demographic characteristics of sample size, or stating study objectives or hypotheses. This checklist allows for the most variability in study design, but as it does not have a quality structure, one was created. If a study had 80% or more of the checklist items completed, it was considered good quality, between 80% and 60%, fair quality, and less than 60%, poor quality and was not included in the review. Five studies were removed for low quality, with two items common among all that were removed: explaining how sample size was determined and describing the generalizability of the results. The quality assessment, with rows in red indicating the study was of poor quality, of each study can be found in Appendices C and D.

## **Data Extraction**

Articles were reviewed, and data items were iteratively added to the extraction list by one researcher. A table was used to list the data from each paper: when and where the study was conducted, what procedures were considered for analysis, if a population was specified for the study, if a condition was used to limit the scope of the study, what types of errors were documented, possible causes of these errors, and methods used in conducting the study. The primary outcomes extracted from each study were the error occurrence studied and the possible causes for error. These findings were subsequently used to develop themes found in the literature and to systematically analyze the errors found in these studies.

### RESULTS

The initial search returned 28,773 articles, 21,929 after removing the duplicates; using the process seen in Figure 1, another 21,781 were removed, meaning 148 remained for full article review. The inclusion criteria were applied to an additional search that was conducted as described in the Methods, resulting in 69 articles to be included in this review.



Figure 3.2: Citation search and review process

Five distinct themes based on the types of errors seen in emergency medical services emerged in these studies: procedure errors (Table 3.2), protocol deviations (Table 3.3), medication errors (Table 3.4), diagnosis accuracy (Table 3.5), and adverse or safety events (Table 3.6). Data for each including population, condition, procedure, method, error occurrence, and possible causes were extracted as can be seen in Tables 3.2-6. However, the variety of methods and procedures used did not allow for meta-analysis.

## Location

Studies included in this review were conducted in many countries, a majority (60%) being from the United States, with most being conducted by a state or local EMS. Although some studies were conducted nationally (10%), the rest were conducted in state communities. Of the studies reviewed, 40% were conducted internationally: 8 in Australia (17%); 5 in Germany or German-speaking countries (7%); 4 in Canada (6%); 2 in Israel (3%); 2 in the United Kingdom; and 1 in each of the following: Finland, Iran, New Zealand, South Africa, Sweden, and Switzerland. The US has 28% studies in pediatric care, the only other country to identify a specific population is Sweden, their only study in this review, which was also a study of pediatric care. Almost all locations in this review isolated certain conditions, the US (40% of studies), Canada (50%), Australia (50%) were the most frequent outside of the UK, Finland, New Zealand, Switzerland, and German-speaking countries for which each of the single studies isolated a condition. Cardiac conditions were the most common, but the relative percentage of studies conducted was similar across countries who isolated any conditions. However, stroke research was only conducted in the

Citation	Population	Condition	Procedure	Method	Error Occurrence		Possible Causes
Eberle et al. 1996*	None Specified	None Specified	Pulse Check	Simulation	83.5% were not able to reach the diagnosis within the guideline time limit	•	Lack of intensive training specifically for pulselessness
Ruppert et al. 1999*	None Specified	None Specified	Breathing Check	Simulation with mannequin	44.4% of the time incorrect technique was used	•	Lack of training for participants except EMS
MacDonald et al. 2001	None Specified	None Specified	AED use	Retrospective Record Search	Almost 4% of all rhythms had errors and 54% of those were operator errors	• •	Patient movement during AED analysis due to CPR Unable to recognize when the AED is analyzing a cardiac rhythm
Barnes et al. 2003	None Specified	None Specified	Blind Tracheal Intubation	Simulation	79% were not successful in blindly placing an endotracheal tube in the 60 second window	•	None Specified
Jemmett et al. 2003	None Specified	None Specified	ET Intubation	Observation	12% of all EMS recorded cases were misplaced	•	No protocol in place for checking tube placement with end-tidal CO2 monitors or esophageal detector devices
Jones et al. 2004	None Specified	None Specified	ET Intubation	Observation	Unrecognized, misplaced ET Tubes inserted in the out-of-hospital setting byground paramedics happened in 5.8% of cases	• •	Limited training or tools No uniform application of verification techniques
McDermott et al. 2005	None Specified	Traffic Fatalities	None Specified	Retrospective Record Search	77% of fatalities had prehospital errors that contributed to death, of the errors found, 11% were human error	• •	Failure to establish IV access Low GCS score for intubation patients
Wang et al. 2006	None Specified	None Specified	ET Intubation	Retrospective Record Search	23% of patients had an ET intubation error occur	• •	Semiconscious and head-injured patients protective reflexes that impede access to airway Limited practice of procedure in clinical field

## *Table 3.2:* Procedure Errors

(Continued on next page)

45

Hein et al. 2008	None Specified	None Specified	Laryngeal mask airway	Retrospective Record Search	Failure due to technique was found in 9 of 285 attempts 0.03%	<ul> <li>Difficulty c</li> <li>insertion</li> <li>Inapproprié</li> </ul>	letermining successful tte size selection
Brice et al. 2008	None Specified	Chest Pain	Demographic reporting	Retrospective Record Search	Overall error rate (name, DOB, SSN, etc.) was 73.9%	<ul> <li>Patient dist</li> <li>EMT more care than re</li> <li>Poor handy</li> </ul>	ress focused on patient cording vriting skills
Williams et al. 2008	None Specified	None Specified	ECG interpretation	Simulation	90% of paramedic students did not correctly identify the ST-segment at least once and 89% incorrectly identified the J-point at least once	• Level of ed learning en	ucation and mode of vironment
Kaserer et al. 2017	None Specified	Trauma	Resuscitation	Retrospective Record Search	17% failure rate with TT, 82% failure with NT	<ul> <li>Limited exident</li> <li>determining</li> <li>pneumotho</li> <li>Inadequate</li> </ul>	perience and difficulty g tension rax in the field catheter length
Alenyo et al. 2018	None Specified	None Specified	Triage	Survey	Over-triage was 31.4% (95% CI, 29.66 to 33.2), and under-triage was 13.8% (95% CI, 12.55 to 12.22)	• Limited tra	ining in triage
Ghiyasvandian et al. 2018	None Specified	None Specified	Ventilation and Intubation	Simulation	The total failure rate in the three stages of ventilation, intubation and airway back-up were 49.4%, 46.7%, and 50%, respectively	<ul> <li>Inappropriá tube</li> <li>Lack of exj</li> </ul>	tte depth of intubation perience with BVM
* Also addressed dia Glasgow Coma Scalı	ıgnosis errors. A e, IO: Intraosse	AED: Automated ous, IV: Intraver	d External Defibrillato nous, NT: Needle Tho	or, BVM: Bag Valve oracentesis, SSN: So	Mask, DOB: Date of Birth, EMT: Emergency M cial Security Number, TT: Chest Tube Thoracost	edical Technician, omy	ET: Endotracheal, GCS:

Citation	Population	Condition	Procedure	Method	Error Occurrence		Possible Causes
Salerno et al. 1991	None Specified	None Specified	None Specified	Retrospective Record Search	16% of ambulance runs were found to contain procedure deviations, 45% were serious or very serious	• •	In 69% of cases with procedure deviations a physician was not contacted for advice Lack of understanding of IV solution or flow rates needed
Liberman et al. 1999	None Specified	None Specified	CPR	Simulation with mannequin	74.2% of compression depths were outside of guidelines, 35.9% were in the wrong position, 60.6% ventilations were incorrect	• •	Insufficient practice of correct compression depth Unable to complete landmarking to find proper position
Kirves et al. 2007	None Specified	Cardiac arrest	Resuscitation	Retrospective Record Search	60% of cases did not meet care guidelines	• •	Patients with neurological diseases and females had were more frequently associated with care not within care guidelines Overvue of adrenaline during care
Boyle 2009	None Specified	Vehicular trauma	None Specified	Retrospective Record Search	Errors per patient in the prehospital were up to 3.20 with 2.12 of those errors contributing to death in the 2003 report	• •	Constant change and introduction of new procedures and drugs Training to support these changes
Lammers et al. 2009	Pediatric	Cardiac Arrest, Asthma, and Sepsis	None Specified	Simulation in Static Ambulance	The average amount of procedure stepped missed by scenario were 54.7% for cardiac arrest, 48.4% for asthma, 52.9% for sepsis	• •	Overconfidence in intubation ability Lack of supporting checklist
Cienki and DeLuca 2012	None Specified	None Specified	Blood Pressure Measure	Ride Along Observation	Lowest rates of adherence to standards were found in palpitating the brachial artery and placing the pressure cuff at 6%	•	Paramedics are unaware of problems with their technique
Lammers et al. 2012	Pediatric	Shaken Baby Syndrome	None Specified	Simulation in prepared room	Participants did not complete 38% of procedure steps identified by researchers	•	Did not use proper equipment for oxygen delivery
Lammers et al. 2014	Pediatric	Cardiac Arrest	None Specified	Simulation in prepared room	Participants did not complete 33% of procedure steps identified by researchers	• • • •	Lack of proper respiration equipment knowledge Poor CPR compressions Inability to place IO line Inaccurate weight estimates and dosing calculations

## Table 3.3: Protocol Deviations

CPR: Cardiopulmonary Resuscitation, IO: Intraosseous, IV: Intravenous

Citation	Population	Condition	Procedure	Method	Error Occurrence	Possible Causes
Hubble et al. 2000	None Specified	None Specified	Medication Dosing	Simulation	The mean score was 51.4%	<ul> <li>Skill decay from drug calculations infrequently performed</li> <li>Lack of calculation aids in field and continued competency training</li> </ul>
Goebel et al. 2004	None Specified	Tachycardia	Adenosine Use	Retrospective Record Search	20% of patients had inappropriate use of adenosine due to rhythm misidentification	<ul> <li>Lack of practice identifying rhythms</li> <li>Limited use of 12-lead electrocardiogram</li> </ul>
Kaji et al. 2006	Pediatrics	Cardiopulm- onary Arrest	Epinephrine Dosing	Retrospective Record Search	In the most recent 2003-2004 cohort 45% of doses were not within 20% of correct dose, as compared to the 1994-97 cohort which had a rate of 66%	<ul> <li>Difficulty of remembering dosing and concentration with different administration routes</li> <li>Higher error rate when Broselow tape was not used to estimate weight</li> </ul>
Vilke et al. 2007	None Specified	None Specified	Drug Administration	Survey	Of responding paramedics 9.1% reported an error made in the last year, 67% were medication related, with the most common being Atropine and epinephrine	<ul> <li>Failure to triple check</li> <li>Infrequent use of medication</li> <li>Dosing calculation</li> </ul>
Hoyle et al. 2012	Pediatrics	None Specified	Drug Administration	Retrospective Record Search	Dosing errors occurred in 34.7% of doses administered during the study period, with epinephrine having the highest incorrect dosage rate	<ul> <li>Lack of pediatric clinical experience</li> <li>Calculation error</li> <li>Low utilization of dosing aids</li> </ul>
al. 2012	None Specified	None Specified	Drug Administration	Retrospective Record Search	12.7% of patient had a drug error occur in care and 25.8% were described as significant	<ul> <li>Longer evacuation time had higher risk for error</li> <li>32.6% of these errors were due to a drug administer that was not indicated for the patient</li> <li>38.7% were dosing errors</li> </ul>

## *Table 3.4:* Medication Errors

(Continues on next page)

Eastwood, et al. 2012	None Specified	None Specified	Medication Dosing	Simulation	Undergraduate paramedics achieved an overall mean score of 39.5%	<ul> <li>Po</li> <li>Kn</li> <li>pra</li> </ul>	or basic math skills towledge decay with lack of totice
Lim et al. 2013	None Specified	Tachycardia	Adenosine use	Retrospective Record Search	20% of patients had inappropriate use of adenosine due to rhythm misidentification	• La Thy ele	ck of practice identifying /thms mited use of 12-lead ctrocardiogram
Lammers et al. 2014	Pediatric	Anaphylaxis	Medication dosing	Simulation	Of the 95% of teams that chose epinephrine, 54% used incorrect dosing or method, other medications used were outside of correct dosing 99% of the time.	• W( • Fai cor	eight estimation error ulty reasoning or mmunication error
Stevens et al. 2015	Pediatric	Cardiac Arrest	Epinephrine and Atropine dosing	Simulation	53% of cases of incorrect doses were errors and 27% of those were critical errors	• Ca • La	lculation of dosing required ck of regularity and practice th pediatric dosing
Coppler et al. 2016	None Specified	None Specified	IV fluid measurement	Observation	The absolute error comparing self- reported to mass-derived fluid volume was median 109 mL [IQR: 41–205 mL], and was less than 250 mL in more than 80% of subjects, which is small	• • Lo	ort transport times w acuity transports
IOR · Interdular	tile Range IV.	Intravienone					

IQR: Interquartile Range, IV: Intravenous

				N K = 41			
Cutation Kothari et al. 1995	roputation None Specified	Stroke	Diagnosis	Retrospective Record Search	Error Occurrence 28% of diagnoses made by EMTs were unconfirmed by Emergency Department physicians	• •	rossible Causes Short interactions with patients Lack of knowledge of the symptoms of a stroke
Hollander et al. 1995	None Specified	None Specified	ECG Interpretation	Retrospective Record Search	10% of ECGs were interpreted incorrectly, 6% of those cases were determined to be clinically important errors	•	Inability to identify atrial fibrillation or flutter and supraventricular tachycardias accounted for the majority of misinterpreted rhythms
*Eberle et al. 1996	None Specified	None Specified	Pulse Check	Simulation	35% of cases were misidentified	•	Lack of intensive training specifically for pulselessness
Linn et al. 1997	None Specified	None Specified	Air Evac Diagnosis	Retrospective Record Search	23% of diagnoses were missed by flight physicians and 74.5% of those could have been made by the flight physician	• • •	Sever injuries and stabilizing the patient may distract from determining all diagnoses Missed measuring blood pressure or temperature Head or spine injuries can mask others
Smith et al. 1998	None Specified	Stroke	Diagnosis	Retrospective Record Search	Paramedics identified a stroke/TIA that was incorrect 23% of diagnoses and did not diagnose 39.5% of patients later diagnosed with stroke/TIA	• •	Documentation deficiencies Lack of training in identifying stroke symptoms
Haynes and Pritting 1999	None Specified	None Specified	None Specified	Retrospective Record Search	25% disagreement	• •	Limited accuracy in treating respiratory distress Low utilization of intravenous line
*Ruppert et al. 1999	None Specified	None Specified	Breath Check	Simulation	19% misidentification of breathing	•	Lack of training for participants except EMS
Eckstein and Suyehara 2002	None Specified	Congestive Heart Failure	Diagnosis	Retrospective Record Search	42% of patients diagnosed by EMS as something other than CHF later determined to be CHF	•	Difficulty determining CHF in elderly or debilitated patients that are dehydrated or have been administered diuretics and nitrates (Continued on next page)

# Table 3.5: Diagnosis Accuracy

None Specified	<ul> <li>of Stringent inclusion criteria for prehospital thrombolysis for safety reasons</li> <li>Patients not presenting the necessary ECG criteria in the early stages of the infarct</li> </ul>	<ul> <li>edic Resolution of signs between</li> <li>ign assessments</li> <li>Paramedics not considering or excluding hypoglycemia as a stroke mimic</li> </ul>	None Specified	<ul> <li>ty • Wide range of diagnoses with difficulty breathing symptom</li> <li>Difficulty examining chest noises ther</li> </ul>	<ul> <li>Tendency to base the assessment on mechanism of injury</li> <li>Delay of presentation of symptoms</li> <li>Poor neurological assessment</li> </ul>	None Specified	(Continued on next page
14% of cases were misdiagnosed	Paramedics incorrectly identified 8% patients, but 0 cases were due to paramedic error	Complete agreement between parame and physician for each neurological s was: facial weakness, 78%; arm weakness, 98%; and speech disturban 89%	6% of patients were incorrectly diagnosed	Paramedics achieved 86.4% sensitivit and 86.6% specificity for cardiac diagnoses, 71.4% and 93.6% for respiratory, and 82.1% and91% for ot	Overassessment of the patient by the transport team occurred in 27 cases (1.9%), and underassessment occurre 18 cases (1.3%)	Cardiologist did not agree with paramedic diagnosis in 10% of cases	
Retrospective Record Search	Simulation	Retrospective Record Search	Observation	Retrospective Record Search	Retrospective Record Search	Retrospective Record Search	
Diagnosis	Determining thrombosis candidacy	FAST assessment	Diagnosis	Diagnosis	Diagnosis	Diagnosis	
Aortic Emergencies	Cardiac Arrest	Stroke/TIA	STEMI	Difficulty Breathing	None Specified	STEMI	
None Specified	None Specified	None Specified	None Specified	None Specified	None Specified	None Specified	
Mitchell and Tallon 2002	Pitt 2002	Nor et al. 2004	Feldman et al. 2005	Ackerman and Waldron 2006	Benner et al. 2006	Cantor et al. 2012	

Williams et al. 2013	None Specified	Acute pulmonary edema	Diagnosis	Retrospective Record Search	71% of patients with APE were missed by paramedics and 59% were misdiagnosed with APE by paramedics	• •	Limited patient history No radiography or echocardiography
Gropen et al. 2014	None Specified	Stroke	Diagnosis	Retrospective Record Search	Of the stroke patients, 28/56 (50%) were missed by EMS (false negatives) and 9/254 (3.5%) were falsely considered to have stroke by EMS (false positives)	• • •	Stroke screening tool used Lack in stroke specific training Inadequate documentation of screening scale
Brandler et al. 2015	None Specified	Stroke	Diagnosis	Retrospective Record Search	38% of stroke patients were missed and 0.3% of patients were incorrectly identified as having a stroke	•	Strokes presenting with more subtle or nonspecific presenting symptoms
Williams et al. 2015	None Specified	Asthma or COPD	Diagnosis	Retrospective Record Search	paramedic diagnosis did not match ED for 59% of asthma patients and 43% of COPD patients	• • •	Lack of clear criteria to identify the diseases No verification of patient medical history Limited diagnostic facilities in the prehospital setting
Christie et al. 2016	None Specified	Dyspnea	Diagnosis	Retrospective Record Search	The overall agreement between paramedic and either ED or hospital diagnosis was 186/293, 64%	• •	Shortness of breath is common to many conditions making it difficult to determine exact diagnosis Data form only identified Chief complaint and not working diagnosis
Latimer et al. 2018	None Specified	Anaphylaxis	Medication decision	Retrospective Record Search	12% of cases paramedic decision did not match physician	•	None Specified
Duignan, et al. 2018	None Specified	None Specified	Tourniquet use	Retrospective Record Search	9.5% of cases tourniquets were not appropriate	•	"shock factor "may have contributed to the decision to apply tourniquets
* Also addressed Emergency Dena	procedure error	s. APE: Acute Puli mergency Medical	monary Edema, C Services FMT	CHF: Congestive He Emergency Medical	art Failure, COPD: Chronic Obstructive Pulmona Technician STFMI ST-elevation Muccardial in	ary Dise	ase, ECG: Electrocardiogram, ED:

Possible Causes	Authors cited high decision density for EMTs as cause for high error rate Providers have limited insight into their own error patterns and do not inform others	Lack of support to identify errors and poorly trained to identify and disclose errors	Lack of standardized training or inadequate training for pediatrics	Communication issues Long hours and complex tasks	Communication deficits No consensus on diagnosis	None Specified	Insufficient training Inattention or thoughtlessness
	• •	•	•	• •	• •	•	• •
Error Occurrence	55% of EMT respondent reported making at least one error in the last year	45% of EMT respondent reported making at least one error in the last year, and while only 7% of the time severe errors, in mild errors more than 50% of errors were identified in treatment administration and failure to	61 events were ported in 6 months, 44% were near misses, 56% were adverse events	Adverse events occurred at a rate of 11.53 per 1,000 air ambulance flights, 21% of those were due to EMT performance, of those 25% could have caused patient harm	More than 7 unsafe acts occurred per scenario, these were more common in trauma and bronchial asthma care scenarios. The highest rate of unsafe acts were during respiration acts at 13% of all twose of acts	An average of 2 incidents were identified per dispatch, 71% of incidents were due to patient management or technique. 10% of incidents were maior or deadly	65% of incidents recorded were due to EMT error and mostly preventable
Method	Survey	Survey	Survey and Interview	Retrospective Record Search	Simulated Ambulance Room	Survey	Survey
Procedure	None Specified	None Specified	None Specified	None Specified	ALS, Asthma, Pulmonary Embolism, Trauma Care	None Specified	CPR
Condition	None Specified	None Specified	None Specified	None Specified	None Specified	None Specified	Cardiac Arrest
Population	None Specified	None Specified	None Specified	None Specified	None Specified	None Specified	None Specified
Citation	Hobgood et al. 2004	Hobgood et al. 2006	Fairbanks et al. 2008	MacDonald et al. 2008	Zimmer et al. 2010	Stella et al. 2010	Hohenstein et al. 2011

# Table 3.6: Adverse or Safety Events

Jallagher and Kupas 2012	None Specified	None Specified	None Specified	Retrospective Record Search	45% of errors reported were related to human judgement or behavior	None Specified
<sup>2</sup> atterson et ıl. 2012	None Specified	None Specified	None Specified	Survey	41.1% of respondents reported at least one error or adverse event in the previous 3 months	<ul><li> Poor sleep quality and fatigue</li><li> High workload during shifts</li></ul>
Weaver et al. 2012	None Specified	None Specified	None Specified	Survey	42% of respondents recorded an affirmative response for EMS-SI items measuring error or adverse event, 89% respondents reported safety-	None Specified
Hohenstein et al. 2014	None Specified	None Specified	None Specified	Survey	78% of incidents were caused by staff lack of knowledge or experience, communication or technique	<ul> <li>Insufficient techniques, wrong diagnosis, and underestimating severity</li> </ul>
van den Berg et al. 2015	Neonatal	None Specified	None Specified	Retrospective Record Search	290 of 833 (35%) Adverse Events were human error related	<ul> <li>Poor control of multiple services helping during transports</li> <li>Continual equipment failure</li> </ul>
Hansen et al. 2016	Pediatric	None Specified	Airway Management	Retrospective Record Search	27.8% of patient cases had errors	<ul> <li>Cardiac arrest significantly increases odds of an airway management error</li> <li>Multiple critical decisions and interventions performed quickly</li> </ul>
Duby et al. 2018	Pediatric	None Specified	None Specified	Retrospective Record Search	2.6 safety events occurred per patient with a range of 0 to 6, 73% of patients had one or more safety event, 38.5% had a severe event	<ul> <li>Used techniques from wrong resuscitation protocols</li> <li>Failure to obtain IV/IO access</li> </ul>
Meckler et al. 2018	Pediatric	None Specified	None Specified	Retrospective Record Search	at least one UNSEM occurred in 69.3% of transports, 33% of these were errors	<ul> <li>Infrequent and insufficient training on pediatric resuscitation</li> <li>Lack of standardized drug concentrations and packaging</li> <li>Difficulty estimating weights and calculating does not</li> </ul>
ALS: Advanced Unintended injut	Life Support, CPR v or consequences	: Cardiopulmo	nary Resuscitation Suboptimal actions	, EMS- SI: Emerger	ncy Medical Services – Safety Inventory, IO: Int gement complications	and curvature acceded assecous, IV: Intravenous, UNSEM:

UK and the US. In the US the most commonly investigated error type was diagnosis accuracy and adverse or safety events following in frequency. The relative amount of studies in each error type did not vary significantly by location.

#### Method

A variety of methods was used to determine the errors made in emergency services in these studies: In person observation, simulation of tasks, surveys or interviews of care providers, and retrospective record search. Most (56%) utilized a retrospective record tool to collect data, reviewing hospital or emergency service patient records to determine medication or procedure errors, adverse events, or departure from procedure. Past research has established that medical errors typically exhibit a low reporting level (Nguyen et al., 2005; Rowin et al., 2008), which appears to be the one most frequently cited in the available literature. Perhaps because of this phenomenon, several studies discussed the use of anonymous reporting systems to capture errors as part of an effort to ensure continuing improvement of the medical systems (Okafor et al., 2015; Stella et al., 2010; Westfall et al., 2004). Only 7% of studies included in this review used observational methods to determine the occurrence of errors in real time. Surveys and interviews were used in a few studies (14%) to obtain perceptions of errors, the frequency errors occurred per individual, or the accuracy of identifying potential errors. An interesting method introduced in some studies (22%) was simulation through the observation of an emergency care scenario in either a parked ambulance or in a simulated environment. This method is frequently used for educational training (McLaughlin et al., 2002; Small et al., 1999) and to complete observation studies particularly in areas such as medical services where observation of real patient care can be difficult to complete (Bond & Spillane, 2002; Hunt et al., 2008). Barriers to observation in patient care could be due to privacy concerns, patients withholding consent for observation or the loss of confidentiality, or the safety of the patient. Another barrier is logistics as in many emergency care scenarios, the stress involved in stabilizing the patient impairs the ability of researchers to observe or interact with caregivers, and in ambulances space is limited. Simulated observation using patient actors or patient mannequins can provide valuable data without the risks associated with live observation of patient care. The methods used to determine error occurrences varied by type of error. Protocol deviations, for example, were determined using mostly simulation (50%), then retrospective reporting (37%), then observation (13%). Using simulation to measure protocol deviations is expected as determining adherence to protocol in an emergency scenario may be difficult as previously discussed and from surveys or interviews would be skewed by bias. Diagnosis accuracy is similarly measured by simulation, retrospective reporting, and observation; however, the use of retrospective reporting is much higher (81%) as diagnoses are commonly documented at each stage of patient care and as such can be simply retrieved from records rather than simulated or realtime observation.

## **Population and Condition**

Almost half of the studies included in this review did not specify the population or the condition studied (48%). Of the studies that did specify outside of general EMS care, 13 studies (19%) limited their population to pediatrics, children under the age of 12, or neonatal. However, it was more common to limit a study by condition (47%); while a wide variety of conditions were considered, most (57%) were cardiac- or stroke-related (chest pain, cardiac arrest, ST-Elevated Myocardial Infarction or STEMI, stroke symptoms). There were also some studies that isolated both a population and a condition (10%) focusing mostly on pediatric cardiac conditions (71%). All studies investigating diagnosis and procedure errors did not limit their population to pediatrics and of the studies that focused on medication errors almost half of them were specifically focused on pediatric care (45%). 75% of all studies of diagnosis accuracy and 63% of all protocol deviation studies specified a condition. This is expected as isolating a diagnosis or identifying an existing protocol to check deviations would be helpful in limiting the scope of an experiment.

## **Possible Causes**

Most studies offered some discussion of possible causes for the errors reported or observed, ranging from causes within the types of errors to those related to conditions or procedures. Generally, possible causes listed concerned training, experience, cognitive demands, oversight, communication, documentation, support, protocols, skill or knowledge, fatigue, and various situational causes (patient movement, symptom masking, pre-existing conditions or medications). The full list of possible causes for each study can be found in the last columns of Tables 2-6, but there were some common causes of error among studies. In at least 43% of the studies, lack of training, knowledge, experience, or understanding was cited as a possible cause for error across all error reporting types. Another common possible cause cited was lack of supporting materials such as checklists or references for certain procedures, including Cardiopulmonary Resuscitation (CPR), intubation, medication dosing, and Intravenous/Intraosseous (IV/IO). Some studies (10%) did not offer any reasons for the errors found; most of these were conducted using a survey or a retrospective record search, methods that limit the ability of researchers to determine the causes of errors as they are done remotely and after errors have occurred. However, many studies using the same methods were able to provide some information concerning why errors were made.

#### DISCUSSION

A review of the literature on error reporting in EMS finds relatively few papers quantifying the frequency and type of errors that occur in the prehospital domain, a situation that may be due to several factors. As mentioned previously, error reporting is known to be limited, especially in the medical field, because of the liability errors represent and a culture of blame rather than continuous improvement (Fairbanks et al., 2008). It may also be difficult to measure errors observationally because of issues such as the Hawthorne effect, a phenomenon in field research that occurs when participants feel they are being observed such that they deviate from normal behavior under observation, (McCambridge et al., 2014) or because of the fast-paced, high-stress environment with little space for inperson observation. Although there are in-depth systematic reviews of studies of patient safety in EMS (Bigham et al., 2012) and research into the recommendations to reduce error through auditing procedures (O'Connor et al., 2002), literature identifying the types of errors and causes is limited, and a systematic review of research specifically of human error in EMS was not found in the search for studies for this article.
## **Types of Error**

Though few studies were found for this review, the error occurrence reported among them is considerable and varies widely within error types and methods. Error reporting or observation in these studies fell into 5 categories: procedure errors, protocol deviations, medication errors, diagnosis accuracy, and adverse or safety events.

#### **Procedure Errors**

Procedure errors found in these studies primarily focused on resuscitation, specifically intubation, with many studies identified in early stages of the review focusing on this medical process but not identifying errors related to the patient. Only 25% of studies did not focus on resuscitation (Alenyo et al., 2018; Brice et al., 2008; Williams et al., 2008). Misplaced intubation tube, incorrect tube selected, or overall intubation failure can affect the patient's breathing ability, but it can also impact the paramedic or the receiving emergency physicians and nurses as their treatment of the patient may be changed by the placement or success of the intubation tube. Failure to intubate can be attributed to factors other than the skill of the medical providers such as patient movement, age, obesity, or unconsciousness (Cook & MacDougall-Davis, 2012; McDermott et al., 2005; Moon et al., 2013). Some studies cited these difficulties as causes for error, others a lack of skill, training, and tools to support this complex procedure (Ghiyasvandian et al., 2018; Jones et al., 2004; Kaserer et al., 2017; Wang et al., 2006). One intervention suggested by a study in this review (Barnes et al., 2003) that shows promising results for difficult airways is an altered facemask called the LMA-Fastrach (Ferson et al., 2001). Changing the design of the tool used creates intervention at the highest level of hazard control priority, a technique

to minimize hazards or accidents through the tiered use of system design to eliminate the hazard, system design for safeguards, and finally training or organizational control (Tweedy, 2005). Another supporting tool suggested in the literature (Jemmett et al., 2003) was end-tidal CO2 monitors to determine correct placement or placement detectors (Bhende et al., 1992; Foutch et al., 1992). Some of these devices have also been tested with paramedics (Donahue, 1994). However, other articles in this review provide support for instituting guidelines or protocols to prevent misplaced or failed intubations (Hein et al., 2008; Jones et al., 2004). New monitoring devices are not the only intervention that could be used at the top hazard control level. Telemedicine systems have been used to connect prehospital providers to information sources, provide support for procedures, and allow rich communication with specialists (Barrett et al., 2016; LaMonte et al., 2004; Pavlopoulos et al., 1998; Xiao et al., 2000). Connection to an airway specialist could improve performance in this complex procedure. This use of telemedicine support for emergency responders has been evaluated. A study comparing a control group using a physician in the room and an intervention group with telemedicine consultation found that the performance of the latter was comparable to the control group and that the telemedicine group collected allergy and current medications more frequently with significant p values of 0.002 and 0.004, respectively (Rörtgen et al., 2013). Furthermore, an investigation of the use of a telemedicine system to support prehospital providers in complex simulated procedures found that the survival rate of the teams with telemedicine connected physicians was significantly higher than for the control group (2 patients deaths for the telemedicine group compared to 16 in the control group) and that performance measured by the correct identification of pathological signs, processes and appropriate interventions was significantly higher in the telemedicine group compared to the control (96%, 98%, 92% versus 79%, 75%, 49%) (Charash et al., 2011). Studies investigating both the technique used in or performance of these tasks as well as the diagnostic accuracy of prehospital providers are found in both the procedure error and diagnostic accuracy themes.

## **Protocol Deviations**

Studies in the protocol deviations focused on care that did not meet or deviated from guidelines, protocols, or checklists used to describe tasks in EMS. Deviations from protocol can involve missed steps or steps completed incorrectly and in either case the coordination of care tasks within the ambulance can be affected and it could also impact how the patient is treated at a hospital. For example in some healthcare systems the protocol for a suspected stroke is to gather blood samples in the ambulance when an IV is placed such that those samples can be processed at the emergency department lab upon arrival, if this step is not completed then that task is delayed and a nurse will have to gather blood from a different vein. The protocols studied in this review varied from general care to cardiac arrest or vehicular trauma to more specific tasks like cardiopulmonary resuscitation (CPR) or blood pressure measurement. The most common type of protocol in the literature was that applied to cardiac arrest symptoms (Kirves et al., 2007; Lammers et al., 2009, 2014b). Most studies referenced implementing changes at the lowest hazard control priority: improving guidelines for divisions of tasks (Lammers et al., 2014b), targeted training or increased practice (Lammers et al., 2009; Liberman et al., 1999; Salerno et al., 1991), or standardization of protocol (Boyle, 2009). One study in this review created a list

of suggestions specific to the tasks monitored in the study that included training and organizational changes, but also safeguards and system changes such as labelling IV bags and medications, developing an EMS specific Broselow tape, and eliminating the use of certain syringes (Lammers et al., 2012).

#### Medication Errors

Medication errors in this review were primarily inappropriate or incorrect dosing of epinephrine, adenosine, or atropine (common medications with adverse effects at incorrect dosages) or errors in dosing calculation and application tests. Again, these errors mostly affect the patient, but the available medications or further dosing of these medications used in later treatment could also affect nurses' and physicians' processes. Many of the remediations suggested in the pediatric medication error literature (45% of studies in this theme) focuses on the use of Broselow tape, a measurement tool made to establish an estimate of children's weight by size to determine appropriate common medication dosing (DeBoer et al., 2005). Another study in this review suggested a safeguard approach to hazard control by supporting use of Broselow tape with color coded syringes, which eliminates any further errors of administration (Stevens et al., 2015). Other interventions suggested in this theme at the lowest hazard control priority were training and practice (Hoyle et al., 2012; Lammers et al., 2014a; Lim et al., 2013) or changes to protocol (Kaji et al., 2006). Telemedicine systems integrated with electronic health records (EHRs) could assist with medication dosing as a patient's EHR would contain his/her weight and any allergy information which can be used to improve dosing procedures. Telemedicine systems could also be integrated to support dosing calculations based on

predetermined formulas as some studies in this review suggested support for error prone calculations (Hubble et al., 2000; Lammers et al., 2014a; Vilke et al., 2007).

# **Diagnosis** Accuracy

Diagnosis accuracy was a significant theme in this review, with 32% of all studies evaluating human error in this way. Studies in this theme detailed the accuracy of prehospital providers in diagnosing strokes, congestive heart failure, acute pulmonary edema, and STEMI compared with emergency department (ED) or hospital discharge diagnosis for prehospital providers ability to determine differences in conditions such as asthma or chronic obstructive pulmonary disease (COPD). Studies in this theme focused on prehospital provider ability and possible causes for misidentification; however, it should be remembered that many of these conditions are difficult to diagnose without support from x-ray or computed tomography (CT) scans or lab results that EDs have access to. In almost 20% of studies possible causes for an incorrect diagnosis included certain injuries, symptoms, seizures or head and spine injuries that could mask others, making a correct diagnosis difficult (Ackerman & Waldron, 2006; Brandler et al., 2015; Christie et al., 2016; Linn et al., 1997). Other studies cited lack of experience and even shock as factors that could create inappropriate diagnosis (Duignan et al., 2018; Gropen et al., 2014; Kothari et al., 1995; Smith et al., 1998). In these cases, consultation with stroke or trauma specialists through telemedicine could assist in diagnosis especially when there is uncertainty in determining procedures and treatment in difficult cases. Additionally, having a telemedicine system that assists medics with completing demographic data collection and merging the new data with existing electronic health records (EHRs) could improve the

ability of prehospital providers to diagnose patients accurately as it would allow for access to all health records and for connecting information from this care instance to the patient profile (Frank & Pape-Haugaard, 2011; Handel & Hackman, 2010; Moloney et al., 2017). Time to diagnosis has also been seen to improve with the use of telemedicine, with one study completing telemedicine diagnosis en route with on-call physicians at a local hospital 18 minutes sooner than the in-hospital diagnosis and that the prehospital diagnosis had a positive predictive value of 56% (Terkelsen et al., 2002). Yperzeele et al. (2014) investigated the use of a traditional ambulance equipped with bidirectional audio and video and automatic transmission of vitals, finding that the diagnoses developed with teleconsulted physicians reached high agreement with final in-hospital diagnoses (0.98), with a high kappa statistic for differencing neurological from non-neurological diseases (0.92).

# Adverse or Safety Events

The final theme, adverse or safety events, was developed from studies that classified human error but typically defined an adverse event as that which could potentially harm the patient. Although such studies are common in the literature, only those that further isolated the events due to the technique, skill, or behavior of the prehospital provider were included in this review. This theme is defined by how general errors are measured. In 63% of studies no specific procedure, population, or conditions were identified to limit the scope. In the remaining 37% most (83%) studied focused on pediatrics. Given the general nature of these errors, interventions provided by individual studies in the literature were not focused on system design. The most common suggestion

was that of increased training and experience for specific procedures (Duby et al., 2018; Fairbanks et al., 2008; Matthew Hansen et al., 2016; Hohenstein et al., 2011; Meckler et al., 2018), however these interventions are at the lowest hazard control priority. Few studies identified organizational changes through standardization of protocols or documentation (Fairbanks et al., 2008; van den Berg et al., 2015; Weaver et al., 2012) or design safeguards through the poka-yoke technique, preventing errors despite intentions or thoughtlessness such as pre-filled syringes or labelling (Hansen et al., 2018; Hohenstein et al., 2011). A more commonplace organizational change was that anonymous error reporting could enhance caregivers, healthcare providers, and policy makers knowledge of the type and preceding conditions errors in emergency medicine (Gallagher & Kupas, 2012; Hobgood et al., 2004, 2006; Hohenstein et al., 2011, 2014; Stella et al., 2010). Providing an open resource for communicating errors and understanding when errors occur can create an awareness for individuals of their own errors. Lack of error awareness in emergency medicine was discussed by studies in this theme which could be supported by telemedicine. Telemedicine systems provide a level of oversight and observation by nurses and physicians that could increase error reporting and make providers more aware of their error patterns, which was a concern for a study in the protocol deviations theme as well (Cienki & DeLuca, 2012).

# **Occurrence** of Error

Overall there was no consistency among error occurrences in the studies selected for review, even within error types. The magnitude of error occurrences in many studies were large, for example missing 33% to 54% of the procedural steps required (Lammers et al., 2009, 2012, 2014b), dosing error occurrences in up to 54% of the cases (Coppler et al., 2016; Eastwood et al., 2012; Goebel et al., 2004; Hoyle et al., 2012; Hubble et al., 2000; Kaji et al., 2006; Lammers et al., 2014a; Lifshitz et al., 2012; Lim et al., 2013; Stevens et al., 2015), and incorrect diagnosis in up to 71% of cases (Ackerman & Waldron, 2006; Brandler et al., 2015; Cantor et al., 2012; Christie et al., 2016; Eberle et al., 1996; Eckstein & Suyehara, 2002; Gropen et al., 2014; Haynes & Pritting, 1999; Hollander et al., 1995; Kothari et al., 1995; Linn et al., 1997; Nor et al., 2004; Ruppert et al., 1999; Smith et al., 1998; Williams et al., 2013, 2015). There were some studies that cited relatively low error occurrences: 4% of cases with misuse of AED (MacDonald et al., 2001), 0 - 14% paramedic misdiagnosis (Benner et al., 2006; Feldman et al., 2005; Latimer et al., 2018; Mitchell & Tallon, 2002; Pitt, 2002), 5.8% cases with misplaced intubation tubes or 0.3% failed intubation cases (Hein et al., 2008; Jones et al., 2004). However, the majority of studies in this review had errors in over 20% of cases analyzed.

# **Possible Causes**

Causes ranged from organizational factors such as fatigue or shift length to those that are more easily preventable such as lack of contact with physicians, lack of support or visual aids, dosing miscalculations, or lack of training. Many of the errors could be mitigated or even prevented with improved training in medical areas like pediatric care or stroke, visual aids for difficult procedures such as endotracheal intubation, or the use of a sophisticated telemedicine system to support providers in prehospital care. The SEIPS 2.0 model can provide some context for the causes of these errors identified in the review. Many studies cited lack of experience, training or skill as a cause for error which is identified within the person and task work system factors in the model. Situational causes can be linked to both the internal environment (movement of the ambulance) and task (inherent difficulty of a task). Lack of oversight, support, or standard protocols can be linked to the external environment or organization. Documentation issues can be linked to tools and technology. All these factors are connected in the SEIPS model meaning the factors impact each other, for example a task factor such as difficult tasks or person factor such as limited experience or training can be impacted by organization or external factors such as protocols or support from specialists. These work factors and their interactions make up the work processes which in turn have measurable outcomes. These outcomes are the results analyzed in the studies in this review which provide insight on the work processes and work system factors in the SEIPS 2.0 model. Interventions discussed in the studies in this review can be understood through the SEIPS 2.0 model by how they impact the work system factors. Improvements in protocols impacts the organization as demonstrated by factor interactions will also impact the person and task factors. An Anonymous error reporting system is an intervention of the tools and technology factor, but it impacts the organization culture and the external environment by creating accountability and awareness of errors and consequences. Telemedicine systems were also suggested interventions. Telemedicine systems use telecommunication technologies, video conferencing and real- time data tracking to support connections between patients or health care providers and specialists or other providers when separated by geographical distance. These systems have been implemented in ambulances with mild success in reducing diagnosis and treatment times and increasing positive task completion (Rogers et al., 2017). The telemedicine interventions discussed in the previously can also be understood within the context of the SEIPS 2.0 model. It is an intervention in tools and technology, but it impacts the task (by providing connection to specialists to support difficult procedures), internal environment (by adding new technology elements to the environment), and person (by expanding the caregivers involved in the work system in the ambulance). Telemedicine also supports teamwork and rich communication with audio and visual connection that provides more detail and earlier in the caregiving process which was cited as a possible cause for error in this review (Lammers et al., 2014a; Russell D. MacDonald et al., 2008; Zimmer et al., 2010). This enhances the collaborative work process and previous research has shown the improved patient outcomes as a result of connecting caregivers through telemedicine.

# **Systematic Review Limitations**

Few articles are available for this systematic review of the literature on human error in ambulance-based care, perhaps due to the factors previously mentioned concerning reporting rates and publication bias in which non-significant results are less likely to be published, and the exclusion of the non-English articles. This review considered only experimental or observation studies measuring error occurrences; reviews, commentaries, or studies that focused only on methods for measuring error in ambulances without data were not included. In addition, studies that did not classify errors in such a way that identified human errors were also not included in this review. As mentioned previously, reporting rates of human error are low, so this may bias the data of retrospective reporting studies.

#### **Directions and Future Work**

Many interventions are proposed in the studies reviewed and above, future studies are needed to assess how these interventions could be integrated into EMS systems that could function efficiently in the high-workload, fast-paced environment of an ambulance. Considerations of these interventions as to how they impact individual caregivers as well as the care process and patient safety should be investigated. Additionally, in at least one study, workload and fatigue were noted as possible causes for adverse and safety events. Studies have been conducted on the stress of prehospital providers, but based on a previous review, none has been conducted including a telemedicine system to determine the workload and workflow with this system (Rogers et al., 2017). Research needs to be conducted to determine how prehospital providers currently work with telemedicine systems before new features are added to prevent errors. Additional functionality without consideration for how these functions would fit into workflow or effect workload is inadvisable until their impact is fully researched and understood.

These studies should also determine how telemedicine systems themselves could cause error, a topic not covered by the current research. Careful consideration of their design for optimal workflow and protection of patient and provider safety and confidentiality is important in these systems. Some telemedicine systems take the format of simple video conferencing systems, while others include more functionality such as multiple viewpoints of the ambulance and patient or transfer of vital signs and ECGs. While the more complex systems provide more functionality, the work environment and the operator must be considered in both the design of the interface and the placement in the ambulance. Off-the-shelf systems for video conferencing may be suitable for home telemedicine or a starting point for ambulance telemedicine, but an entire systems approach is needed for these telemedicine systems to be effective in supporting physician and prehospital provider care for patients and in reducing errors.

## CONCLUSION

This review systematically investigated the errors made by prehospital providers in ambulances via a variety of methods. The reviews, though few, generated established types of errors found in ambulance-based care: procedure errors, protocol deviations, medication errors, diagnosis accuracy, and adverse or safety events. Though interventions to mitigate error were identified in some studies, inferences from the SEIPS 2.0 model were made to develop suggestions for mitigations to common causes of human error described in this review. This literature explored human error, provided error occurrences, and frequently described the causes of errors. Further research is needed to determine how to effectively minimize these errors through system design, environment design, or training to improve patient safety in the prehospital environment.

70

#### CHAPTER FOUR

# ANALYSIS OF AN AMBULANCE-BASED TELEMEDICINE-INTEGRATED SETTING FOR STROKE CARE

#### **INTRODUCTION**

## **Telemedicine in Stroke Care**

Strokes are one of the most serious healthcare issues in the United States: they represent the fifth leading cause of death, resulting in 1 death about every 4 minutes and annually cost \$34 billion in health care, medication, and lost work (Benjamin et al., 2017; Heron, 2019). To improve stroke outcomes, healthcare professionals seek to minimize the time to treatment to restore blood flow to the affected tissue (Saver, 2006). Strokes caused by a blood clot (ischemic strokes) result in brain tissue death as blood cannot circulate to the affected area. The earlier patients receive treatment, the more likely they are to have improved clinical outcomes and reduced long-term disability and treatment complications (Saver et al., 2013). Telemedicine, the use of telecommunications technology to remotely connect trained specialists to patients, has been recommended by the American Heart Association and the American Stroke Association to improve stroke care (Schwamm et al., 2009). Indeed, research shows that telemedicine has improved patient triage, increased accuracy of diagnosis, and most importantly, reduced time to treatment (LaMonte et al., 2004; Rogers et al., 2017). The quick access to stroke specialists or neurologists afforded telemedicine improves the process of correctly diagnosing a stroke (Barrett et al., 2016; Terkelsen et al., 2002). Telemedicine further coordinates receiving hospitals to allow for expedited care as ambulance-based telemedicine can permit stroke consultations en-route. This can decrease time between first provider contact and definitive therapy, as specialists can rapidly identify and triage stroke patients.

During prehospital stroke consultations, telemedicine helps coordinate patient care by a remote neurologist, the nurse at the receiving hospital and the paramedic in the back of the ambulance. A small rural county in South East United States in conjunction with a tertiary care hospital has implemented a pilot telemedicine system called REACH in their emergency department and ambulances which facilitates earlier contact with a neurology specialist, which the rural county hospital does not have access to on site. The REACH telemedicine platform serves two functions: as a tri-directional visual communication with the secondary hospital, the tertiary hospital, and the ambulance and as a tool to document information about the patient case (Adams et al., 2012). All caregivers have the same interface in REACH, which includes a view of the patient in the ambulance, the hospital, and the neurologist, and an information screen where patient history, vitals, assessment items, imaging, and other information can be accessed. A screenshot of the telemedicine system from the neurologist's view is shown in Figure 4.1.

The typical stroke care process with ambulance based telemedicine includes an Emergency Medical Service (EMS) doing a quick assessment of the patient and alerting a receiving hospital of the incoming patient and their information. Then once in the ambulance and connected to the telemedicine system a neurologist can conduct a detailed assessment until the patient arrives to the Emergency Department (ED) and admitted.

72



Figure 4.1: REACH interface (Participants' faces masked for confidentiality)

Neurologists awareness is of paramount importance for this system to be effective. The neurologist must gather information from the patient and paramedic to maintain an awareness of patient status and develop a diagnosis and care plan. Development and maintenance of this awareness is necessary specifically for ischemic strokes, which account for 87% of all strokes, as they are treated with a life-saving treatment referred to as a tissue plasminogen activator (tPA) (Benjamin et al., 2017). This treatment has multiple conditions and contraindications affecting its use, the most common being that the treatment must be given within 0 to 4.5 hours of stroke symptom onset (Powers et al., 2018). As such an awareness of the time from the symptom onset must be maintained. Time is not the only factor that neurologists must be aware of, a hemorrhage, high blood pressure or the use of blood thinners all prohibit a patient from receiving tPA (Fugate & Rabinstein, 2015). In addition, mimics for strokes such as low blood sugar or seizures must be ruled out.

Telemedicine has been researched extensively as a tool to improve time to treatment and ultimately patient care outcomes (Rogers et al., 2017); however, the current research lacks an understanding of the effect of this system on the caregivers involved. The objectives of this study are to (1) understand the tasks needed to complete a stroke assessment using telemedicine in an ambulance and how those tasks are completed; (2) determine the usability issues in the telemedicine system and the possible errors that could be made; and (3) provide recommendations to improve the telemedicine design to mitigate usability issues and errors. To understand how these recommendations impact the work system, work processes and ultimately outcomes in the healthcare system, the Systems Engineering Initiative for Patient Safety (SEIPS) 2.0 model was used as method to accomplish the third objective in addition to the heuristic evaluation and SHERPA. This model describes the work system factors in a healthcare system, how they interact to create work processes, how work system factors and work processes impact outcomes, and how outcomes feed back to the work processes and the work system (Holden et al., 2013). To accomplish these objectives, a Hierarchical Task Analysis was completed and used as input for a heuristic evaluation and in a Systematic Human Error Reduction and Prediction Approach (SHERPA). It is crucial to analyze how the telemedicine system is used by caregivers because the caregivers in this work system ultimately determine patient safety and patient outcomes. They need tools that support their processes easily and without causing additional human error. The analyses completed in this study were selected to investigate the tasks, human error, and usability because of their inexpensive cost and rich data results, as well as the prevalence of their use in the human factors field and the healthcare domain. In addition, SHERPA has been validated and found to be one of the most effective methods for determining human error in complex work systems (Kirwan, 1992; Salmon et al., 2002). Subsequent sections detail the background for this study including the effects of human error in stroke care and the approaches used to accomplish the objectives of this study; further sections detail the data collection and analysis methodologies, the results of the analyses, and a discussion of the results and recommendations for design based on the analyses.

## Background

There is potential for human error in almost all healthcare delivery as evidenced by the reports from the Institute of Medicine (Institute of Medicine (US) Committee on Quality of Health Care in America, 2014). Several studies have been conducted on prehospital care and human error within stroke care (Brandler et al., 2015; Kothari et al., 1995; Nor et al., 2004; Smith et al., 1998), and there have also been studies on the use of telemedicine to improve stroke care (Barrett et al., 2016; LaMonte et al., 2004). However, there is no current research studies that has investigated potential errors using a telemedicine system with a focus on the caregiver. This lack of telemedicine research in stroke care focusing on the caregivers in the process and the ways to support their use of the system needs to be addressed given the importance of preventing usability issues and error in this domain. This research is doubly important in emergency medicine, specifically in telemedicine-integrated ambulance based environments as the use of a telemedicine system in such a constrained, high-stress environment may result in errors on the part of the health-care providers given the cognitive, physical, and temporal demands of this environment (Goldberg et al., 2017; Hobgood et al., 2006; Mould-Millman et al., 2018; Patterson et al., 2012; Rogers et al., 2017).

The consequences of error in medicine vary by procedure or patient from insignificant to significant harm and even death. Errors in determining if a patient is having a stroke and its severity can delay patient care, impacting the likelihood of improved outcomes. The potential for critical errors in prescribing tPA is especially high given the many conditions tied to its use. Telemedicine can help address this issue by providing access to specialists with the best training to assess these patients; however, the complex tasks involved and the interactions between team members in a stroke assessment make this process at risk for human error. Potential errors for nurses who administer the neurologist recommended tPA involve dosage calculation, mixing the medicine and intravenous administration. In addition, even after the neurologist recommends tPA, changes in blood pressure may contraindicate administration. The nurses administering the medication and the neurologists are responsible for ensuring that the blood pressure is within an acceptable range. Minor errors in this work system including incorrect history or patient information, misspelled medications, and the inability to locate important information could disrupt the patient care workflow and result in prolonged treatment times. Thus, it is vitally important that the work system be analyzed to determine where and how errors could occur. Insights provided by such an investigation can pinpoint the causes and allow designers to improve the system to prevent these errors, or they may highlight changes in the training and organization to improve system awareness, thus reducing the potential for errors.

Many human error investigation techniques are available for predicting human error in a process in any given domain, however, the consensus among researchers is that the Systematic Human Error Reduction and Prediction Approach (SHERPA) is more effective for exploring human error when compared to other tools like HEIST and Human Error HAZOP (Kirwan, 1992; Salmon et al., 2002). SHERPA was originally created for use in the nuclear power domain (Embrey, 1986), but has been used widely as a human error investigation tool in many other domains (Guarascio et al., 2019; Khandan et al., 2017; Stanton et al., 2002). It has been used extensively in healthcare for determining error modes for surgery (Mohammadfam & Saeidi, 2015), addressing errors in medication administration (Lane et al., 2006), or generally investigating care areas such as nursing stations or emergency wards (Ghasemi et al., 2015; Kermani et al., 2016). Even given the large variety of complex tasks represented in these studies, the consistent finding was that the SHERPA has the capability to determine errors and create strategies for improvement such that the errors found in these processes are either eliminated or the consequences are reduced. Based on the extensive use of SHERPA in healthcare and the consensus of SHERPA as an effective human error identification technique, it was used to investigate the possible human error in this stroke assessment process. However, SHERPA looks at a broad work system to determine errors that could be made, to understand one of the most important components, telemedicine itself, a more focused analysis is needed. To determine the issues specifically with the interface, one validated tool is heuristic evaluation, which is a simple and effective tool for determining the usability violations in a system design that could contribute to human error.

Just one of the many tools of an interface designer, user experience designer, or human factors researcher, heuristic evaluation is a usability inspection method for determining problems in a design. Heuristic evaluation, as currently used by designers and researchers, was developed by Jakob Nielsen in collaboration with Rolf Molich in 1990 (Nielsen & Molich, 1990). In 1994, Nielsen refined the heuristics based on a factor analysis of usability problems to create the 10 usability heuristics now used worldwide (Nielsen, 1994b). In healthcare, heuristic evaluation has been used extensively to improve the design of medical devices, medical web pages, and persuasive health technology (Allen et al., 2006; Kientz et al., 2010; Zhang et al., 2003) and in the design cycle of new healthcare technologies (Choi & Bakken, 2010; Preece et al., 2013). Others have also used it to evaluate telemedicine systems (Agnisarman et al., 2017; Lathan et al., 1999; Tang et al., 2006). Heuristic evaluation is an inexpensive tool that has been used in the development and improvement of many health care systems as it has been shown to find more usability issues than user testing using a fraction of the time and cost (Thyvalikakath et al., 2009).

To investigate the mechanisms for improving this ambulance-based telemedicineenabled stroke assessment by understanding the human error and usability issues in the system, three analyses were selected as the tools for describing the work system process and developing design improvements: Hierarchical Task Analysis; heuristic evaluation, which focuses on the design and use of the interface based on usability heuristics; and SHERPA, which focuses on the reduction of human error in the system as a whole. These methods were selected to answer the following questions because they are powerful assessment tools for discovering and designing improvements to address issues in design and because they are used frequently as human factors tools in healthcare.

- 1. What tasks are involved in stroke assessment in a telemedicine integrated ambulance?
- 2. What are the usability issues in the telemedicine interface?
- 3. What are the potential errors that can be made using the telemedicine system?
- 4. How can the predicted errors and heuristic violations be mitigated through enhanced system design?

### **METHOD**

To answer research questions and accomplish objectives, 4 methodologies were used: observations of the stroke caregiving process, HTA, Heuristic Evaluation, and SHERPA. These methods will be described further in the sections below.

## **Participants**

To develop an understanding of the telemedicine system and the tasks completed during a stroke evaluation, nurses from the secondary hospital in the rural South Carolina county studied here, paramedics from the local Fire and EMS, and neurologists from the tertiary hospital were recruited to participate in telemedicine sessions such that researchers could observe their process. One of each type of caregiver-- a neurologist, nurse, and paramedic--was required for each session. In addition, an EMT was recruited to drive the ambulance during the session and a standardized patient was recruited from the secondary hospital. First, four pilot observation sessions were completed to test the observation protocol created. At the conclusion of those sessions, it was determined that the protocol did not require any changes. Then, nine final observation sessions were then completed using the same protocol as the pilot observations, resulting in a total of 13 sessions and 39 caregiver participants.

# Procedure

The simulated stroke sessions began with the ambulance arriving at the ED of the secondary hospital where the standardized patient and a paramedic were loaded into the back of a functioning ambulance with real equipment, which included two stationary GoPro cameras to view the participants as they drove a loop around the area. The paramedic was told to connect to REACH and treat the simulated patient per the normal caregiving process. Two researchers observed the nurse and neurologist caregivers, one at the secondary hospital and one at the tertiary hospital. Similar to the paramedic, the nurse and neurologist were instructed to complete their normal work tasks as if they were assessing the symptoms of an actual patient. The sessions concluded when a full neurological assessment was complete, and the patient arrived back at the ED. At the conclusion of each session, the caregivers were asked to complete a Retrospective Think Aloud (RTA) session which involved watching either the screen recording of the REACH session (for the nurses and neurologists) or the video of the observation (for the paramedics) and commenting on what they were thinking during the session. These sessions gave the researchers insights into the mental processes involved in the caregiving tasks as well as when and why the order of tasks was created or changed. A full process flow diagram of the observation study can be found in Figure 4.2. These observations were used to determine the tasks required to assess a stroke patient in a telemedicine integrated ambulance, the instances of and possibility for error, and the issues that the caregivers experienced with the current telemedicine system. These insights were used as input for several analyses conducted by the research team to further understand the structure of the caregiving process, the usability issues of the system, and the probability and consequences of human error in this system.



Figure 4.2: Process Flow Diagram of the Sessions

#### **Analysis Techniques**

## Development of Hierarchical Task Analysis (HTA)

First, the development of a structure of the tasks including their goals and subgoals was necessary to create the subsequent complete structure of the complex process that requires several caregivers to complete and to serve as inputs in other analyses conducted. From the simple observations, an HTA was performed for each role, from the time a caregiver logged on to the telemedicine system to the patient leaving the ambulance after completing a stroke assessment. Given that many steps and protocols either followed a specified order or occurred simultaneously, a clear order is needed to fully understand both how the stroke assessment process should be completed and how the tasks and protocols deviated in various situations. Based on our observations, the researcher was able to list all tasks for all caregivers and establish a beginning and end point for the assessment. The RTA was used to detail the thought processes or protocols that supported this order and to identify deviations from the steps listed. After the list was created for each role, the tasks were grouped into goals and subgoals, and protocols were developed to identify how each goal and subgoal was completed based on the tasks. At this stage, the task lists were shared with the caregivers to obtain their feedback. The caregivers corrected the step lists, adding or removing steps when necessary and clarifying the protocols. Based on this feedback, a unique HTA diagram was developed for each caregiver role. These role specific HTAs served as the input for the heuristic evaluation as each role was evaluated separately. To create the input for the SHERPA, the tasks were combined into an overall process to create a Team Hierarchical Task Analysis. Again, the tasks were organized into goals and subgoals and each Subgoal was assigned a team member. New protocols were created for the overall process to reflect the timing of the tasks including all caregivers. An additional step was completed to identify where teamwork was required and the criterion measure to determine the goal was completed.

#### Heuristic evaluation

To conduct a heuristic evaluation, simplified task lists of the 3 caregivers input into an Excel sheet along with the knowledge requirements for each step, given that the evaluators, while experts in the usability heuristics used, may not have had the medical knowledge to complete the tasks unassisted. This Excel spreadsheet can be found in Appendix H. Nielsen's 10 heuristics, which are standard for this evaluation, were used in this study (Nielsen, 1994b). The heuristics and their description can be found in Table 4.1.

Three experts from Clemson University completed the evaluation (two men, one woman), each with experience in completing heuristic evaluations of other systems and with training in usability and design. These evaluators included one assistant professor from the Human-Centered Computing Department with a Ph.D. in Information Sciences and Technology, and two Ph.D. students, both with a Master of Science Degree in Industrial Engineering. The choice to use 3 experts was made based on Nielsen's recommendations for conducting heuristic evaluations (Nielsen, 1992; Nielsen & Landauer, 1993). Each evaluator completed the tasks for all 3 caregiver roles--paramedic, nurse, and neurologist--in the REACH system interface, subsequently describing the violations of the heuristics at each step and assigning each a severity on the scale of 0 to 4 developed by Nielsen, with 0 indicating not a usability problem, 1 a cosmetic problem only

that does not need to be addressed unless extra time is available for the project, 2 a minor usability problem that is given a low priority for addressing, 3 a major usability problem that is important to fix and thus given a high priority, and 4 a usability catastrophe that is imperative to fix before the product can be released (Nielsen, 1994a).

Number	Heuristic	Description				
1	Visibility of system status	The system should always keep users informed about what is going on, through appropriate feedback within reasonable time.				
2	Match between system and the real world	The system should follow real-world conventions, making information appear in a natural and logical order.				
3	User control and freedom	Support undo and redo without having to go through an extended dialogue.				
4	Consistency and standards	Follow platform conventions.				
5	Error prevention	Either eliminate error-prone conditions or check for them and present users with a confirmation option before they commit to the action.				
6	<b>Recognition rather than recall</b>	Minimize the user's memory load by making objects, actions, and options visible.				
7	Flexibility and efficiency of use	Allow users to tailor frequent actions.				
8	Aesthetic and minimalist design	Dialogues should not contain information which is irrelevant or rarely needed.				
9	Help users recognize, diagnose, and recover from errors	Error messages should be expressed in plain language (no codes), precisely indicate the problem, and constructively suggest a solution.				
10	Help and documentation	Any help and documentation should be easy to search, focused on the user's task, list concrete steps to be carried out, and not be too large.				

Table 4.1. Nielsen's 10 Heuristics adapted from (Nielsen, 1994b)

Each evaluator worked independently to assess all heuristic violations. Once the list was complete, a single researcher compiled all violations, averaging the severity rating for duplicate violations.

### **Development of SHERPA**

To evaluate and predict human error in this process, a Systematic Human Error Reduction and Prediction Approach (SHERPA) was conducted by a single researcher. The first input for this analysis was the HTA developed for the overall process; then all tasks in the subgoals were classified according to a behavior taxonomy developed by Stanton (2006) as either Action, Retrieval, Checking, Selection, or Information Communication. Each behavior in a task involved different types of errors, all of which were described and labeled by behavior type according to the SHERPA error mode taxonomy shown in Table 4.2. If multiple error modes could be applied to the task, each was listed. For example, when entering a password to log in, a user could omit the action step or have an incomplete password or wrong password; in this case Action type error modes A7-9 were applied. The consequences and probability of making an error along with the criticality of the error were identified based on the observations and interactions with the caregivers. Probability is measured on a 3-item scale: Low (hardly ever happens), Medium (happens once or twice), or High (happens frequently), with criticality being measured on a similar scale, ranging from Low indicating a non-critical incident to High indicating a critical incident. It was also determined if the user could recover from each error and at which task step. For example, if the patient's name was incorrectly entered during patient creation, this error could be corrected later in the step asking for updated patient demographics. Finally, for each error at least one remedy strategy was identified based on four categories proposed by Stanton (2006): equipment, training, procedures, and organizational. The research team adapted these remediation strategies so that they could reasonably be implemented in the healthcare system. The items from the SHERPA technique (error type, error mode, consequence, recovery, probability, criticality, and remediation) were collected in tabular form to organize the information for each bottom level task in the HTA.

#### RESULTS

The results from the three analyses, which are presented in the sections below in the order in which they were conducted, detail the structure of the caregiving task and the number and type of issues found within the system based on the use of the heuristic evaluation and SHERPA.

# HTA

The results of the HTA answer research question 1. The overall process of care for a stroke patient was described by the caregivers. When a person potentially experiencing a stroke calls 911, the local EMS sends one of their ambulances with an Emergency Medical Technician (EMT) and paramedic team to assess the patient. On arrival, they typically conduct a Rapid Arterial oCclusion Evaluation (RACE), a simple 5-item scale used to help identify a possible stroke. Based on the RACE score and the overall evaluation of the patient, the EMT and paramedic move the patient into the ambulance, call the receiving secondary hospital to begin a stroke consult and log on to the REACH program from a laptop in the ambulance.

Error Type	Code	Error Mode			
Action	A1	Operation too long/short			
	A2	Operation mistimed			
	A3	Operation in wrong direction			
	A4	Operation too little/much			
	A5	Misalign			
	A6	Right operation on wrong object			
	A7	Wrong operation on right object			
	A8	Operation omitted			
	A9	Operation incomplete			
	A10	Wrong operation on wrong object			
Checking	C1	Check omitted			
	C2	Check incomplete			
	C3	Right check on wrong object			
	C4	Wrong check on right object			
	C5	Check mistimed			
	C6	Wrong check on wrong object			
Retrieval	R1	Information not obtained			
	R2	Wrong information obtained			
	R3	Information retrieval incomplete			
Information	I1	Information not communicated			
Communication	I2	Wrong information communicated			
	13	Information communication incomplete			
Selection	S1	Selection omitted			
	S2	Wrong selection made			

Table 4.2. SHERPA Error Mode Taxonomy

The nurse at the receiving Emergency Department (ED) creates the patient encounter, and the system alerts the neurologist on call at the tertiary care hospital of the awaiting consult. The paramedic gathers vital information and stabilizes the patient until the neurologist and nurse can be connected. The neurologist obtains more health history and information about the patient and begins an assessment using the National Institutes of Health Stroke Scale (NIHSS). When the ambulance arrives at the secondary hospital, the patient is taken directly for a Computed Tomography (CT) brain scan, which is immediately uploaded to the telemedicine system such that the neurologist can make a treatment decision that may include transfer to the tertiary hospital for more intensive care. However, to evaluate the ambulance based assessment the start and endpoints of the telemedicine system use were used to scope the HTA. The HTA was developed first by role and then as an overall process conducted by a team. Overall, the goal of the team process is to care for the patient; however, individually for the nurse and neurologist, this goal takes different forms. The main goal of the nurse is to prepare for the arrival of the patient at the ED, while the neurologist determines the patient care plan. In the overall HTA, the subgoals for the process are distributed among caregiver roles as described below.

## Log on to REACH

All participants must first log into the REACH system to begin the stroke consult, with the nurse typically logging on first after being alerted by the EMS and the neurologists last after a patent case has been created and the EMS has connected. Each caregiver has an individual user ID and password for accessing the system. In addition, the interface accessed is tied to the caregiver role: nurses begin with a create patient screen, neurologists a list of available and past consults, and paramedics with the main REACH screen and the accompanying video screens and data input.

## Create a patient case

Nurses are contacted by EMS with the basic information needed to create the stroke consult on the REACH system, including the medic unit the EMS is calling from, an estimated age of the patient, and symptoms including the RACE score the EMS has collected. The nurse is responsible for inputting as much information into the patient's case as quickly as possible so that the neurologist can select the patient when the alert is received. Without a patient file, neither the neurologist nor the paramedic can access the REACH consult. In addition, inputting as much information as possible alleviates the need for updates once the case is created; however, according to the nurses in this study, time is the most important element in this process.

# Gather and record demographics

Typically, the paramedic or EMT obtains a name and the date of birth from the patient or family when they arrive on scene, depending on the severity of the patient and if their family is present at the scene. If the medical personnel is able to obtain this information before the nurse or neurologist is connected to the REACH consult, then either caregiver asks the patient or the paramedic for this information once connected. It is the nurse's responsibility to record this information either when creating the patient case or in the data input screen when connected to the consult. This step is particularly important for the nurses who must connect this REACH chart information to the patient's Electronic Health Record (EHR) at the receiving hospital.

## Gather and record vitals

Once the patient is moved into the ambulance, the paramedics connect the patient to a vitals monitor that collects blood pressure, heart rate, and blood oxygenation (SpO<sub>2</sub>%). They also obtain the patient's respiration rate by counting breaths over a 15 or 30 second interval to determine breaths per minute, and a glucose measure, usually obtained from a blood sample. The paramedic typically communicates these vitals over the audio connection in REACH, and the nurse records them in the system. These vital signs are important for the neurologist to determine if the patient has had a stroke, a low blood sugar incident or a stroke mimic as well as to form a mental model of patient status. Current vitals are updated either upon request from the neurologist or when they change drastically and are communicated verbally by the paramedic.

# Gather and record patient history

Once the neurologist is connected, they collect information to create the patient history. If the nurse has obtained the medications the patient is taking, any allergies, and the medical or surgical history before the neurologist accesses the REACH consult, they review this information either with the patient or the nurse. Subsequently, the neurologist summarizes the patient history in REACH for the patient's chart. This process is necessary for the neurologist to develop the mental model of the patient's status and the severity of the patient's stroke. This initial step of recording the history is necessary for neurologist to form a complete view of the patient before they begin to evaluate the patient based on the stroke scale.

## Conduct stroke assessment

The primary tool used by neurologists to determine the type, location and severity of a stroke is the NIHSS. This is the most complex process in a stroke consult because it requires the neurologist to communicate tasks to the patient and the paramedic as they are not physically in the ambulance. One of the most difficult items on the scale involves assessing the patient's peripheral vision and reaction to movement; to do so, the paramedic must place their hands outside the direct view of the patient but within a reasonable distance for the patient to see them peripherally; they then ask the patient the number of fingers they are holding up on each hand. The proper positioning of the paramedic in relation to the patient is not only difficult to communicate but also difficult to judge in the telemedicine system. There are many items on the scale that require paramedic assistance, and while some have experience or training in conducting these, it is not standard for them. As the patient completes the movements or responses needed to the best of their ability, the neurologist selects the level for each scale item to create the scale score, which is a measure of the severity of the stroke. Neurologists can do this as they complete each assessment or in batches, and while they usually complete this assessment in the order suggested by the NIH, they may deviate based on patient status, patient symptoms, or personal preference. As the neurologists enter selections, the REACH system records the score so that nurses can follow the assessment as the score increases. This monitoring helps the nurses prepare for the patient's arrival as they can determine the neurologist's judgement of the severity of the patient's condition. The neurologist uses this assessment to develop a diagnosis and further develop a plan for the patient before they arrive at the ED. This process improves the efficiency and timing of the treatment and, more importantly the patient outcomes as quicker treatment saves brain tissue.

An excerpt of the nurse's role in the HTA can be seen in Figure 4.3. The full role specific HTAs can be found in Appendix E. An overall HTA diagram can be seen in Appendix F, and the Team HTA table can be found in Appendix G. The role specific HTAs were used for evaluating the heuristic violations of the interface by providing the overall goals for each user's tasks that evaluators then used to organize their analysis. The overall HTA was required input for SHERPA as errors are evaluated at the task level.



Figure 4.3. HTA Excerpt (See Appendices A and B for Detailed Task Analyses)

# **Heuristic evaluation**

The heuristic evaluation answered research question 2. A total of 129 usability violations were found by the three reviewers for the 3 roles studied: 10 violations for the paramedic role, 69 for the nurse, and 46 for the neurologist. The average severity rating for each role was 3.375 for the paramedic role, 2.28 for the nurse, and 2.35 for the neurologist. The further breakdown of the violation counts and severities for each role can be seen in Table 3. Ninety-five percent of the violations found were unique; however there were many instances where a similar violation description was used, but evaluators assigned different violation types. In the log on task, for example, one evaluator described a violation to the Visibility of System Status as "No update on what (ID/password) was entered wrong," while another described a violation to Help Users Recognize, Diagnose and Recover from Errors as "No error message." The most frequent classification of severity was a minor usability problem with 45 violations rated at this level (35%), followed by 43 violations rated as a major usability problem (33%), 18 as a usability catastrophe (14%), 22 as a cosmetic problem (17%), and only one violation being rated as not a usability problem (1%). When the common violations were merged and severity ratings averaged, there were 123 violations among the three caregiving roles. All heuristics were used, but the most commonly used was the Visibility of System Status heuristic, which accounted for 24% of all violations, followed by Consistency and Standards with 20% and Error Prevention with 15%. A summary of the severity ratings by heuristic violated can be found in Table 4.3 and Figure 4.4.

	Paramedic		Nurse		Neurologist	
	Count	Average Severity	Count	Average Severity	Count	Average Severity
Visibility of System Status	3	3.67	19	2.53	8	3.00
Match Between System and Real World	0		6	2.17	4	1.25
User Control and Freedom	0		2	2.5	3	2.33
Consistency and Standards	0		14	2	10	2.25
Error Prevention	1	3	9	2.5	8	3.00
Recognition Rather than Recall	0		4	2.25	3	1.33
Flexibility and Efficiency of Use	2	3	3	2.33	5	2.00
Aesthetic and Minimalist Design	0		5	1.2	3	1.67
Help users recognize, diagnose, and recover from						
errors	2	3.5	5	3	2	3.50
Help and Documentation	0		2	2	0	

## Table 4.3. Summary of Heuristic Violations

Some examples of violations found in this evaluation are that the distance from the labels to data input were too large making it unclear what label belonged to each text box, the interface for selecting consults was confusing as there did not seem to be a difference between active and pending consults, and that the consults and some tabs in the interface are not clearly clickable, inhibiting the function of the interface. The next most violated heuristic was Consistency and Standards. Some examples of these violations include inconsistent use of radio buttons, multiple conflicting data inputs for the same information, inconsistent date formatting, and inconsistent use of bolding and highlighting.

The role tasks with the most violation was the nurse role. Many violations for this role focused on the data formatting as the nurse role has several steps that require text input. Another frequently mentioned violation in this role was focused on the edit button used to change the demographics of the patient after a patient case is created, violating the Visibility of System Status heuristic as it was unclear that the edit button found when
hovering over the patient's name, date of birth, and many other demographic data would allow the user to edit all information in one menu. This required the user to search after selecting the appropriate menu for the information that required updating rather than providing a singular input for that data. Error messages for missing or incorrect information when creating a patient were either not functioning or unclear which violates the Help Users Recognize, Diagnose, and Recover from Errors heuristic. A full list of violations by role can be found in Appendix I.



Figure 4.4. Heuristic Evaluation Severity Summary

#### SHERPA

The goal of these analyses were to better understand the tasks and process of each caregiver in the system and determine where environment, training, or system design remediations could be provided to address the errors made. This was a first step for the research team to develop system improvements to support the caregivers in using this telemedicine system for stroke assessment. A sample of the SHERPA table created for the two main goals of the process including the HTA can be found in Table 4. This analysis answered research question 3. At least one error was found for each of the 97 lowest level tasks (1.35 per task on average), with a total of 131 being documented. The potential errors found included almost all task types, with 49% being information communication errors, 35% action errors, 10% retrieval errors, 6% selection errors and 0% checking errors. The prevalence of information communication errors can be explained by the fact that most data needed for stroke assessment are collected from the paramedic and patient verbally. As a result, the most common error is miscommunication or not hearing the response to a communicated command. There are also many action errors for tasks involving recording vitals and creating the patient case. No errors were assessed as high probability or high criticality based on the information collected from the RTA; however, these were only assessed based on the task directly affecting the process. Most errors were found to occur at medium probability (53%), and overall most errors were considered of medium criticality (55%). In addition, 77% of medium probability errors were also considered medium criticality, and 73% of low probability errors were considered low criticality. Almost all errors (90%) were recoverable immediately after they were made, and for the remaining 10%, recovery steps were identified later when a process was repeated or the error was accessed in later steps. The full SHERPA table can be found in Appendix J.

#### DISCUSSION

The objectives of this study were to develop a detailed understanding of the task structure and flow for completing a stroke assessment in a telemedicine integrated ambulance, determine the usability issues of the REACH telemedicine system interface and possible human errors that could occur when using this system, and suggest remediations for usability issues and human error. The first objective was met using observational data to perform an HTA outlining the goals, subgoals, steps, and workflow for each caregiver role. The heuristic evaluation and SHERPA met the second objective, the former by identifying 129 usability violations and the latter by detailing the possible human error occurrences for each observed task. Both heuristic evaluation and SHERPA suggested remediations to improve usability and safe performance, addressing the third objective.

While there have been HTAs, heuristic evaluations and usability assessments conducted on telemedicine systems (Agnisarman et al., 2017; Lathan et al., 1999; Narasimha & Agnisarman, 2016; Tang et al., 2006), studies of human error in stroke care (Gropen et al., 2014; Hughes et al., 2015; Kothari et al., 1995; Mould-Millman et al., 2018; Nor et al., 2004), and limited study of human error using telemedicine (Dharmar et al., 2013), there is little human-centered research on the errors made using telemedicine for stroke care and none using SHERPA to evaluate human error. This study addresses that

Task No.	Task	Error Mode	Error Description	Consequence	Recovery Task No.	Probability	Criticality	Remediation
2	Create Patient							
2.1	Enter medic unit as last name	A6, 8	Medic unit not correctly recorded in last name or placed elsewhere	Cannot create patient 1	Immediately	L	М	Highlight required field, Suggestions for how to fill
2.2	Enter current date as first name	A6, 8	Date not correctly recorded in first name or placed elsewhere	Cannot create patient	Immediately	L	L	Highlight required field, Suggestions for how to fill
2.3	Enter DOB greater than current date	A6, 8	DOB not correctly formatted or omitted	Cannot create patient	3.2.2	М	L	Create rigid formatting, Highlight required field
2.4	Enter Gender	A6	Gender not selected	Cannot create patient	3.2.2	L	L	Highlight required field, Suggestions for how to fill
2.5	Enter EMS as Room/Bed	A8	EMS not written as Room	Cannot create patient	3.2.2	L	М	Highlight required field
2.6	Select Stroke – Urgent in bottom selection box	A6, 8	Reason for consult not selected or stroke not selected	<ul> <li>Cannot create patient with omitted,</li> <li>Will not be allocated to the right</li> <li>people if incorrectly selected</li> </ul>	3.2.2	L	М	Highlight required field, Suggestions for how to fill
2.7	Enter Reason for Consult as Stroke Alert	A8	No reason for consult written	Cannot create patient	3.2.2	L	L	Highlight required field, Suggestions for how to fill
2.8	Click Submit Button	A8	Button not clicked	Cannot create patient	Immediately	L	L	Enlarge button
3	Update Demographics							
3.1	Gather Demographics							
3.1.1	Ask patient for name	11,3	Patient did not hear paramedic or vice versa	Cannot properly update patient case	Immediately	R	L	Enhance audio quality of transmission, Closed captioning, Visual aid of normal requests made by caregivers
3.1.2	Ask patient for birthday	11,3	Patient did not hear paramedic or vice versa	Cannot properly update patient case	Immediately	Z	L	Enhance audio quality of transmission, Closed captioning, Visual aid of normal requests made by caregivers
3.2	Record Demographics		;	•	:			
3.2.1	Click Edit button as it appears in the header next to name	A8	Button not clicked	Cannot properly update patient case	Immediately	L	L	Highlight required field, Suggestions for how to fill
3.2.2	Change name	A6, 8	Name not correctly recorded in correct sections or placed elsewhere	1 Cannot properly update patient case	Immediately	L	L	Highlight required field, Suggestions for how to fill
3.2.3	Change DOB	A6, 8	DOB not correctly formatted or omitted	Cannot properly update patient case	5.1.2	L	L	Create rigid formatting, Highlight required field
3.2.4	Click Submit	A8	Button not clicked	Cannot properly update patient	Immediately	Г	Γ	Enlarge button
Task	No. Taken from HTA							

# Table 4.4: Sample of SHERPA Table

need by providing a detailed analysis of the types, consequences, probability, criticality, and remediations of human error in a telemedicine system for stroke care. Results of the heuristic evaluation revealed many violations of common design heuristics. Most prominently the Visibility of System Status, Consistency and Standards, and Error Prevention heuristic violations identified issues with error messages, the layout of the tab structure, data input formats, information architecture, and overall page formatting. Information architecture and the design of the tabs for navigating the system are both important for building a solid foundation for finding information and allowing users to move through the interface. Information architecture involves different models depending on the use of the system; an application focused on fixed steps, for instance, would have a different information architecture from one for browsing. Creating a structure that assists users and is consistent and easy to learn is important in the development of an information architecture (Danaher et al., 2015). This architecture can also be supported by page formatting, consistency in type face and the use of bold in headings, and effective organization which can guide a user through the system and make items on a page easy to find. Not only should the structure be easy to identify and follow, it should also be easy to access, meaning it should not require additional effort to determine the state of the system and how to navigate it, this is consistent with previous usability findings (Johnson et al., 2005). Other usability studies have also found similar issues with data input formatting as we found here; with open-ended data input or a nonrestricted format, users often ignore formatting, leading to errors in the system (Lai, 2007). The system studied here also exhibited a lack of error messages or a lack of specificity in the error messages. Error messages are critical feedback for the user, allowing them to move forward in a system and correct future behavior. These messages should be informative, identifying the location of the error and how to fix it, and succinct so that the correction is easy to understand and implement quickly (Bargas-Avila et al., 2010).

The lower number of heuristic violations found for the paramedic role was due to the limited number of interface interactions for that role. The paramedic has only two goals, to log on and select the correct patient case, compared to the nurse and neurologist, each of whom has four goals. The reason for the fewer interactions with the interface is that paramedics are focused solely on patient care. In many cases, the EMT completes the REACH setup before the patient is in the ambulance so that the paramedics only needs to interact with the other caregivers via the two way audio connection in the telemedicine system. They cannot input demographic information or vitals while attending to the patient, meaning the protocol dictates that they communicate this information verbally. This impacts the possibility for human error in this process as the noise level in the ambulance can be high with road noise and sirens and the audio connection through the telemedicine can be impaired because of poor data connections as the ambulance drives through rural areas.

The higher number of usability violations found for the nurse compared to the neurologist can be explained by the amount of free data entry required in this role. Nurses are required to enter dates, names, and vital signs during their tasks, none of which have formatting suggestions in the system, violating heuristics such as Visibility of System Status, Consistency and Standards, and Recognition Rather than Recall. In comparison, although the neurologist role includes data input, the majority is limited to checking boxes and radio buttons to record the patient history, current medications, or allergies or using the drop-down menus selections in the NIHSS. Even though boxes, radio buttons, and drop-down menus are rigid forms of data input which can be restrictive when extra detail is needed, they prevent confusion and errors.

Findings from the SHERPA included the common possible human errors in the telemedicine system for stroke assessment, most of which were information communication or action errors, specifically, miscommunicated, misheard, or unheard communications and incorrect formatting or selection of data input. Communication errors in this system can be caused by unheard or unclear audio, which then requires the communications to be repeated. More common with complex communications through computer systems is misunderstood communications, in which the messenger does not provide adequate information or the receiver does not comprehend the message communicated (Morrow et al., 2005; Jordan, 1996). Data input formatting issues were also an issue found in the heuristic evaluation of this system, with incorrect date formats or misunderstood data input labels with a lack of formatting being were some of the issues seen from the SHERPA. While these errors may seem minor, missing or incorrect data can negatively impact patient care (Cebul et al., 2011; Wells et al., 2013). Other healthcare informatics research finding high error and data input has concluded that drop down selection reduces input errors (Devine et al., 2010). These simple changes to the formatting to allow for restricted input can reduce human error in data input.

The consequences for most of the tasks in this process were that the information would not be documented correctly or that a full assessment of the patient could not be formed. While these may not be as severe as causing patient harm, given that tPA is not given until a CT scan can be conducted, the further consequences of incomplete patient documentation could lead to incomplete links to the EHRs, difficulty connecting to a neurologist, or an incorrect NIHSS, all of which could affect the nurses' and ED physicians' care plan in the hospital or the search strategies of the neurologist when searching the CT imaging for the clots or bleeds that potentially caused the stroke.

Almost all errors could be recovered immediately as the system allows for free data entry after creating the patient case. After the case is created, updating demographics requires entering a sub menu, which is necessary later in the process, meaning that some create patient tasks have a recovery mechanism under the update demographics goal. Other vital sign documentation tasks have recovery steps when the neurologist completes an overview of the patient history. These are advantages in the system, as free data manipulation in the system gives users the chance to correct mistakes and update information to its most current state.

# Remediations

Remediations developed based on heuristic evaluation and SHERPA findings are organized below by work system category from the SEIPS 2.0 model (Holden et al., 2013).

# **Tools and Technology**

*Highlighting and labelling:* Consistent use of color, bolding and labelling formats should be used in the design of the system not only to create a cohesive aesthetic but also

to provide a sense of hierarchy and organization. Violations of the Aesthetic and Minimal Design and Consistency and Standards heuristics would be mediated by simple changes in the visual formatting of the labels used in the display. The location and text of the labels were also a focus of violations to the Aesthetic and Minimal Design, Visibility of System Status, Match Between System and Real World, and Consistency and Standards heuristics. Data labels for all input should be located close to the input boxes and be clear, concise and consistent. This not only allows the users to identify quickly and correctly where the data should be recorded but also is aesthetically pleasing on an interface. In terms of the SEIPS 2.0 model, changing the labelling and use of color on the interface impacts the work system through the tools and technology factor, but a clear organization can also impact how caregivers complete the task. The use of color, bolding, and size can provide a subtle structure to data input tasks and impact how users complete tasks or organize their work. This, in turn, can make users more efficient by making the organization of the system transparent.

*Error messages:* A common violation described in the heuristic evaluation involved data input tasks with requirements that have to be met before the user can log on, create a patient case or move through the system. Frequently there was little to no information about which items were the issue or how to fix those items. Providing salient error messages that help the user to recover is crucial to the usability of a system. A portion of this study focuses on methods for preventing human error; however, a system design needs to account for the possibility of human error and assist the users in correcting their mistakes. Error messages should follow design standards and include the item that needs to be changed, the severity

of the error and suggestions for how to fix it. As this change affects only the interface of the telemedicine design, it impacts only the tools and technology factor. However, as the SEIPS 2.0 model describes, changes to the tools and technology in the interface impact the task, internal environment, organization, external environment, and person factors in the work system.

*Layout:* The organization of the data input screens in this telemedicine system were the consistent source of violations in the neurologist's role. This role is the only one that consistently uses the tab structure in the REACH system to complete their tasks. Many evaluators reported that the visual design of the tabs did not make it clear that they were able to click on them to access different data inputs, especially in the History tab, which includes multiple tabs as subsections as seen in Figure 5. This issue violated the Consistency and Standards, Flexibility and Efficiency of Use, Visibility of System Status, and Error Prevention heuristics. Each item in an interface should have a clear purpose and communicating that purpose through design is crucial to usability. This is typically done though platform conventions. For example, links should be underlined in a highlighted color or clickable tabs or buttons should have some coloring or positioning to indicate that they are selected or can be selected. In this system the tab selection is inverted; the selected tab is a lighter color than the others, which doesn't follow conventions as the selected tab should be darker and have other features to distinguish that it has been selected such as an outline. The main tab structure has an issue as well based on the task flow observed in the mock stroke sessions and confirmed in the RTA with the caregivers; the neurologists use the history tab first and then input medications and allergies on the onset tab, but the onset tab is first, not second, in the tab structure. The tab structure should follow the task flow, and items like medical history and current medications that are completed together should be supported in the layout. Evaluators also mentioned that the organization of the tabs and subtabs made moving through the system confusing as there were multiple scrolling sections in the display, violating the Flexibility and Efficiency of Use heuristic. A simpler design of tabs organized by the task flow of the user would improve user experience and efficiency as organization prevents extraneous movement in the system. Similar to the labelling remediation, these changes in layout affect the SEIPS model work system through the tools and technology factor and the task factor because of the impact on the task organization mentioned previously.

*Formatting:* Given the large amount of data entered into this system, many task remediations focused on changes to the telemedicine system such that the data required to be entered in the process is clear by highlighting areas for data input. Another remediation suggested was rigid or suggested formatting. For data input such as dates and times, rigid formatting was suggested, meaning segmenting the data input such that the data are entered in a format consistently (for example separate data inputs for month, date and year), could eliminate confusion and maintain consistency in data input.



Figure 4.5: History Tab in REACH (Participants' faces masked for confidentiality)

This change reduces the number of errors involving switching day and month or entering month as an abbreviated word rather than a number. Suggested formatting could be used for such data as patient blood pressure, heart rate, or blood oxygenation by providing an example of a typical input with the correct units. For example, near the blood pressure data input, a text label could identify "132/88 mmHg" as an example of blood pressure formatting. This works well to mediate errors and address heuristic violations. Consistent and rigid formatting supports recognition rather than recall and helps users prevent errors and recover from them quickly as the formatting should make mistakes obvious. Again, this remediation focuses on the design of the interface and as such directly impacts the tools and technology work system factor in the SEIPS 2.0 model.

Tasks

Automation: For tasks that require manipulation of controls, such as the camera focused on the patient, automation could be implemented. A simple command or control would focus the camera on an area of the patient rather than requiring the neurologist to manually control the camera by clicking directional and zoom arrows. For assessment items that require a close view of the patient's facial movements, a control could allow the camera to detect the patient's face and zoom in, or for items requiring visuals of leg movements, the camera could move to the lower portion of the patient. Additionally, this functionality could be automated such that as assessment items are completed, the view the neurologist needs for the next item would be implemented as the previous assessment item was completed. Automating the movement within this system with set view controls could eliminate over or under correction of the camera location and reduce the instances that an assessment item is incorrectly recorded due to lack of detailed visibility. Further automating to set the views as neurologists work through assessment items would allow for a consistent assessment order for the neurologist and parametrics and simplify the movements needed to complete the task. Implementation of automation would impact the tools and technology work system factor, and automation based on tasks step would impact the task work system factor. The task change may also impact the communication in the system as a standardized task flow would limit the need for the neurologist to communicate the assessment needed, thus impacting the person work system factor.

#### Internal Environment

*Audio/Video improvements:* There are many communication tasks in this system that could be impacted by low quality data connections or audio equipment. Not being able

to hear or understand from either the neurologist or paramedic perspective requires multiple repetitions of information or physical movement in the ambulance to view the neurologist. Improvements to the equipment in the ambulance could eliminate or reduce the instances of miscommunications or work arounds. Improved microphones and speakers could make the audio connection clearer for both the neurologist and paramedic, reducing the need for repeated communications. Easily accessible views of the neurologist from the patient's side would allow the paramedic to make use of nonverbal communication with them. Based on the RTA protocol applied in this study, many neurologists mentioned that they would mime movements they wanted paramedics to use in the assessment whether they could be seen or not.

Often the paramedics leave the laptop connected to the REACH program in the front of the ambulance as it is the most stable place for it and movement is restricted by the wired connections to the laptop. Many paramedics mentioned their frustration that when instruction is needed from the neurologist, they must leave the patient's bedside to access the laptop. A screen at the rear of the ambulance so that patient care tasks are not interrupted would be ideal for providing a visual of the neurologist for that visual communication. In addition, closed captioning on a visible screen for the paramedics and neurologists could possibly provide some support in understanding audio degraded by poor data connections. These improvements primarily impact the tools and technology work system factor. However, the implementation of these tools affects the physical layout of the ambulance; thus, the internal environment work system factor is impacted as well. In addition, the support of communication and use of the system without moving from the patient's bedside could impact the task factor, decreasing the effort or repetition of communication needed to complete the assessment items.

#### **Organization**

*Training:* Consistency in procedure steps and increased familiarity with the assessment steps are the training remediations suggested by this analysis. Specifically based on the NIHSS errors in communication, unheard or misheard assessment steps and under communicated instructions for physical assessment could be reduced with training for paramedics on the tasks they need to assist with in the assessment and knowing the order in which to expect those assessment tasks. This could reduce the instances of repeated communication from the neurologist and, thus, the probability of miscommunication. In particular, training in the movements needed for the assessment as well as their correct application could eliminate the need for the neurologist to communicate techniques, that are often either unheard or misunderstood. This remediation focuses on the changes to the organization work system factor, but this training could impact the person factor by providing knowledge about the assessment and the task factor by changing the sequence and reducing the difficulty of the assessment tasks.

Error remediation techniques suggested in this analysis primarily focus on interface system improvements rather than the use of training or environmental changes. This is due to the constraints of the physical environment in ambulances, meaning additional equipment would need to be carefully considered for necessity and placement, and the mobile and complex nature of the nursing stations makes it difficult to implement lasting environmental changes. In addition, all caregivers in this process have rigorous training that they complete for many aspects of their jobs, meaning additional training for this single process should be kept to a minimum.

Many remediations were developed so that the process of completing the assessment is consistent, correct, and as quick as possible. To do this, the main problems of data collection and data input needed to be assessed. Supporting the communications of the paramedic and the neurologist through the telemedicine system is the primary remediation for errors of information communication. Setting a protocol and order for data collection can create redundancy in the communications. The remediation from the NIHSS assessment requiring training of neurologists to complete the assessment in a specific order for each patient would allow for consistent evaluation and better prediction of commands for the paramedic or nurse. This remediation is two-fold: it prevents lost or incomplete assessment items by following a checklist and making use of retrieval cues which improves performance (Hales & Pronovost, 2006), and it eliminates the need for redundant communication and alleviates dependence on audio connection as all members of the team would be able to predict the next item to be assessed. Most of the other remediations focus on formatting suggestions, which allow for consistency and provides an error check for users before they enter the data.

## Limitations

Analyses were conducted based on observations of mock sessions and supported with the questioning of caregivers to allow for further clarification on the order of tasks or deviations. The research team was not able to conduct live observations of a real stroke patient, meaning the errors developed and the probability of these errors occurring are the estimation of the research team. However, the team was able to obtain data about the consistent tasks that are completed by controlling the patient and the time to the hospital. This study was also conducted with one telemedicine system and with one county hospital system, the errors and usability found in this system may not be represented in others and the process of stroke care with telemedicine systems may differ slightly in other hospital groups.

#### CONCLUSION

Completing a stroke assessment in an ambulance using telemedicine is a complex process that is prone to error, as in all processes both simple and complex. Many usability violations were found in the interface, revealing problems with the system's design concerning information architecture, error messages, and page formatting. Predicted errors included miscommunication, omitted or incomplete steps, incorrect data entry, and insufficient assessment. The consequences were discussed with limited probability of patient harm, but some with implications to the patient care process beyond the ambulance. Finally, remediations were suggested for the heuristic violations and errors, mainly focusing on the support of clear communication, task flow, information retrieval, and data formatting. These remediations will be considered as the research team further investigates the use of this telemedicine system for prehospital stroke assessment and suggestions for system improvement.

#### CHAPTER FIVE

# UNDERSTANDING THE DEMANDS PLACED ON CAREGIVERS IN A TELEMEDICINE-INTEGRATED AMBULANCE-BASED SETTING FOR STROKE CARE

#### **INTRODUCTION**

#### **Telemedicine Use for Strokes**

Using a telemedicine system in ambulances can significantly impact patient care by improving triage and expediting diagnosis and treatment (LaMonte et al., 2004; Rogers et al., 2017). Reduced time to diagnosis and treatment is crucial for time-sensitive conditions or incidents such as a stroke where "time is brain" (Hill & Hachinski, 1998). Stroke is the leading cause of disability and the fifth leading cause of death in the United States (Benjamin et al., 2017; Heron, 2019). While the broad category stroke comprises multiple types, ischemic strokes account for 87% of the occurrences of this disease (Benjamin et al., 2017). Tissue plasminogen activator (t-PA) is considered the most effective treatment for stroke care. However, to be effective, it must be administered within 3 hours of symptom onset. Using telemedicine en-route to the hospital to consult with stroke specialists or neurologists can improve the possibility of correctly diagnosing an ischemic stroke (Barrett et al., 2016; Terkelsen et al., 2002), allowing treatment to begin in the ambulance or streamlined upon arrival to the emergency department (ED).

Extensive research in the use of telemedicine systems to diagnose strokes shows improved patient treatment time and diagnosis accuracy, with some of these studies focusing on the usability of these systems (Mandellos et al., 2004; Sibert et al., 2008; Takeuchi et al., 2015; Valenzuela Espinoza et al., 2016; Xiao et al., 2000; Yperzeele et al., 2014). Limited research has been conducted to understand the impact of telemedicine on all caregivers involved in stroke care and on team distributed cognition. In light of the significant potential benefits of telemedicine systems for ambulance-based stroke care, its effects on clinicians must be studied to ensure the benefits are realized at reasonable cost to its users.

Stroke care typically involves a series of caregivers, determined by the hospital system, support staff, and how the patient arrives at the hospital. For patients arriving by ambulance, paramedics and Emergency Medical Technicians (EMTs) assess typically using a Rapid Arterial oCclusion Evaluation (RACE) scale to determine stroke and triage a patient (Pérez de la Ossa, Natalia et al., 2014). This scale is a simplified version of the National Institutes of Health Stroke Scale (NIHSS), which is the national standard for physicians and can be conducted through telemedicine with consulting neurologists with specialized stroke care training (Leifer et al., 2011; Ortiz & Sacco, 2014). For example, the emergency nurses at Tidelands Georgetown Memorial Hospital, a secondary hospital in a rural county in South Carolina, use a telemedicine system called REACH to connect an incoming ambulance from the local Emergency Medical Services (EMS) and a neurologist on call located at the Medical University of South Carolina (MUSC) approximately 50 miles away. Once the patient has arrived at the ED the patient may undergo a Computed Tomography (CT) scan, a procedure critical for determining the location and type of stroke

required before treatment. A full description of the process flow for a stroke case using a telemedicine-integrated ambulance can be found in Figure 5.1.

#### **Theoretical Background**

The Systems Engineering for Patient Safety, or SEIPS 2.0 model, applies a system approach to healthcare to investigate how the work systems ultimately shape outcomes (Holden et al., 2013). This model has been used to systematically improve healthcare by applying human factors principles and focusing on human system integration. According to this model, the work system comprises six components: Tools and Technology, Organization, Tasks, Internal Environment, External Environment and Person(s). These components create processes that can be categorized as Professional Work, Collaborative Professional - Patient Work, and Patient Work, processes commonly referred to as workflow in many healthcare systems. They then create outcomes that can be categorized as Patient, Professional or Organizational and further described as desirable or undesirable and proximal, the direct result of a work process, or distal, a result of causal actions and observed over time. All stages of this model involve feedback or adaptation: based on the outcomes observed, appropriate changes are made to work processes or the work system.



Figure 5.1. Emergency Stroke Care Process Flow

This model is frequently used in research as a framework for investigating work in a healthcare system. It can be applied to simple systems such as patient interaction with a computer system intervention (Martinez et al., 2017), implementing a new patient portal (Walker et al., 2018), or primary care processes (Wooldridge et al., 2017) to better understand the work system. The versatility of this model allows it to be applied to many systems ranging in complexity, patient interaction, and personnel, usually as a framework for understanding the interactions among the elements in a process by mapping the existing system or a potential system to the model.

This research investigates how telemedicine in an internal environment of an ambulance interacts with the organization, task, external environment, and human (EMTs or paramedics, nurses, and emergency physicians) components of a work system. This interaction was observed through the workflow or work processes identified in this model. Assessment of patient condition, communication of symptoms, and development of diagnosis and treatment plan are classified by this model as Collaborative Professional -- Patient work, with the assumption that the patient is conscious. Observations of mock telemedicine stroke evaluations were conducted with a patient actor and professional caregivers to determine the workflow and observe the interactions among the latter. Follow-up interviews were conducted to improve our understanding of the interaction among the components in this work system and how this interaction creates the current workflow as well as perceptions of how that workflow impacts the outcomes. This research considered how changes to the work system affect professional outcomes through system testing and provide feedback for future development of these work systems.

## **Research Questions**

This research study investigated the following research questions:

1. What are the cognitive demands placed on caregivers while completing a stroke assessment in a telemedicine-integrated ambulance-based setting?

2. What are the barriers and facilitators for using telemedicine for stroke assessment in a telemedicine integrated ambulance?

#### METHOD

We used a qualitative research design consisting of observational studies and interviews to collect and analyze data during simulated stroke caregiving sessions in an ambulance-based telemedicine setting. This study was approved by the Institutional Review Board at IRB2018-465 and consent was gathered from participants prior to completing the study. The studies were conducted from November 7, 2019 to January 26, 2020.

# Participants

Participants were recruited through purposive sampling from two major hospitals and a fire and EMS department in the southeast United States. Nine observation sessions were conducted with 3 caregivers recruited for each session, a nurse, paramedic, and neurologist for a total of 27 caregivers. In addition, an EMT to drive the ambulance and a standardized patient to allow for a simulated stroke were recruited for each session. Paramedics and nurses were recruited by their supervisors using word of mouth and the neurologists by email. The only inclusion criteria for these participants were their occupations: paramedics were currently employed in Georgetown County EMS; nurses were on staff at Tidelands Georgetown Memorial Hospital, and neurologists were completing a neurology Telehealth rotation at MUSC.

# Equipment

A Dell 13-in laptop provided by MUSC with the REACH program installed was used for these specialists as seen in Figure 5.2a. Nurses used one of two available mobile REACH carts, which included a 24-in monitor, a keyboard, and a connected camera and overhead speaker as shown in Figure 5.2b.

The ambulance was outfitted with a Dell 13-inch laptop with the REACH program installed, connected to a wireless router to create a data connection when driving, an overhead speaker installed in the ceiling above the patient and a camera at the back facing the patient. GoPro Hero7 cameras were mounted inside the ambulance at the front and rear facing the patient from the back and the paramedic workstation at the front. The telemedicine set-up in the ambulance can be seen in Figure 5.2c. A functioning ambulance was used in these sessions and driven during the observations to simulate real conditions. The Xbox Game Bar was installed on the neurologist's and nurse's computers to record their screens during the session. The same laptops were used to deliver the surveys, and the follow- up interviews were recorded with an Olympus Digital Voice Recorder when conducted in person and using WebEx Virtual Meeting when conducted virtually.



*Figure 5.2*: from top down left to right a) Neurologist Set-Up (Participant's face masked for confidentiality), b) Nurse REACH Cart, c) Telemedicine Setup Diagram, d) Ambulance Overhead Speaker, e) Ambulance Laptop, f) Ambulance Camera

# Procedure

On the day of the study, the participants were informed that the purpose of the research was to obtain information to help the researchers understand the system in order to generate ideas for improvement. One researcher was located at MUSC with the neurologist participants, while another researcher was located at Tidelands hospital with the paramedic, patient, and nurse as they completed the study. All participants completed their normal workflow, see Figure 5.1, during the observation while researchers observed. The sessions were also audio and video recorded. Participants then completed the National Aeronautics and Space Administration Task Load Index (NASA-TLX), International Business Machine Corporation Computer System Usability Questionnaire (IBM-CSUQ), and Team Effectiveness questionnaires. Interviews were conducted with each of the 3 participants (paramedic, nurse, and neurologist) for each of the 9 observation sessions completed. It took approximately 2 hours to complete the study. Interview transcripts were not returned to the participants; however, the results of the thematic analysis were shared with them to update them on the project and obtain their feedback.

#### Measures

The NASA-TLX was used to measure the caregivers' perceived workload during the simulation (Hart & Staveland, 1988). The NASA-TLX survey measures the six workload subscales of Mental Demand, Physical Demand, Temporal Demand, Performance, Effort, and Frustration on a scale from 0 to 100; these values along with the weightings of the importance of the subscales for each individual are then combined into a Total Workload score. The IBM-CSUQ was used to measure the participants' perception of the usability of the telemedicine system (Lewis, 2018). This survey measures usability based on level of agreement to 16 statements on a scale of 1 to 7, an overall score and 3 subscales (System Usefulness, Information Quality, Interface Quality). The Team Effectiveness scale, (Rentsch & Klimoski, 2001) which was used to measure quantitatively how individuals felt their team worked together, measures the level of agreement to eight statements on a scale of 1 to 5, with each statement reflecting different aspects of teamwork, such as team member commitment or satisfaction with performance; however, the agreement rankings can be averaged for an overall measure of team effectiveness.

Interview questions were adapted from an interview protocol developed by the Center for Quality and Productivity Improvement at the University of Wisconsin-Madison (*Interview Guide for Clinicians and Office Staff*, 2016), with questions specifically pertaining to teamwork, distributed cognition, barriers, and facilitators being added to the protocol. The questions were then pilot tested with neurologists, nurses, and paramedics for length and repetition. Adapted protocols used in the study can be found in Appendix K and L respectively. Notes were taken during the interviews to guide additional questions or focus the subsequent thematic analysis. The interview audio files were transcribed and analyzed to determine if the responses contained any common themes.

# Analysis

The survey data were analyzed based on the descriptive statistics for each measurement. Two coders were assigned to descriptively code the interview data to ensure researcher bias did not influence the codes developed, and prior to consensus the researchers assessed intercoder agreement using the ATLAS.ti 8 program. This analysis function in the code management program assesses coder agreement in the use of a single code over all documents using Krippendorff's alpha. Open coding was used to create a code book from the pilot interviews. Once the code book was defined, the two coders used it to classify the responses of the 27 interviews. Themes were developed from the interviews by taking quotations from each caregiving role and from each descriptive code, and summarizing the quotations for each code. Next, the two researchers further summarized and organized the themes into major and minor themes and categorized them as barriers or facilitators to using telemedicine. A flat rather than a hierarchical structure was used for the descriptive coding.

The interviews were on average 36 minutes long. The total audio time for all 27 interviews conducted was approximately 16 hours and 13 minutes. Due to the complex nature of responses, and the number of documents, ATLAS.ti 8 was used in all levels of coding. Data saturation of responses was discussed with the research team. It was determined that after eight sessions we had reached repetitive responses.

#### RESULTS

The objectives of this study were to determine the cognitive demand of caregivers and the barriers and facilitators to using telemedicine to complete ambulance-based telemedicine stroke assessments. These were met by collecting quantitative data about the workload, team effectiveness, and usability, which are presented below in sections, and qualitative interview data to further describe the demands, barriers, and facilitators. The descriptive coding results and the themes developed from that coding process are organized by the barriers and facilitators they describe in the following sections.

# Workload

The average total workload on a scale from 0 to 100 for all roles was 55.15 (SD = 16.68), and did not vary among roles. On average, users rated mental demand as 65.67 (SD = 25.67), physical demand as 17.33 (SD = 13.61), temporal demand as 71.00 (SD = 23.94), performance as 36.59 (SD = 31.25), effort as 38.48 (SD = 31.25), and frustration as 28.85 (SD = 20.49). More role specific changes were found in the sub-measures of workload as can be seen in Figure 5.3.



Figure 5.3: NASA-TLX Average Participant Response by Caregiving Role

# **Team Effectiveness**

Mean team effectiveness was high overall; on a scale of 0 to 5, participants rated their team effectiveness in the simulated stroke session at 4.08 (SD = 0.81), with nurses rating higher on average than the other roles as seen in Figure 5.4.



Figure 5.4: Average Team Effectiveness by Caregiving Role

# Usability

Usability received high ratings from the participants, with overall usability being rated as 5.46 (SD = 0.97) on a scale from 0 to 7, system usability as 5.70 (SD = 0.99), interface quality as 5.30 (SD = 1.28), and information quality as 5.24 (SD = 1.06). The nurses' ratings were higher than the other caregivers in all categories as seen in Figure 5.5.



Figure 5.5: Average Usability Ratings by Caregiving Role

# **Interview Analysis**

# **Descriptive** Coding

The pilot interviews were inductively coded to describe the responses of caregivers. This process resulted in 52 codes, which were then used to deductively code the 27 final interviews. No new codes were added to the codebook during the deductive coding as the final interviews did not introduce any new content. Five codes were removed from the codebook during coding of the final interviews as they were not used to classify any responses. The average intercoder agreement for the deductive coding of the 27 interviews was 0.71. Consensus was then reached on the quotation size and code for each for all the final interviews.

# Themes

From the 2,255 quotations for 47 codes and 78 unique themes, 10 major themes were developed. The themes, their associated codes, and illustrative quotes can be seen in Table 5.1.

Codes difficulty with system, technical difficulty,	Themes	Example Quotes
teamwork, visual ussessment difficulty, frustration, technical upport, workarounds, noise, delays, cost, system improvement	Issues in the Telemedicine System	"common to have a poor connection, and then I can't hear them very well, or they can't hear me very well or the camera freezes. That's probably the most frustrating."- (T9) "it's very hard to get someone who knows how to check visual fields well and that's frustrating because I cannot do it at all" - (T5)
tech integration, work station, flexibility, elework, consult space	Work Station and Consult Space Changes	"I now have more flexibility of whether I work from home or if I'm working from my office during the daytime. It does give me that benefit." - T8 "the computer system is hard to access from [the left side of the ambulance] but someone has to decide which side of the patient do I need to be on and why." - P8
ocumentation, usability, workload	Simplistic Documentation	"I think it's easy to complete the consult or there's a lot of parts that are not free text that I can just click. That part I like a lot" - $T7$ "we actually have to leave the cart and go over to our computer to pull up the patient's old records and stuff to see if there was a previous stroke or to see how many times they've been for the same problem." - N7
loss of face to face, patient interaction, increased computer interaction	Patient Interaction Changes	"You can't lay a comforting hand on their shoulder for example and if the patient's not used to the concept of telemedicine and the idea that they have to talk to a computer screen and that person over there that they can barely see in here is their doctor, that still actually puts some people off" - T2
rikflow, learning curve, dfing issues, efficiency, ttient care, environment awareness, visual communication, rustration, usefulness, multitasking ability, access to the patient, trigger factors	Efficiency and Focus	"If we see something in the back of the ambulance where the neurologist says, 'Hey, go ahead and have TPA ready.' When that patient gets there, then we've got it mixed and we're ready to run." - N1 "telemedicine is not very efficient in a patient to doctor interaction. Because there's a lot of things that you can do when a patient is physically there, in examination of that patient, which you cannot do during over telemedicine, is physically there.'
'isual communication, access to the patient, fficiency, patient care, teamwork, task awareness, communication	Team Visualization and Communication	"I think it's a great way to communicate because you're actually looking at the neuro doctor, you're looking at the patient, and all three of us are on one page. You also have your attending doctor standing in the room. I just think it's a great way for everybody to communicate, it's like sitting down at the dinner table, and everybody eating and discussing what's going on, 'Okay, what do we do next?''' - N7

# Table 5.1: Description Codes for Themes and Quotations

(Continues on next page)

<ul> <li>benefit of neurologist, patient care, Ac errors, efficiency, usefulness Speciausability, satisfaction, use for other System conditions, best features and S.</li> <li>learning curve, teamwork, training, Trai efficiency Exp</li> <li>work station, privacy Priv</li> </ul>	ccess to ialist Care m Usability satisfaction ining and perience vacy and	"you're taking a physician who specializes in this area, this is all they do, they eat sleep and breathe stroke patients, neurological patients and they're going to be the ones to have the final say so and this is what's best for the patient." - N3 "I think that it's good that we have telemedicine, I think that we could probably tweak the system a little bit and make it more effective in terms of making it physically easier for the people that do the consult." - T2 "Ithe more you use, though you're going to be with the process itself and that streamlines itself." - P8 "If you're having to do a consult on someone who's in a room that's just separated by a curtain, then it could leak
Surr	Consult roundings	that patient's privacy and their information just because it's not in a more secluded area or private area, but generally when we use the cart, we are trying to put these patients the best we can, 95% of the time in a room that has a door surrounded by walls so that their information is kept private for them and it abides by the laws." -N9

#### Barriers

Issues in the system was a major theme for both the paramedic and neurologist caregivers, with comments from six of the latter (46%) and all paramedic participants (13) contributing to this theme. The source of these issues when using the telemedicine system consistently involved difficulties with the internet connection or device malfunction. Loss of service or spotty and delayed communication as the ambulance moves into an area with a limited data connection can make many processes difficult; for example, it may make patient responses difficult to see or understand or freeze communications between the neurologist and the paramedic and patient.

According to one neurologist, it is

"common to have a poor connection, and then I can't hear them very well, or they can't hear me very well or the camera freezes. That's probably the most frustrating." - T9

The work system factors mentioned in this comment focus primarily on the tools and technology in the system. These issues with the data connection were more frequently seen in the paramedic interviews as these caregivers are responsible for connecting the system to REACH, and any problems often interfere with patient care. A paramedic described this issue from that perspective, saying

"Maybe a little bit of frustration in the essence of sometimes the computer's slow or if the computer's not up to date or up to speed that's behind on the internet. Then at that point it's a frustration as I'm trying to do everything else on top of that." - P3

This comment again addresses tools and technologies but also the task factor as it emphasizes the work paramedics must do concurrently with technical support. Another paramedic described the thought process as they make decisions about technical difficulties:

"Do we call it in and plug it up or do we do we not? How do we get there the fastest? What is in the best interest?" - P5

This paramedic is describing the concern for patient outcomes, the tasks they must complete to care for the patient and weighing the potential benefit of using the tools and technology available against the time it will take to address the issue. More than one neurologist also mentioned having to communicate tasks for the paramedics that they would normally accomplish themselves, saying

"It's very hard to get someone who knows how to check visual fields well and that's frustrating because I cannot do it at all, it has to be the person who's examining there next to the patient and then sometimes I'm just not sure if the patient has a deficit or not because they just can't get it right. That is frustrating." - T5

The neurologists mentioned issues with completing their tasks because they have to assist

team members to do so. This was seen as due to a lack of training or experience in these

members:

"I think it's primarily experience-based. Sometimes interestingly, we will find that there's a nurse who's actually doing it with us who is new. You can tell that they don't have any idea of what to do. - T8

"I think it's experience-based. I think ones that do it enough or who are taught well understand that. I've certainly done a lot of consults where people either maybe they're not comfortable or they just don't want to do what they're asked to do or

they've just not been experienced enough to do things to really understand the intricacies of a stroke consult that they require a little bit more with us pushing them or as

king them to do things." - T3
These neurologists' comments reflect the person work system factor as the training or experience of a team member can impact their process or sense of the ability of the team to work together effectively. Paramedics experienced this issue when receiving instructions from the neurologists as well:

"Then you get into a discussion about how to do a test or perform or procedure or whatever it is the doctor wants us to try and do and it's like, 'Oh, I don't know what the heck you're talking about.' Now you get frustrated because the doctor gets frustrated because you two aren't on the same page anymore." - P2

In addition, noise also played a factor in communication:

"I think what was happening is that they were in order to remove background noise, they were actually doing that process twice. That actually made it really difficult for us to hear the nurses and the paramedics." - T8

This comment best represents the internal environment work system factor as the immediate surroundings such as road noise impact effective communication with other caregivers. However, both the paramedics and nurses felt that the issues they had with communicating or with delays in the system improved with experience:

"If you've done it a few times, every time you do it, you get a little bit more efficient. We pick up on certain things that'll make things a little better the next time." - P1

"Depending on what neurologist it is and what part of the brain is affected or what they're concerned about with the brain, it's depending on which cart they're going to use. Once you do enough, you know what cart they're going to or what test they're going to do before it ever happens. It's definitely where you're building routine." -N6

These caregiver comments about how experience with the system or tasks improves efficiency supports the task and person work system factors.

In addition to the issues of communicating through telemedicine, 38% of neurologist participants (5) also mentioned that they felt a loss of personal connection during the virtual consults:

"Before it was all about physically being present at the bedside and being able to adjust some of the factors along the way physically. Like if I didn't think the patient was being moved along enough in the process, I would physically move them in the process but with telemedicine it's different." - T2

The disconnect felt in telemedicine is due primarily to tools and technology, but it also impacts the task factor as these participants mention they feel limited in the tasks they can complete while distant from the patient. This adds to the feeling of disconnect for both neurologists and patients:

"You can't lay a comforting hand on their shoulder for example and if the patient's not used to the concept of telemedicine and the idea that they have to talk to a computer screen and that person over there that they can barely see in here is their doctor, that still actually puts some people off" - T2.

However, neurologists mentioned that while this limits the interaction or the level of

relationship you can have with the patient, it allows them to speak directly to patients who

would not otherwise receive specialist care:

"It allows me to talk to patients directly, which before I wouldn't do it when we're just using the phone. When I see a patient via telemedicine and I decide that I'm going to transfer the patient somewhere, whether it's here or somewhere else, I have to interact with a lot of people to make that transfer happen. Yes, there's some interactions that have to happen, otherwise it would not happen." - T7

"It's obviously different if you're seeing someone in person, a patient in person, touching the patient, having more connection than when you're just a face on a card. It's better than not having for sure, not having any connection, these patients not having access to any specialized care." - T5

Paramedics experienced additional physical barriers, specifically, issues in the internal environment as a result of the placement and implementation of the technology according to 62% of these caregivers:

"The computer system is hard to access from [the left side of the ambulance] but someone has to decide which side of the patient do I need to be on and why." - P8

Further, the layout of the telemedicine ambulance can hinder visibility of the REACH system:

"If you're sitting by the patient, monitors come in the way, but that's where they are mounted for our trucks. You just have to peer around it. Other than that, that's the one knock I would give on the way our ambulances are organized." - P5

In addition, any movement in the ambulance to improve visibility of the REACH monitor

is constrained:

"Anytime you're walking around the patient because you're trying not to fall on them and you're moving. Those are places that are constrained around the patient, around the computer, no. You don't want to make anything any harder than it is in the back of the ambulance." - P5

Paramedics mentioned developing a more mobile system or one that would mount in the

back of the ambulance where most patient care tasks occur would improve visibility.

"I would prefer to have a monitor that I could see the patient from where I'm sitting and the screen but we're making do." - P6

Other paramedics mentioned a preference for a smaller mobile system:

"I think it's okay. If it can go to a tablet or something that is smaller." - P7

"if you were able to take technology and put it into a tablet and stick it up on the back wall, that would be where my number one preference to have it, not only because it would be easy to access" - P2

Of all the caregiving roles, the nurses have the least critical need of the REACH

system; thus, these themes were not brought up by nurses in the interviews. However, one

theme was considered a barrier for 30% of the nurse participants (4). Documentation in healthcare requires detailed notes on patient history, symptoms, medications, and the steps taken in a consult, and stroke care is no different:

"There's a lot more documentation so that you have to have it right in telemedicine, you have to have it the same in your notes, everything has to be across the board at the same time. It seems like it's a lot." - N4

"I wouldn't say that it decreases the amount of paperwork that we have to do because there's still a lot of charting that goes along with stroke patients and documentation." - N9

Nurses are responsible for creating the patient chart in REACH as well as for documentation in both REACH and in the hospital's electronic health record (EHR), creating duplicate tasks in their workflow:

"We actually have to leave the cart and go over to our computer to pull up the patient's old records and stuff to see if there was a previous stroke or to see how many times they've been for the same problem. That's the only thing that I feel would be a little better is if you could access your patient information at the cart" - N7

These comments not only address the impact that the telemedicine interface has on the

work system but also the impact on their tasks as their work must be duplicated in an EHR

chart. A nurse also mentioned the increased accuracy and detail in their documentation

afforded by the telemedicine system:

"No, I cannot say I spend less time. It just allows me to be more accurate with my documentations... a lot of the information that we obtain through the telemedicine and what we're relaying with the physician allows us to go back and use this information while we're charting and documenting in our computer systems. I feel it's a benefit." - N5

Given the need for patient privacy or confidentiality in healthcare, the researchers included an interview question about this issue when using a telemedicine system. Their decision to do so is supported by findings from a review of previous telemedicine implementation studies (Rogers et al., 2017). Some caregivers, approximately 18% of all participants, conceded that privacy could potentially be a barrier:

"It can be if you're not in a private room just because some of the rooms in our department are-- I guess they're blocked off by curtains and not walls or doors. If you're having to do a consult on someone who's in a room that's just separated by a curtain, then it could leak that patient's privacy and their information just because it's not in a more secluded area or private area." - N9

The perspective here focuses on the internal environment as nurses must consider or alter it to protect patient privacy. However, most caregivers (82%) stated it was not a concern for them or their patients as exemplified in the comment below :

"I don't think anybody worries about it and I never have had a patient say, 'I'm not comfortable with this because of security or privacy,' or anything like that." - P2.

In addition, some nurses communicated that from a benefits versus cost perspective, privacy is not an issue:

"Why wouldn't the patient give you consent to do something that's going to help them? To me that's important." - N7

"Is privacy really that much important when you're trying to save someone's life?" - N5

# Facilitators

The system is not without its benefits; in fact, facilitators were more frequently found than barriers in these interviews, and even when the barriers described previously were mentioned, they were more often than not followed by a caveat, suggesting that they were fairly insignificant compared to the benefits to the patient and the overall health system.

One of the major themes classified as a facilitator in the caregiving process for 69% of the caregivers was the visual connection to the patient and to the other members of the

caregiving team which enhanced the awareness of each member's care tasks and the

environment as seen in the responses to the question below:

"You're able to develop an awareness of what your other team members are doing. Do you think that that's enhanced by the telemedicine at all?" Interviewee: "Yes, because I can see them. Not just talking to them, you can see each other." - T6

"Just being able to see everything, witness everything from the ride here to when they get here and to the time we go upstairs ship them to MUSC, You get to be part of the care from stage 1 of it. It's great that you can be that part " - N4

In part, this ability to see one another helped highlight that they all used the same protocol,

rather than using behavior triggers to complete tasks:

"We have certain things that the HealthStream classes that we take that have given you step by step on what to look for as far as stroke protocol and how to assess it. We're doing the same exact things that he's actually doing" - N4

The confidence that this awareness gave to the team is enhanced by the neurologist's

communication with the caregivers as seen in the comment below:

*"The neurologist is telling you this is what will take place. You know what's coming up and everything." - P7* 

Even with the protocol and communication, being aware of other caregivers' tasks can

improve teamwork as all members are better able to predict future tasks:

"As you start flowing through and you understand what the doctor is looking for, yes, I think I have a better understanding and you can make some anticipation sometimes as to what's going to happen next." - P3

"I think with the system because it's been now running for a while, I think the nurses at the sites and us, we all know what everybody's role is, so the nurse already will, most of the time, the nurse will already plug in the vital's data into the system and will put the last known normal in the system and then the radiology tech will push the patient's CAT scan into the system so we can review it. Then I know that my job is to enter everything else." - T8 These comments on the visual communication aspect of the telemedicine support the impact of the tools and technologies on the communication and coordination among the team members, in turn, impacting how tasks are completed in the system.

The communication among caregiving team members is not only enhanced by being able to see one another with the new technology but caregivers specifically cited that live constant contact with their fellow caregivers improved their process, impacting the tasks as they can communicate with the various medical professionals with greater access. This openness primarily affects the communication between the paramedics and the specialist that is a direct result of telemedicine:

"It is nice to be able to see, you've got the nurse and you have open line of communication with the nurse. ... Then the doctor, to be able to get information from the doctor. I think in that aspect, it's a good teamwork and there's not really anything that impedes." - P2

"Once you are connected, it's constant communication so there is potential to improve communication for sure." - P8

This open line of communication coupled with being able to see the other caregivers, both

a direct result of telemedicine, is seen as improving the caregiving process and, in turn,

patient care:

"I think it's a great way to communicate because you're actually looking at the neuro doctor, you're looking at the patient, and all three of us are on one page. You also have your attending doctor standing in the room. I just think it's a great way for everybody to communicate, it's like sitting down at the dinner table, and everybody eating and discussing what's going on, 'Okay, what do we do next?'" - N7

The open communication in the telemedicine system also creates a greater sense of teamwork as seen in the comment below:

"To provide the appropriate care for the patient, you have to work as a team to get to the essential end goal. You have to come in and essentially come in as a team in a group effort, and get everything done for this patient that needs to be done. Like I said, those first even couple minutes are just so hectic that one person cannot do that by themselves, and everybody have an understanding of that. Everybody comes in and work as a team, and we get the task accomplished. I definitely think that the telemedicine actually promotes the teamwork instead of inhibits it." - N9

In addition, to open communication impacting the teamwork, having a visual of the patient

during the care tasks in the ambulance was also seen as a facilitator:

*"Absolutely very useful and beneficial to have it in the ambulance because it will allow us to see them en route." - T8* 

"They're seeing what's taking place with the patient. They don't have to depend on me to say, 'Well, she can't lift her arm,' or, 'She can hold it out.' They're seeing the drifts. They're seeing it one-on-one just like we are. I just think for a patient, it's better, " - N7

While the neurologists mentioned interacting with a patient virtually can mean loss

of personal connection, in contrast, the nurses and paramedics (30%) indicated that they

believed their interaction with the patient has not changed with the use of telemedicine.

Paramedics may have to get the system up and running and interact with the neurologist,

but because it is mostly hands-free, their interaction with the patient has remained the same

as seen in the comments below:

"Being hands-free, it's very easy to move about and get things done and get the tasks done that they want by the audio." - P5

"I don't think it really has an effect on how I care for my patients. It might add a little bit to it because they do have the neurologist on them, but it doesn't change how I act as a patient or whether I put an IV in them." - P9

For nurses as well, they interact with the patients the same, both while following the treatment in the ambulance and upon arrival at the ED:

"I would talk to them the same way whether they're in front of me or on the computer." - N2

Especially for the nurses, the knowledge of steps taken with the patient allows them to prepare for the patient's arrival to the ED, increasing the efficiency of the care in the hospital. Increased efficiency was a major theme for both nurses and paramedics, with 77% of nurses and 100% of paramedics describing this theme in their interviews. One nurse talked about being aware of their place in the process is helpful:

"I'm able to see what's going on. On the nursing standpoint, I'm able to see what has already been done and what assessments have occurred. That's something that to me." - N5

Others described how they can prepare better with the telemedicine:

"Yes, we're going to start the process of everything we've talked about up until this point and have everything ready. As soon as they get here, the patient can be received and have a head CT down immediately as soon as they get here." - N3

"If we see something in the back of the ambulance where the neurologist says, 'Hey, go ahead and have TPA ready.' When that patient gets there, then we've got it mixed and we're ready to run." - NI

In addition, the neurologists see the increased focus in the assessment provided by telemedicine as an improvement to efficiency as well:

"Things are very, very quick and very short, very, very focused. You spend less time with the patient but in a more focused and acute manner." - T6

Again based on these comments, the telemedicine technology is seen by these healthcare professionals as impacting the task factor by increasing efficiency.

However, as was mentioned in the barrier themes, conducting the assessment through another person can be inefficient. Many neurologists followed this comment by adding that conducting this assessment with the visual, live communication is more efficient than bringing a patient to MUSC or trying to conduct it over the phone:

"Much better because you can communicate much better. You can assess the patient better and know what's going on so you get the patient to the right type of treatment sooner as opposed to getting the patient somewhere else and then assessing the patient and get and then getting the patient somewhere else. Yes, so it's much better." - T1

Not only was improved efficiency in general seen to improve patient care, conducting these assessments improves the care of the patients as they receive access to a specialist, another major theme across all caregiving roles, with 97% of all participants commenting that they can see the improvement in process efficiency because of the capability of being able to connect to a neurologist, ultimately improving patient outcomes:

"The patients are getting the care quicker. They're getting a neurologist right then and there. They're getting diagnosed quicker, they're getting treatments done quicker, they're getting into rehab, or getting the procedures that they know they need to be completed quicker." - N6

"You're taking a physician who specializes in this area, this is all they do, they eat sleep and breathe stroke patients, neurological patients and they're going to be the ones to have the final say so and this is what's best for the patient." - N3

"Being more thorough, getting door to needle times reduced, seeing if they need TPA or if they have a brain bleed or getting them into the door to a CT scan and then already having the drugs pulled up to give them as soon as they need can absolutely have effect on the outcome." - P6

Beyond improving the quality of care, caregivers also identified how they think access to

a specialist reduces human error and increases patient safety:

"I think it improves safety and reduces the scenarios because you're bringing expert, coordinated, scripted care in places that did not have it." - T7

"Maybe [the neurologist] picking up on something that you missed or something that you haven't had an opportunity to do when you, of course, get to the patient." - P1.

Caregivers have also seen how patients appreciate the use of the system

"When a patient hears, 'I'm a neurologist. I'm a specialist,' it's like, 'I am getting the best care I can get." - N7

Some patients even return to the hospital system to say how much the telemedicine has

helped them:

"I've had some patients that were a successful TPA drug given candidate and everything turned out right. They were just really pleased that it went the way it went. We don't always get feedback from the patients that receive the medication. Some of them we never hear from again but the ones that get discharged from MUSC who come back two months walking and talking and they want to say thank you, they tell us about everything that happened and how appreciative they are about it." - N3

In addition to the benefits to the patient care process and to the patients themselves,

there are benefits for caregivers. While paramedics' tasks and work location do not change much with the addition of telemedicine, 69% of the neurologists and 30% of the nurses stated they gained flexibility in their work station because of the addition of telemedicine. This theme of flexibility describes the changes in the internal environment created with the use of telemedicine. Neurologists are able to work outside of the hospital, providing patient care from wherever they are:

"I now have more flexibility of whether I work from home or if I'm working from my office during the daytime. It does give me that benefit." - T8

They also have flexibility in their scheduling with telemedicine:

"I can pick the days I want to work or not work or I can pick the place where I'd work with, where I work from or not. I can fix how many days that I want to work. I guess it gives me a lot of flexibility." - T7

Nurses have increased flexibility with the introduction of the REACH system as everything they need for a stroke consult is located on a rolling cart. This allows them flexibility in a busy ED where patients are continually coming in and out rather than constraining them to a few beds reserved for stroke care:

"That cart can go anywhere in the hospital. Regardless if it's ER, whether it's coming into the ER or a patient is already there, it's giving improved patient care because it is mobile. It can go to east, it can go to new onset or This thing is going to get moved to east because that's our stroke ward." - N6

In addition to increased flexibility, caregivers also experience simpler documentation procedures as a result of telemedicine implementation, which impacts the tasks of the caregivers. For paramedics, documentation is limited as the nurses can see all the tasks they complete in the ambulance during travel to the ED and take responsibility for documenting them; however, only 2 paramedics (15%) noted these changes to paperwork:

"The nurse kind of acts as our scribe, so I don't have to worry about the paperwork or documentation or anything like that, so that's a good system we have in place." - P2

For 30 % of the neurologists documentation is easier as they are doing it in real time:

"We'll do the full exam because you're also documenting and ticking these boxes in real-time as you're doing the exam versus when you do it in person where you have to do the exam and then when you eventually go and sit in front of a computer, that's when you actually start documenting." - T8

Both nurses and neurologists also mention the improvement to documentation and their

overall workflow as the telemedicine system provides a centralized, easy-to-use format for

documenting and finding patient information:

"I think it's easy to complete the consult or there's a lot of parts that are not free text that I can just click. That part I like a lot." - T7

"We know everything's got to go in that page and that's the information that the neurologist needs to make his decision as to what happens with that patient so I

think it's a great way to just localize one center for all the information that's needed for the patient." - N7

Further, 97% of participants representing all caregiver roles stated they were

satisfied with the current telemedicine system, but improvements can always be made.

Many cited that it was simple to use and easy to learn:

"Pretty satisfied. It's easy to use. It's typically easy to get a neurologist on. I think it's a great benefit to have. It's another resource tool. I think it's definitely a very useful tool." - N8

Several neurologists, however, were more critical of the system:

"From 0 to 10, I'm like five. I think we have improved a lot of things, like the access to patients first. Our access to the patients, but I think we still have like a good way to go in terms of improving both technically and in terms of appropriateness of use of it." - T5

"I think that it's good that we have telemedicine, I think that we could probably tweak the system a little bit and make it more effective. Some of the things that I have already mentioned, but also in terms of making it physically easier for the people that do the consult." - T2

Overall, the interviews supported a system that caregivers find effective, but one that has areas needing improvement.

### DISCUSSION

Telemedicine supports collaborative work processes with visual communications and constant contact. Thus, it is logical that caregivers in these interviews discussed how this system improved teamwork and ultimately efficiency and patient care. This perception is supported by past research on stroke telemedicine, which has shown reduced door to needle times (Adams et al., 2012; Belt et al., 2016; LaMonte et al., 2004), and overall improved patient recovery and mortality (Levine & Gorman 1999; Wechsler et al., 2017). The themes found in this research are tied to the SEIPS 2.0 model work system factors: Tools and Technology, Tasks, Person, Internal Environment, External Environment, and Organization, while the changes in telemedicine that support collaborative work can be linked to the work system factors: Task and Tools and Technology. Tools and Technology is impacted by the integration of the telecommunications system in the ambulance and this introduction changes how caregivers communicate and tasks are accomplished. As the interviews found, the tasks became more efficient, and their workflow adapted to streamlined processes such as taking a patient directly for a CT scan or the more in-depth NIHSS that the paramedics do not normally conduct.

More importantly, the interviews point to one possible reasoning for improved patient care outcomes in current research; not only access to a specialist but also an increased level of teamwork among caregivers including the specialist was perceived to improve patient outcomes. Research in the field of Computer Supported Collaborative Work (CSCW) can provide some clarity about how visual access to fellow team members increases performance and teamwork (Schlichter et al., 1996). When people are geographically distributed, they must rely on technology to support their communication and coordination. Research in CSCW, which primarily focuses on the development of collaborative virtual environments (Fleury et al., 2015), is based on the principle that providing a visual representation of the work environment helps the interaction among team members. Telemedicine provides this environment through the video feeds of the patient and caregivers as well as a central interface for charting that can be edited by all. Conducting collaborative work tasks in these virtual environments has been shown to be equal in performance to in-person collaboration (Narasimha et al., 2019). In addition, the awareness of the patients and their status that is created is valuable to the nurses as they can better prepare the ED for patient arrival, thus increasing efficiency and decreasing door to needle time.

The improvement in patient care and safety resulting from access to a specialist was a theme in these interviews as well. This supports previous research on telemedicine stroke care research that found increased diagnosis accuracy with access to a neurologist (Lumley et al., 2020). Many caregivers mentioned that multiple eyes on the patient or oversight by the specialist could reduce medical errors and increase patient safety, a finding supported by human error research, which has found, for example, that inadequate supervision is a pre-condition for errors (Shappell & Wiegmann, 2000). By providing for specialists to be involved in the stroke assessment through telemedicine, they can also serve in a supervisory capacity to deter and potentially eliminate unsafe acts and improve patient safety.

The training and experience of the caregivers was found to impact how other caregivers completed their tasks and the overall teamwork. These comments offered two perspectives: a lack of experience or training can inhibit efficiency or teamwork and increased training and experience can improve the collaborative work. Teamwork research supports this finding as well, suggesting that experience can improve task performance and teamwork and further that team training should be developed based on the experience levels of the members (Deering et al., 2011; Kleij, 2007; Rentsch et al., 1994).

Generally, lack of training can be frustrating for fellow caregivers, but the specifics from the interviews focused on neurologists being frustrated because of the need for repetitive communication. If the paramedics did not have the training needed to complete an assessment item on the NIHSS, the neurologists had to explain where to stand or what movements to make, often multiple times, before they were able to complete the assessment or if neurologists needed updated vital signs because the patient's condition changed, the specialist had to signal the paramedic to update them. This situation can create strain between the neurologist and the paramedic as they collaborate in the telemedicine system, ultimately resulting in the undesirable professional outcomes of distrust between caregivers and increased stress. The source of the frustration for the paramedics in this collaborative process is that neurologists are inconsistent in the order of steps required for the assessments; thus, they cannot learn from experience. Research on the use of checklists indicates that providing a structured list of tasks can improve both task efficiency and performance (Kelleher et al., 2014). In addition, providing standardized order lists can improve patient care outcomes (Berenholtz et al., 2004; Pronovost et al., 2003) and reduce the number of medical errors (Hales et al., 2008; Hales & Pronovost, 2006).

These findings also support another important theme from the interviews: the usability and satisfaction with the system. Nurses like the current system primarily because it acts as a checklist for their tasks. They appreciate the layout of the small segment of the screen they are responsible for because it enables them to check through the items they need to ask and document and easily see if any of these items have not been completed. However, neurologists were more critical of the layout, logical because they had more

interaction with the interface than the caregivers. They had issues with the tab layout, which did not follow their typical process, and the amount of text input needed for entering medications or some history information. The tools and technology implemented in the work system should be designed such that they support the task, not requiring additional training or expertise to use the technology. As much as possible, the design, especially for complex tasks and work processes, should be simple and easy to use so as not to create additional burdens on caregivers. Neurologists and nurses both mentioned that they like how easy the checkbox input is to use and generally thought the system was easy to use, despite the specific comments from neurologists; these results, however do not support the heuristic evaluation of this system in previous research, which has found multiple violations, especially concerning data input that had no suggestions for formatting and the complex tab structure design.

Another impact of the introduction of the telemedicine was on the physical environment, the internal environment work system factor presented in the SEIPS 2.0 model. Many paramedics mentioned that the addition of the laptop, video and microphone components not only potentially creates extra work in the event of technical difficulties but also creates additional movement in a constrained environment. They described how the movements needed for the assessment or to see the telemedicine laptop to interact with the neurologist can be difficult in the limited space in an ambulance, a constraint they did not have to consider previously. Careful planning and consideration of the placement of these laptops for both accessibility during the tasks and technical limitations such as wiring is needed to eliminate additional movement or obstructed visibility in a confined and busy space. Work has been done in the field of architecture to determine ambulance design with caregivers; however, most of it has been conducted in a static environment (Goodwin et al., 2017). This movement can also cause harm to both the patient and the paramedic. As the ambulance is moving, the paramedics could fall or be rocked into the patient bed or other equipment. In addition, some ambulances have only limited handholds for stabilizing and preventing falls; however, these were not considered in light of the movements needed to see the telemedicine screen or adjust equipment when needed.

Opinions in the interviews were mixed about workload: some neurologists felt that tasks became more difficult as they could not do them themselves, and paramedics were frustrated with unclear communication from the neurologists. In contrast, the nurses felt telemedicine made their jobs easier. These trends are seen in the effort and frustration scales, with nurses rating effort and frustration below midway on the NASA-TLX and neurologists and paramedics at or above midway. However, overall the average workload was midway on the 0-100 scale, which corresponds to the interview responses as most caregivers said that their workload did not increase or decrease with the use of telemedicine.

Higher contributors to total workload in this task were mental demand and temporal demand. Neurologists rating an average mental demand of 73.7, this can be best explained by the comments from neurologists as to the level of mental analysis of patient responses needed to diagnose the patient and create a care plan. Temporal demand was rated highest by nurses on average with a rating of 78.9 in comparison to neurologist and paramedics who on average rated temporal demand as 70.6 and 63.6, respectively. The difference in

average ratings by roles here can be explained by the nurses interview responses, their process is to create the patient case as soon as possible such that neurologists can be connected to the patient in the ambulance which could increase the feeling of time pressure. After this point their task is to observe the neurologist assessment and patient responses until addressed or the patient arrives to the ED. This waiting period as they watch the tPA window close can emphasize the lack of ability to speed up this process and impact temporal demand as well.

Teamwork was discussed by all caregivers in the interviews as something that was improved in the telemedicine system due to the open communication among the caregivers and a shared visual of the patient and their fellow members. In addition, most caregivers felt that they were able to coordinate and communicate effectively as well as develop a shared understanding of team tasks, related to items on the team effectiveness scale. Data from the survey, while not as detailed as the findings from the interviews, were consistent with the interview findings. First, the comments about usability, that the flow of the system and how to input data were easy to use, are supported by the high overall and system usability ratings from the IBM-CSUQ survey. Caregivers also rated team effectiveness, a quantitative measurement developed to rate teamwork, fairly high at 4 out of 5.

## CONCLUSION

The quantitative data from this study determined that the demands of the telemedicine system on the caregivers was moderate; however, the technical support required, the communication with caregivers about vital signs and the required assessment tasks, and documentation are not supported by the current system, creating cognitive

demands that should be addressed through system design. Further, barriers and facilitators were determined. Barriers included cognitive demands as well as the physical constraints of the ambulance, the loss of personal connection to patients, and maintaining awareness of the surroundings to protect patient privacy. These barriers indicate how telemedicine has made the work of caregivers more difficult. Facilitators, ways telemedicine has improved the caregivers' jobs, were much more common; these included flexibility in the consult space, the usability of the system, the streamlining of tasks, improved efficiency, an improved view of the patient and caregivers, and access to a specialist.

While current research in the field of telemedicine has found improved patient outcomes and process efficiencies, many of these studies do not focus on the people in the work system, the caregivers, who have to make sense of and use the system in their workflow The interviews and themes developed here provide a new perspective on the demands placed on stroke caregivers using telemedicine and the ways they think telemedicine has improved or made more difficult how they complete their tasks. These findings not only investigate problems with design and usability but also the difficulties with communicating with team members and troubleshooting the system. Improvements to stroke care are not just found in the numbers; caregivers felt more connected to their team members and indicated that as they conducted telemedicine consults, they began to learn more about how to treat strokes from interactions with the specialists. Based on these barriers and facilitators from the interviews, future research can be conducted to determine how system design can eliminate such barriers as redundant communication of vitals or the unorganized layout for data input as well as support the facilitators such as process efficiency and open communication for better task awareness. Further, these findings can inform design to support the distributed cognition of caregivers by focusing on their responses about task and environment awareness, for example that consistency in the assessment tasks can allow for a better understanding of the tasks in the consult and prediction of role in the assessment.

### CHAPTER SIX

# PRIORITIZED INFORMATION DISPLAY FOR ENHANCING DISTRIBUTED COGNITION IN A TELEMEDICINE-INTEGRATED AMBULANCE-BASED SYSTEM FOR STROKE CARE

### **INTRODUCTION**

Technology platforms and software systems have seen increasing implementation in ambulance-based telemedicine (Rogers et al., 2017), with research on their efficacy finding that for a variety of healthcare conditions, including stroke care, patient care, task performance, and error management have improved. In fact, the use of telemedicine early in the care process has been suggested as a method for improving stroke care (Schwamm et al., 2009). This dissertation focused on REACH, the system used by the Medical University of South Carolina (MUSC) in coordination with several rural counties in South Carolina. It consists of a laptop, including speakers and a camera, connecting the ambulance both to a neurologist, also on a laptop, and to a rolling computer cart for nurses, enabling communication among these healthcare professionals as patients are transferred to the emergency department (ED). As this description suggests, the REACH interface involves three user roles: a nurse, neurologist, and paramedic. Initially, the nurse is the user who creates the patient case, while the neurologists and paramedics can view only those patients with a patient case. However, once connected to the patient, the REACH interface appears the same for all users and updates as they input information.

Not all users have the same interactions with the interface. Paramedics simply connect to patient case, and no further input to the system is required of them unless a technical issue arises. They communicate all information to the neurologist and nurse verbally and interact with the patient in the ambulance. Nurses are responsible only for updating the patient information in the "nurses lane" on the left side of the data input as seen in Figure 6.1 and monitoring the patient responses and neurologist communications. Neurologists interact with the REACH interface the most as they are responsible for developing the patient history, documenting it in the system and conducting a neurological exam to diagnose the patient's condition using the National Institutes of Health Stroke Scale (NIHSS). The neurologists' tasks require them to use the full range of tabs on the right side of the interface and monitor the patient in the video screens on the left as seen in Figure 6.1.

<b>Marke</b>	CHREACHHEALTH Support: (655) 894-1500 Start: 09:27:52 E	musc-gmh-te lly Blac 8 Jan 1985 (34y) ST TStoke Urgent	Ale MR#: Unk	I settings   about   feedback   logood Enc/Acct#: Unk Room/Bed: tr 1 TES @
	3 Hr Window 00:59:01	ONSET HISTOR IMAG	IN NIHSS+ LABS D	X/tPA PLAN F/ CONSU
	4.5 Hr Window 02:29:01	Start/Stop Camera Start Camera Stop Camera Reason for Consult	Onset Summary Last Known Well 07:30 02 Dec 2019 In ER Door	Onset Observed By OPatient
	1 Hr Door To Needle Unknown	Chief Complaint 34 yr old male onset 2 hrs ago suddon onset	ER Doc Seen CT Ordered	Family     Friend     Inferred     Clinical Staff
	Treating Physician	headache left sided weakness	Sent to CT	Pain Intensity Pain
	Nurse		Read/Reported	Location S
MUSC Health	*Last Known Well 07:30 02 Dec 2019	Telemedicine Informed Consent Obtained	Initiated Call to Provider	OLeft ORight OAmbi
and the second sec	Door *BP (mm/Hg)	Medications & Allergie	s	
VidyoPower Version (2.0.5.40 REACH Health,	HR(bpm) RR (bpm)	<ul> <li>Notes</li> <li>Family Contact Information</li> </ul>	a	Everyth

*Figure 6.1:* REACH interface (Participants faces masked for confidentiality)

The previous work conducted in this dissertation examined the usability issues, possible human error occurrences, the demands on the caregivers, and the barriers to and facilitators in the REACH system when conducting a stroke assessment in a telemedicineintegrated ambulance. The overall impression of the REACH system provided by the caregivers in previous studies was positive, and the benefits of the system were thoroughly represented both by the caregiver participants and research investigating time to treatment and patient outcomes (Adams et al., 2012). However, the usability evaluation and SHERPA of the REACH system indicated issues in the design of the user interface (Rogers et al., 2020). In addition, analysis of the interviews found barriers to the use of this telemedicine system that increased caregiver workload (Rogers et al., 2020). These previous findings in conjunction with the lack of research on the role of caregivers in telemedicine stroke care point to a need for system design. While previous work details the current system and its strengths and weaknesses, further research needs to be conducted to redesign the system interface to minimize the demands and workload on caregivers while supporting their processes and the distributed cognition necessary to care for stroke patients.

### **Distributed Cognition**

Distributed Cognition (DC) Theory is applied to enhance our understanding of the organization of cognitive systems, not only the individual's but the interactions with other people and resources in the environment (Hollan et al., 2000). In this way, Distributed Cognition Theory, at its most basic level, is similar to situation awareness with its focus on the interactions between the individual, external factors, and cognitive processes (Hollan et al., 2000; Endsley, 1995). Hollan et al. (2000) defines distributed cognition as cognitive

processes occurring across members of a social group across time and involving coordination of material and environmental structures. This theory assumes that systems including more than one person have cognitive properties that differ from the processes of the individuals within that system (Rogers & Ellis, 1994), and as such, it was developed to understand the cognitive processes that occur in a system, defining cognition as an emergent property of interactions between people and their environment rather than within the singular person (Liu et al., 2008). While researchers have developed methodologies to analyze complex systems, few have been applied to multiple domains (Jenkins et al., 2011; Patrick et al., 2006; Walker et al., 2010; Waterson, 2009).

For example, Blanford and Furniss (2006) developed Distributed Cognition for Teamwork, or DiCoT, a methodology for applying distributed cognition to the study of teamwork. This methodology defines three themes of distributed cognition, determined from DC literature, that form the basis for analysis: physical layout, information flow, and artifacts. These themes include principles, 18 total, that can be used to assess the system in terms of how it supports distributed cognition and how to improve system design. Some of the principles in the physical layout theme include Don Norman's (1993) perceptual and naturalness principles, the arrangement of equipment, and situation awareness, while the information flow theme includes the principles information hubs, buffering (an affordance for delaying the processing of new information until an appropriate time), and information transformation, and the principles in the artifacts theme include coordination of resources, creating scaffolding (for example, setting reminders to complete a task at a later time), and mediating artifacts (things brought into the system to coordinate task completion). Using DiCoT, Blanford and Furniss (2006) analyzed the London ambulance service using observations and interviews to determine how the system for receiving and dispatching ambulance service calls could be improved. More recently, researchers have used DiCoT to assess systems and provide a basis for design recommendations. For example, in the mobile health care domain, McKnight and Doherty (2008) examined patient processing in a cancer treatment center, finding that the principles within the themes were effective in identifying the shortcomings of the system in general, understanding the technology locations, and determining the needs of the people in the system. A DiCoT model-based evaluation of a training simulation used to prepare medical professionals in emergency and disaster management found a need for greater flexibility in the interaction with artifacts in the system and better management of the physical layout to avoid unintended information movements (Rybing et al., 2016).

## **Proposed Interface Design**

Observation and interview themes and principles from the DiCoT model were used to define features to support distributed cognition and facilitators of the current system and address its barriers. While all caregivers' tasks were considered in the design changes and many of the features impact multiple caregivers, the display of the neurologist is the primary focus of the interface features as they have the most interaction with it. Four primary features were developed based on the data from the observations and the theoretical frameworks used in this study. Figure 6.2 shows the proposed interface design.



Figure 6.2: Proposed Interface Design

The first new feature provides real-time vital signs displayed on a graphical timeline shown in Figure 6.2. This feature was developed to support the creation of neurologist's mental model and its subsequent updating, thus reducing the workload for both the paramedic by eliminating the need to repeat vital signs and the neurologist asking for a vital sign update. All caregiving roles communicated in the interview that updating vital signs was important and could be simplified. Paramedic 4 provided an example of the issue in the REACH interface, saying "Because he's, 'Did you check the BGL?' 'Whoops. We'll check it again doc. We got you." Neurologist 2 emphasized that "having the vitals be able to be seen on a display. Those would be important things," and Nurse 1 offered more specific feedback supporting this idea, "Then being able for them to automatically or Bluetooth-feed their vitals and stuff in or even their patient information." Updating vitals in real time improves information flow through the use of automatic mechanisms,

supporting the information flow principle, information movement, in the DiCoT model (Furniss & Blandford, 2006). Providing an information mechanism supports this general principle, but automating this flow of communication can alleviate the demands on caregivers as the mechanisms of information flow, thus reducing the workload and the redundant communication.

However, updating vitals is not the only information can be displayed using a timeline, the current system provides a tPA window countdown from the last known well time as shown in Figure 6.1. Neurologist 8 specifically mentioned the usefulness of these markers of time: "*I like the fact that it auto-calculates the window times to four and a half hours and three hours and one hour door to needle once you plug in the last known normal time.*" However, representing this time to end of window value can be improved by providing a graphical representation of the current time to that window, which can be implemented through the vital sign monitoring. The feedback from providing a clear and active representation of the closing of this window will ensure the tPA window is met if a patient is a candidate.

The DiCoT model was also used to determine features to be implemented in conjunction with the principle to support the use of artifacts, specifically the goal parity principle (Furniss & Blandford, 2006). This principle states that providing explicit representation of the current state in relationship to a goal state can aid cognition. The proposed representation of the tPA window addressed this principle by providing a visual representation not just of the distance to the goal but also progress to it. Another proposed design element in the vital monitor that applies this principle is the color coding of the vital signs signaling if the value is within normal bounds or if the patient needs stabilization. For example, if the systolic blood pressure rises above 180 or the diastolic above 105, the neurologist can recommend actions to lower these numbers as they too high for tPA. Providing a cue to determine the status of the patient and how it compares to the goal (normalized vital values) again supports the goal parity principle and allows for an effective and efficient check to update the mental model of the patient's status.

Design Elements	DiCoT Model Principle	Barrier or Facilitator
Real Time Graphic Vitals	Information flowInformation Movement Principle	Barrier for Neurologist and Paramedic Repetitive Communication required to update vitals
Color Coded Vitals	Artifacts Goal Parity Principle	
tPA Window Timeline	Artifacts Goal Parity Principle	Facilitator for Neurologist Increased awareness of patient with current telemedicine system
Centralized Patient Info	Artifacts Coordination of Resources Principle	
Task Progress Bar	Information Flow Trigger Factors Principle Artifacts Scaffolding Principle	Barrier for Neurologist and Paramedic Unclear communication and expectations in tasks
Simplified and Ordered Documentation	Artifacts Scaffolding Principle	Facilitator for Neurologist Checkbox layout for creating efficient documentation
Enlarged View of Patient		Facilitator for Neurologist Visual of the patient and caregivers to increase the awareness of team members' care tasks

Table 6.1: DiCoT Principles and Barriers or Facilitators Used for Design Features

Improving how vital signs are represented also changed how patient information in general is presented; in the current system demographics, vital signs, history, and current medications and allergies are presented separately. Patient history and medications and

allergies are also presented such that navigation in the system is required to view them as they are found under either the history tab or onset tab, both of which cannot be viewed at all times as seen in Figure 6.3. Demographics provide an initial perception of the patient; for instance, age can play a role in decision making: for a person under 35 having a stroke, a neurologist could look to drugs, medications, or underlying conditions for the cause. Vital signs, medications, and patient history can influence the decision to give a patient tPA treatment for a stroke, and all should be represented and accessible at all times. To do so, a feature was added to summarize demographics, patient history and current medications in one place; furthermore, this information can easily be updated using selections in the data input screen. This feature is supported by another artifact principle in the DiCoT model, the coordination of resources (Furniss & Blandford, 2006), which states that an information structure that assists in the coordination of plans, goals, affordance, history, action-effect, and current state can aid in cognition. Providing information about patient history that is constantly available supports its use to inform future decision making. In addition, the presentation of this information together and the collocation of the monitoring of the patient's vital signs supports the design principle, proximity compatibility (Wickens et al., 2015): activities or information used in the same process should be represented together or in close proximity to one another.

ONSET	HISTO	IMAGIN	NIHSS+	LABS	Dx/tPA	PLAN F,	CONSU	ONSET	HISTO	IMAGIN	NIHSS+	LABS	Dx/tPA	PLAN F,	CONSU
										oagulation	Disorder"			iabetes	
											,			neurysm	History.*
History				Review o	of Systems									cher Family	liscory
Past Med	lical Histor	v		Past Sur	gical Histo	v		Social H	isto <b>ry</b>						
РМН				PSH				Occupat	ion			Alcohol U	se		
		one Repor	ted			one Reporte	ed	Tobacc	Use					enies	
🖂 Diabetes				Coronary Artery Bypass Graft			○ Denies ○ Former ○ Unknown ○ < 1 PPD			Yes <sup>∞</sup>					
	Hypertension Coronary Artery Disease			Aortic/Mitral Valve Replacement			0 1 PPD 0 2 PPD 0 3 PPD 0 4 PPD 0 5 PPD			Sübstance Use					
		troke		13	Endarterectomy						Unknown				
		ancer			Recent Surgery*								□ Ye	25 <sup>*</sup>	
		eizure				ther Surgica istory*	al	History	Notes			<u> </u>			=
Atrial Fibrillation			istory			History	Notes	Í							
		arotid Arte	ery Disease	FH											
	<b>—</b> 1-	traccasial				one Penorte	be								
ONSET	HISTOR	IMAGIN	NIHSS+	LABS	Dx/tPA	PLAN F/	CONSU	ONSET	ISTOR	IMAGIN	NIHSS+	LABS	Dx/tPA	PLAN F,	CONSU
Start/Sto	p Camera	0	nset Summa	rγ				Warfari		rgies					
Start Car	nera	к	Last		Onset	ed		(Couma	din)			Dye Aller	rgy		
Stop Can	nera		ell 07:30	02 Dec 2019	By				OYe	15			ON	5	
Reason f	or Consult	D	oor		_	O Patient			ON	ot Specifie	d	Drug Allergies			
Chief Complain	34 yr old male onse	t 2 St	R Doc een			() Friend		Heparin	2			Food			
	hrs ago	C	r rdered			O Interred		(nepun	OYes			Allergies Env.			
	headache Sent to		Pain	Pain V			O No O Not Specified			Allergies					
	weakness	C	r r		Pain	,		Dabigat	ran	ve opreente		Home Medicatio	ons		
		R	ead/Reporte me	d	Location	norr	_	(Prada)	(a)						
Telemed	icine Inforn	ned In	itiated all to		nanueu	OLeft (	Right		ON	:\$ D				-	
Consent	Obtained	P	rovider		-	() Ambi			ON	ot Specifie	d			I	
L	0							Rivarox (Xarelto	aban )						
Medi	cations & Al	lergies							OYe	is		FR	_		
> Notes						O No Not Specified			d	Medications					
Family C	ontact Infor	mation						Other An	nti-	- specific	-				
Toning Co	anact milor	mation						coagular	its /			1			

Figure 6.3: REACH Patient History and Medication Input

A common theme among paramedics was the need for consistency and structure in the neurologists' assessment of the patient so that they could better anticipate the actions they needed to perform for the neurologist. As Paramedic 2 stated,

One of the harder things for us is the fact that it's not standardized. Each individual doctor does it a little different every time they do it, so I would like to see a standard-- Give me a script, give the doctor a script, give the nurse script.

*Everybody knows what questions are going to be asked, in what order they're going to be asked, what every test is, what the expectation of the test is.* 

Further, Paramedic 6 mentioned how consistency could improve efficiency: "If there is more of a set guideline of, 'This is the order, this is what we're looking for, here is this process that were looking for, this is what I want the medic to do during these sessions.' If it was more everything is based on, then we would be more efficient in getting the same result." This standardization of the process was supported by Neurologist, 7 who said, "I think I would have a checklist in place. The medic, just go see a checklist and gives me all the pertinent information that I would need as opposed to just a medic trying to talk, 'Oh, this is a guy with this and that.' I think he should have a specific script to follow." While this can be supported with scripts or protocols as suggested by caregivers, interface design can provide a visual representation of the task list and the progress within a task to further enforce the use of the list. This design can be developed by applying gamification principles like progression mechanics to engage users to use the task progress map. A structure of the tasks and an indication of those completed can provide reinforcement of achievement and communicate progression (Robson et al., 2015).

The use of a progress map also supports the artifact principle, creating scaffolding, in the DiCoT model (Furniss & Blandford, 2006). According to this principle, the use of scaffolding or a physical representation of where we are in a task can simplify cognitive tasks. This design feature supports this principle by providing a visual of the tasks to be completed, determined from a task analyses of the stroke caregiving process, and an indication of our location in it. In addition, as paramedics stated, a consistent process and knowledge of upcoming tasks can improve their process and efficiency, both of which support the information flow principle, behavioral trigger factors, in the DiCoT model (Furniss & Blandford, 2006). This principle explains how individuals involved in a task can operate without a communicated plan, only responding to factors in their system. This scaffolding can provide the factors that trigger behaviors for paramedic caregivers such as moving in relation to the patient to test the peripheral vision for the neurologist.

The task monitoring feature created a need to change how data input should be presented to ensure that the protocol and movement through tasks are also supported by the design for imputing patient information and the neurologist' stroke assessment. For this reason, the data input process was redesigned. First, the neurologists' need for a large visual of the patient was made clear through observations and their responses in the interviews. Neurologist 6 explained how she resizes the windows in the current system: "I'm in a consult, the first thing that I do is I grab that far-- let's say left because I'm a physician, but the right side of the patient, and I pull it all the way to the right. Basically, it's almost all patient and then on the right is the NIH stroke scale, really skinny." However, other neurologists were not aware of this feature, often attempting to move the camera for a better view of the patient, something not currently possible in the ambulance but supported in hospital nursing carts. To support this use of the system, the proposed design prioritized making the data input a smaller portion of the screen and the visual of the patient, the primary focus for caregivers.

For this change to be effective, the data input itself was simplified, not merely made smaller. Data input items from the system were compiled into screens similar to its current tab structure, with changes based on the task analysis conducted and on feedback from the neurologist. One notable change was to move the medications and allergies data input into the history data input as these tasks are conducted simultaneously. This change addresses the frustration with the current system expressed by Neurologist 6:

Yes, we always ask what's your past medical history, surgical history, social history? What medications are you on? Do you have any allergies? It's not in a different section of the whole encounter. It's all right there, so you have to click around a lot. A lot of times when you ask people what's their past medical history, they'll be like, 'Oh, I don't have one. Oh well, maybe I have hypertension. I'm on Lisinopril for hypertension and I'm on Metformin.' Then you're like, so now you're in the medicine list, but now they're naming of medicines that prove that they have diabetes.

However, many neurologists also mentioned a facilitator in the current system being the checkboxes that allow for quick data input. As Neurologist 7 stated, "*I think it's easy to complete the consult or there's a lot of parts that are not free text that I can just click. That part I like a lot.*" In the current system, patient history utilizes this checkbox design, while medications and allergies are provided as text input as seen in Figure 6.3.

The new data input design uses non-free text input for most inputs with optional text boxes for unique responses. As a result of previous research identifying issues with the tab structure and the design of the current system and the need to align data input to the task progress monitor, the navigation through data input screens was also changed. The grouping of the all items for a task and the prescribed navigation of the new design are supported by pagination buttons. Users can move to the next screen when they are ready to complete the next tab by clicking the next button. These screens are presented in the order of the task flow in a circular menu, meaning that users can move forward or backward through the pages with no end points. As they reach the final page for input based on the task list, they can return to the first data input page by clicking the next, or forward, button. Examples of these screens can be seen in Figure 6.4. This feature supports the artifact principle, create scaffolding (Furniss & Blandford, 2006). The titles of the buttons and the title of the current page are additional place markers indicating where users are in the task flow as they progress through the tasks completing data input and moving to the next screen.

The foundation for the features developed for this design is a focus on creating a view of the system based on distributed cognition principles that emphasizes and prioritizes the information found to be important in previous research. For this reason, the new design proposed here is called the Prioritized Information Display and will be referred to by this name throughout the study.

Medical History		Surgical and Family History					
Patient History		Past Surgical History Dense Reported Coronary Artery Bypass Graft Carotid Endarterectomy Recent Surgeries* Other Surgical History*	Family History None Reported Stroke Coronary Artery Disease Hypoardial Infarction Hyportension Diabetes				
Past Medical History None Reported Arr Diabetes Type 1 Int Diabetes Type 2 Hey Coronary Artery Disease Va Stroke Tot Conner Akter Seizure	ial Fribiliation accanial Hemmonhage percholestrolemea fighend Vascular Disease vular Heart Disease accoo Use* hotol Use*		Aneurysm     Other Family History*				
Patient Info	Surgical & Family History	Nedical History	Medications				
Medications		Allergies					
Anti-Coagulants		Drug	Allergies				
Wafarin (Cournadin) R Heparin A Dabigatran (Pradaxa) Othe Ans Congulant Arts Panaiers	ivaroxaban (Xarelto)	Ace Inhibitor Alteplase er Drug Allergies					
Wafarin (Courmadin) R Hepsrin A Dabigatran (Pradaxa) Other Area Coognident Area Plantes Home Medications	ivaroxaban (Xareho)	Ace Inhibitor Alteplase er Drug Allergies					
Wafarin (Couradin) R Hoppin Dabigatran (Pradaxa) Dabigatran (Pradaxa) Home Medications Linisoprit (Printa) Cale Medicrain (Claucophage) G Atorvastatin (Lipitor) Atorvastatin (Lipitor)	Ivaroxaban (Xareho)	Ace Inhibitor Alteplase er Drug Altergies					
Wafarin (Courmadini) R C Wafarin (Courmadini) R C Debugatran (Pradaxa) Debu Arei Cosegularet Areis Plaastes Home Medications Linisopril (Prinivit) C Cu Metformin (Glucophage) G G Acovastani (Lipitor) A A Copisopril (Plavis) R C Copisopril (Plavis) R C HCTZ (Microzote) S S	Ivaroxaban (Karelio)	Ace Inhibitor Alteplase er Drug Altergies //ronment Altergies // Altergies					
Wafarin (Cournadin)     R       Heparin     A       Dabigatran (Pradaxa)     A       Other Area Coegularity Area Phanishis     A       Linisopri (Prinnk)     Ccc       Metformin (Glucophage)     G1       Atorxatatin (Lipitor)     Arr       Apprino     Im       Clopidoprel (Planki)     Rc       HCTZ (Microzide)     Sa       Metoprolo (Lopressor. Toprol)     Other Medications	Ivaroxaban (Carelto)	Ace Inhibitor Alteplase er Drug Allergies //ronment Allergies // Allergies // Allergies					

Figure 6.4: History, Medication, and Allergy Input for Proposed Design

# **Research Questions**

Based on the principles in the DiCoT model and previous research conducted on the current system, design features were developed to support distributed cognition and the demands of caregivers while improving general usability. Distributed Cognition Theory, specifically the DiCoT model, was selected to address the research questions for several reasons. First, because of the distributed location of the telemedicine team members geographically, the information flow and interaction with the artifacts in the DiCoT model helped us evaluate crucial elements of this complex system. Second, the focus on the interaction between not only the individuals in their environment but also the team
members inherent in distributed cognition will be useful for assessing how the system should be designed to accommodate for these interactions. While team situation awareness describes the knowledge gathering process of teams, it primarily focuses on the interactions between team members and the environment as they contribute to decision making. Finally, this study aims to improve the workload of the caregivers. The diagnosis and treatment of stroke patients require many decisions, but the focus of this research goes beyond the decision-making processes that a situation awareness model describes. The focus of this study is on cognitive processes on a broader scale, and the elements required to facilitate those processes so that multiple caregivers can contribute to providing quality patient care. Specifically, this research aims to investigate how interface design can support the distributed cognition of caregivers and improve the demands made on and the workload of the caregivers in the system. It investigates the following questions:

1. What interface features can support distributed cognition in telemedicineintegrated ambulances for stroke assessment?

2. How can the usability of the telemedicine system interface be improved and the workload of caregivers be reduced for telemedicine stroke assessment?

#### METHODOLOGY

### Participants

As the design features proposed here are primarily to support neurologists, their testing in comparison with the current system used neurologists or physicians with neurology experience as participants as they are the caregivers with experience in stroke caregiving and the NIHSS specifically. In addition, two internal medicine residents were also recruited as they had completed a stroke rotation during their residency. All other participants were neurology residents. To further ascertain they had the experience to complete a stroke evaluation, they were specifically asked if they had experience in treating stroke patients. To ensure an unbiased assessment of the current system and the Prioritized Information Display, these neurologists were recruited from multiple hospital systems in South Carolina and were additionally screened to eliminate participants with prior experience with REACH. Twenty participants meeting these criteria were recruited through contacts in Clemson University's Department of Public Health. The participant demographic information is shown in Table 6.2 below.

Characteristic	n	%
Gender		
Male	13	65%
Female	7	35%
Age		
20-30	5	25%
31-40	14	70%
41<	1	5%
Years of Experience		
>1	7	35%
1-5	5	25%
5-10	6	30%
10<	2	10%

Table	62.	Participant	Demographi	cs
1 4010	0.2.	1 un norpune	Demographi	•0

#### **Experimental setup**

A Zoom video conference call was used to connect the research moderator with the participants, with this platform's screen sharing and record meeting functions being used to capture the participants' experience in the telemedicine systems. Participants were asked to use a PC with a monitor measuring at least 13" diagonally. Their screen resolution ratios were checked to verify a setting of 16:9 or 16:10 to ensure a clear view of the system. Surveys were conducted through Qualtrics by emailing anonymous links before the study. The simulation, built in the Unity development program, includes both telemedicine interfaces and a wireless connection to the moderator. Interviews were conducted and recorded through the Zoom video conference system.

### **Experimental Design**

### Independent variables

This study used a repeated measures experimental design with a single independent variable involving 2 conditions: the current telemedicine system, REACH, and the new design, he Prioritized Information Display. All participants completed a stroke assessment in both conditions, which were counterbalanced to distribute order effects.

### Task Design

Further, two stroke diagnoses were used for the patient and paramedic responses. One involved a large vessel stroke, called a Middle Cerebral Artery (MCA) stroke, on the right hemisphere of the brain, presenting as major motor and sensory deficits including a forced gaze deviation and inattention to any stimuli on the left side of the patient, but with the patient still able to respond to commands from the participants, allowing them to evaluate the condition. The other was a small vessel stroke in the right hemisphere, which again allows for similar commands to be followed by the participants but presents significantly different symptoms, meaning a different diagnosis was required. Symptoms for the small vessel stroke includes some sensory and motor deficits, but not as severe as the Right MCA stroke, and limb ataxia which is a delay in coordination movements such as touching their finger to their nose. Both conditions were determined in collaboration with a telemedicine neurologist on the research team with extensive stroke experience and were considered equal in difficulty and interaction with the patient, but allowed participants to arrive at two separate diagnoses. These two diagnoses were also counterbalanced to account for learning effects.

### **Dependent** Variables

*Workload:* The National Aeronautics and Space Administration's Task Load Index (NASA-TLX) was used to measure the workload of the participants after each condition (Hart & Staveland, 1988).

*Usability:* The International Business Machines Computer Systems Usability Questionnaire (IBM-CSUQ) was used to measure the usability of each condition (Lewis, 2018).

*Performance:* The time taken to complete the assessment was measured by the simulation software, starting from the time the participants entered the simulation condition to the time the simulation ended excluding any surveys for each condition. The errors made during each assessment were defined as either conceptual or documentation errors. Conceptual errors are defined as an error in selecting or misunderstanding the patient

history or the NIHSS levels even when the correct information was provided in the patient responses. Documentation errors are defined as omitted or documentation of relevant history or medications in an incorrect location.

*Situation awareness:* Scores on a situation awareness global assessment technique or SAGAT were assessed twice during each assessment (Endsley, 1988), once after completing the patient history documentation and once after completing the assessment.

*Preference:* During the interviews, participants were asked which system they preferred for evaluating a stroke patient in an ambulance, their responses representing the measurement.

Interview responses were obtained to determine support for the distributed cognition in each system in addition to evaluations of the system to support the quantitative findings.

### Procedure

Participants were contacted to schedule a two-hour window for the study. Prior to their scheduled time, participants were sent an email with a Zoom meeting link to connect to the research moderating the study, links to the surveys in the study and a file for downloading the simulation program. This program provided the REACH interface and the Prioritized Information Display in a format that could be easily downloaded to most PC laptops. Before the participants began the study, they connected with the researcher moderator through Zoom to ensure the program was downloaded and that their screen resolution was set to the correct image ratio for viewing the simulation.

The procedure task flow of the study can be found in Figure 6.5. First, the researcher provided an overview of the tasks to be completed; then the informed consent and

demographic survey was administered through one of the survey links provided to the participant through email. Next, the participants were shown a brief training session, which included screenshots of the simulator for both conditions and identified where participants could find and document information during the study. After the training, participants began the simulation program and were directed to the simulation they were to complete. In it, the participants were asked to complete a patient history and input a full NIHSS exam. They obtained the information they needed to complete the assessment through pre-recorded video responses from a standardized patient, for this study a retired nurse, and an assistant acting as a paramedic. The videos were recorded in an environment emulating an ambulance with the camera view and positioning of the patient and paramedic similar to the views currently available in the REACH system.

The patient and paramedic were presented with questions or commands from a list generated from the task analysis completed in previous research (Rogers et al. 2020) and commands observed in previous simulated stroke sessions (Rogers et al. 2020). This list was also confirmed with a consulting telemedicine neurologist to ensure all appropriate questions and responses for the two diagnoses were included. Participants in the study completed the assessment by communicating verbally to the patient or paramedic, and the moderator selected the response video that best matched the request or question through a Wizard of Oz connection to the simulation (Steinfeld et al., 2009). Participants were made aware that responses were simulated, and if a question was asked that was outside the scope of the simulator, the moderator alerted the participant to ask yes or no questions to collect the information. In addition, pilot testing of the system with 3 consulting physicians was

conducted to ensure responses and transitions were appropriate for the session. When the participant completed the NIHSS exam and arrived at a diagnosis for the patient, the simulation was stopped. Once during the simulation and once at the end, participants were asked to answer questions about the patient and their diagnosis to assess their situation awareness, following the SAGAT protocol.

After the simulation, the participant was directed to complete a survey, which contained the NASA-TLX and IBM-CSUQ assessments, evaluating only their experience with the telemedicine system, not the content of the video simulations. The participant was then asked to complete the second simulation and the surveys in the same manner. When both simulations and follow-up surveys were complete, the participant was asked to complete a short semi-structured interview about their experiences in the two systems focusing on any difficulties, their feedback on the design, their identification of the features supporting distributed cognition, and their preferences. After the interview was completed, the participants were emailed their incentive, a \$100 Amazon gift card.



Figure 6.5: Study Procedure Task Flow

#### Measures

During the simulation SAGAT was used to measure the participant's situation awareness (Endsley, 1988). Two freezes were conducted, once after the patient information was obtained and inputted (10 seconds after moving to the NIHSS data collection) and once after the simulation ended. At each freeze questions pertaining to Level 1, 2, and 3 situation awareness were asked about patient information including vitals, medications, the impact of the history on the assessment, and the diagnosis. These questions were developed in coordination with a telemedicine neurologist on the research team and focused only on items for which an awareness would need to be maintained by the participants for assessing a stroke. The questions asked for each diagnosis and freeze can be found in Appendix N. SAGAT was used as a measure to assess situation awareness since currently there is no validated quantitative measure of distributed cognition.

Time to complete the assessment tasks was measured during the simulation as well as any conceptual or documentation errors made during the assessment. The NASA-TLX survey was used to determine the participant's perceived workload for each design condition, and the IBM-CSUQ survey was used to determine the perceived usability of each condition. Interviews were conducted using a retrospective think-aloud protocol to determine participant's feedback on the design, asking, for example, what they thought about the layout of the information in the first interface and to assess how they perceived the support for distributed cognition in each interface through such questions as "How did each system perform in creating scaffolding (a reminder of where you are in the task)?" The semi-structured interview protocol can be found in Appendix O. The interviews were audio recorded and later transcribed for content analysis.

#### Analysis

All quantitative dependent variables were analyzed using paired t-tests to determine the differences, if any, between the two systems. The SAGAT responses were analyzed by taking the average accuracy for each freeze and for each situation awareness level within each freeze and conducting a paired t-test on the average accuracy to determine if the responses at each freeze vary by system. Additionally, preference between the two systems determined in the interviews were analyzed with a Chi-Square goodness of fit test to determine if the occurrence of preference between systems observed was statistically different from the assumption that systems are preferred equally, meaning 50% preferred each system. Finally, a content analysis was conducted on the transcripts of the follow-up interviews to determine commonalities in them, the feedback for both systems and how both systems support distributed cognition. The transcripts were organized into Excel spreadsheets by question groups, and for each group coding categories were identified for each participant answer. For example, for each DiCoT model question and for each participant answer, the categories included which system supports the principle, the features cited, which system is preferred, and the recommendations made. Two coders independently evaluated each participant's answers. Sample pilot coding was conducted using 20% of the interviews to determine if the categories were appropriate. The overall percentage agreement for the pilot coding was 82%, and categories were determined to effectively summarize the data. Final coding was conducted using the same categorization for the rest of the interviews. The final overall percentage agreement was 84%. Percentages of answers for each category and a list of features or recommendations were generated for each question group for the analysis of the participant interviews.

#### RESULTS

The first objective of this study was to determine the effects of the system design on the perceived workload, situation awareness, and task performance of participants for each condition, and a second was to determine the effect of system design on the usability and support for distributed cognition for each condition.

#### Workload

#### **Total Workload**

Total workload experienced by the participants during each stroke evaluation was measured subjectively using the NASA-TLX tool. A paired t-test was conducted to determine the difference between the system conditions on the subjective total workload experienced by participants. As shown in Figure 6.6, the average total workload for the REACH condition (M = 41.30, SD = 19.89) was higher than that for the Prioritized Information Display condition (M = 38.23, SD = 19.99). However, no significant difference in the total workload was observed among the participants t(19) = 1.34, p = 0.19.

# Mental Demand

The perceived mental demand experienced by the participants during each stroke evaluation was measured subjectively using the NASA-TLX tool. A paired t-test was conducted to determine the difference between system conditions on the perceived mental demand experienced by participants. As shown in Figure 6.6, mental demand was higher for the REACH condition (M = 46.65, SD = 29.52) than for the Prioritized Information Display condition (M = 46.15, SD = 28.95). However, no significant difference in the mental demand was observed among the participants t(19) = 0.18, p = 0.86.

#### **Physical Demand**

The perceived physical demand experienced by the participants during each stroke evaluation was measured subjectively using the NASA-TLX tool. A paired t-test was conducted to determine the difference between system conditions on the perceived physical demand experienced by participants. As shown in Figure 6.6, the physical demand was higher for the REACH condition (M = 12.30, SD = 13.33) than for the Prioritized Information Display condition (M = 9.95, SD = 13.69). However, no significant difference in the physical demand was observed among the participants t(19) = 1.18, p = 0.25.

### **Temporal Demand**

The perceived temporal demand experienced by the participants during each stroke evaluation was measured subjectively using the NASA-TLX tool. A paired t-test was conducted to determine the difference between system conditions on the perceived temporal demand experienced by the participants. As shown in Figure 6.6, temporal demand was higher for the REACH condition (M = 49.05, SD = 27.37) than for the Prioritized Information Display condition (M = 44.00, SD = 27.66). However, no significant difference in the temporal demand was observed among the participants t(19) = 1.34, p = 0.19.

### Performance

The performance perceived by the participants during each stroke evaluation was measured subjectively using the NASA-TLX tool, with higher performance ratings indicating a lower perceived performance. A paired t-test was conducted to determine the difference between system conditions on the perceived performance of participants. As shown in Figure 6.6, perceived performance was lower for the REACH condition (M = 35.75, SD = 26.07) than for the Prioritized Information Display condition (M = 29.90, SD = 22.79). However, no significant difference in the perceived performance was observed among the participants t(19) = 1.88, p = 0.08.

### Effort

The effort perceived by the participants during each stroke evaluation was measured subjectively using the NASA-TLX tool. A paired t-test was conducted to determine the difference between system conditions on the participant's perceived effort. As shown in Figure 6.6, perceived effort was lower for the REACH condition (M = 35.7, SD = 25.75) than for the Prioritized Information Display condition (M = 37.80, SD = 25.39). However, no significant difference in the perceived effort was observed among the participants t(19) = -0.73, p = 0.47.

### Frustration

The frustration perceived by the participants during each stroke evaluation was measured subjectively using the NASA-TLX tool. A paired t-test was conducted to determine the difference between system conditions on the perceived frustration of the participants. As shown in Figure 6.6, perceived frustration was higher for the REACH condition (M = 35.75, SD = 21.29) than the Prioritized Information Display condition (M

= 28.30, SD = 17.62). However, no significant difference in the perceived frustration was observed among the participants t(19) = 1.72, p = 0.10.



Figure 6.6: Average Perceived Workload

# Usability

# Overall

The perceived usability of the system was measured subjectively using the IBM-CSUQ tool. A paired t-test was conducted to determine the difference between system conditions on the perceived usability. As shown in Figure 6.7, perceived usability was lower for the REACH condition (M = 5.61, SD = 1.07) than for the Prioritized Information Display condition (M = 5.71, SD = 0.98). However, no significant difference in the perceived usability was observed among the participants t(19) = -0.62, p = 0.54.

#### System Usability

The perceived system usability of the system was measured subjectively using the IBM-CSUQ tool. A paired t-test was conducted to determine the difference between system conditions on the perceived system usability. As shown in Figure 6.7, perceived system usability was lower for the REACH condition (M = 5.68, SD = 1.17) than for the Prioritized Information Display condition (M = 5.89, SD = 1.01). However, no significant difference in the system usability was observed among the participants t(19) = -1.20, p = 0.24.

#### Information Quality

The perceived information quality of the system was measured subjectively using the IBM-CSUQ tool. A paired t-test was conducted to determine the difference between system conditions on the perceived information quality. As shown in Figure 6.7, perceived information quality was lower for the REACH condition (M = 5.53, SD = 0.97) than for the Prioritized Information Display condition (M = 5.60, SD = 0.91). However, no significant difference in the information quality was observed among the participants t(19) = -0.40, p = 0.69.

#### Interface Quality

The perceived interface quality of the system was measured subjectively using the IBM-CSUQ tool. A paired t-test was conducted to determine the difference between system conditions on the perceived interface quality. As shown in Figure 6.7, perceived interface quality was higher for the REACH condition (M = 5.63, SD = 1.13) than for the Prioritized Information Display condition (M = 5.46, SD = 1.23). However, no significant difference in the interface quality was observed among the participants t(19) = 0.73, p = 0.48.



*Figure 6.7*: Average Usability

### Situation Awareness

The SAGAT responses were recorded as 0 (if the response was incorrect) and 1 (if the response was correct). Each SAGAT freeze was analyzed separately. The averages of the responses for each level of situation awareness and the overall average were calculated to create the accuracy measures. The accuracy measures for each level and overall were then analyzed to determine the difference between the system conditions.

# SAGAT Freeze 1

*Overall:* The overall SAGAT accuracy for the participants for Freeze 1 was calculated using the average response to all of the SAGAT questions. A paired t-test was conducted to determine the difference between system conditions on the overall SAGAT accuracy. As shown in Figure 6.8, overall SAGAT accuracy was lower for the REACH

condition (M = 0.85, SD = 0.15) than the Prioritized Information Display condition (M = 0.91, SD = 0.11). However, no significant difference in the overall SAGAT accuracy was observed among the participants t(19) = -1.76, p = 0.09.

*Level 1:* The Level 1 SAGAT accuracy of participants was calculated using the average response to only Level 1 SAGAT questions. A paired t-test was conducted to determine the difference between system conditions on the Level 1 SAGAT accuracy. As shown in Figure 6.8, Level 1 SAGAT accuracy was lower for the REACH condition (M = 0.82, SD = 0.18) than for the Prioritized Information Display condition (M = 0.90, SD = 0.09). However, no significant difference in the Level 1 SAGAT accuracy was observed among the participants t(19) = -2.03, p = 0.06.

*Level 2:* The Level 2 SAGAT accuracy of the participants was calculated using the average response to only Level 2 SAGAT questions. A paired t-test was conducted to determine the difference between system conditions on the Level 2 SAGAT accuracy. As shown in Figure 6.8, Level 2 SAGAT accuracy was the same for both the REACH condition (M = 0.88, SD = 0.22) and the Prioritized Information Display condition (M = 0.88, SD = 0.22). No significant difference in the Level 2 SAGAT accuracy was observed among the participants t(19) = 0.00, p = 1.00.

*Level 3:* The Level 3 SAGAT accuracy of the participants was calculated using the average response to only Level 3 SAGAT questions. A paired t-test was conducted to determine the difference between system conditions on the Level 3 SAGAT accuracy. As shown in Figure 6.8, Level 3 SAGAT accuracy was lower for the REACH condition (M = 0.95, SD = 0.22) than for the Prioritized Information Display condition (M = 1.00, SD =

0.00). However, no significant difference in the Level 3 SAGAT accuracy was observed among the participants t(19) = -1.00, p = 0.33.



Figure 6.8: Average Accuracy of SAGAT Freeze 1

# SAGAT Freeze 2

*Overall:* The overall SAGAT accuracy of the participants for Freeze 2 was calculated using the average response to all SAGAT questions. A paired t-test was conducted to determine the difference between system conditions on the overall SAGAT accuracy. As shown in Figure 6.9, the overall SAGAT accuracy was lower for the REACH condition (M = 0.84, SD = 0.13) than for the Prioritized Information Display condition (M = 0.89, SD = 0.10). However, no significant difference in the overall SAGAT accuracy was observed among the participants t(19) = -1.28, p = 0.22.

*Level 1:* The Level 1 SAGAT accuracy of the participants was calculated using the average response to only Level 1 SAGAT questions. A paired t-test was conducted to determine the difference between system conditions on the Level 1 SAGAT accuracy. As shown in Figure 6.9, Level 1 SAGAT accuracy was lower for the REACH condition (M = 0.76, SD = 0.19) than for the Prioritized Information Display condition (M = 0.85, SD = 0.15). However, no significant difference in the Level 1 SAGAT accuracy was observed among the participants t(19) = -1.58, p = 0.13.

*Level 2:* The Level 2 SAGAT accuracy of the participants was calculated using the average response to only Level 2 SAGAT questions. A paired t-test was conducted to determine the difference between system conditions on the Level 2 SAGAT accuracy. As shown in Figure 6.9, for the Prioritized Information Display condition (M = 1.00, SD = 0.00). No significant difference in the Level 2 SAGAT accuracy was observed among the participants t(19) = 0.00, p = 1.00.

*Level 3:* The Level 3 SAGAT accuracy of the participants was calculated using the average response to only Level 3 SAGAT questions. A paired t-test was conducted to determine the difference between system conditions on the Level 3 SAGAT accuracy. As shown in Figure 6.9, Level 3 SAGAT accuracy was the same for the REACH condition (M = 0.93, SD = 0.18) as for the Prioritized Information Display condition (M = 0.93, SD = 0.18) as for the Prioritized Information Display condition (M = 0.93, SD = 0.18). No significant difference in the Level 3 SAGAT accuracy was observed among the participants t(19) = 0.00, p = 1.00.



Figure 6.9: Average Accuracy of SAGAT Freeze 2

# Performance

### Time

The time taken to complete the assessment was measured by the simulator in seconds from when the participants entered the system to when they verbally stated they had completed their assessment without the time taken to complete the SAGAT freezes. A paired t-test was conducted to determine the difference between system conditions on the time taken to complete the assessment. A significant difference in time taken was observed among the participants t(19) = 2.10, p = 0.049. More time was needed for REACH condition (M = 791.05, SD = 189.62) than for the Prioritized Information Display condition (M = 731.81, SD = 128.21).

# Errors

*Total Errors:* Errors were documented by the moderator during the study session, with occurrences being totaled for two types of errors, documentation and conceptual. A paired t-test was conducted to determine the difference between system conditions on the total number of error occurrences. As shown in Figure 6.10, total error occurrences were higher for the REACH condition (M = 0.95, SD = 0.89) than for the Prioritized Information Display condition (M = 0.45, SD = 0.76). However, no significant difference in the total error occurrences was observed among the participants t(19) = 2.03, p = 0.06.

*Documentation Errors:* Documentation errors were documented by the moderator during the study session. A paired t-test was conducted to determine the difference between system conditions on the number of documentation error occurrences averaged over participants. A significant difference in documentation error per participant was observed t(19) = 2.98, p = 0.01. As shown in Figure 6.10, documentation error occurrences were higher for the REACH condition (M = 0.85, SD = 0.75) than for the Prioritized Information Display condition (M = 0.30, SD = 0.57).

*Conceptual Errors:* Conceptual errors were documented by the moderator during the study session. A paired t-test was conducted to determine the difference between system conditions on the number of conceptual error occurrences averaged over participants. As shown in Figure 6.10, conceptual error occurrences were fewer for the REACH condition (M = 0.10, SD = 0.31) than the Prioritized Information Display condition (M = 0.15, SD = 0.37). However, no significant difference in the conceptual error occurrences was observed among the participants t(19) = -0.44, p = 0.67.



Figure 6.10: Average Errors by Type

# Preference

In the interview, participants were asked which system they preferred for telemedicine stroke assessment in an ambulance setting. A Chi Square goodness of fit test was conducted on the number of participants who responded that they preferred REACH and the number who preferred the Prioritized Information Display to determine if there was a difference between the two systems. The number of participants who preferred REACH was 6 compared to 14 who preferred the Prioritized Information Display. However, there was no significant difference between the observed preference counts compared to the expectation that they would be evenly distributed  $\chi^2(1, N=20) = 3.2, p = 0.07$ . The reasons for preferring the Prioritized Information Display varied among the participants. Of those participants who explained the reason for their preference, 66% mentioned the guided

navigation, the completion of the tasks in a predefined order and the use of only next buttons to navigate in the system. Other reasons mentioned included the streamlined presentation of the documentation, the task progress bar, and the vitals presentation. Participant 15 described their preference, saying

I like the second system [Prioritized Information Display] a little better just because the workflow is linear and they are streamlined that until you complete one task you're not navigating to the other. Having a big graphical interface for the vitals, I like the second system [Prioritized Information Display] better. – P15

Approximately a third of the participants who preferred the REACH system offered a reason. The ability to input more documentation on one screen, or general usability, were the only explanations given when asked for their preference. Of the participants who preferred REACH, two participants (11 and 17) mentioned that it was closer to the system they currently use or their typical process:

The first system [REACH], it was more similar to what I've used in the past. I think it was about as equally organized, so it was more familiar to me than the second system [Prioritized Information Display]. - P17

### **Distributed Cognition**

The responses to interview questions asking participants if the systems supported principles of distributed cognition were divided fairly evenly as at least 45% of participants believed that both systems supported each principle. To assess support of the information flow principle of information movement, the first question focused on obtaining the information needed in each system. Most participants, 70%, felt both systems supported

this principle; however, 55% expressed a preference for the Prioritized Information Display, with 15% preferring REACH and 30% having no preference. In addition, 5% of the participants thought only REACH supported this principle, 20% only the Prioritized Information Display, and 5% neither. Participant 7 explained their thoughts on the information gathering in each system, saying

I think both of them allowed you to gather information correctly. The second one [Prioritized Information Display] though, because it put everything on the main page, you could quickly see what drugs they were taking, if they had any allergies, with thrombolytic, and blood thinners and stuff. That's just information that you would remember as soon as you ask the patient, but say if you weren't to remember that, the second system [Prioritized Information Display] would show that on the page with the little boxes in the middle page. I think it gives you a big picture when you're trying to come up with a diagnosis and just make an assessment. -P7

The most common features mentioned as specifically supporting this principle were the streamlined interface (25% of participants) and the large graphical representation of vitals in the Prioritized Information Display (20%).

A second information flow principle, information transformation, was assessed by asking participants about the level of detail in each system, with 75% of the participants thinking both systems were equal in level of detail, 10% only REACH had sufficient detail, and 15% only the Prioritized Information Display had sufficient detail. In addition, Additionally, 35% of the participants indicated no preference in the level of detail in either systems, 30% indicating that they preferred the Prioritized Information Display and 25%

that they preferred REACH. The primary feature mentioned in reference to the level of detail was the streamlined layout of the Prioritized Information Display (20% of participants), with others stating more specifically that the organization of the documentation for components like medications or patient history in the Prioritized Information Display was an important feature (20%). Participant 6 described their thoughts on the relative level of detail, saying

I don't recall seeing the antiplatelet questions on the [REACH] like there was on the second one [Prioritized Information Display]. I think [REACH] was decent, but it certainly seemed to be missing some key things. It just didn't seem like it was developed by a neurologist if you will. [chuckles] They were both adequate. I think the second one [Prioritized Information Display] was more relevant to what I need as a neurologist trying to determine whether a patient is, A, having a stroke and, B, a TPA candidate or a thrombectomy candidate. – P6

Trigger factors was another principle assessed during the interviews, with the participants being asked if any features of either interface triggered or cued them to complete a task. Only 45% of the participants expressed that both systems had some form of trigger factor, 25% thought neither did, 25% said only the Prioritized Information Display, and 5% said only the REACH system had trigger factors. Half of the participants expressed a preference for the Prioritized Information Display in terms of supporting trigger factors, the same 5% who stated only REACH had trigger factors preferred that system, and 45% had no preference. Features that participants identified as supporting trigger factors were the labeling (35% of participants) and guided navigation (15%) in the

Prioritized Information Display, and the tPA window countdown clock (5%) and documentation headers (5%) in REACH. Participant 1 describes the features they felt supported trigger factors:

I know both systems had all of those columns [surgical, family history]. It would remind us to make sure we ask those questions. The first one [Prioritized Information Display] was just better labeled, I think. – P1

Principles in the artifacts component of the DiCoT model were also assessed, the first being creating scaffolding. The participants were asked if any features in either system created a reminder of where they were in their task list, with 50% saying both had some features, 35% only the Prioritized Information Display, 5% only REACH, and 10% thought neither system included scaffolding features. Most participants believed that the Prioritized Information Display offered better scaffolding (60%), while 10% felt REACH did and 30% didn't state a preference for either system. Several of the Prioritized Information Display features that participants identified as creating scaffolding were its guided navigation (45% of participants) and the task progress bar (10%). Participant 18 describes the guided navigation in creating scaffolding:

I don't remember any alert or anything, but I do feel like system two [Prioritized Information Display], just navigating through it, it was like a list reminder. It was all in perfect order. – P18

The REACH system features that were identified included the tab navigation structure (5%) and the collection of all patient history on one tab (5%).

The second artifact principle assessed was representation goal-parity, which describes the way a system can represent how near or far a user is from a goal in their tasks. Participants were asked how both systems performed in the representation of their place in relation to such goals as the end of the tPA window, normal patient vitals, or the end of the assessment. Most participants, 65%, stated that both systems supported this principle, 10% only the Prioritized Information Display, 10% only REACH, and 15% stated that neither system supported this principle. Most participants, 60%, thought REACH performed better in supporting this principle, with 65% of participants specifically listing the tPA countdown clock as the feature they believed supported representation goal-parity. Participant 2 described this feature:

I think the first system [REACH] was a little easier for me personally to figure out the time to three hours and training time to four and a half hours. I tried to look for something comparable in the second system [Prioritized Information Display], and I noticed that it was a little different. Maybe the time the thrombolysis incorporating the last known well equals of the time of symptom onset, and then directly giving us a countdown, I think is a very neat feature. – P2

Those that thought the Prioritized Information Display performed better (25%) listed the graphical representation of vitals (15% of participants), color coding of vitals (20%), and the visual representation of the tPA window (20%). Participant 15 described their use of the timeline:

The second system [Prioritized Information Display], I was able to track it. ... I think I have to also keep a tab on the blood pressure because if the pressure is so high, you cannot administer the drug. Rather than focusing on the patient until the time and then I decide to do the tPA, then if the blood pressure is 250/110, then, I'm stuck on how to decide on giving those medications first before I can administer the tPA.... I think having that vital graph or trend, information for the physician also helps him as the EMT to work on the blood pressure with the patient. Following my commands, I can actually do the NIH myself. We can prepare the patient to as close as possible to get the medication as soon as the CT scan is done, because we actually have a good control of the blood pressure and things like that. I think it allows us some freehand when the physician is able to have that vital trend information and presented to him in a nicer way. That's how I felt the system to update the job in a more elegant fashion compared to system one [REACH]. – P15

### **Benefits of Telemedicine**

Almost all participants, 95%, saw a general benefit in using these telemedicine systems for ambulance-based stroke caregiving. The most common benefit listed was increased efficiency in patient care (60%). Participant 14 describes the benefit of using either of these telemedicine systems:

Being able to evaluate the patient before they even come in is a huge advantage. Being able to preemptively prepare and to have a-- For example, if they are applicable for the tPA, like to get them right away and be able to effectively treat them without wasting any time. I think it could be a huge advantage moving forward to be able to utilize that. – P14 No participants discussed a specific benefit of the REACH system over the Prioritized Information Display; however several participants, 15%, described benefits of the Prioritized Information Display specifically, listing a streamlined view of documentation, guided navigation, and larger view of the patient creating a more reliable assessment. Participant 8 describes their thoughts on the relative benefits between the systems:

Obviously, for both of them, I feel like the time aspect. They're pretty quick, you really get through and get it in NIHSS score quickly, and then you're able to see that tPA window, make sure you know what you're dealing with, whatever a stroke or not pretty quickly with both systems, honestly. I will say that the first one felt quicker [Prioritized Information Display], just because there's less to distract me from it and my goal is to be efficient. The first one [Prioritized Information Display] I feel like is doing a better job and my goal was to get a great history and chart awesomely. – P8

#### DISCUSSION

This study investigated the effect of the design changes implemented in the Prioritized Information Display on the perceived workload, situation awareness, and performance of the participants and the usability evaluation of the system. These changes were developed based on the information flow and artifact principles of the DiCoT model and feedback from neurologist, nurse, and paramedic caregivers currently using telemedicine for stroke care in ambulances. In general, trends were seen in improved workload, usability, situation awareness, and task performance for the Prioritized Information Display compared to the REACH telemedicine system currently in use. However, the dependent variables of time to assessment and documentation errors were statistically significant.

NASA-TLX workload measures six subscales, one being mental demand. On average mental demand was higher for participants using the REACH telemedicine system; several theories could explain this change. First, one of the main differences between the REACH system and the Prioritized Information Display is the relative video sizes of the patient and paramedic in the ambulance setting, with the latter providing a larger view. In the interviews, 50% of the participants indicated that the visibility of patients' movements impacted their assessment, with the smaller video making it more difficult to rely on their view of the patient. Participant 14 described their experience:

For the [Prioritized Information Display], the other thing that I preferred about it was the fact that the video is bigger. Because on the [REACH], I was having trouble trying-- I was going more by what the paramedic was saying, then being able to actually see myself what was going on. Whereas in the [Prioritized Information Display], I thought having the bigger video was definitely an advantage. - P14

As many participants mentioned in the interviews in the previous study, a video of the patient improves the ability to consult over the phone; although patients are not seen face to face by neurologists, video allows for quality communication. A study conducted on distanced collaborating teams of software engineers found using video in addition to audio adds to or improves the ability to show understanding, forecast responses, give non-verbal

information, enhance verbal descriptions, manage pauses and express attitudes (Isaacs & Tang, 1993). In addition, incorporating quality video for increased visibility is a contributing factor to mental demand as the better the quality, the better movements can be seen and interpreted, making the assessment easier and more reliable. Similar effects of video quality have been found in studies on video conferencing systems for digital learning investigating the bandwidth and size presentation for non-verbal communication and quick visual confirmation of movements (Watson & Sasse, 1996).

Mental demand could also be increased in the REACH system because of the difficulty participants had in finding documentation. Of those mentioning having any issues in the systems they evaluated (55% of participants), 72% specifically mentioned having difficulty finding medication documentation or the correct location for documenting the patient history in REACH, citing the unclear tab labels. Participant 10 describes their difficulty in documentation despite the training they received:

Despite you telling me where they were, when I got to it [in REACH], it wasn't obvious to me which tab to push to get to those allergies and medication so I just scripted them. - P10

Design and labelling of navigation structure are essential for clear and efficient movement through any system. These results are consistent with a usability study of a tab navigation system for a college library website that found increased specificity of tab labeling created more use of the pages in the structure (Pittsley & Memmott, 2012).

Another subscale of workload, physical demand, was also seen to increase when using the REACH system. One possible reason identified by participants in their interviews was the amount of scrolling and movement between tabs required to accomplish the tasks in REACH compared to the Prioritized Information Display, which concentrated the input on a small area of the screen. According to Participant 8, it was hard to manipulate the display in REACH:

Honestly, I feel like the information was harder. I had a scroll bar in the left, it was hard. I had to manipulate. I just was not able to see everything right away as readily as the previous one [Prioritized Information Display]. - P8

Almost half the users in this study (45%) were using trackpads to complete their stroke evaluation. Although we did not see an increase in the amount of time taken to complete the assessment when using a trackpad compared to a mouse, research in ergonomics investigating the two found that users indicated that the scrolling techniques on their trackpads were exhausting and jumpy compared to mouse control or positioning with the trackpad (Bial et al., 2010).

Temporal demand, or time pressure felt by participants, another subscale of workload, also was higher for the REACH system, perhaps because of the clutter of the screen, a problem mentioned by 45% of participants. As Participant 15 describes,

System 1 [REACH] it's always been problem is like a lot of boxes, colors are not that good because there's no differentiation between what is what. Everything is one color. It's a very busy interface, whether it is scrolling off a lot of up or down. I think, for physicians, especially, you're evaluating the patient, I think if it's less of a use of a mouse, or some sort of scrolling, it really helps because sometimes they don't know how to scroll that they miss that there is a scrolling element in an interface, sometimes they may not be able to see where to document information. That can create frustration and can increase the amount of pressure they feel compared to the linear system where you present one information or one element of the clinical evaluation at once. - P15

REACH also includes additional documentation items not pertinent to acute stroke consult, such as the review of the systems assessment, that could contribute to the increase in time pressure. Participant 6 expressed their opinion about the additional documentation required:

I think if you're going to have review systems on there, then have the ones that are only pertinent to your stroke alert or then just other. There were some things in there that I was like, "What? No, this a--"They had sleep apnea, nocturnal dyspnea in there. I was just like in a way, it makes you sometimes feel like you have to ask that question, but then it's like, first of all, sleep apnea is a diagnosis and not a review systems. - P6

Perceived performance on the tasks is another measure that was higher in the REACH telemedicine system. This result could be tied to the increased number of documentation errors made in this system when participants either omitted documentation or documented in the incorrect input. Participants' interviews pointed to time as the cause for these errors, the tradeoff being the time to find the correct documentation area and completing the patient evaluation. Participant 1 explained their thought process for finding medication documentation:

Yes, I was just putting it wherever I could because I just wanted to make sure she was not on any blood thinners and to remind myself what all she is on. That's why I just wrote it down wherever I could find an empty box, I guess, to type in . - P1

Difficulty finding documentation could be linked to the issues with the tab labelling described earlier when discussing mental demand, but the medication and allergies documentation was located at the bottom of the screen. In a study of web surveys comparing scrolling to clicking through pages to complete a questionnaire, participants in the scrolling condition recorded higher rates of item non-response, meaning within pages requiring scrolling, documentation could be difficult to find and, thus, not worth the effort to complete (Peytchev et al., 2006).

The final subscale of workload found to be higher for the REACH system was frustration. This finding is also supported by comments made by the participants in the interviews. Scrolling was a theme that emerged again in these comments as seen in Participant 15's description of the stress involved when scrolling in REACH

There is a lot of up and down scrolling that needs to happen [in REACH], which again especially when you're trying to figure out and you're also trying to perform a stressful assessment, but with another person. Having less grueling and less information on one screen with that load helps. - P15

A usability study of scrolling websites found similar low ratings for navigation and orientation, suggesting confusion and loss of orientation when using scrolling designs (Koukouletsos et al., 2014). The tab structure in REACH was also identified as a source of frustration for participants. As Participant 9 said about their experience with REACH,

Interviewee: Yes. I was trying to figure out where everything was and everything. Fonts were a little bit tiny for the vitals and all that.

Interviewer: When you were talking about trying to figure out where everything was, do you feel like that made it take longer or was that more frustrating for you having to-

Interviewee: Yes, I think more frustrating. - P9

Unnecessary documentation was another aspect of REACH that was mentioned in connection with frustration in this study. Participant 6, in particular, mentioned the review of systems as a frustrating aspect of REACH:

That actually frustrated me because when I was going through it, the things that were listed on the review systems didn't even seem very relevant. - P6

Findings from research in form design recommend showing only input for those fields which are necessary (Beaumont et al., 2002). Form design research also provides some context for why unnecessary documentation in REACH may increase frustration as REACH provides many options for documentation with no indication of required input tasks in the system. Providing formatting or indication of required input is a recommendation based on findings from usability testing of documentation forms (Linderman & Fried, 2004).

The only subscale with a higher workload in the Prioritized Information Display than in REACH was effort. Few participants mentioned any increase in effort to complete their assessment during the interview, one participant (11) commented that the guided navigation system required more clicks in the system: It [Prioritized Information Display] felt more clicky like I have to go from, for example, this screen for medical history and then I have to do surgical and family like the histories are not all on one screen as the meds. At least practically I think of my history is all one thing. - P11

The documentation for patient history in the Prioritized Information Display was divided into two documentation screens to avoid scrolling, medical history on conditions and substance use on one page and family and surgical history on the other. Usability research supports the use of pagination rather than scrolling to improve data input completion and time to complete data input (Peytchev et al., 2006).

Usability was also, on average, rated higher in the Prioritized Information Display (M=5.71) than in the REACH system (M=5.61). The first of the indices used to measure usability was system usability. These data collected through the IBM-CSUQ are supported by findings from the interviews. When participants were asked about the ease of use for each system individually, the Prioritized Information Display was referenced as easy to use by 100% of participants compared to 55% for the REACH system. Participant 13 describes the ease of use of each, saying

It [Prioritized Information Display] was easier to use, and the layout was easier to interpret relative to the second system [REACH]. The second system had more, but there was too much going on at the screen at one time. - P13

Functionality, however, participants felt was about equal in the systems as referenced by Participant 5:

Like I said, I feel like functionally, they're pretty similar where they both do the job.

*- P5* 

Information Quality, another IBM-CSUQ index, was rated higher for the Prioritized Information Display. Although most participants thought the level of detail in the systems was the same (75%), several (10%) believed there was more detail in the Prioritized Information Display, citing the checkbox items as Participant 8 did:

I think when I was trying to pick options, it was nice and there wasn't a lot of options [in Prioritized Information Display]. There weren't too many quick bubbles because otherwise, it takes me— in some ways, I could type faster hypertension than I could if I needed to look down the list and find hypertension and click it, but there weren't too many options which made it more amenable to that. - P8

Others (15%) felt there was more detail in REACH, with more data input locations as Participant 7 describes:

I feel like there were more options to select on that, not on the scale, but on the history tabs, I feel because I'm trying to think if I had to run through a more elaborate list on the first one [REACH]. I'm just trying to think, because the second one [Prioritized Information Display] seemed more concise and also with that--You can correct me if I'm wrong. Maybe that was because things were sort of squished together, so it'd look like a smaller, list but for the first one, I felt like there was a larger list to go through. - P7

However, the relevancy of the extra data input was questioned by some participants, in particular Participant 6:
That actually frustrated me because when I was going through it [REACH], the things that were listed on the review systems didn't even seem very relevant. - P6 Usability guidelines for designing web-based forms provides some support for this finding, with Bargas-Avila et al. (2010) recommending streamlining the level of detail and asking for documentation of only relevant information.

Interface quality was the only usability index that was higher on average for the REACH system than the Prioritized Information Display. A possible reason for this finding is that some participants (10%) preferred the aesthetics of REACH: Participant 4 offers their opinion of this system:

### Much sleeker, seems more professional - P4

Participant 5 echoes this sentiment in their response when asked about the layout of REACH:

# It was more aesthetically pleasing - P5

A more common response about the features in the system that may explain this finding was the preference for the countdown clock in REACH compared to the visual tPA window in the Prioritized Information Display; 65% of participants stated they like the countdown clock when asked about maintaining awareness of patient status as Participant 1 explains:

*Like I said, just digital numbers, there's just one more hour while they're within the time window. That just is more clearly communicating. - P1* 

Some participants (25%) also preferred the location of vitals next to the view of the patient in REACH compared to as the bottom of the screen in the Prioritized Information Display as explained by Participant 14: I preferred having all the vitals in that one area in the second one [REACH. . . . I think that I just felt it easier to just look at it. I think just visually it was easier to have all that information lined under where I was looking at the patient. - P14

Situation awareness for each system, was measured twice in this study, once after completing the patient history, medication, and allergy documentation and again at the end of the assessment. Endsley's model of situation awareness describes 3 levels of situation awareness: Level 1, the perception of elements in the environment; Level 2, the comprehension of those elements in relation to current knowledge; and Level 3, the ability to predict future states of a system from the comprehension of those elements (Endsley, 1995). During the first freeze the average accuracy of all situation awareness questions was higher for the Prioritized Information Display, but the largest difference was found for the Level 1 questions which focused on accurately identifying the current vital signs of the patient, the medications the patient was taking or the patient's condition. This result was expected as the presentation of vitals in the Prioritized Information Display was not only larger and more obvious but also more informative, showing the trend of the vital signs over time and using color coding to provide guidance as to the normal their ranges. This finding, based on the accuracy of the participants' responses, was further confirmed by interviews. Participant 6 describes the advantages of the display of the vitals in the Prioritized Information Display:

That's where having the vital signs be much more prevalent or obvious is helpful. I was definitely paying more attention to the changes in the blood pressure on the second one [Prioritized Information Display]. Whereas on the first one [REACH], it was a little bit even harder for me to remember to look at it. - P6

Similar results have been found in assessments of remote experts consulting on trauma care. For example, a study conducted by Nilsson et al. (2013) found a live presentation of vitals was the most important information for consulting trauma experts to determine a patient's condition. The same analysis can be applied to the increase in accuracy found in the Level 1 questions in the second freeze. In addition to the presentation of the vitals, having a summary available of all documentation including the patient's conditions, medications, allergies, and current NIHSS was identified as one possible reason for increased accuracy in Level 1 situation awareness. Participant 5 describes their use of these patient summaries:

Actually, something that was good about this [Prioritized Information Display] is how it, say for the meds, what you clicked on was saved there in that bar as you scrolled on and were still being listed and the allergies had no, no, no. That was good. I liked that.- P5

In addition, Participant 19 specifically mentioned using the same patient information summary location for keeping track of the NIHSS score, supporting the increased accuracy for the Prioritized Information Display found by the Freeze 2, Level 1 situation awareness questions:

I guess in the second system [REACH], the NIH score was there as well. I think the first system [Prioritized Information Display], when I did the NIH was a little bit easier for me to keep in track of than the second one [REACH]. - P19

This finding is supported by researchers investigating the situation awareness of users in multiple systems. First, the constant availability of critical information is a recommendation resulting from research on team situation awareness in the operating room, with studies focusing on developing interfaces that integrate all critical information into a single screen (Lai et al., 2006) and others on maintaining a persistent view of demographic and case information (Levine et al., 2005). This support for the continual view of task information can be found in other domains as well. Quick glances at updated documentation supports peripheral awareness or background situation awareness of the patient case while focused on documentation and forming assessment, a recommendation from a study on displays allowing pilots to be aware of all background tasks simultaneously, supporting their situation awareness (Wickens, 2000).

Accuracy in the responses to the Level 2 situation awareness questions in the first freeze was found to be equal between the two systems, with the incorrect answers being the response to the question, "does the patient have a stroke indicating history?" This result may be explained by a personal bias to determine whether patient history is indicative of a stroke rather than an effect of the system and is further supported by the fact that most incorrect answers to this question for both systems came from the same participants (10, 14,16).

Level 3 situation awareness question accuracy was higher in the Prioritized Information Display, again the constant display of patient information could be a reason for this result. In this freeze the only question asked at this level was for participants to determine if anything in the patients' history would prevent them from being a tPA medication candidate. This difference could again be explained by the continual presence of the patient history, medications, and vitals as seen in Level 1, all of which would contribute to more effectively determining the patient's candidacy for this treatment.

In the second freeze the situation awareness Level 2 question was a single question asking participants to present their preliminary diagnosis of the patient. All participants were able to determine from the assessment that the patient was having a stroke, meaning the accuracy was equal between the systems for this question.

During the second freeze the Level 3 question asked participants to identify that if the patient is indeed having a stroke, was it caused by a clot in a large or small blood vessel. The accuracy of the answers to this question were equal for both systems (92.5%). As mentioned in the experimental design of this study, two stroke diagnoses were used to create patient cases for participants to assess in the telemedicine systems. Each participant evaluated both patients, the pairing of the cases and the systems being counterbalanced with no cases repeated in a single session. All incorrect responses to the concerned diagnosing Case 2, the small vessel stroke, with the participants mistakenly determining the patients were having a large vessel stroke; thus, this incorrect answer is due to diagnosis, not the system, as it occurred equally for REACH and the Prioritized Information Display.

According to Endsley's situation awareness model, higher levels are not achieved without lower levels first being reached (Endsley, 1988). However, the results from this study do not support this model as the average accuracy of participants answering questions at the perception level is lower than the accuracy of questions at the prediction level. These findings may be explained by participants responding in the interview that typically exact knowledge of vital signs is not required to determine tPA candidacy, just knowledge that vital signs are under certain limits. For example, most participants indicated that only if systolic blood pressure is above 200 mmHg should action be taken and tPA reconsidered. Thus, accuracy at the granularity selected for these questions, whether the blood pressure was under 95, between 95 and 120, between 120 and 170, or above 170, does not suggest that a Level 3 prediction cannot be made. In further research, the responses to the SAGAT questions should be tailored to this level of detail to better measure situation awareness levels.

Time to complete the assessment was significantly longer in the REACH system than for the Prioritized Information Display. Many of the reasons for the increased workload can be used to analyze this finding as well, in particular the participants' comments about scrolling in the REACH system. The study of the design of a web survey referenced previously also provides context for the longer time taken using the REACH system due its scrolling design. In that study the time to complete the survey was significantly longer than for survey using page navigation (Peytchev et al., 2006). In addition to scrolling, completing extra documentation tasks in REACH and the time taken to find documentation of the medications and allergies that was frequently described as an issue in the REACH system could also contribute to longer assessment time in that system.

Average documentation errors were also significantly higher in the REACH system. Similar to the discussion of the performance on the workload subscale performance, participants had trouble determining where the medications and patient history should be documented and, thus, did so in the incorrect location. Participant 10 described their experience with this issue:

It wasn't obvious to me which tab to push to get to those allergies and medication so I just scripted them. - P10

Participant 19 also experienced this documentation problem, explaining their thought process:

Your time is of the essence, so you don't really want to spend a lot of time looking for things. You just want to move on to figure out your management strategy. - P19

Few conceptual errors were made, meaning the participants did not mis-select or misunderstand the patient history or NIHSS levels even when the correct information was provided to them through the responses. These errors primarily occurred when the patient response was slurred, a symptom confirmed by the paramedic, yet the participants selected that their voice was clear; they also incorrectly indicated no ataxia, or drift, when it presented in patients' arm movements. The overall trend indicated a higher average number of conceptual errors in the REACH system than for the Prioritized Information Display. Based on the examples of conceptual errors made, the video size may contribute to the misinterpretation of movements in the ataxia assessment for REACH. Participant 15 emphasized the need to be able to visually confirm movements during the assessment:

For me, I think the impression that I think she's [paramedic] not giving me the adequate feedback or the correct feedback because I do my observation. I think that might be a little more- NIHSS is all about the level of visibility or that level of the symptoms. Right?

Interviewer: Right. It's easier to see it yourself rather than get that feedback? Interviewee: Yes. - P15

Other participants, including Participant 12, also mentioned the large video size in the Prioritized Information Display resulting in more reliable assessments:

I feel like the exams would be more reliable using the second one [Prioritized Information Display] because of the video being larger. - P12

However, these interview responses contradict the finding of higher average conceptual errors in the Prioritized Information Display.

The effects of the display features discussed in the introduction were confirmed in the interviews, with participants specifically mentioning the increased view of the patient, the streamlined information, the guided navigation, and the size of the trends in and summary of the vitals as improving experience and supporting distributed cognition. However, only a few mentioned the task progress bar and the visual tPA window as contributing to distributed cognition. When directly asked about the former, most participants indicated that they either didn't notice it or noticed it but followed the guided navigation or were focused on documentation. Participant 19 described their use of the progress task bar:

Yes, I didn't really use it. I noticed that. I think it's because it's a-- Whenever I'm doing a stroke scale on a normal patient, I'm not paying attention to anything else except for the information I need, because time is just such a huge, relevant factor. - P19

For the second feature, the tPA window, participants overwhelmingly preferred having a numerical countdown rather than a visual representation. Although some liked the use of the visual tPA window for the same reasons it was chosen for the design, those comments were often qualified with the idea that the digital clock was useful because it provided a cut-off, as can be seen in Participant 8's response:

It was certainly easier to check on for that TPA window or making sure we're eligible [in Prioritized Information Display]. Then, the second one [REACH], what I really liked was that just like stark, 'This is the TPA window time. This is your cutoff.' Being a little bit more obvious than stop this bar where it says the time or the end. - P8

# Limitations

This study was conducted during the COVID-19 pandemic, and for this reason all study participation was virtual, with the participants completing it on their home computers using simulation files and surveys sent to their email and connected with a researcher through an Internet connection and a Zoom conference call. The size of the screens, the computer setup, and the mouse or trackpad use were, thus, unable to be controlled in this study. However, conducting the study under these conditions is perhaps how most neurologists conduct stroke telemedicine sessions. During the interviews, data about the computer setup were collected to determine if trends could be found in time to complete or workload; however, no such trends were found.

For this study to be conducted virtually, a simulated patient and paramedic were needed to provide the responses for the participants to conduct their assessment realistically. Although all questions that could be asked were carefully considered and confirmed with multiple physicians, some participants asked questions that the researchers did not anticipate. However, the video responses increased the consistency of the responses for each assessment, addressing one limitation resulting from using live actors. In addition, due to the COVID-19 pandemic, these recorded responses could not be conducted in an ambulance. While the positioning of the patient and paramedic were made to simulate an ambulance environment as much as possible, factors such as noise or movement were not able to be replicated.

Limiting the scope of this experiment to neurologists was necessary, and conducting the study virtually meant that the design changes that could be made were limited to the interface. Physical changes to the ambulance equipment or layout and the neurologist workstation were not considered in this study. As such, further research is needed to determine the effectiveness of the telemedicine systems in the actual environment as the physical layout could impact how these systems are used.

## CONCLUSION

This study investigates the effect of a redesign of a telemedicine system for stroke care in ambulances based on the perceived usability of the system, the perceived workload and situation awareness of the neurologists, and task performance. In addition, the designs examined were evaluated for their support of distributed cognition principles. The findings suggest that participants performed better, experiencing lower workload and higher situation awareness in the redesigned system. This system also received higher ratings for usability, and the features were identified more frequently as supporting distributed cognition than the system currently in use at a major hospital system to connect neurologists to rural hospitals and the EMS.

The system, however, could be improved. For example, the participants preferred the countdown clock feature of the current system, although overall they preferred the redesigned system. Thus, further research should be conducted on the Prioritized Information Display, with additional iterations improving it based on the results from this study including the recommendations from the participants for a countdown clock. In addition, this study was conducted using a small convenient sample of neurologists or physicians with neurology experience from one state and virtually moderated with simulated videos of a patient and paramedic in a simulated ambulance environment. Further research of the application of this system in its actual environment with a wider sample is necessary to confirm the results of this research.

#### CHAPTER SEVEN

### CONCLUSION

Telemedicine systems, specifically those used in ambulances, have the potential to improve stroke care with improved time to treatment, diagnosis accuracy, and patient outcomes. The first systematic review in this dissertation investigated the outcomes of the implementation of telemedicine systems in ambulances. It found that most of the current research has focused on the outcomes of the system in terms of efficiency and improved patient care; however, few have considered the caregivers using the telemedicine systems to meet, diagnose, and treat patients.

To investigate the impact of such systems on the caregivers in ambulances, the second review was of the types and occurrences of human error in ambulance-based care, with the results suggesting these could be categorized as procedure errors, protocol deviations, medication errors, diagnosis accuracy, and adverse or safety events with their occurrences varying among these categories. Many of the studies included in this review along with those found in the telemedicine implementation study focused on stroke care. However, the review of human error found no studies investigating the error occurrences within telemedicine systems.

To address this limitation, the first study in this dissertation focused on telemedicine in ambulances designated for stroke care based on the reviews conducted. Specifically, it investigated the tasks involved in stroke care by observing caregivers conducting simulated stroke consults using REACH, one of the current telemedicine systems. Further, this task analysis was used as input for a heuristic evaluation and in a Systematic Human Error Reduction and Prediction Approach (SHERPA) to evaluate this telemedicine system based on usability and possibility for human error. The heuristic evaluation found several violations, including the visibility of the system status, impacting the usability of the REACH system. The SHERPA also indicated potential errors when using the REACH system, many of which could be remediated with basic display changes such as a streamlined and correctly formatted data input.

To develop design improvements to this telemedicine system required further study of the caregivers' experience with the REACH telemedicine system, specifically of the barriers and facilitators to using the current telemedicine system implemented in stroke caregiving. To do so, 13 observations of simulated stroke consults were conducted with neurologists, nurses, and paramedics using the REACH telemedicine system. After these simulations of stroke consults, the participants completed surveys on their perceived workload, the usability of the system, and their evaluation of teamwork, followed by semistructured interview with each of the three caregivers in each team, 39 participants in total. The data collected were then analyzed for themes documenting the barriers and facilitators identified by the caregivers. The barriers found included frustration with equipment and with their training, both of which increased their perceived cognitive demand; the loss of the personal connection between the neurologists and the patients; and the physical constraints in the ambulance. The facilitators found were more numerous, including live and visual communication that increased teamwork and efficiency, the ease of access to specialists, increased flexibility, and high overall satisfaction and usability.

The final study evaluated a prototype design developed for neurologists in stroke care based on the feedback from the interviews of the caregivers conducted in the previous study and the principles of the distributed cognition for teamwork model, the DiCoT, specifically those in the information flow and artifacts components of the model. This proposed display included a streamlined, linear navigation through documentation tasks, graphical representation of vital trends, a dynamic task progress bar, a summary of the documentation, and larger video display for patient playback. In a study using a withinsubjects design, 20 neurologists participated in an evaluation of both the REACH telemedicine system and the prototype, named the Prioritized Information Display for its focus on streamlined and emphasized presentation of only necessary documentation and data. The results found reduced workload, higher usability, higher situation awareness, and improved task performance, including time to complete assessment and fewer errors, for the Prioritized Information Display. During the interviews conducted after evaluating both systems, the neurologists indicated the features in the Prioritized Information Display that supported distributed cognition, most notably the streamlined layout of the system and the guided navigation. Most participants also stated that they preferred using the Prioritized Information Display when evaluating stroke patients in an ambulance setting.

The findings from this research can be used to develop ambulance-based stroke telemedicine systems, important because such care is difficult, given the time constraints to limit brain tissue death and the need to communicate complicated assessment tasks between neurologists in the hospital and the paramedics at the scene. The changes made in the proposed design were found to reduce the time to complete assessment and documentation errors in addition to indicating increased usability and situation awareness and decreased workload. The most frequently mentioned improvement identified by the participants was the linear, guided navigation though stroke caregiving tasks. Simplifying the documentation and cognitive effort required to move to the next step allowed them to focus on their assessment and only the tasks necessary to treat the patient. In addition, the increased size of the video showing the patient and paramedic in the ambulance improved their view of the patient's movements and made them more confident and their assessments more reliable. Some participants also mentioned that the constantly updating vitals, graphical representation of trends in addition to current values, and color coding of nonnormal vital signs all contributed to higher awareness of the patient status. Most caregivers also agreed that these features more effectively supported distributed cognition during this process than REACH.

However, the neurologist participants in this study offered further recommendations. Many noted that the location of vitals in the lowest point of the screen with no connection to the patient video and documentation area where their attention was focused created difficulty for both noticing and maintaining awareness of the vital signs. Participants also frequently mentioned their preference for the countdown clock in the REACH telemedicine system, which showed a numeric value of time left to the end of the 3 hour and 4.5 hour tPA window, over the visual representation in the Prioritized Information Display. Some participants also mentioned that while the detailed patient information summary was useful for giving them an update on the documentation that had been completed, they expected this information to be at the top of the screen as many Electronic Health Record (EHR) systems place demographic information about a patient in screen headers. Based on these recommendations further research should be conducted to determine if changes such as duplicating the countdown next to the end of the vitals timeline, duplicating vitals near the patient video, and moving the patient information bar to the top of the screen, as shown in a mockup of a new iteration in Figure 7.1, will further improve the stroke caregiving process for neurologists and possibly other members of the caregiving team as well.



Figure 7.1: New Iteration of Prioritized Information Display

The broader applications of the findings from this research could be used to improve telemedicine or collaborative interfaces in other areas. Identifying task specific protocols and tailoring interfaces to only what is necessary streamline the amount of

information presented, facilitating faster and more effective movement through tasks. This efficiency is especially important in emergency care where caregivers are under both time pressure and the environmental pressures of the limited resources in most ambulances. Streamlining documentation and creating rigid formats for input were two of the recommendations found in earlier studies in this dissertation, and the results from this final study indicate they also appear to reduce errors in a high risk and high stress situation. Creating an easy, uncomplicated navigation that requires minimal effort from the user to understand the structure of the system and how to move from one task step to the next was a useful feature of this system that could be applied in a variety of collaborative interfaces. Further, supporting scaffolding artifacts, interface features such as labelling or progress maps, can assist users working both alone or as a team to coordinate tasks or self-regulate their progress. Finally, and perhaps the most obvious, the largest possible view of the main source of information or the ability to zoom or focus on specific areas, both for the view of a patient and a collaborative document, is key to improving task performance, the usability of the system, and collaboration on a task. These recommendations can not only have a broader impact on the design of telemedicine and other collaborative systems but they also increase knowledge about the features required in these systems and adds to the human factors literature. Further, the research presented in this dissertation provide not only recommendations for telemedicine systems in general, but the research methods used to evaluate and determine improvements are generalizable to other interface designs in other domains. A focus on design for the user, understanding their needs and any barriers or facilitators of current systems to generate design recommendations and iteratively implement changes with user evaluation is a critical research method for any interface design project. Additionally, the models and theories presented in this dissertation, the distributed cognition theory and the distributed cognition for teamwork or DiCoT model, can be applied to many systems involving collaboration of teams. The principles with in the DiCoT model that were used to guide the redesign of the telemedicine systems will be applicable for teams collaborating regardless of the artifacts or technology used in the future.

### **Limitations and Future Research**

The studies in this dissertation, like all research, have several limitations, especially Studies Two and Three in Chapters Five and Six, respectively. These two studies are limited because of their use of convenient sampling of neurologists and other caregivers in local healthcare systems. Further study of the telemedicine systems in other healthcare systems or other telemedicine systems for stroke care should be conducted. The primary limitations of the final study, Study Three, were the restrictions necessitated for conducting the study virtually because of the COVID-19 pandemic. Selections of pre-recorded videos were used in lieu of live patient actors, and the study was moderated over Zoom Video Conferencing. Further studies should be conducted to investigate the prototype system in comparison to the REACH system in an ambulance-based setting and with live simulated stroke patients. There is also a need to test new iterations of the prototype developed for the final study including the recommendations from the caregivers. In addition, future iterations and subsequent recommendations should be tested using telemedicine caregiving processes in other emergency situations such as cardiac arrests. The limitation of using a

simulated study method and therefore reducing the scope to designing for and testing with only neurologists and only within the interface also needs to be assessed with further design studies. As mentioned through this dissertation, teamwork of these caregivers must be supported within these designs and as such should be considered in testing design recommendations. Through a communications analysis of the observation sessions conducted in Chapter Five, it was also found that the patient should be treated as a member of this team and considered in system design as they contribute to the communication of important information for stroke assessment (Joseph et al., in press). For this reason, further studies should investigate how to support the communication and experience of the patient as well. Suggestions were made by paramedics when completing interviews during the observation study in Chapter Five, mentioning that patients can be disoriented by the disembodied voice of a neurologist with no visual representation in the ambulance and that a video screen facing the patient could improve their experience. This would further assist neurologists to be able to non-verbally communicate movements required for assessment. Considerations not only for interface design to support all team members, but system design such as adding video of the neurologist for the patient should be carefully considered in future research.

# **My Contributions**

During my four years as a Ph.D. student at Clemson University, I have been involved in several research projects investigating human factors problems. Through these projects I have developed skills in multiple research approaches including systematic review and meta-analysis, content analysis, surveys, interviews, and controlled behavioral experiments. I have conducted several studies on the effect of latency, display design, and automation in teleoperated robots used for search and rescue. One journal and conference publications resulted from this research project (Khasawneh et al., 2019; Rogers et al., 2017). I have also investigated how the design of Unmanned Aerial Vehicles (UAVs) control interfaces impact trust and decision making (Rogers et al., 2018, 2019), assisted with a review of usability studies of telemedicine systems for geriatric patients (Narasimha et al., 2016), and I participated in a study on how Alzheimer's patient caregivers reach out to others on a peer support portal (Scharett et al., 2017).

The first systematic review in this dissertation was published in *Telemedicine and eHealth* (Rogers et al., 2017). The task analysis and SHERPA in the first study were published as in the *Proceedings of the Human Factors and Ergonomics Society's Annual Meeting* (Rogers et al., 2020).

### REFERENCES

- Ackerman, R., & Waldron, R. L. (2006). Difficulty Breathing: Agreement of Paramedic and Emergency Physician Diagnoses. Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors, 10(1), 77–80.
- Adams, R. J., Debenham, E., Chalela, J., Chimowitz, M., Hays, A., Hill, C., Holmstedt, C., Jauch, E., Kitch, A., Lazaridis, C., & Turan, T. N. (2012). REACH MUSC: A Telemedicine Facilitated Network for Stroke: Initial Operational Experience. *Frontiers in Neurology*, 3, 33.
- Agnisarman, S., Narasimha, S., Chalil Madathil, K., Welch, B., Brinda, F., Ashok, A., & McElligott, J. (2017). Toward a More Usable Home-Based Video Telemedicine System: A Heuristic Evaluation of the Clinician User Interfaces of Home-Based Video Telemedicine Systems. *JMIR Human Factors*, 4(2), e11.
- Agnisarman, S. O., Chalil Madathil, K., Smith, K., Ashok, A., Welch, B., & McElligott, J. T. (2017). Lessons learned from the usability assessment of home-based telemedicine systems. *Applied Ergonomics*, 58, 424–434.
- Alenyo, A. N., Smith, W. P., McCaul, M., & Van Hoving, D. J. (2018). A Comparison Between Differently Skilled Prehospital Emergency Care Providers in Major-Incident Triage in South Africa. *Prehospital and Disaster Medicine*, 33(6), 575–580.
- Allen, M., Currie, L. M., Bakken, S., Patel, V. L., & Cimino, J. J. (2006). Heuristic evaluation of paper-based Web pages: a simplified inspection usability methodology. *Journal of Biomedical Informatics*, *39*(4), 412–423.
- Band, R. A., Gaieski, D. F., Hylton, J. H., Shofer, F. S., Goyal, M., & Meisel, Z. F. (2011). Arriving by emergency medical services improves time to treatment endpoints for patients with severe sepsis or septic shock. *Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine*, 18(9), 934–940.
- Bång, M., & Timpka, T. (2003). Cognitive tools in medical teamwork: the spatial arrangement of patient records. *Methods of Information in Medicine*, 42(4), 331–336.
- Bargas-Avila, J. A., Brenzikofer, O., Roth, S. P., Tuch, A. N., Orsini, S., & Opwis, K. (2010). Simple but Crucial User Interfaces in the World Wide Web: Introducing 20 Guidelines for Usable Web Form Design. https://edoc.unibas.ch/18353/1/20100620163248 4c1e269053732.pdf

- Barnes, D. R., Reed, D. B., Weinstein, G., & Brown, L. H. (2003). Blind tracheal intubation by paramedics through the LMA-Unique. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 7(4), 470–473.
- Barrett, K. M., Pizzi, M. A., Kesari, V., TerKonda, S. P., Mauricio, E. A., Silvers, S. M., Habash, R., Brown, B. L., Tawk, R. G., Meschia, J. F., Wharen, R., & Freeman, W. D. (2016). Ambulance-based assessment of NIH Stroke Scale with telemedicine: A feasibility pilot study. *Journal of Telemedicine and Telecare*. https://doi.org/10.1177/1357633X16648490
- Beaumont, A., James, J., Stephens, J., & Ullman, C. (2002). Usable Forms for the Web. Glasshaus.
- Belt, G. H., Felberg, R. A., Rubin, J., & Halperin, J. J. (2016). In-Transit Telemedicine Speeds Ischemic Stroke Treatment: Preliminary Results. *Stroke; a Journal of Cerebral Circulation*, 47(9), 2413–2415.
- Benjamin, E. J., Blaha, M. J., Chiuve, S. E., Cushman, M., Das, S. R., Deo, R., de Ferranti, S. D., Floyd, J., Fornage, M., Gillespie, C., Isasi, C. R., Jiménez, M. C., Jordan, L. C., Judd, S. E., Lackland, D., Lichtman, J. H., Lisabeth, L., Liu, S., Longenecker, C. T., ... American Heart Association Statistics Committee and Stroke Statistics Subcommittee. (2017). Heart Disease and Stroke Statistics-2017 Update: A Report From the American Heart Association. *Circulation*, 135(10), e146–e603.
- Benner, J. P., Brauning, G., Green, M., Caldwell, W., Borloz, M. P., & Brady, W. J. (2006). Disagreement between transport team and ED staff regarding the prehospital assessment of air medically evacuated scene patients. *Air Medical Journal*, 25(4), 165–169.
- Berenholtz, S. M., Milanovich, S., Faircloth, A., Prow, D. T., Earsing, K., Lipsett, P., Dorman, T., & Pronovost, P. J. (2004). Improving care for the ventilated patient. *Joint Commission Journal on Quality and Safety*, 30(4), 195–204.
- Bergrath, S., Czaplik, M., Rossaint, R., Hirsch, F., Beckers, S. K., Valentin, B., Wielpütz, D., Schneiders, M.-T., & Brokmann, J. C. (2013). Implementation phase of a multicentre prehospital telemedicine system to support paramedics: feasibility and possible limitations. *Scandinavian Journal of Trauma, Resuscitation and Emergency Medicine*, 21, 54.
- Bergrath, S., Rörtgen, D., Rossaint, R., Beckers, S. K., Fischermann, H., Brokmann, J. C., Czaplik, M., Felzen, M., Schneiders, M.-T., & Skorning, M. (2011). Technical and organisational feasibility of a multifunctional telemedicine system in an emergency medical service an observational study. *Journal of Telemedicine and Telecare*, 17(7), 371–377.

- Bernard, S. A., Nguyen, V., Cameron, P., Masci, K., Fitzgerald, M., Cooper, D. J., Walker, T., Std, B. P., Myles, P., Murray, L., David, Taylor, Smith, K., Patrick, I., Edington, J., Bacon, A., Rosenfeld, J. V., & Judson, R. (2010). Prehospital rapid sequence intubation improves functional outcome for patients with severe traumatic brain injury: a randomized controlled trial. *Annals of Surgery*, 252(6), 959–965.
- Bhende, M. S., Thompson, A. E., Cook, D. R., & Saville, A. L. (1992). Validity of a disposable end-tidal CO2 detector in verifying endotracheal tube placement in infants and children. *Annals of Emergency Medicine*, 21(2), 142–145.
- Bial, D., Block, F., & Gellersen, H. (2010). A study of two-handed scrolling and selection on standard notebook computers. *Proceedings of HCI 2010 24*. https://www.scienceopen.com/hosted-document?doi=10.14236/ewic/HCI2010.42
- Bigham, B. L., Buick, J. E., Brooks, S. C., Morrison, M., Shojania, K. G., & Morrison, L. J. (2012). Patient safety in emergency medical services: a systematic review of the literature. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 16(1), 20–35.
- Board on Health Care Services & Institute of Medicine. (2012). *The role of Telehealth in an evolving health care environment: Workshop summary*. National Academies Press.
- Bond, W. F., & Spillane, L. (2002). The use of simulation for emergency medicine resident assessment. Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine, 9(11), 1295–1299.
- Boyle, M. J. (2009). Comparison overview of prehospital errors involving road traffic fatalities in Victoria, Australia [Review of *Comparison overview of prehospital* errors involving road traffic fatalities in Victoria, Australia]. Prehospital and Disaster Medicine, 24(3), 254–261.
- Brandler, E. S., Sharma, M., McCullough, F., Ben-Eli, D., Kaufman, B., Khandelwal, P., Helzner, E., Sinert, R. H., & Levine, S. R. (2015). Prehospital Stroke Identification: Factors Associated with Diagnostic Accuracy. *Journal of Stroke and Cerebrovascular Diseases: The Official Journal of National Stroke Association*, 24(9), 2161–2166.
- Brennan, T. A., Leape, L. L., Laird, N. M., Hebert, L., Localio, A. R., Lawthers, A. G., Newhouse, J. P., Weiler, P. C., & Hiatt, H. H. (1991). Incidence of adverse events and negligence in hospitalized patients. Results of the Harvard Medical Practice Study I. *The New England Journal of Medicine*, 324(6), 370–376.

- Brice, J. H., Friend, K. D., & Delbridge, T. R. (2008). Accuracy of EMS-recorded patient demographic data. Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors, 12(2), 187–191.
- Brice, J. H., Garrison, H. G., & Evans, A. T. (2000). Study Design and Outcomes in Out-Of-Hospital Emergency Medicine Research: at End-Year Analysis. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 4(2), 144–150.
- Callaham, M. (1997). Quantifying the scanty science of prehospital emergency care. *Annals of Emergency Medicine*, *30*(6), 785–790.
- Cantor, W. J., Hoogeveen, P., Robert, A., Elliott, K., Goldman, L. E., Sanderson, E., Plante, S., Prabhakar, M., & Miner, S. (2012). Prehospital diagnosis and triage of ST-elevation myocardial infarction by paramedics without advanced care training. *American Heart Journal*, 164(2), 201–206.
- Cebul, R. D., Love, T. E., Jain, A. K., & Hebert, C. J. (2011). Electronic health records and quality of diabetes care. *The New England Journal of Medicine*, *365*(9), 825– 833.
- Chalil Madathil, K., Koikkara, R., Obeid, J., Greenstein, J. S., Sanderson, I. C., Fryar, K., Moskowitz, J., & Gramopadhye, A. K. (2013). An investigation of the efficacy of electronic consenting interfaces of research permissions management system in a hospital setting. *International Journal of Medical Informatics*, 82(9), 854–863.
- Chalil Madathil, K., Koikkara, R., Gramopadhye, A. K., & Greenstein, J. S. (2011, September). An empirical study of the usability of consenting systems: iPad, Touchscreen and paper-based systems. In *Proceedings of the human factors and ergonomics society annual meeting (Vol. 55, No. 1, pp. 813-817)*. Sage CA: Los Angeles, CA.
- Charash, W. E., Caputo, M. P., Clark, H., Callas, P. W., Rogers, F. B., Crookes, B. A., Alborg, M. S., & Ricci, M. A. (2011). Telemedicine to a moving ambulance improves outcome after trauma in simulated patients. *The Journal of Trauma*, 71(1), 49–55.
- Chaturvedi, H. K., & Shweta, R. C. (2015). Evaluation of inter-rater agreement and interrater reliability for observational data: An overview of concepts and methods. *Journal of the Indian Academy of Applied Psychology*, *41*, 20–27.
- Choi, J., & Bakken, S. (2010). Web-based education for low-literate parents in Neonatal Intensive Care Unit: development of a website and heuristic evaluation and usability testing. *International Journal of Medical Informatics*, *79*(8), 565–575.

- Cho, S. J., Kwon, I. H., & Jeong, J. (2015). Application of Telemedicine System to Prehospital Medical Control. *Healthcare Informatics Research*, 21(3), 196–200.
- Christie, A., Costa-Scorse, B., Nicholls, M., Jones, P., & Howie, G. (2016). Accuracy of working diagnosis by paramedics for patients presenting with dyspnoea. *Emergency Medicine Australasia: EMA*, 28(5), 525–530.
- Cienki, J. J., & DeLuca, L. A. (2012). Agreement between emergency medical services and expert blood pressure measurements. *The Journal of Emergency Medicine*, 43(1), 64–68.
- Cohen, T., & Patel, V. L. (2014). A Framework for Understanding Error and Complexity in Critical Care. In V. L. Patel, D. R. Kaufman, & T. Cohen (Eds.), Cognitive Informatics in Health and Biomedicine: Case Studies on Critical Care, Complexity and Errors (pp. 17–34). Springer London.
- Cook, T. M., & MacDougall-Davis, S. R. (2012). Complications and failure of airway management. In *British Journal of Anaesthesia* (Vol. 109, pp. i68–i85). https://doi.org/10.1093/bja/aes393
- Coppler, P. J., Padmanabhan, R., Martin-Gill, C., Callaway, C. W., Yealy, D. M., & Seymour, C. W. (2016). Accuracy of Prehospital Intravenous Fluid Volume Measurement by Emergency Medical Services. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 20(1), 125–131.
- Curry, G. R., & Harrop, N. (1998). The Lancashire telemedicine ambulance. *Journal of Telemedicine and Telecare*, 4(4), 231–238.
- Czaplik, M., Bergrath, S., Rossaint, R., Thelen, S., Brodziak, T., Valentin, B., Hirsch, F., Beckers, S. K., & Brokmann, J. C. (2014). Employment of telemedicine in emergency medicine. Clinical requirement analysis, system development and first test results. *Methods of Information in Medicine*, 53(2), 99–107.
- Danaher, B. G., Brendryen, H., Seeley, J. R., Tyler, M. S., & Woolley, T. (2015). From black box to toolbox: Outlining device functionality, engagement activities, and the pervasive information architecture of mHealth interventions. *Internet Interventions*, 2(1), 91–101.
- Davis, P., Lay Yee, R., Briant, R., Schug, S., & Scott, A. (2001). Adverse events in New Zealand public hospitals: principal findings from a national survey. https://researchspace.auckland.ac.nz/handle/2292/15593

- DeBoer, S., Seaver, M., & Broselow, J. (2005). Color Coding to Reduce Errors: The Broselow–Luten system streamlines pediatric emergency treatment. *American Journal of Nursing*, 105(8), 68–71.
- Deering, S., Johnston, L. C., & Colacchio, K. (2011). Multidisciplinary teamwork and communication training. *Seminars in Perinatology*, *35*(2), 89–96.
- Devine, E. B., Hansen, R. N., Wilson-Norton, J. L., Lawless, N. M., Fisk, A. W., Blough, D. K., Martin, D. P., & Sullivan, S. D. (2010). The impact of computerized provider order entry on medication errors in a multispecialty group practice. *Journal of the American Medical Informatics Association: JAMIA*, 17(1), 78–84.
- Dharmar, M., Kuppermann, N., Romano, P. S., Yang, N. H., Nesbitt, T. S., Phan, J., Nguyen, C., Parsapour, K., & Marcin, J. P. (2013). Telemedicine consultations and medication errors in rural emergency departments. *Pediatrics*, 132(6), 1090–1097.
- Dietrich, M., Walter, S., Ragoschke-Schumm, A., Helwig, S., Levine, S., Balucani, C., Lesmeister, M., Haass, A., Liu, Y., Lossius, H.-M., & Fassbender, K. (2014). Is prehospital treatment of acute stroke too expensive? An economic evaluation based on the first trial. *Cerebrovascular Diseases*, 38(6), 457–463.
- Donahue, P. L. (1994). The oesophageal detector device. An assessment of accuracy and ease of use by paramedics. *Anesthesia*, 49(10), 863–865.
- Duby, R., Hansen, M., Meckler, G., Skarica, B., Lambert, W., & Guise, J.-M. (2018). Safety Events in High Risk Prehospital Neonatal Calls. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 22(1), 34–40.
- Duignan, K. M., Lamb, L. C., DiFiori, M. M., Quinlavin, J., & Feeney, J. M. (2018). Tourniquet use in the prehospital setting: Are they being used appropriately? *American Journal of Disaster Medicine*, 13(1), 37–43.
- Eastwood, K., Boyle, M. J., & Williams, B. (2012). Undergraduate paramedic students cannot do drug calculations. *World Journal of Emergency Medicine*, *3*(3), 221–226.
- Eberle, B., Dick, W. F., Schneider, T., Wisser, G., Doetsch, S., & Tzanova, I. (1996). Checking the carotid pulse check: diagnostic accuracy of first responders in patients with and without a pulse. In *Resuscitation* (Vol. 33, Issue 2, pp. 107–116). https://doi.org/10.1016/s0300-9572(96)01016-7
- Eckstein, M., & Suyehara, D. (2002). Ability of paramedics to treat patients with congestive heart failure via standing field treatment protocols. *The American Journal of Emergency Medicine*, 20(1), 23–25.

- Embrey, D. E. (1986). SHERPA: A systematic human error reduction and prediction approach. *Proceedings of the International Topical Meeting on Advances in Human Factors in Nuclear Power Systems*. https://inis.iaea.org/Search/search.aspx?orig q=RN:18074340
- Endsley, M. R. (1995). Toward a Theory of Situation Awareness in Dynamic Systems. *Human Factors*, 37(1), 32–64.
- Endsley, M. R. (1988). Situation awareness global assessment technique (SAGAT). Proceedings of the IEEE 1988 National Aerospace and Electronics Conference, 789–795.
- Fairbanks, R. J., Crittenden, C. N., O'Gara, K. G., Wilson, M. A., Pennington, E. C., Chin, N. P., & Shah, M. N. (2008). Emergency medical services provider perceptions of the nature of adverse events and near-misses in out-of-hospital care: an ethnographic view. Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine, 15(7), 633–640.
- Feldman, J. A., Brinsfield, K., Bernard, S., White, D., & Maciejko, T. (2005). Real-time paramedic compared with blinded physician identification of ST-segment elevation myocardial infarction: results of an observational study. *The American Journal of Emergency Medicine*, 23(4), 443–448.
- Felzen, M., Brokmann, J. C., Beckers, S. K., Czaplik, M., Hirsch, F., Tamm, M., Rossaint, R., & Bergrath, S. (2016). Improved technical performance of a multifunctional prehospital telemedicine system between the research phase and the routine use phase--an observational study. *Journal of Telemedicine and Telecare*, 1357633X16644115.
- Ferson, D. Z., Rosenblatt, W. H., Johansen, M. J., Osborn, I., & Ovassapian, A. (2001). Use of the intubating LMA-Fastrach in 254 patients with difficult-to-manage airways. *Anesthesiology*, 95(5), 1175–1181.
- Field, M. J. (1996). *Telemedicine: A Guide to Assessing Telecommunications for Health Care*. National Academies Press.
- Fleury, C., Ferey, N., Vezien, J.-M., & Bourdot, P. (2015). Remote Collaboration across Heterogeneous Large Interactive Spaces. Second VR International Workshop on Collaborative Virtual Environments (3DCVE). https://doi.org/10.1109/3DCVE.2015.7153591
- Foutch, R. G., Magelssen, M. D., & MacMillan, J. G. (1992). The esophageal detector device: a rapid and accurate method for assessing tracheal versus esophageal intubation in a porcine model. *Annals of Emergency Medicine*, 21(9), 1073–1076.

- Frank, L., & Pape-Haugaard, L. (2011). Integration of health records by using relaxed ACID properties between hospitals, physicians and mobile units like ambulances and doctors. *International Journal of Handheld Computing Research (IJHCR)*, 2(4), 29– 41.
- Frexia, F., Triunfo, R., Cabras, F., Mascia, C., Zanetti, G., Montis, S., & Tumbarello, R. (2014). Preliminary Cost-Benefit Analysis of a Real-Time Telemedicine System. 23– 17.
- Fugate, J. E., & Rabinstein, A. A. (2015). Absolute and Relative Contraindications to IV rt-PA for Acute Ischemic Stroke. *The Neurohospitalist*, *5*(3), 110–121.
- Furniss, D., & Blandford, A. (2006). Understanding emergency medical dispatch in terms of distributed cognition: a case study. *Ergonomics*, 49(12-13), 1174–1203.
- Gallagher, J. M., & Kupas, D. F. (2012). Experience with an anonymous web-based state EMS safety incident reporting system. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association* of State EMS Directors, 16(1), 36–42.
- Gardtman, M., Waagstein, L., Karlsson, T., & Herlitz, J. (2000). Has an intensified treatment in the ambulance of patients with acute severe left heart failure improved the outcome? *European Journal of Emergency Medicine: Official Journal of the European Society for Emergency Medicine*, 7(1), 15–24.
- Ghasemi, M., Khoshakhlagh, A. H., Mahmudi, S., & Fesharaki, M. G. (2015). Identification and assessment of medical errors in the triage area of an educational hospital using the SHERPA technique in Iran. *International Journal of Occupational Safety and Ergonomics: JOSE*, 21(3), 382–390.
- Ghiyasvandian, S., Khazaei, A., Zakerimoghadam, M., Salimi, R., Afshari, A., & Mogimbeigi, A. (2018). Evaluation of Airway Management Proficiency in Pre-Hospital Emergency Setting; a Simulation Study. *Emergency (Tehran, Iran)*, 6(1), e58.
- Goebel, P. J., Daya, M. R., & Gunnels, M. D. (2004). Accuracy of arrhythmia recognition in paramedic treatment of paroxysmal supraventricular tachycardia: a ten-year review. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 8(2), 166– 170.

- Goldberg, S. A., Porat, A., Strother, C. G., Lim, N. Q., Wijeratne, H. R. S., Sanchez, G., & Munjal, K. G. (2017). Quantitative Analysis of the Content of EMS Handoff of Critically III and Injured Patients to the Emergency Department. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 21(1), 14–17.
- Goodwin, D., Biesbroek, S., & Thorkelson, B. (2017). Collaborative Design of Canada's First Stroke Ambulance. 48th Annual Conference of the Association of Canadian Ergonomists, Banff, Alberta, Canada.
- Gropen, T. I., Gokaldas, R., Poleshuck, R., Spencer, J., Janjua, N., Szarek, M., Brandler, E. S., & Levine, S. R. (2014). Factors related to the sensitivity of emergency medical service impression of stroke. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 18(3), 387–392.
- Guarascio, M., Fargnoli, M., Lombardi, M., Puri, D., & Garzia, F. (2019). Use of SHERPA for the Prevention of Human Errors Among Agricultural Machinery Users. WIT Transactions on The Built Environment, 189, 1–12.
- Hadeed, G. J., & Hadeed, S. J. (2011). Prehospital telemedicine–digital ambulances. In R. Latifi (Ed.), *Telemedicine for Trauma, Emergencies, and Disaster Management*. Artech House.
- Hales, B. M., & Pronovost, P. J. (2006). The checklist—a tool for error management and performance improvement. *Journal of Critical Care*, *21*(3), 231–235.
- Hales, B., Terblanche, M., Fowler, R., & Sibbald, W. (2008). Development of medical checklists for improved quality of patient care. *International Journal for Quality in Health Care: Journal of the International Society for Quality in Health Care / ISQua*, 20(1), 22–30.
- Handel, D. A., & Hackman, J. L. (2010). Implementing electronic health records in the emergency department. *The Journal of Emergency Medicine*, 38(2), 257–263.
- Hansen, M., Eriksson, C., Skarica, B., Meckler, G., & Guise, J.-M. (2018). Safety events in pediatric out-of-hospital cardiac arrest. *The American Journal of Emergency Medicine*, 36(3), 380–383.
- Hansen, M., Meckler, G., Lambert, W., Dickinson, C., Dickinson, K., Van Otterloo, J., & Guise, J.-M. (2016). Patient safety events in out-of-hospital paediatric airway management: a medical record review by the CSI-EMS. *BMJ Open*, 6(11), e012259.

- Hart, S. G., & Staveland, L. E. (1988). Development of NASA-TLX (Task Load Index): Results of Empirical and Theoretical Research. In P. A. Hancock & N. Meshkati (Eds.), *Advances in Psychology* (Vol. 52, pp. 139–183). North-Holland.
- Haynes, B. E., & Pritting, J. (1999). A rural emergency medical technician with selected advanced skills. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 3(4), 343–346.
- Hazlehurst, B., Gorman, P. N., & McMullen, C. K. (2008). Distributed cognition: an alternative model of cognition for medical informatics. *International Journal of Medical Informatics*, 77(4), 226–234.
- Hein, C., Owen, H., & Plummer, J. (2008). A 12-month audit of laryngeal mask airway (LMA) use in a South Australian ambulance service. *Resuscitation*, *79*(2), 219–224.
- Heron, M. (2019). Deaths: Leading Causes for 2017. *National Vital Statistics Reports*, 68(6). https://www.cdc.gov/nchs/data/nvsr/nvsr68/nvsr68\_06-508.pdf
- Hill, M. D., & Hachinski, V. (1998). Stroke treatment: time is brain. *The Lancet*, 352, S10–S14.
- Hobgood, C., Bowen, J. B., Brice, J. H., Overby, B., & Tamayo-Sarver, J. H. (2006). Do EMS personnel identify, report, and disclose medical errors? *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 10(1), 21–27.
- Hobgood, C., Xie, J., Weiner, B., & Hooker, J. (2004). Error identification, disclosure, and reporting: practice patterns of three emergency medicine provider types. *Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine*, 11(2), 196–199.
- Hohenstein, C., Hempel, D., Schultheis, K., Lotter, O., & Fleischmann, T. (2014). Critical incident reporting in emergency medicine: results of the prehospital reports. *Emergency Medicine Journal: EMJ*, 31(5), 415–418.
- Hohenstein, C., Rupp, P., & Fleischmann, T. (2011). Critical incidents during prehospital cardiopulmonary resuscitation: what are the problems nobody wants to talk about? *European Journal of Emergency Medicine: Official Journal of the European Society* for Emergency Medicine, 18(1), 38–40.
- Holden, R. J., Carayon, P., Gurses, A. P., Hoonakker, P., Hundt, A. S., Ozok, A. A., & Rivera-Rodriguez, A. J. (2013). SEIPS 2.0: a human factors framework for studying and improving the work of healthcare professionals and patients. *Ergonomics*, 56(11), 1669–1686.

- Hollander, J. E., Delagi, R., Sciammarella, J., Viccellio, P., Ortiz, J., & Henry, M. C. (1995). On-line Telemetry: Prospective Assessment of Accuracy in an All-volunteer Emergency Medical Service System. *Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine*, 2(4), 280–286.
- Hollan, J., Hutchins, E., & Kirsh, D. (2000). Distributed cognition: toward a new foundation for human-computer interaction research. ACM Transactions on Computer-Human Interaction: A Publication of the Association for Computing Machinery, 7(2), 174–196.
- Hoyle, J. D., Davis, A. T., Putman, K. K., Trytko, J. A., & Fales, W. D. (2012). Medication dosing errors in pediatric patients treated by emergency medical services. *Prehospital Emergency Care: Official Journal of the National Association* of EMS Physicians and the National Association of State EMS Directors, 16(1), 59– 66.
- Hubble, M. W., Paschal, K. R., & Sanders, T. A. (2000). Medication calculation skills of practicing paramedics. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 4(3), 253–260.
- Hughes, C. M. L., Baber, C., & Bienkiewicz, M. (2015). The application of SHERPA (Systematic Human Error Reduction and Prediction Approach) in the development of compensatory cognitive rehabilitation strategies for stroke patients with left and right brain damage. *Ergonomics*. https://www-tandfonlinecom.libproxy.clemson.edu/doi/abs/10.1080/00140139.2014.957735
- Hunt, E. A., Walker, A. R., Shaffner, D. H., Miller, M. R., & Pronovost, P. J. (2008). Simulation of in-hospital pediatric medical emergencies and cardiopulmonary arrests: highlighting the importance of the first 5 minutes. *Pediatrics*, 121(1), e34– e43.
- Institute of Medicine. (2012). *The Role of Telehealth in an Evolving Health Care Environment: Workshop Summary*. National Academies Press.
- Institute of Medicine (US) Committee on Quality of Health Care in America. (2014). *To Err is Human: Building a Safer Health System* (L. T. Kohn, J. M. Corrigan, & M. S. Donaldson (eds.)). National Academies Press (US).
- Interview Guide for Clinicians and Office Staff. (2016). University of Madison -Wisconsin Center for Quality and Productivity Improvement . https://cqpi.wisc.edu/wpcontent/uploads/sites/599/2016/07/Interview\_Guide\_for\_Clinicians\_and\_Office\_Staf f.pdf

- Isaacs, E. A., & Tang, J. C. (1993). What video can and can't do for collaboration: a case study. Proceedings of the First ACM International Conference on Multimedia, 199– 206.
- Jemmett, M. E., Kendal, K. M., Fourre, M. W., & Burton, J. H. (2003). Unrecognized misplacement of endotracheal tubes in a mixed urban to rural emergency medical services setting. Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine, 10(9), 961–965.
- Jenkins, D. P., Salmon, P. M., Stanton, N. A., Walker, G. H., & Rafferty, L. (2011). What could they have been thinking? How sociotechnical system design influences cognition: a case study of the Stockwell shooting. *Ergonomics*, 54(2), 103–119.
- Johnson, C. M., Johnson, T. R., & Zhang, J. (2005). A user-centered framework for redesigning health care interfaces. *Journal of Biomedical Informatics*, *38*(1), 75–87.
- Jones, J. H., Murphy, M. P., Dickson, R. L., Somerville, G. G., & Brizendine, E. J. (2004). Emergency physician-verified out-of-hospital intubation: miss rates by paramedics. Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine, 11(6), 707–709.
- Joseph, A., Chalil Madathil, K., Jafarifiroozabadi, R., Rogers, H., Mihandoust, S., Khasawneh, A., McNeese, N., Holmstedt, C., & McElligott, J. (in press). Communication and teamwork during telemedicine-enabled stroke care in an ambulance. *Human Factors*.
- Jurriën, M., van't Hof, A. W. J., Dill, T., Heestermans, T., van Werkum, J. W., Mosterd, A., van Houwelingen, G., Koopmans, P. C., Stella, P. R., Boersma, E., & Others. (2010). Effect of early, pre-hospital initiation of high bolus dose tirofiban in patients with ST-segment elevation myocardial infarction on short-and long-term clinical outcome. *Journal of the American College of Cardiology*, 55(22), 2446–2455.
- Kable, A. K., Gibberd, R. W., & Spigelman, A. D. (2002). Adverse events in surgical patients in Australia. International Journal for Quality in Health Care: Journal of the International Society for Quality in Health Care / ISQua, 14(4), 269–276.
- Kaji, A. H., Gausche-Hill, M., Conrad, H., Young, K. D., Koenig, W. J., Dorsey, E., & Lewis, R. J. (2006). Emergency medical services system changes reduce pediatric epinephrine dosing errors in the prehospital setting. *Pediatrics*, 118(4), 1493–1500.
- Kaserer, A., Stein, P., Simmen, H.-P., Spahn, D. R., & Neuhaus, V. (2017). Failure rate of prehospital chest decompression after severe thoracic trauma. *The American Journal of Emergency Medicine*, 35(3), 469–474.

- Kelleher, D. C., Carter, E. A., Waterhouse, L. J., Parsons, S. E., Fritzeen, J. L., & Burd, R. S. (2014). Effect of a checklist on advanced trauma life support task performance during pediatric trauma resuscitation. *Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine*, 21(10), 1129–1134.
- Kermani, A. L. I., Mazloumi, A., & Kazemi, Z. (2016). Identification of nurses' errors in the emergency ward, using SHERPA technique. *International Journal of Occupational Hygiene*, 8(1), 54–61.
- Khandan, M., Yusefi, S., Sahranavard, R., & Koohpaei, A. (2017). SHERPA technique as an approach to healthcare error management and patient safety improvement: A case study among nurses. *Health Scope*, 6(2). http://jhealthscope.com/en/articles/13123.html
- Khasawneh, A., Rogers, H., & Bertrand, J. (2019). Human adaptation to latency in teleoperated multi-robot human-agent search and rescue teams. *Automation in Construction*. https://www.sciencedirect.com/science/article/pii/S0926580517310890
- Kientz, J. A., Choe, E. K., Birch, B., Maharaj, R., Fonville, A., Glasson, C., & Mundt, J. (2010). Heuristic evaluation of persuasive health technologies. *Proceedings of the 1st ACM International Health Informatics Symposium*, 555–564.
- Kirves, H., Skrifvars, M. B., Vähäkuopus, M., Ekström, K., Martikainen, M., & Castren, M. (2007). Adherence to resuscitation guidelines during prehospital care of cardiac arrest patients. *European Journal of Emergency Medicine: Official Journal of the European Society for Emergency Medicine*, 14(2), 75–81.
- Kirwan, B. (1992). Human error identification in human reliability assessment. Part 1: Overview of approaches. *Applied Ergonomics*, 23(5), 299–318.
- Kleij, R. v. (2007). Overcoming distance in virtual teams: Effects of communication media, experience, and time pressure on distributed teamwork [TNO Defensie en Veiligheid]. https://repository.tudelft.nl/view/tno/uuid:713a44b9-17fc-4013-bb17-45462eb82ec4
- Kleindorfer, D., Xu, Y., Moomaw, C. J., Khatri, P., Adeoye, O., & Hornung, R. (2009). US geographic distribution of rt-PA utilization by hospital for acute ischemic stroke. *Stroke; a Journal of Cerebral Circulation*, 40(11), 3580–3584.
- Kobb, R., Hoffman, N., Lodge, R., & Kline, S. (2003). Enhancing elder chronic care through technology and care coordination: report from a pilot. *Telemedicine Journal* and E-Health: The Official Journal of the American Telemedicine Association, 9(2), 189–195.

- Koch, S. (2006). Home telehealth--current state and future trends. *International Journal* of Medical Informatics, 75(8), 565–576.
- Koikkara, R., Greenstein, J. S., & Madathil, K. C. (2015). The Effect of Graphic Organizers on the Performance of Electronic Consenting Systems. Proceedings of the Human Factors and Ergonomics Society ... Annual Meeting Human Factors and Ergonomics Society. Meeting, 59(1), 466–470.
- Kothari, R., Barsan, W., Brott, T., Broderick, J., & Ashbrock, S. (1995). Frequency and accuracy of prehospital diagnosis of acute stroke. *Stroke; a Journal of Cerebral Circulation*, 26(6), 937–941.
- Koukouletsos, K., Moustris, K., & Paliatsos, A. G. (2014). *The impact of long scrolling web pages on users*. International Scientific Conference eRA-9. http://era.teipir.gr/sites/default/files/ict\_ii\_session\_0.pdf
- Kunisawa, S., Kobayashi, D., Lee, J., Otsubo, T., Ikai, H., Yokota, C., Minematsu, K., & Imanaka, Y. (2014). Factors associated with the administration of tissue plasminogen activator for acute ischemic stroke. *Journal of Stroke and Cerebrovascular Diseases: The Official Journal of National Stroke Association*, 23(4), 724–731.
- Kwak, M. J., Kim, J. M., Shin, I. H., Shin, S. D., Song, K. J., Suh, G. J., & Kim, H. C. (2009). Real-time medical control using a wireless audio-video transmission device in a pre-hospital emergency service in Korea. *Journal of Telemedicine and Telecare*, 15(8), 404–408.
- Kwan, J., Hand, P., & Sandercock, P. (2004). Improving the efficiency of delivery of thrombolysis for acute stroke: a systematic review. *QJM: Monthly Journal of the Association of Physicians*, *97*(5), 273–279.
- Kyriacou, E., Pavlopoulos, S., Berler, A., Neophytou, M., Bourka, A., Georgoulas, A., Anagnostaki, A., Karayiannis, D., Schizas, C., Pattichis, C., Andreou, A., & Koutsouris, D. (2003). Multi-purpose HealthCare Telemedicine Systems with mobile communication link support. *Biomedical Engineering Online*, *2*, 7.
- Lai, F., Spitz, G., & Brzezinski, P. (2006). Gestalt operating room display design for perioperative team situation awareness. *Studies in Health Technology and Informatics*, 119, 282–284.
- Lai, T.-Y. (2007). Iterative refinement of a tailored system for self-care management of depressive symptoms in people living with HIV/AIDS through heuristic evaluation and end user testing. *International Journal of Medical Informatics*, 76 Suppl 2, S317–S324.

- Lammers, R., Byrwa, M., & Fales, W. (2012). Root causes of errors in a simulated prehospital pediatric emergency. Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine, 19(1), 37–47.
- Lammers, R., Byrwa, M., Fales, W., & Hale, R. (2009). Simulation-based assessment of paramedic pediatric resuscitation skills. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 13(3), 345–356.
- Lammers, R., Willoughby-Byrwa, M., & Fales, W. (2014a). Medication errors in prehospital management of simulated pediatric anaphylaxis. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 18(2), 295–304.
- Lammers, R., Willoughby-Byrwa, M., & Fales, W. (2014b). Errors and error-producing conditions during a simulated, prehospital, pediatric cardiopulmonary arrest. *Simulation in Healthcare: Journal of the Society for Simulation in Healthcare*, 9(3), 174–183.
- LaMonte, M. P., Xiao, Y., Hu, P. F., Gagliano, D. M., Bahouth, M. N., Gunawardane, R. D., MacKenzie, C. F., Gaasch, W. R., & Cullen, J. (2004). Shortening time to stroke treatment using ambulance telemedicine: TeleBAT. *Journal of Stroke and Cerebrovascular Diseases: The Official Journal of National Stroke Association*, 13(4), 148–154.
- Lane, R., Stanton, N. A., & Harrison, D. (2006). Applying hierarchical task analysis to medication administration errors. *Applied Ergonomics*, 37(5), 669–679.
- Lathan, C. E., Sebrechts, M. M., Newman, D. J., & Doarn, C. R. (1999). Heuristic evaluation of a web-based interface for internet telemedicine. *Telemedicine Journal: The Official Journal of the American Telemedicine Association*, *5*(2), 177–185.
- Latimer, A. J., Husain, S., Nolan, J., Doreswamy, V., Rea, T. D., Sayre, M. R., & Eisenberg, M. S. (2018). Syringe Administration of Epinephrine by Emergency Medical Technicians for Anaphylaxis. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association* of State EMS Directors, 22(3), 319–325.
- Leape, L. L. (2002). Reporting of adverse events. *The New England Journal of Medicine*, 347(20), 1633–1638.

- Leifer, D., Bravata, D. M., Connors, J. J., III, Hinchey, J. A., Jauch, E. C., Johnston, S. C., Latchaw, R., Likosky, W., Ogilvy, C., Qureshi, A. I., & Others. (2011). Metrics for measuring quality of care in comprehensive stroke centers: detailed follow-up to Brain Attack Coalition comprehensive stroke center recommendations: a statement for healthcare professionals from the American Heart Association/American Stroke Association. *Stroke; a Journal of Cerebral Circulation, 42*(3), 849–877.
- Levine, Steven R., & Gorman, Mark. (1999). "Telestroke": The Application of Telemedicine for Stroke. *Stroke; a Journal of Cerebral Circulation*, *30*(2), 464–469.
- Levine, W. C., Meyer, M., Brzezinski, P., Robbins, J., Lai, F., Spitz, G., & Sandberg, W. S. (2005). Usability factors in the organization and display of disparate information sources in the operative environment. AMIA ... Annual Symposium Proceedings / AMIA Symposium. AMIA Symposium, 1025.
- Lewis, J. R. (2018). Measuring Perceived Usability: The CSUQ, SUS, and UMUX. *International Journal of Human–Computer Interaction*, *34*(12), 1148–1156.
- Liberman, M., Lavoie, A., Mulder, D., & Sampalis, J. (1999). Cardiopulmonary resuscitation: errors made by pre-hospital emergency medical personnel. *Resuscitation*, *42*(1), 47–55.
- Lifshitz, A. E., Goldstein, L. H., Sharist, M., Strugo, R., Asulin, E., Bar Haim, S., Feigenberg, Z., Berkovitch, M., & Kozer, E. (2012). Medication prescribing errors in the prehospital setting and in the ED. *The American Journal of Emergency Medicine*, 30(5), 726–731.
- Liman, T. G., Winter, B., Waldschmidt, C., Zerbe, N., Hufnagl, P., Audebert, H. J., & Endres, M. (2012). Telestroke ambulances in prehospital stroke management: concept and pilot feasibility study. *Stroke; a Journal of Cerebral Circulation*, 43(8), 2086–2090.
- Lim, C. A. E., Kaufman, B. J., O'Connor, J., Jr, & Cunningham, S. J. (2013). Accuracy of weight estimates in pediatric patients by prehospital Emergency Medical Services personnel. *The American Journal of Emergency Medicine*, 31(7), 1108–1112.
- Linderman, M., & Fried, J. (2004). Defensive Design for the Web: How to improve error messages, help, forms, and other crisis points. New Riders Publishing.
- Linn, S., Knoller, N., Giligan, C. G., & Dreifus, U. (1997). The sky is a limit: errors in prehospital diagnosis by flight physicians. *The American Journal of Emergency Medicine*, 15(3), 316–320.
- Lippman, J. M., Smith, S. N. C., McMurry, T. L., Sutton, Z. G., Gunnell, B. S., Cote, J., Perina, D. G., Cattell-Gordon, D. C., Rheuban, K. S., Solenski, N. J., Worrall, B. B., & Southerland, A. M. (2016). Mobile Telestroke During Ambulance Transport Is Feasible in a Rural EMS Setting: The iTREAT Study. *Telemedicine Journal and E-Health: The Official Journal of the American Telemedicine Association*, 22(6), 507– 513.
- Liu, Z., Nersessian, N., & Stasko, J. (2008). Distributed cognition as a theoretical framework for information visualization. *IEEE Transactions on Visualization and Computer Graphics*, 14(6), 1173–1180.
- Lumley, H. A., Flynn, D., Shaw, L., McClelland, G., Ford, G. A., White, P. M., & Price, C. I. (2020). A scoping review of pre-hospital technology to assist ambulance personnel with patient diagnosis or stratification during the emergency assessment of suspected stroke. *BMC Emergency Medicine*, 20(1), 30.
- MacDonald, R. D., Banks, B. A., & Morrison, M. (2008). Epidemiology of adverse events in air medical transport. Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine, 15(10), 923–931.
- MacDonald, R. D., Swanson, J. M., & Mottley, J. L. (2001). Performance and error analysis of automated external defibrillator use in the out-of-hospital setting. *Annals* of Emergency Medicine. https://www.sciencedirect.com/science/article/pii/S019606440165649X
- Madathil, K. C., Greenstein, J. S., Juang, K. A., Neyens, D. M., & Gramopadhye, A. K. (2013). An Investigation of the Informational Needs of Ovarian Cancer Patients and Their Supporters. *Proceedings of the Human Factors and Ergonomics Society ... Annual Meeting Human Factors and Ergonomics Society. Meeting*, 57(1), 748–752.
- Madathil, K. C., Koikkara, R., Obeid, J., Greenstein, J. S., Sanderson, I. C., Fryar, K., ... & Gramopadhye, A. K. (2013). An investigation of the efficacy of electronic consenting interfaces of research permissions management system in a hospital setting. *International journal of medical informatics*, 82(9), 854-863.
- Madathil, K. C., Rivera-Rodriguez, A. J., Greenstein, J. S., & Gramopadhye, A. K. (2015). Healthcare information on YouTube: A systematic review. *Health Informatics Journal*, 21(3), 173–194.
- Mandellos, G. J., Lymperopoulos, D. K., Koukias, M. N., Tzes, A., Lazarou, N., & Vagianos, C. (2004a). A novel mobile telemedicine system for ambulance transport. Design and evaluation. *Conference Proceedings: ... Annual International Conference* of the IEEE Engineering in Medicine and Biology Society. IEEE Engineering in Medicine and Biology Society. Conference, 2004, 3080–3083.

- Mandellos, G. J., Lymperopoulos, D. K., Koukias, M. N., Tzes, A., Lazarou, N., & Vagianos, C. (2004b). A novel mobile telemedicine system for ambulance transport. Design and evaluation. *The 26th Annual International Conference of the IEEE Engineering in Medicine and Biology Society*, 2, 3080–3083.
- Mardon, R. E., Khanna, K., Sorra, J., Dyer, N., & Famolaro, T. (2010). Exploring relationships between hospital patient safety culture and adverse events. *Journal of Patient Safety*, 6(4), 226–232.
- Martinez, V. I., Marquard, J. L., Saver, B., Garber, L., & Preusse, P. (2017). Consumer Health Informatics Interventions Must Support User Workflows, Be Easy-To-Use, and Improve Cognition: Applying the SEIPS 2.0 Model to Evaluate Patients' and Clinicians' Experiences with the CONDUIT-HID Intervention. *International Journal* of Human–Computer Interaction, 33(4), 333–343.
- McCambridge, J., Witton, J., & Elbourne, D. R. (2014). Systematic review of the Hawthorne effect: new concepts are needed to study research participation effects. *Journal of Clinical Epidemiology*, *67*(3), 267–277.
- McDermott, F. T., Cooper, G. J., Hogan, P. L., Cordner, S. M., & Tremayne, A. B. (2005). Evaluation of the prehospital management of road traffic fatalities in Victoria, Australia. *Prehospital and Disaster Medicine*, 20(4), 219–227.
- McKnight, J., & Doherty, G. (2008). Distributed Cognition and Mobile Healthcare Work. *Proceedings of the 22Nd British HCI Group Annual Conference on People and Computers: Culture, Creativity, Interaction - Volume 2*, 35–38.
- McLaughlin, S. A., Doezema, D., & Sklar, D. P. (2002). Human simulation in emergency medicine training: a model curriculum. *Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine*, 9(11), 1310–1318.
- Meckler, G., Hansen, M., Lambert, W., O'Brien, K., Dickinson, C., Dickinson, K., Van Otterloo, J., & Guise, J.-M. (2018). Out-of-Hospital Pediatric Patient Safety Events: Results of the CSI Chart Review. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 22(3), 290–299.
- Mitchell, A. D., & Tallon, J. M. (2002). Air medical transport of suspected aortic emergencies. In *Air Medical Journal* (Vol. 21, Issue 3, pp. 34–37). https://doi.org/10.1016/s1067-991x(02)70037-0
- Mohammadfam, I., & Saeidi, C. (2015). Evaluating human errors in cataract surgery using the SHERPA technique. *Journal of Ergonomics*, 2(4), 41–47.

- Moloney, E., Rice, S., Levell, N., Craig, D., & Stoniute, J. (2017). VP208 Informing An Economic Model For Hyperhidrosis: A Clinical Survey. *International Journal of Technology Assessment in Health Care*, 33(S1), 246–247.
- Moon, H.-Y., Baek, C. W., Kim, J.-S., Koo, G. H., Kim, J.-Y., Woo, Y.-C., Jung, Y. H., Kang, H., Shin, H.-Y., & Yang, S.-Y. (2013). The causes of difficult tracheal intubation and preoperative assessments in different age groups. *Korean Journal of Anesthesiology*, 64(4), 308–314.
- Morrow, D., North, R., & Wickens, C. D. (2005). Reducing and Mitigating Human Error in Medicine. *Reviews of Human Factors and Ergonomics*. https://doi.org/10.1518/155723405783703019
- Mosley, I., Nicol, M., Donnan, G., Patrick, I., Kerr, F., & Dewey, H. (2007). The impact of ambulance practice on acute stroke care. *Stroke; a Journal of Cerebral Circulation*, 38(10), 2765–2770.
- Mould-Millman, N.-K., Meese, H., Alattas, I., Ido, M., Yi, I., Oyewumi, T., Colman, M., Frankel, M., & Yancey, A., 2nd. (2018). Accuracy of Prehospital Identification of Stroke in a Large Stroke Belt Municipality. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association* of State EMS Directors, 1–9.
- Nahm, M., Nguyen, V. D., Razzouk, E., Zhu, M., & Zhang, J. (2010). Distributed cognition artifacts on clinical research data collection forms. *Summit on Translational Bioinformatics*, 2010, 36–40.
- Narasimha, S., & Agnisarman, S. (2016). An investigation of the usability issues of home-based video telemedicine systems with geriatric patients. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*. https://journals.sagepub.com/doi/abs/10.1177/1541931213601412
- Narasimha, S., Agnisarman, S., Madathil, K. C., Gramopadhye, A. K., Welch, B., & Mcelligott, J. (2016). An Investigation of the Usability Issues of Home-based Video Telemedicine Systems with Geriatric Patients. *Proceedings of the Human Factors* and Ergonomics Society Annual Meeting, 60(1), 1804–1808.
- Narasimha, S., Chalil Madathil, K., Agnisarman, S., Rogers, H., Welch, B., Ashok, A., Nair, A., & McElligott, J. (2016). Designing Telemedicine Systems for Geriatric Patients: A Review of the Usability Studies. *Telemedicine Journal and E-Health: The Official Journal of the American Telemedicine Association*. https://doi.org/10.1089/tmj.2016.0178

- Narasimha, S., Dixon, E., Bertrand, J. W., & Chalil Madathil, K. (2019). An empirical study to investigate the efficacy of collaborative immersive virtual reality systems for designing information architecture of software systems. *Applied Ergonomics*, 80, 175–186.
- National Academy of Sciences. (2014). *The Role of Telehealth in an Evolving Health Care Environment: Workshop Summary*. National Academies Press (US).
- National Research Council, Division of Behavioral and Social Sciences and Education, Committee on Human Factors, & Committee on Human-System Design Support for Changing Technology. (2007). *Human-System Integration in the System* Development Process: A New Look. National Academies Press.
- Newlin, T., McCall, T., Ottmar, P., Welch, B., & Khairat, S. (2018). Assessing the Satisfaction of Citizens Using Teleconsent in Clinical Research. *Studies in Health Technology and Informatics*, 247, 685–689.
- Nguyen, Q.-T., Weinberg, J., & Hilborne, L. H. (2005). *Physician event reporting: training the next generation of physicians*. University of Southern California Los Angeles School of Medicine. http://www.dtic.mil/docs/citations/ADA434759
- Nielsen, J. (1994a). Heuristic evaluation. In *Usability inspection methods* (pp. 25–62). John Wiley & Sons, Inc.
- Nielsen, J. (1994b). Usability Engineering. Morgan Kaufmann.
- Nielsen, J. (1992). Finding usability problems through heuristic evaluation. *Proceedings* of the SIGCHI Conference on Human Factors in Computing Systems, 373–380.
- Nielsen, J., & Landauer, T. K. (1993). A mathematical model of the finding of usability problems. *Proceedings of the INTERACT '93 and CHI '93 Conference on Human Factors in Computing Systems*, 206–213.
- Nielsen, J., & Molich, R. (1990). Heuristic Evaluation of User Interfaces. 249–256.
- Nilsson, M., Yngling, A., Groth, K., Hammarqvist, F., & Jernling, C. (2013). Remote supported trauma care: Understanding the situation from afar. *Proceedings of the 26th IEEE International Symposium on Computer-Based Medical Systems*, 65–70.
- Nor, A. M., McAllister, C., Louw, S. J., Dyker, A. G., Davis, M., Jenkinson, D., & Ford, G. A. (2004). Agreement between ambulance paramedic- and physician-recorded neurological signs with Face Arm Speech Test (FAST) in acute stroke patients. *Stroke; a Journal of Cerebral Circulation*, 35(6), 1355–1359.

- Norman, D., & Dunaeff, T. (1993). *Things That Make Us Smart: Defending Human Attributes In The Age Of The Machine*. Basic Books.
- Norris, R. M. (1998). Fatality outside hospital from acute coronary events in three British health districts, 1994-5. *BMJ*, *316*(7137), 1065.
- O'Connor, R. E., Slovis, C. M., Hunt, R. C., Pirrallo, R. G., & Sayre, M. R. (2002). Eliminating Errors in Emergency Medical Services: Realities and Recommendations. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 6(1), 107–113.
- Okafor, N. G., Doshi, P. B., Miller, S. K., McCarthy, J. J., Hoot, N. R., Darger, B. F., Benitez, R. C., & Chathampally, Y. G. (2015). Voluntary Medical Incident Reporting Tool to Improve Physician Reporting of Medical Errors in an Emergency Department. *The Western Journal of Emergency Medicine*, 16(7), 1073–1078.
- Ortiz, G. A., & Sacco, R. L. (2014). National Institutes of Health Stroke Scale (NIHSS). In Wiley StatsRef: Statistics Reference Online. https://doi.org/10.1002/9781118445112.stat06823
- Pamela Jordan, R. H. T. (1996). Refining the Categories of Miscommunication. In Proceedings of AAAI Workshop on Detecting, Repairing, and Preventing Human-Machine Miscommunication, CA. http://citeseerx.ist.psu.edu/viewdoc/summary?doi=10.1.1.52.677
- Papai, G., Racz, I., Czuriga, D., Szabo, G., Edes, I. F., & Edes, I. (2014). Transtelephonic electrocardiography in the management of patients with acute coronary syndrome. *Journal of Electrocardiology*, 47(3), 294–299.
- Patrick, J., James, N., & Ahmed, A. (2006). Human processes of control: tracing the goals and strategies of control room teams. *Ergonomics*, 49(12-13), 1395–1414.
- Patterson, P. D., Weaver, M. D., Frank, R. C., Warner, C. W., Martin-Gill, C., Guyette, F. X., Fairbanks, R. J., Hubble, M. W., Songer, T. J., Callaway, C. W., Kelsey, S. F., & Hostler, D. (2012). Association between poor sleep, fatigue, and safety outcomes in emergency medical services providers. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 16(1), 86–97.
- Pavlopoulos, S., Kyriacou, E., Berler, A., Dembeyiotis, S., & Koutsouris, D. (1998). A novel emergency telemedicine system based on wireless communication technology-AMBULANCE. *IEEE Transactions on Information Technology in Biomedicine: A Publication of the IEEE Engineering in Medicine and Biology Society*, 2(4), 261– 267.

- Pérez de la Ossa, Natalia, Carrera, David, Gorchs, Montse, Querol, Marisol, Millán, Mònica, Gomis, Meritxell, Dorado, Laura, López-Cancio, Elena, Hernández-Pérez, María, Chicharro, Vicente, Escalada, Xavier, Jiménez, Xavier, & Dávalos, Antoni. (2014). Design and Validation of a Prehospital Stroke Scale to Predict Large Arterial Occlusion. *Stroke; a Journal of Cerebral Circulation*, 45(1), 87–91.
- Pettker, C. M., Thung, S. F., Norwitz, E. R., Buhimschi, C. S., Raab, C. A., Copel, J. A., Kuczynski, E., Lockwood, C. J., & Funai, E. F. (2009). Impact of a comprehensive patient safety strategy on obstetric adverse events. *American Journal of Obstetrics* and Gynecology, 200(5), 492.e1–e8.
- Peytchev, A., Couper, M. P., McCabe, S. E., & Crawford, S. D. (2006). Web Survey Design: Paging versus Scrolling. *Public Opinion Quarterly*, 70(4), 596–607.
- Pitt, K. (2002). Prehospital selection of patients for thrombolysis by paramedics. *Emergency Medicine Journal: EMJ*, 19(3), 260–263.
- Pittsley, K. A., & Memmott, S. (2012). Improving Independent Student Navigation of Complex Educational Web Sites: An Analysis of Two Navigation Design Changes in LibGuides. *Information Technology and Libraries*, 31(3).
- Ponathil, A., Ozkan, F., Welch, B., Bertrand, J., & Chalil Madathil, K. (2020). Family health history collected by virtual conversational agents: An empirical study to investigate the efficacy of this approach. *Journal of Genetic Counseling*. https://doi.org/10.1002/jgc4.1239
- Powers, W. J., Rabinstein, A. A., Ackerson, T., Adeoye, O. M., Bambakidis, N. C., Becker, K., Biller, J., Brown, M., Demaerschalk, B. M., Hoh, B., Jauch, E. C., Kidwell, C. S., Leslie-Mazwi, T. M., Ovbiagele, B., Scott, P. A., Sheth, K. N., Southerland, A. M., Summers, D. V., Tirschwell, D. L., & American Heart Association Stroke Council. (2018). 2018 Guidelines for the Early Management of Patients With Acute Ischemic Stroke: A Guideline for Healthcare Professionals From the American Heart Association/American Stroke Association. *Stroke; a Journal of Cerebral Circulation, 49*(3), e46–e110.
- Preece, M. H. W., Hill, A., Horswill, M. S., Karamatic, R., Hewett, D. G., & Watson, M. O. (2013). Applying heuristic evaluation to observation chart design to improve the detection of patient deterioration. *Applied Ergonomics*, 44(4), 544–556.
- Pronovost, P., Berenholtz, S., Dorman, T., Lipsett, P. A., Simmonds, T., & Haraden, C. (2003). Improving communication in the ICU using daily goals. *Journal of Critical Care*, 18(2), 71–75.

- Quain, D. A., Parsons, M. W., Loudfoot, A. R., Spratt, N. J., Evans, M. K., Russell, M. L., Royan, A. T., Moore, A. G., Miteff, F., Hullick, C. J., Attia, J., McElduff, P., & Levi, C. R. (2008). Improving access to acute stroke therapies: a controlled trial of organised pre-hospital and emergency care. *The Medical Journal of Australia*, 189(8), 429–433.
- Rafter, N., Hickey, A., Condell, S., Conroy, R., O'Connor, P., Vaughan, D., & Williams, D. (2015). Adverse events in healthcare: learning from mistakes. *QJM: Monthly Journal of the Association of Physicians*, 108(4), 273–277.
- Rajkomar, A., & Blandford, A. (2012). Understanding infusion administration in the ICU through Distributed Cognition. *Journal of Biomedical Informatics*, 45(3), 580–590.
- Rasmussen, J. (1983). Skills, rules, and knowledge; signals, signs, and symbols, and other distinctions in human performance models. *IEEE Transactions on Systems, Man,* and Cybernetics, SMC-13(3), 257–266.
- Reason, J. (1990). The contribution of latent human failures to the breakdown of complex systems. *Philosophical Transactions of the Royal Society of London. Series B, Biological Sciences*, 327(1241), 475–484.
- Reason, J. (1990). Human Error. Cambridge University Press.
- Rentsch, J. R., Heffner, T. S., & Duffy, L. T. (1994). What You Know is What You Get from Experience: Team Experience Related to Teamwork Schemas. *Group & Organization Management*, 19(4), 450–474.
- Rentsch, J. R., & Klimoski, R. J. (2001). Why do great minds think alike: antecedents of team member schema agreement. In *Journal of Organizational Behavior* (Vol. 22, Issue 2, pp. 107–120). https://doi.org/10.1002/job.81
- Rigby, K., Clark, R. B., & Runciman, W. B. (1999). Adverse events in health care: setting priorities based on economic evaluation. *Journal of Quality in Clinical Practice*, 19(1), 7–12.
- Rivard, P. E., Luther, S. L., Christiansen, C. L., Shibei Zhao, Loveland, S., Elixhauser, A., Romano, P. S., & Rosen, A. K. (2008). Using patient safety indicators to estimate the impact of potential adverse events on outcomes. *Medical Care Research and Review: MCRR*, 65(1), 67–87.
- Robson, K., Plangger, K., Kietzmann, J. H., McCarthy, I., & Pitt, L. (2015). Is it all a game? Understanding the principles of gamification. *Business Horizons*, 58(4), 411– 420.

- Rogers, H., Al Ghalayini, M., & K., C. M. (2018). A preliminary study to investigate the sensemaking process of UAV reports by operators after periods of disconnect for threat assessment. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*.
- Rogers, H., Chalil Madathil, K., Joseph, A., Holmstedt, C., McNeese, N. J., & McElligott, J. T. (2020). Evaluation and Prediction of Human Error in Ambulance-Based Telemedicine Stroke Assessment. *Proceedings of Human Factors and Ergonomics Society Annual Meeting*, 64.
- Rogers, H., Chalil Madathil, K., Joseph, A., Holmstedt, C., Qanungo, S., McNeese, N. J., Morris, T., Holden, R. J., & McElligott, J. T. (2020). *Demands Placed on Caregivers in a Telemedicine Integrated Ambulance-Based Setting for Stroke Care.*
- Rogers, H., Khasawneh, A., Bertrand, J., & Chalil Madathil, K. (2017). An investigation of the effect of latency on the operator's trust and performance for manual multi-robot teleoperated tasks. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*.
- Rogers, H., Khasawneh, A., Bertrand, J., & Chalil Madathil, K. (2019). Understanding Reliance and Trust in Decision Aids for UAV Target Identification. *Proceedings of* the Human Factors and Ergonomics Society Annual Meeting, 63(1), 1953–1954.
- Rogers, H., Madathil, K. C., Agnisarman, S., Narasimha, S., Ashok, A., Nair, A., Welch, B. M., & McElligott, J. T. (2017). A Systematic Review of the Implementation Challenges of Telemedicine Systems in Ambulances. *Telemedicine Journal and E-Health: The Official Journal of the American Telemedicine Association*, 23(9), 707–717.
- Rogers, Y., & Ellis, J. (1994). Distributed Cognition: An Alternative Framework for Analysing and Explaining Collaborative Working. *Journal of Information Technology Impact*, 9(2), 119–128.
- Rörtgen, D., Bergrath, S., Rossaint, R., Beckers, S. K., Fischermann, H., Na, I.-S., Peters, D., Fitzner, C., & Skorning, M. (2013). Comparison of physician staffed emergency teams with paramedic teams assisted by telemedicine--a randomized, controlled simulation study. *Resuscitation*, 84(1), 85–92.
- Rowin, E. J., Lucier, D., Pauker, S. G., Kumar, S., Chen, J., & Salem, D. N. (2008). Does error and adverse event reporting by physicians and nurses differ? *Joint Commission Journal on Quality and Patient Safety / Joint Commission Resources*, 34(9), 537– 545.

- Ruppert, M., Reith, M. W., Widmann, J. H., Lackner, C. K., Kerkmann, R., Schweiberer, L., & Peter, K. (1999). Checking for breathing: evaluation of the diagnostic capability of emergency medical services personnel, physicians, medical students, and medical laypersons. *Annals of Emergency Medicine*, 34(6), 720–729.
- Rybing, J., Nilsson, H., Jonson, C.-O., & Bang, M. (2016). Studying distributed cognition of simulation-based team training with DiCoT. *Ergonomics*, *59*(3), 423–434.
- Saler, M., Switzer, J. A., & Hess, D. C. (2011). Use of telemedicine and helicopter transport to improve stroke care in remote locations. *Current Treatment Options in Cardiovascular Medicine*, 13(3), 215–224.
- Salerno, S. M., Wrenn, K. D., & Slovis, C. M. (1991). Monitoring EMS protocol deviations: a useful quality assurance tool. *Annals of Emergency Medicine*, 20(12), 1319–1324.
- Salmon, P., Stanton, N. A., Young, M. S., Harris, D., Demagalski, J., Marshall, A., Waldman, T., & Dekker, S. (2002). Using existing HEI techniques to predict pilot error: A comparison of SHERPA, HAZOP and HEIST. *Proceedings of the HCI Aero* 2002 Conference. AAAI, 129–130.
- Sanderson, I. C., Obeid, J. S., Madathil, K. C., Gerken, K., Fryar, K., Rugg, D., Alstad, C. E., Alexander, R., Brady, K. T., Gramopadhye, A. K., & Moskowitz, J. (2013).
  Managing clinical research permissions electronically: A novel approach to enhancing recruitment and managing consents. *Clinical Trials*, 10(4), 604–611.
- Sasser, S. M., Hunt, R. C., Faul, M., Sugerman, D., Pearson, W. S., Dulski, T., Wald, M. M., Jurkovich, G. J., Newgard, C. D., Lerner, E. B., & Centers for Disease Control and Prevention (CDC). (2012). Guidelines for field triage of injured patients: recommendations of the National Expert Panel on Field Triage, 2011. MMWR. Recommendations and Reports: Morbidity and Mortality Weekly Report. Recommendations and Reports / Centers for Disease Control, 61, 1–20.
- Saver, J. L. (2006). Time is brain--quantified. *Stroke; a Journal of Cerebral Circulation*, 37(1), 263–266.
- Saver, J. L., Fonarow, G. C., Smith, E. E., Reeves, M. J., Grau-Sepulveda, M. V., Pan, W., Olson, D. M., Hernandez, A. F., Peterson, E. D., & Schwamm, L. H. (2013). Time to treatment with intravenous tissue plasminogen activator and outcome from acute ischemic stroke. *JAMA: The Journal of the American Medical Association*, 309(23), 2480–2488.

- Scharett, E., Lopes, S., Rogers, H., Bhargava, A., Ponathil, A., Chalil Madathil, K., & Dye, C. (2017). An Investigation of Information Sought by Caregivers of Alzheimer's Patients on Online Peer-Support Groups. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 61(1), 1773–1777.
- Scharett, E., Madathil, K. C., Lopes, S., Rogers, H., Agnisarman, S., Narasimha, S., Ashok, A., & Dye, C. (2017). An Investigation of the Information Sought by Caregivers of Alzheimer's Patients on Online Peer Support Groups. *Cyberpsychology, Behavior and Social Networking*, 20(10), 640–657.
- Schlichter, J., Koch, M., & Bürger, M. (1996). Workspace awareness for distributed teams. Annual Asian Computing Science Conference, 199–218.
- Schwamm, L. H., Holloway, R. G., Amarenco, P., Audebert, H. J., Bakas, T., Chumbler, N. R., Handschu, R., Jauch, E. C., Knight, W. A., Levine, S. R., Mayberg, M., Meyer, B. C., Meyers, P. M., Skalabrin, E., Wechsler, L. R., & on behalf of the American Heart Association Stroke Council and the Interdisciplinary Council on Peripheral Vascular Disease. (2009). A Review of the Evidence for the Use of Telemedicine Within Stroke Systems of Care: A Scientific Statement From the American Heart Association/American Stroke Association. *Stroke; a Journal of Cerebral Circulation*, 40(7), 2616–2634.
- Shappell, S. A., & Wiegmann, D. A. (2000). The Human Factors Analysis and Classification System--HFACS (No. DOT/FAA/AM-00/7; Publications). Embry-Riddle Aeronautical University. https://commons.erau.edu/publication/737/
- Shappell, S., & Wiegmann, D. (2000). The Human Factors Analysis and Classification System--HFACS (No. DOT/FAA/AM-00/7). Federal Aviation Administration, Civil Aeromedical Institute. https://commons.erau.edu/cgi/viewcontent.cgi?article=1777&context=publication
- Sibert, K., Ricci, M. A., Caputo, M., Callas, P. W., Rogers, F. B., Charash, W., Malone, P., Leffler, S. M., Clark, H., Salinas, J., Wall, J., & Kocmoud, C. (2008). The feasibility of using ultrasound and video laryngoscopy in a mobile telemedicine consult. *Telemedicine Journal and E-Health: The Official Journal of the American Telemedicine Association*, 14(3), 266–272.
- Small, S. D., Wuerz, R. C., Simon, R., Shapiro, N., Conn, A., & Setnik, G. (1999). Demonstration of high-fidelity simulation team training for emergency medicine. *Academic Emergency Medicine: Official Journal of the Society for Academic Emergency Medicine*, 6(4), 312–323.

- Smith, W. S., Isaacs, M., & Corry, M. D. (1998). Accuracy of paramedic identification of stroke and transient ischemic attack in the field. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 2(3), 170–175.
- Stanton, N. A. (2006). Hierarchical task analysis: developments, applications, and extensions. *Applied Ergonomics*, *37*(1), 55–79.
- Stanton, N. A., Young, M. S., Salmon, P., Marshall, A., Waldman, T., & Dekker, S. (2002). Predicting pilot error: assessing the performance of SHERPA. *Human Decision Making and Control*, 47–51.
- Steinfeld, A., Jenkins, O. C., & Scassellati, B. (2009). The oz of wizard: simulating the human for interaction research. *Proceedings of the 4th ACM/IEEE International Conference on Human Robot Interaction*, 101–108.
- Stella, J., Bartley, B., & Jennings, P. (2010). Introduction of a prehospital critical incident monitoring system--final results. *Prehospital and Disaster Medicine*, 25(6), 515– 520.
- Stevens, A. D., Hernandez, C., Jones, S., Moreira, M. E., Blumen, J. R., Hopkins, E., Sande, M., Bakes, K., & Haukoos, J. S. (2015). Color-coded prefilled medication syringes decrease time to delivery and dosing errors in simulated prehospital pediatric resuscitations: A randomized crossover trial. *Resuscitation*, 96, 85–91.
- Takeuchi, I., Fujita, H., Yanagisawa, T., Sato, N., Mizutani, T., Hattori, J., Asakuma, S., Yamaya, T., Inagaki, T., Kataoka, Y., Ohe, K., Ako, J., & Asari, Y. (2015). Impact of Doctor Car with Mobile Cloud ECG in reducing door-to- balloon time of Japanese ST-elevation myocardial infarction patients. *International Heart Journal*, 56(2), 170–173.
- Tang, Z., Johnson, T. R., Tindall, R. D., & Zhang, J. (2006). Applying heuristic evaluation to improve the usability of a telemedicine system. *Telemedicine Journal* and E-Health: The Official Journal of the American Telemedicine Association, 12(1), 24–34.
- Terkelsen, C. J., Nørgaard, B. L., Lassen, J. F., Gerdes, J. C., Ankersen, J. P., Rømer, F., Nielsen, T. T., & Andersen, H. R. (2002). Telemedicine used for remote prehospital diagnosing in patients suspected of acute myocardial infarction. *Journal of Internal Medicine*, 252(5), 412–420.
- Thomas, E. J., & Brennan, T. A. (2000). Incidence and types of preventable adverse events in elderly patients: population based review of medical records. *BMJ*, *320*(7237), 741–744.

- Thomas, E. J., & Petersen, L. A. (2003). Measuring errors and adverse events in health care. *Journal of General Internal Medicine*, 18(1), 61–67.
- Thyvalikakath, T. P., Monaco, V., Thambuganipalle, H., & Schleyer, T. (2009). Comparative study of heuristic evaluation and usability testing methods. *Studies in Health Technology and Informatics*, *143*, 322.
- Tweedy, J. T. (2005). Healthcare Hazard Control. In *Healthcare Hazard Control and Safety Management* (p. 60). CRC Press.
- Valdez, R. S., Holden, R. J., Caine, K., Madathil, K., Mickelson, R., Lovett Novak, L., & Werner, N. (2016). Patient Work as a Maturing Approach Within HF/E: Moving Beyond Traditional Self-Management Applications. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 60, 657–661.
- Valenzuela Espinoza, A., Van Hooff, R.-J., De Smedt, A., Moens, M., Yperzeele, L., Nieboer, K., Hubloue, I., de Keyser, J., Convents, A., Fernandez Tellez, H., Dupont, A., Putman, K., Brouns, R., & PreSSUB-Consortium. (2016). Development and Pilot Testing of 24/7 In-Ambulance Telemedicine for Acute Stroke: Prehospital Stroke Study at the Universitair Ziekenhuis Brussel-Project. *Cerebrovascular Diseases*, 42(1-2), 15–22.
- van den Berg, J., Olsson, L., Svensson, A., & Håkansson, S. (2015). Adverse events during air and ground neonatal transport: 13 years' experience from a neonatal transport team in Northern Sweden. *The Journal of Maternal-Fetal & Neonatal Medicine: The Official Journal of the European Association of Perinatal Medicine, the Federation of Asia and Oceania Perinatal Societies, the International Society of Perinatal Obstetricians, 28*(10), 1231–1237.
- Vilke, G. M., Tornabene, S. V., Stepanski, B., Shipp, H. E., Ray, L. U., Metz, M. A., Vroman, D., Anderson, M., Murrin, P. A., Davis, D. P., & Harley, J. (2007). Paramedic self-reported medication errors. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association* of State EMS Directors, 11(1), 80–84.
- von Elm, E., Altman, D. G., Egger, M., Pocock, S. J., Gøtzsche, P. C., Vandenbroucke, J. P., & STROBE Initiative. (2007). The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. *Epidemiology*, 18(6), 800–804.
- Walker, D. M., Hefner, J. L., Sieck, C. J., Huerta, T. R., & McAlearney, A. S. (2018). Framework for Evaluating and Implementing Inpatient Portals: a Multi-stakeholder Perspective. *Journal of Medical Systems*, 42(9), 158.

- Walker, G. H., Stanton, N. A., Baber, C., Wells, L., Gibson, H., Salmon, P., & Jenkins, D. (2010). From ethnography to the EAST method: a tractable approach for representing distributed cognition in Air Traffic Control. *Ergonomics*, 53(2), 184– 197.
- Walshe, K. (2000). Adverse events in health care: issues in measurement. *Quality in Health Care: QHC*, 9(1), 47–52.
- Walter, S., Kostopoulos, P., Haass, A., Keller, I., Lesmeister, M., Schlechtriemen, T., Roth, C., Papanagiotou, P., Grunwald, I., Schumacher, H., Helwig, S., Viera, J., Körner, H., Alexandrou, M., Yilmaz, U., Ziegler, K., Schmidt, K., Dabew, R., Kubulus, D., ... Fassbender, K. (2012). Diagnosis and treatment of patients with stroke in a mobile stroke unit versus in hospital: a randomised controlled trial. *Lancet Neurology*, 11(5), 397–404.
- Wang, H. E., Lave, J. R., Sirio, C. A., & Yealy, D. M. (2006). Paramedic intubation errors: isolated events or symptoms of larger problems? *Health Affairs*, 25(2), 501– 509.
- Wang, X., Liu, K., You, L.-M., Xiang, J.-G., Hu, H.-G., Zhang, L.-F., Zheng, J., & Zhu, X.-W. (2014). The relationship between patient safety culture and adverse events: a questionnaire survey. *International Journal of Nursing Studies*, 51(8), 1114–1122.
- Waterson, P. (2009). A systems ergonomics analysis of the Maidstone and Tunbridge Wells infection outbreaks. *Ergonomics*, 52(10), 1196–1205.
- Watson, & Sasse. (1996). Evaluating audio and video quality in low-cost multimedia conferencing systems. *Interacting with Computers*, 8(3), 255–275.
- Weaver, M. D., Wang, H. E., Fairbanks, R. J., & Patterson, D. (2012). The association between EMS workplace safety culture and safety outcomes. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 16(1), 43–52.
- Wechsler, L. R., Demaerschalk, B. M., Schwamm, L. H., Adeoye, O. M., Audebert, H. J., Fanale, C. V., Hess, D. C., Majersik, J. J., Nystrom, K. V., Reeves, M. J., Rosamond, W. D., Switzer, J. A., American Heart Association Stroke Council, Council on Epidemiology and Prevention, & Council on Quality of Care and Outcomes Research. (2017). Telemedicine Quality and Outcomes in Stroke: A Scientific Statement for Healthcare Professionals From the American Heart Association/American Stroke Association. *Stroke; a Journal of Cerebral Circulation, 48*(1), e3–e25.

- Wells, B. J., Chagin, K. M., Nowacki, A. S., & Kattan, M. W. (2013). Strategies for handling missing data in electronic health record derived data. *EGEMS (Washington, DC)*, 1(3), 1035.
- Wendt, M., Ebinger, M., Kunz, A., Rozanski, M., Waldschmidt, C., Weber, J. E., Winter, B., Koch, P. M., Freitag, E., Reich, J., & Others. (2015). Improved prehospital triage of patients with stroke in a specialized stroke ambulance: results of the pre-hospital acute neurological therapy and optimization of medical care in stroke study. *Stroke; a Journal of Cerebral Circulation*, STROKEAHA – 114.
- Westfall, J. M., Fernald, D. H., Staton, E. W., VanVorst, R., West, D., & Pace, W. D. (2004). Applied strategies for improving patient safety: a comprehensive process to improve care in rural and frontier communities. *The Journal of Rural Health: Official Journal of the American Rural Health Association and the National Rural Health Care Association*, 20(4), 355–362.
- Wickens, C. D. (2000). The trade-off of design for routine and unexpected performance: Implications of situation awareness. In M. R. Endsley & D. J. Garland (Eds.), *Situation Awareness Analysis and Measurement* (pp. 211–225). Erlbaum Mahwah, NJ.
- Wickens, C. D., Hollands, J. G., Banbury, S., & Parasuraman, R. (2015). Spatial Cognition, Navigation, and Manual Control. In *Engineering psychology and human performance*. Psychology Press.
- Williams, B., Boyle, M., & Lord, B. (2008). Paramedic identification of electrocardiograph J-point and ST-segments. *Prehospital and Disaster Medicine*, 23(6), 526–529.
- Williams, T. A., Finn, J., Celenza, A., Teng, T.-H., & Jacobs, I. G. (2013). Paramedic identification of acute pulmonary edema in a metropolitan ambulance service. *Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors*, 17(3), 339–347.
- Williams, T. A., Finn, J., Fatovich, D., Perkins, G. D., Summers, Q., & Jacobs, I. (2015). Paramedic Differentiation of Asthma and COPD in the Prehospital Setting Is Difficult. Prehospital Emergency Care: Official Journal of the National Association of EMS Physicians and the National Association of State EMS Directors, 19(4), 535– 543.
- Woloshynowych, M., Rogers, S., Taylor-Adams, S., & Vincent, C. (2005). The investigation and analysis of critical incidents and adverse events in healthcare. *Health Technology Assessment*, 9(19), 1–143, iii.

- Wooldridge, A. R., Carayon, P., Hundt, A. S., & Hoonakker, P. L. T. (2017). SEIPSbased process modeling in primary care. *Applied Ergonomics*, 60, 240–254.
- Xiao, Y., Gagliano, D., LaMonte, M., Hu, P., Gaasch, W., Gunawadane, R., & Mackenzie, C. (2000). Design and evaluation of a real-time mobile telemedicine system for ambulance transport [1] Work reported here was partially supported by National Library of Medicine. *Journal of High Speed Networks*, 9(1), 47–56.
- Yperzeele, L., Van Hooff, R.-J., De Smedt, A., Valenzuela Espinoza, A., Van Dyck, R., Van de Casseye, R., Convents, A., Hubloue, I., Lauwaert, D., De Keyser, J., & Brouns, R. (2014). Feasibility of AmbulanCe-Based Telemedicine (FACT) study: safety, feasibility and reliability of third generation in-ambulance telemedicine. *PloS One*, 9(10), e110043.
- Zhang, J., Johnson, T. R., Patel, V. L., Paige, D. L., & Kubose, T. (2003). Using usability heuristics to evaluate patient safety of medical devices. *Journal of Biomedical Informatics*, 36(1-2), 23–30.
- Zimmer, M., Wassmer, R., Latasch, L., Oberndörfer, D., Wilken, V., Ackermann, H., & Breitkreutz, R. (2010). Initiation of risk management: incidence of failures in simulated Emergency Medical Service scenarios. *Resuscitation*, 81(7), 882–886.

APPENDICES

### **APPENDIX A**

#### **Database Search Strategy for Chapter 3**

PubMed 1988 to present (initial search 23 January 2019)

1. ("emergency medical technicians" [MeSH Terms] OR "allied health personnel"[MeSH Terms] OR "ambulances"[MeSH Terms] OR "emergency medical services"[MeSH Terms] OR "emergency medical technicians"[MeSH Terms] OR "emergency treatment" [MeSH Terms]) AND ("medical errors" [MeSH Terms] OR "drug-related side effects and adverse reactions"[MeSH Terms] OR "medical errors" [MeSH Terms] OR "risk management" [MeSH Terms]) AND ("1988"[Date - Publication] : "3000"[Date - Publication]) 2. (("prehospital" [Title/Abstract] OR "pre hospital" [Title/Abstract] OR "prehospital" [Title/Abstract]) OR ("Emergency Medical Services" [Title/Abstract] OR "EMS"[Title/Abstract]) OR "paramedic"[Title/Abstract] OR ("Emergency Medical Technician"[Title/Abstract] OR "EMT"[Title/Abstract]) OR "emergency care"[Title/Abstract] OR "ambulance"[Title/Abstract]) AND (error\*[Title/Abstract] OR ("incident reporting"[Title/Abstract] OR "incident" [Title/Abstract]) OR "mistake" [Title/Abstract] OR "adverse event"[Title/Abstract] OR "safety event"[Title/Abstract] OR "mistake" [Title/Abstract] OR "failure" [Title/Abstract]) AND ("1988" [Date -Publication] : "3000"[Date - Publication])

Web of Science 1988 to present (initial search 18 January 2019) Databases= WOS, BCI, BIOSIS, CCC, DRCI, DIIDW, KJD, MEDLINE, RSCI, SCIELO, ZOOREC Timespan=1988-2018 Search language=English

((TS="prehospital" OR TS="pre hospital" OR TS="pre-hospital" ) OR (TS="Emergency Medical Services" OR TS="EMS") OR TS="paramedic" OR (TS="Emergency Medical Technician" OR TS="EMT") OR TS="emergency care" OR TS="ambulance") AND (TS=error\* OR (TS="incident reporting" OR

TS="incident") OR TS="mistake" OR TS="adverse event" OR TS="safety event" OR TS="mistake" OR TS="failure")

Academic One File 1988 to present (initial search 23 January 2019) ((Abstract (prehospital) Or Abstract (Emergency Medical) Or Abstract (paramedic) Or Abstract (emergency care) Or Abstract (ambulance)) And (Abstract (error) Or Abstract (mistake) Or Abstract (incident) Or Abstract (adverse event) Or Abstract (failure)))

ACM Digital Library 1988 to present (initial search 18 January 2019) "query": { recordAbstract:(prehospital, pre-hospital, pre hospital, EMS, emergency medical\*, ambulance, EMT, paramedic, emergency care) AND recordAbstract:(error, incident, mistake, adverse event, safety event, failure) } "filter": {"publicationYear":{ "gte":1988, "lte":2019 }}, {owners.owner=HOSTED}

AgeLine 1988 to present (initial search 23 January 2019)

((AB"prehospital" OR AB"pre hospital" OR AB"pre-hospital") OR (AB"Emergency Medical Services" OR AB"EMS") OR AB"paramedic" OR (AB"Emergency Medical Technician" OR AB"EMT") OR AB"emergency care" OR AB"ambulance") AND (ABerror\* OR (AB"incident reporting" OR AB"incident") OR AB"mistake" OR AB"adverse event" OR AB"safety event" OR AB"mistake" OR AB"failure")

Alt HealthWatch 1988 to present (initial search 23 January 2019)

((AB"prehospital" OR AB"pre hospital" OR AB"pre-hospital") OR
(AB"Emergency Medical Services" OR AB"EMS") OR AB"paramedic" OR
(AB"Emergency Medical Technician" OR AB"EMT") OR AB"emergency care"
OR AB"ambulance") AND (ABerror\* OR (AB"incident reporting" OR
AB"incident") OR AB"mistake" OR AB"adverse event" OR AB"safety event"
OR AB"mistake" OR AB"failure")

Applied Science and Technology 1988 to present (initial search 23 January 2019)
(( AB"prehospital" OR AB"pre hospital" OR AB"pre-hospital") OR
(AB"Emergency Medical Services" OR AB"EMS") OR AB"paramedic" OR
(AB"Emergency Medical Technician" OR AB"EMT") OR AB"emergency care"
OR AB"ambulance") AND (ABerror\* OR (AB"incident reporting" OR
AB"incident") OR AB"mistake" OR AB"adverse event" OR AB"safety event"
OR AB"mistake" OR AB"failure")

BioOne 1988 to present (initial search 23 January 2019) ((ABSTRACT:(prehospital) OR ABSTRACT:(Emergency Medical) OR ABSTRACT:(Emergency Care) OR ABSTRACT:(Ambulance) OR ABSTRACT:(paramedic)) AND (ABSTRACT:(error) OR ABSTRACT:(incident) OR ABSTRACT:(mistake) OR ABSTRACT:(failure) OR ABSTRACT:(adverse event))

Health Reference Center Academic 1988 to present (initial search 23 January 2019) ((Abstract (prehospital) Or Abstract (Emergency Medical) Or Abstract (paramedic) Or Abstract (emergency care) Or Abstract (ambulance)) And (Abstract (error) Or Abstract (mistake) Or Abstract (incident) Or Abstract (adverse event) Or Abstract (failure)))

PsychINFO 1988 to present (initial search 23 January 2019) (( AB"prehospital" OR AB"pre hospital" OR AB"pre-hospital") OR (AB"Emergency Medical Services" OR AB"EMS") OR AB"paramedic" OR (AB"Emergency Medical Technician" OR AB"EMT") OR AB"emergency care" OR AB"ambulance") AND (ABerror\* OR (AB"incident reporting" OR AB"incident") OR AB"mistake" OR AB"adverse event" OR AB"safety event" OR AB"mistake" OR AB"failure")

ScienceDirect 1988 to present (initial search 18 January 2019): search was limited to 8 Boolean connectors

title-abs-key ("prehospital" OR "Emergency Medical Services" OR "paramedic" OR "Emergency Medical Technician" OR "emergency care" OR "ambulance") AND (error OR "incident" OR "mistake")

## Appendix B

### **Results of Search**

Database	Search	Counts	Date
Web of Science	Торіс	7992	1/23/19
Science Direct	Title/Abstract/Keywords	483	1/23/19
Applied Science and Technology	Abstract	49	1/23/19
PsychINFO	Abstract	144	1/23/19
ACM Digital Library	Abstract	4594	1/23/19
PubMed	Mesh Terms	12443	1/23/19
PubMed	Title/Abstract	3051	1/23/19
Health Reference Center Academic	Abstract	1	1/23/19
BioOne	Abstract	0	1/23/19
Alt HealthWatch	Abstract	4	1/23/19
AgeLine	Abstract	11	1/23/19
Academic OneFile	Abstract	1	1/23/19
Total		28773	
Duplicates		6844	
Without Duplicates		21929	

# Appendix C

## Adjusted STROBE Checklist

Section	No.	Item
Abstract	1	Either: (1) Indicate the study's design with a commonly used term in the title or the abstract or (2) Provide in the abstract an informative and balanced summary of what was done and what was found
Introduction	2	Explain the scientific background and rationale for the investigation being reported
	3	State specific objectives, including any prespecified hypotheses
Methods	4	Present key elements of study design early in the paper
	5	Describe the setting, locations, and relevant dates, including periods of recruitment, exposure, follow-up, and data collection
	6	Give the eligibility criteria, and the sources and methods of selection of participants
	7	Clearly define all outcomes, exposures, predictors, potential confounders, and effect modifiers. Give diagnostic criteria, if applicable
	8	For each variable of interest, give sources of data and details of methods of assessment (measurement). Describe comparability of assessment methods if there is more than one group
	9	Describe any efforts to address potential sources of bias
	10	Explain how the study size was arrived at
Optional if	11	
quantitative variables		Explain how quantitative variables were handled in the analyses. If applicable, describe which groupings were chosen and why
Results	12	Report numbers of individuals at each stage of study—eg numbers potentially eligible, examined for eligibility, confirmed eligible, included in the study, completing follow- up, and analyzed and Give reasons for non-participation at each stage
	13	Give characteristics of study participants (eg demographic, clinical, social) and information on exposures and potential confounders, Indicate number of participants with missing data for each variable of interest
	14	Report numbers of outcome events or summary measures
	15	Give unadjusted estimates and, if applicable, confounder-adjusted estimates and their precision (eg, 95% confidence interval). Make clear which confounders were adjusted for and why they were included, Report category boundaries when continuous variables were categorized, If relevant, consider translating estimates of relative risk into absolute risk for a meaningful time period
Discussion	16	Summarize key results with reference to study objectives
	17	Discuss limitations of the study, taking into account sources of potential bias or imprecision. Discuss both direction and magnitude of any potential bias
	18	Give a cautious overall interpretation of results considering objectives, limitations, multiplicity of analyses, results from similar studies, and other relevant evidence
	19	Discuss the generalizability (external validity) of the study results

## Appendix D

## **Quality Assessment**

A	V											Oua	litv It	em							Total Oual.
Autions	i cai	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	
Hein, Owen, and Plummer Alenyo,	2008	1	0	1	1	1	1	1	1	0	1	1	1	1	1	0	1	1	1	0	0.789
Smith, McCaul and Van Hoving	2018	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
Haynes and Pritting Eckstein	1999	1	1	1	1	1	1	1	1	0	1		1	0	1	0	1	1	1	1	0.833
and Suyehara Goebel	2002	1	1	1	1	1	1	0	0	0	1		1	1	1	0	1	1	1	0	0.722
Daya, and Gunnels	2004	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0	1	1	1	0	0.895
Friend, and Delbridge	2008	1	1	1	1	1	1	1	1	0	1		1	1	1	1	1	1	1	1	0.944
Smith, Isaacs, and Corry	2009	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0	1	0	1	0	0.842
Coppler et al.	2016	1	1	1	1	1	1	1	1	0	1	1	1	1	1	0	1	1	1	0	0.842
Lim et al.	2013	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0	1	1	1	1	0.947
Christie, Costa- Scorse, Nicholls, Jones, and	2016	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0	0.947
Howie Kirves et al.	2007	1	1	1	1	1	1	0	1	0	0	1	1	1	1	0	1	1	0	0	0.684
van den Berg, Olsson, Svensson, and Hakansson	2015	1	1	1	1	1	1	1	1	1	1		0	1	1	1	1	0	1	0	0.833
Rudd, Martin, Harrison, and Price	2016	1	0	1	1	1	1	0	0	0	0		1	1	1	0	1	1	0	0	0.556
Nor et al.	2004	1	1	1	1	1	1	1	1	0	1		1	1	1	1	1	0	1	0	0.833
Cienki and DeLuca	2012	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0	0	0.895
Mitchell and Tallon	2002	1	1	1	1	1	1	1	0	0	1		1	1	1	0	1	0	0	0	0.667

Patterson et al.	2012	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
Reed, Weinstein, and Brown	2003	1	1	1	1	1	1	1	1	0	0	0	0	1	1	1	1	1	1	1	0.789
Liberman, Lavoie, Mulder, and	1999	1	1	1	1	1	1	1	1	1	0		1	1	1	1	1	1	1	0	0.889
Sampalis Ruppert et	1999	1	1	1	1	1	1	1	1	0	1	1	1	1	1	0	1	0	0	0	0.737
Eberle et al.	1996	1	1	1	1	1	1	1	1	0	1	1	1	1	1	0	1	0	1	0	0.789
stevens et al.	2015	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
Boyle	2009	1	1	1	1	1	0	0	0	0	1		0	1	1	0	1	1	1	0	0.611
Hohenstein , Hempel, Schultheis, Lotter, and Fleischman	2014	1	1	0	1	1	1	1	1	1	0		1	1	1	0	1	0	1	0	0.722
Hohenstein , Rupp, and Fleischman	2011	1	0	1	1	1	1	1	0	0	0		1	1	1	0	1	0	1	0	0.611
11																					
Holliman, Wuerz, and Meador	1994	1	1	1	1	1	0	0	0	1	0		1	1	1	0	0	1	0	0	0.556
Holliman, Wuerz, and Meador Ackerman and Waldron	1994 2006	1	1	1	1	1	0	0	0	1	0		1	1	1	0	0	1	0	0	0.556 0.778
Holliman, Wuerz, and Meador Ackerman and Waldron Benner et al. Hobgood,	1994 2006 2006	1 1 1	1 0 1	1 1 1	1 1 1	1 1 1	0 1 1	0 1 1	0 1 1	1 1 0	0 0 1		1 1 1	1 1 1	1 1 1	0 0 0	0 1 1	1 1 0	0 1 1	0 0 0	0.556 0.778 0.778
Holliman, Wuerz, and Meador Ackerman and Waldron Benner et al. Hobgood, Bowen, Brice, Overby, and	1994 2006 2006 2006	1 1 1	1 0 1	1 1 1	1 1 1	1 1 1	0 1 1 1	0 1 1 1	0 1 1 1	1 1 0	0 0 1 1		1 1 1	1 1 1	1 1 1	0 0 0	0 1 1 1	1 1 0	0 1 1 1	0 0 0	0.556 0.778 0.778 0.944
Holliman, Wuerz, and Meador Ackerman and Waldron Benner et al. Hobgood, Bowen, Brice, Overby, and Tamayo- Sarver Fairbanks	<ul> <li>1994</li> <li>2006</li> <li>2006</li> <li>2006</li> </ul>	1 1 1 1 1	1 0 1 1	1 1 1 1 1	1 1 1 1 1	1 1 1	0 1 1 1 1	0 1 1 1	0 1 1 1 1	1 1 0 0	0 0 1 1	1	1 1 1	1 1 1	1 1 1	0 0 1	0 1 1 1	1 1 0 1	0 1 1 1 1 1	0 0 1	0.556 0.778 0.778 0.944
Holliman, Wuerz, and Meador Ackerman and Waldron Benner et al. Hobgood, Bowen, Brice, Overby, and Tamayo- Sarver Fairbanks et al.	<ul> <li>1994</li> <li>2006</li> <li>2006</li> <li>2006</li> <li>2008</li> <li>2005</li> </ul>	1 1 1 1	1 0 1 1 1	1 1 1 1	1 1 1 1	1 1 1 1	0 1 1 1 1	0 1 1 1 1	0 1 1 1 1	1 1 0 0	0 0 1 1 1	1	1 1 1 1	1 1 1 1 1 1 1	1 1 1 1 1 1 1	0 0 1 0	0 1 1 1 1 1 1	1 1 0 1	0 1 1 1	0 0 1 1	0.556 0.778 0.778 0.944 0.947
H Holliman, Wuerz, and Meador Ackerman and Waldron Benner et al. Hobgood, Bowen, Brice, Overby, and Tamayo- Sarver Fairbanks et al. Kaji et al. Jones, Murnby,	1994 2006 2006 2006 2008 2008	1 1 1 1 1	1 0 1 1 1 1	1 1 1 1 1	1 1 1 1 1	1 1 1 1 1	0 1 1 1 1 1 1	0 1 1 1 1 1	0 1 1 1 1 1	1 0 0 1 0	0 1 1 1 1	1	1 1 1 1 1	1 1 1 1 1	1 1 1 1 1	0 0 1 0 1	0 1 1 1 1 1	1 0 1 1 1	0 1 1 1 1 1 1	0 0 1 1 0	0.556 0.778 0.778 0.944 0.947 0.895
Holliman, Wuerz, and Meador Ackerman and Waldron Benner et al. Hobgood, Bowen, Brice, Overby, and Tamayo- Sarver Fairbanks et al. Kaji et al. Jones, Murphy, Dickson, Somerville, and Brizendine	<ul> <li>1994</li> <li>2006</li> <li>2006</li> <li>2008</li> <li>2006</li> <li>2004</li> </ul>	1 1 1 1 1	1 0 1 1 1 1 1	1 1 1 1 1 1	1 1 1 1 1 1	1 1 1 1 1 1	0 1 1 1 1 1	0 1 1 1 1 1 1	0 1 1 1 1 1 1	1 0 0 1 0	0 1 1 1 1 1	1 1	1 1 1 1 1 1	1 1 1 1 1 1	1 1 1 1 1 1	0 0 1 1 1 1	0 1 1 1 1 1 1	1 0 1 1 1 1	0 1 1 1 1 1 1	0 0 1 1 0	0.556 0.778 0.778 0.944 0.947 0.895 0.947

Hobgood, Xie, Weiner, and Hooker	2004	1	0	1	1	1	1	0	1	0	0		1	1	1	0	1	1	1	0	0.667
Lammers, Willoughb y-Byrwa, and Fales	2014	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0	1	1	1	0	0.895
Ghiyasvan dian et al. McDermott	2018	1	1	1	1	1	1	1	1	0	1	1	1	1	1	1	1	1	1	0	0.895
, Cooper, Hogan, Cordner, and Tremavne	2005	1	0	0	1	1	1	1	1	0	0	1	1	1	1	0	1	0	1	0	0.632
Gallagher and Kupas	2012	1	1	1	1	1	0	1	1	1	0	0	1	1	1	0	1	1	1	0	0.737
Gropen et al.	2014	1	1	1	1	1	1	1	1	0	0		1	1	1	1	1	1	1	1	0.889
Kaserer et al.	2017	1	1	1	1	1	1	1	1	0	0		1	1	1	0	1	1	1	1	0.833
Glaeser, Hellmich, Szewczuga , Losek, and Smith	1993	1	1	1	1	1	0	1	0	0	0		1	1	1	0	1	0	0	0	0.556
Kothari, Barsan, Brott, Broderick, and	1995	1	1	1	1	1	1	1	1	0	0	1	1	1	1	0	1	1	1	1	0.842
Ashbrock Zimmer et al.	2010	1	1	1	1	1	1	1	1	1	0		1	1	1	1	1	1	1	1	0.944
Bartley, and Jennings	2010	1	1	1	1	1	1	1	1	0	0		1	1	1	1	1	1	1	0	0.833
Hubble, Paschal, and Sanders	2000	1	1	1	1	1	1	1	1	1	0	1	0	1	1	0	1	1	1	1	0.842
Hoyle Jr. et al.	2012	1	1	1	1	1	1	1	1	0	0		1	1	1	1	1	1	1	0	0.833
Willoughb y-Byrwa and Fales	2014	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0	1	1	1	1	0.947
Einan- Lifshitz et al.	2011	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0	1	1	1	0	0.895
Salerno, Wrenn, and Slovis	1991	1	0	1	1	1	1	1	0	1	0		1	1	1	0	1	0	0	0	0.611
Hollander, Delagi, Sciammare Ila,	1995	1	1	1	1	1	1	1	1	0	0	1	1	1	1	0	1	1	1	1	0.842

Viccellio, Ortiz, and

Meckler et al.	2018	1	1	1	1	1	1	1	1	1	0		1	1	1	0	1	1	1	0	0.833
Wang, Cook, Chang, Yealy, and Lave	2009	1	0	1	1	1	0	1	0	0	0		1	0	1	1	1	1	0	0	0.556
Williams et al.	2015	1	1	1	1	1	1	1	1	0	0	1	1	1	1	1	1	1	1	0	0.842
Williams, Finn, Celenza, Teng, and Jacobs	2013	1	1	1	1	1	1	1	1	0	0	1	1	1	1	0	1	1	1	0	0.789
Williams, Boyle, and Lord	2008	1	1	1	1	1	0	1	1	0	0		1	1	1	1	1	1	1	1	0.833
Wang, Lave, Sirio, and Yealy	2006	1	1	1	1	1	1	1	1	0	0		1	1	1	0	1	1	1	0	0.778
Vilke et al.	2006	1	1	0	1	1	1	1	1	0	0		1	1	1	0	1	1	1	1	0.778
Hansen et al. MacDonald	2016	1	1	1	1	1	1	1	1	1	0	1	1	1	1	1	1	1	1	1	0.947
, Swanson, Mottley, and	2001	1	1	1	0	1	1	1	1	0	0		1	1	1	1	1	1	1	0	0.778
Weinstein																					
Weinstein Potter et al.	2013	1	0	0	0	1	0	1	1	0	0		1	0	1	0	1	0	1	0	0.444
Weinstein Potter et al. Cantor et al.	2013 2012	1	<b>0</b> 1	0 0	0 0	1	<b>0</b> 1	1	1	0 0	0		1 1	0 0	<b>1</b> 1	0 0	<b>1</b> 1	0 1	1 1	0 0	0.444 0.611
Weinstein Potter et al. Cantor et al. Pitt	2013 2012 2002	1 1 1	0 1 0	0 0 1	0 0 1	1 1 1	0 1 1	1 1 1	1 1 1	0 0 0	0 0 1	1	1 1 1	0 0 1	1 1 1	0 0 1	1 1 1	0 1 0	1 1 0	0 0 0	0.444 0.611 0.737
Weinstein Potter et al. Cantor et al. Pitt Brandler et al. Feldman	2013 2012 2002 2015	1 1 1	0 1 0 1	0 0 1 1	0 0 1 1	1 1 1	0 1 1 1	1 1 1	1 1 1	0 0 0 1	0 0 1 1	1 1	1 1 1	0 0 1 1	1 1 1	0 0 1 1	1 1 1	0 1 0 1	1 1 0 1	0 0 0 1	0.444 0.611 0.737 1
Weinstein Potter et al. Cantor et al. Pitt Brandler et al. Feldman, Brinsfield, Bernard, White, and Maciejko	2013 2012 2002 2015 2005	1 1 1 1	0 1 0 1	0 0 1 1	0 0 1 1	1 1 1 1	0 1 1 1	1 1 1 1	1 1 1	0 0 1	0 0 1 1	1	1 1 1 1	0 0 1 1	1 1 1 1	0 0 1 1	1 1 1 1	0 1 0 1	1 1 1 1	0 0 1 1	0.444 0.611 0.737 1 0.944
Weinstein Potter et al. Cantor et al. Pitt Brandler et al. Feldman, Brinsfield, Bernard, White, and Maciejko Lammers, Byrwa, and Fales	2013 2012 2002 2015 2005 2012	1 1 1 1	0 1 0 1 1	0 0 1 1 1	0 0 1 1 1	1 1 1 1	0 1 1 1 1 1	1 1 1 1	1 1 1 1 1	0 0 1 1	0 1 1 0	1 1	1 1 1 1	0 0 1 1 1	1 1 1 1	0 0 1 1 1 2	1 1 1 1	0 1 1 1 1	1 0 1 1	0 0 1 1	0.444 0.611 0.737 1 0.944 0.947
Weinstein Potter et al. Cantor et al. Pitt Brandler et al. Feldman, Brinsfield, Bernard, White, and Maciejko Lammers, Byrwa, and Fales Duby et al.	2013 2012 2002 2015 2005 2012 2012	1 1 1 1 1	0 1 0 1 1 1 1	0 0 1 1 1 1 1	0 0 1 1 1 1 1	1 1 1 1 1 1 1	0 1 1 1 1 1 1 1	1 1 1 1 1 1 1	1 1 1 1 1 1 1	0 0 1 1 1 1	0 1 1 0 1 0	1 1	1 1 1 1 1 1	0 0 1 1 1 1 1	1 1 1 1 1 1 1	0 0 1 1 1 1 0 0	1 1 1 1 1 1	0 1 0 1 1 1 1	1 0 1 1 1	0 0 1 1 1	0.444 0.611 0.737 1 0.944 0.947 0.889
Weinstein Potter et al. Cantor et al. Pitt Brandler et al. Feldman, Brinsfield, Bernard, White, and Maciejko Lammers, Byrwa, and Fales Duby et al. Hansen, Eriksson, Skarica, Meckler, and Guise	2013 2012 2002 2015 2005 2012 2018 2018	1 1 1 1 1 1 1	0 1 1 1 1 1 1	0 1 1 1 1 1 1 0	0 1 1 1 1 1 1	1 1 1 1 1 1 1	0 1 1 1 1 1 1 1	1 1 1 1 1 1 1	1 1 1 1 1 1 1	0 0 1 1 1 1 1	0 1 1 0 1 0 0	1 1	1 1 1 1 1 1 1	0 1 1 1 1 1 1 1	1 1 1 1 1 1 1	0 1 1 1 0 0 1	1 1 1 1 1 1 1	0 1 0 1 1 1 1 1	1 0 1 1 1 1 1	0 0 1 1 1 1 0	0.444 0.611 0.737 1 0.944 0.947 0.889 0.833

Latimer et al.	2018	1	1	0	1	1	1	1	1	0	0		1	1	1	0	1	1	0	1	0.722
Weaver, Wang, Fairbanks, Patterson, and Patterson	2012	1	0	1	1	1	1	1	1	1	0	1	1	1	1	1	1	1	1	1	0.895
Linn, Knoller, Giligan, and Dreifus Duignan	1997	1	0	1	1	1	1	1	1	0	0		1	0	1	0	1	0	1	0	0.611
Lamb, DiFiori, Quinlavin, and Feeney	2018	1	1	0	1	1	1	1	1	0	0	1	1	1	1	0	1	1	1	0	0.737
Eastwood, Boyle, and Williams	2012	1	1	1	1	0	1	1	1	0	0		1	1	1	0	1	1	1	1	0.778
Jemmett, Kendal, Fourre, and Burton	2003	1	1	1	1	1	1	1	1	0	0	1	1	0	1	1	1	1	1	0	0.789

### Appendix E

#### **Role Specific HTA Diagrams**



Paramedic HTA (Sheet 2)





Nurse HTA

267



					Г				1	
		Fill out REACH assessment	Plan 4.2. Do 4.2.1 - 4.2.4, repeat until all assessments are completed	4.1.10. Speech Cards (see sheet 4)	9. Do hen 1 any	itant	4.1.6.4. Ask patient to turn their palms up to the ceiling	4.1.8.5. Ask patient to touch their left heel to right shin		4.2.4. Write unusual responses or Speech Responses under Notes
		4.2.]		tia 4.1.9. Sensation	an 4.1.8. Do Plan 4.1.5 1.8.1, then 4.1.9.1, th 1.8.2 - 4.1.9.2 - 4.1.9.2 - 1.8.5 in any 0.0 der	etic to close their etight	4.1.6.3. Ask patient to lift their left ann	4.1.8.4. Ask patient to touch their right heel to left shiin		4.2.3. Select assessment level of patient
				7. Leg rement 4.1.8. Atax	Plan 4.1.7. Do         Plan 4.1.7.           4.1.7.1, then         4.1.4.           4.1.7.2         4.1.7.2           4.1.7.3 in any         4.1.0           order         00	k patient teeth eyebrow	to lift their right am	1.8.3. Ask patient to ouch their left index finger to nose	4.1.9.7. Ask parametic to pir prick left leg	4.2.2. Select drop down menu
(Sheet 2)	5			4.1.6. Arm Movement Mov	Do         Plan 4.1.6. Do           4.1.6.1. then         4.1.6.2.           y         4.1.6.2.           y         4.1.6.4 in any	et screen to [4.1.5.2. As]	ve 4.1.6.1. Set screen 1 patient's full body	8.2. Ask patient to their right index to nose	k 4.1.9.6. Ask pin parametic to pi prick rightleg	4.2.1. Scroll to assessment item
Neurologist HTA 4. Assess the Patient	Plan 4. Do 4. 1 while 4			4.1.5. Facial Palsy	Plan 4.1.5. T 4.1.5.1, then 4.1.5.2 – 4.1.5.4 in an order	sk patient d release ds 4.1.5.1. S patien	4.1.3.4. Ask patient to mo heir eyes to the left, keepi their head straight	.8.1. Set screen to titlent's full body tou	sk 4.1.9.5. As paramedic to prick left ar	
		ct NIHSS	Plan 4.1. Do 4.1.1, then do 4.1.2 – 4.1.10 in any order	4.1.4. Visual Field (see sheet 3)	14.1.3. Do 3.1. then 3.2 – 3.4 in any 1	atiant to to grip an their eyes han	atient to move their ight, keeping their d straight	4.1 ft leg	sk 4.1.9.4. As pin paramedic to eek prick right a	
		4.1. Condu		f 4.1.3. Eye Movement	4.1.2. Do 1.1. 5 in any orde	ent 4.1.2.3. Ask p open and close	ient 4.1.3.3. Ask p /cs eyes to the ri heau	ient [4.1.7.3. Ask p to lift their le	4.1.9.3. As paramedic to prick left ch.	
				4.1.2. Level o Consciousness	Plan 4.1.2 4.1.2 order	t 4.1.2.2. Ask pati to state their ag	to to open their events	to y to lift their right	to 4.1.9.2. Ask paramedic to pi prick right chee	
				4.1.1. Navigate to NIHSS+ tab		4.1.2.1. Ask patien to state the curren month	4.1.3.1. Set screen patient's face	4.1.7.1. Set screen patient's full bod	4.1.9.1. Set screen patient's full body	

		4.1.4.7. Ask patient to state how many fingers they see
		4.1.4.6. Ask paramedic to hold up one or two fingers on the left side of the patient's peripheral vision
		4.1.4.5. Ask patient to state how many fingers they see
4.1.4. Visual Field	Plan 4.1.4. Do 4.1.4.1 – 4.1.4.7	4.1.4.4. Ask paramedic to hold up one or two fingers on the right side of the patient's peripheral vision
		4.1.4.3. Ask patient to look at paramedic's face
		4.1.4.2. Set screen to see paramedic and patient
		4.1.4.1. Ask paramedic to move next to the patient

Neurologist HTA (Sheet 3)

Neurologist HTA (Sheet 4)



## Appendix F

#### Team HTA Diagram





6.2.4. Write urmsual responses or Speech Responses under Notes -6.2.4, repeat until all assessments are Plan 6.2. Do 6.2.1 6.1.6.4. Ask patient to 6.1.8.5. Ask patient to touch their left heel to turn their palms up to 6.1.10. Speech Cards (see sheet 5) the ceiling right shin completed 6.2. Fill out REACH assessment Plan 6.1.9. Do 6.1.9.1, then 6.1.9.2 – 6.1.9.2 – order 6.1.5.4. Ask patient to close their eyes 6.1.9. Sensation tight 6.1.6.3. Ask patient to lift their left arm 6.1.8.4. Ask patient to touch their night heel to 6.2.3. Select assessment level of patient left shin Plan 6.1.8. Do 6.1.8.1, then 6.1.8.2 – 6.1.8.5 in any 6.1.5.3. Ask patient to raise their order 6.1.8. Ataxia paramedic to pin prick left leg eyebrows 6.1.6.2. Ask patient to lift their right ann 6.1.9.7. Ask 6.1.8.3. Ask patient to touch their left index 6.2.2. Select drop Plan 6.1.7. Do 6.1.7.1, then 6.1.7.2 – 6.1.7.3 in any finger to nose down menu order 6.1.5.2. Ask patient to show teefh 6.1.7. Leg Movement 6.1.6.1. Set screen to paramedic to pin prick rightleg patient's full body 6.1.9.6. Ask Plan 6.1.6. Do 6.1.6.1, then 6.1.6.2 -6.1.6.4 in any 6.2.1. Scroll to assessment item 6.1.8.2. Ask patient to touch their night index order finger to nose 6.1.6. Arm Movement 6.1.5.1. Set screen to patient's face Team HTA (Sheet 3) 6.1.3.4. Ask patient to move their eyes to the left, keeping their head straight paramedic to pin prick left arm Plan 6.1.5. Do 6.1.5.1, then 6.1.5.2 -6.1.5.4 in any Plan 6. Do 6.1 while 6.2 6.1.9.5. Ask 6. [Neurologist] Assess the Patient order 6.1.5. Facial Palsy 6.1.8.1. Set screen to patient's full body 6.1.2.4. Ask patient to grip and release 6.1.9.4. Ask paramedic to pin prick nght arm hands 6.1.4. Visual Field (see sheet Plan 6.1. Do 6.1.1, then do 6.1.2 – 6.1.10 in any order Ask patient to move their eyes to the right, keeping their head straight 9 6.13.1, then 6.13.2 – 6.13.4 in any Plan 6.1.3. Do open and close their eyes 6.1. Conduct NIHSS 6.1.2.3. Ask patient to 6.1.7.3. Ask patient to lift their left leg paramedic to pin prick left cheek 6.1.9.3. Ask order 6.1.3. Eye Movement Plan 6.1.2. Do 6121-6125 in any 6.1.3.2. Ask patient to open their eyes wide 6.1.7.2. Ask patient to lift their right leg 6.1.9.2. Ask paramedic to pin prick night cheek 6.1.2.2. Ask patient 6.1.2. Level of Consciousness to state their age order 6.1.2.1. Ask patient to state the current 6.1.3.1. Set screen to 6.1.7.1. Set screen to 6.1.9.1. Set screen to patient's full body patient's full body 6.1.1. Navigate to patient's face NIHSS+tab month
		6.1.4.7. Ask patient to state how many fingers they see
		6.1.4.6. Ask paramedic to hold up one or two fingers on the left side of the patient's peripheral vision
		6.1.4.5. Ask patient to state how many fingers they see
6.1.4. Visual Field	Plan 6.1.4. Do 6.1.4.7 6.1.4.7	6.1.4.4. Ask paramedic to hold up one or two fingers on the right side of the patient's peripheral vision
		6.1.4.3. Ask patient to look at paramedic's face
		6.1.4.2. Set screen to see paramedic and patient
		6.1.4.1. Ask paramedic to move next to the patient

Team HTA (Sheet 4)

Team HTA (Sheet 5)



# Appendix G

### **Team HTA Table**

Subgoal and Plan	Teamwork Description
0. Assess Possible Stroke Patient Plan: When Stroke Alert is called, do 1 and 2. Then when connected to patient, do 3 - 4 in any order. Then when neurologist is connected, do 5 then 6	Goal: determine if a patient is experiencing or has experienced a stroke to further inform care decisions and alert emergency department Teamwork: This requires input from EMS, collecting patient information and vitals, and neurologist, conducting assessment. Nurses are also involved to manage recording of information and for successful handoff to ED Plan: log on and then create the patient in the system, then collect necessary information from patient or EMS to conduct an assessment Criterion measure: determination of stroke and handoff to ED
1. [Everyone] Log on to REACH Plan: Do 1.1-1.3	Does not require teamwork
2. [Nurse] Create Patient Plan: Do 2.1-2.8	Does not require teamwork
3. [Nurse] Update Demographics Plan: While 3.1 do 3.2	Goal: update patient case to reflect actual patient demographics Teamwork: to gather demographics from paramedics Plan: ask paramedic verbally over REACH audio system for information Criterion measure: patient name and accurate date of birth identified and updated in REACH
3.1. [Paramedic] Gather Demographics Plan: If patient is alert and can respond, do 3.1.1-3.1.2, else ask caregiver	Does not require teamwork
3.2. [Nurse] Record Demographics Plan: Do 3.2.1-3.2.4	Goal: record accurate patient demographics in REACH Teamwork: to gather demographics from Paramedics Plan: ask paramedic verbally over REACH audio system for information Criterion measure: patient name and date of birth are changed in REACH
4. [Nurse] Update Vitals Plan: While 4.1, do 4.2	Goal: update patient case to reflect actual patient vitals Teamwork: to gather vitals from paramedics Plan: ask paramedic verbally over REACH audio system for information after seeing paramedic collect information visually Criterion measure: patient case vitals are complete and correct
4.1. [Paramedic] Gather Vitals Plan: Do 4.1.1 then 4.1.2, while doing 4.1.3 and 4.1.4	Does not require teamwork

<ul><li>4.2. [Nurse] Record Vitals</li><li>Plan: Do 4.2.1, then in any order do 4.2.2-</li><li>4.2.7</li></ul>	Goal: record accurate patient vitals in REACH Teamwork: to gather vitals from paramedics Plan: ask paramedic verbally over REACH audio system for information Criterion measure: patient vitals are changed in REACH
5. [Neurologist] Update Patient History Plan: While 5.1 ,do 5.2	Goal: update patient case to include a detailed patient history Teamwork: to gather and understand patient history from paramedics Plan: ask paramedic verbally over REACH audio system for information Criterion measure: patient history is complete and accurate
5.1. [Neurologist] Gather Patient History Plan: Do 5.1.1, then in any order do 5.1.2- 5.1.8	Goal: record accurate patient history in REACH Teamwork: to gather history from paramedics Plan: ask paramedic verbally over REACH audio system for information Criterion measure: patient history are updated in REACH
5.2. [Neurologist] Record Patient History Plan: Do 5.2.1-5.2.3	Does not require teamwork
6. [Neurologist] Assess Patient with NIHSS Plan: While 6.1, do 6.2	Goal: To determine if patient is experiencing or has experienced a stroke and the severity of that stroke Teamwork: to gather and understand information from the patient and patient responses Plan: ask patient directly for information and have paramedic help in assessment Criterion measure: NIHSS score is completed
6.1 [Neurologist] Conduct NIHSS Plan: Do 6.1.1, then do 6.1.2-6.1.10 in any order	Goal: evaluate severity of a potential stroke with a series of sub scales Teamwork: to gather assessments through paramedic interaction with the patient Plan: ask patient directly for information and have paramedic help in assessment Criterion measure: all assessment categories are evaluated
6.1.2 [Neurologist] Level of Consciousness Plan: Do 6.1.2.1-6.1.2.4 in any order	Does not require teamwork
6.1.3 [Neurologist] Eye Movement Plan: Do 6.1.3.1-6.1.3.4 in any order	Does not require teamwork
6.1.4 [Neurologist] Visual Field Plan: Do 6.1.4.1-6.1.4.7	Goal: evaluate the width of the patients visual field Teamwork: gather patient peripheral vision ability using paramedic interaction with the patient Plan: Explain to paramedic what motions are needed to help in assessment and evaluate patient response Criterion measure: determine if patient can see in peripheral vision, and if not, determine deficit

6.1.5 [Neurologist] Facial Palsy Plan: Do 6.1.5.1, then 6.1.5.2-6.1.5.4 in any order	Does not require teamwork
6.1.6 [Neurologist] Arm Movement Plan: Do 6.1.6.1, then 6.1.6.2-6.1.6.4 in any order	Does not require teamwork
6.1.7 [Neurologist] Leg Movement Plan: Do 6.1.7.1, then 6.1.7.2-6.1.7.3 in any order	Does not require teamwork
6.1.8 [Neurologist] Ataxia Plan: Do 6.1.8.1, then 6.1.8.2-6.1.8.5 in any order	Does not require teamwork
6.1.9 [Neurologist] Sensation Plan: Do 6.1.9.1, then 6.1.9.2-6.1.9.7 in any order	Goal: evaluate the patients sensation reactions Teamwork: gather patient limb sensation response using paramedic interaction with the patient Plan: Explain to paramedic what limbs to pin prick to help in assessment and evaluate patient response Criterion measure: determine if patient has normal stimuli reactions in all limbs
6.1.10 [Neurologist] Language and Processing Cards Plan: Do 6.1.10.1 then 6.1.10.2-6.1.10.3, if more assessment is needed do 6.1.10.4 then 6.1.10.5	Goal: evaluate the speech and mental processing of the patient Teamwork: gather patient mental and speech ability using paramedic interaction with the patient Plan: Explain to paramedic which cards to use to evaluate patient and confirm correct responses to assess patient responses Criterion measure: determine if patient has speech or cognitive deficits
6.1.10.2 [Neurologist] Object Description Plan: Do 6.1.10.2.1-6.1.10.2.3	Goal: evaluate the ability of the patient to recall object description Teamwork: use paramedic to show patient the object card Plan: Explain to paramedic which card to use and confirm correct responses to assess patient responses Criterion measure: determine if patient has cognitive deficits
6.1.10.3 [Neurologist] Scene Description Plan: Do 6.1.10.3.1-6.1.10.3.2	Goal: evaluate the ability of the patient to recall scene description Teamwork: use paramedic to show patient the scene card Plan: Explain to paramedic which card to use and confirm correct responses to assess patient responses Criterion measure: determine if patient has cognitive deficits
6.1.10.4 [Neurologist] Sentence Repeat Plan: Do 6.1.10.4.1-6.1.10.4.3	Goal: evaluate the ability of the patient to correctly read sentences out loud Teamwork: use paramedic to show patient the sentence card Plan: Explain to paramedic which card to use and confirm correct responses to assess patient responses Criterion measure: determine if patient has speech deficits

6.1.10.5 [Neurologist] Dysarthria Plan: Do 6.1.10.5.1-6.1.10.5.3	Goal: evaluate the ability of the patient to correctly read phrases out loud Teamwork: use paramedic to show patient the phrase card Plan: Explain to paramedic which card to use and confirm correct responses to assess patient responses Criterion measure: determine if patient has speech deficits
6.2 [Neurologist] Record Assessment Plan: Do 6.2.1-6.2.3, then do 6.2.4 as needed	Does not require teamwork

# Appendix H

#### Heuristic Evaluation Form

Nurse Role Tasks			
Start	Task No	Task	Knowledge Requirement
You get a call from a dispatcher	1.0	Log on to REACH system	User ID: nurse@email.com and Password: 1234
alert, Now do	2.0	Create patient	
tasks 1 and 2	2.1	Enter Medic unit as last name	
	2.2	Enter to days data as first name	
	23	Enter todays date as first name	Stroke Alert: "We have a possible
	2.5	Enter DOB greater than today's date	stroke in Medic 1, 60 year old Female
	2.4	Enter Gender	with Last Known Well at 8 this
	2.5	Enter EMS as Room/Bed	consult and notify MUSC"
	2.6	Select "Stroke – Urgent" in bottom selection box	
	2.7	Enter Reason for Consult as "Stroke Alert"	
Once EMS logs onto REACH do	3.0	Demographics	
task 3 and 4	3.1	Update Patient Name	"Patient's Name is Ann Job, date of
	3.2	Update Patient Date of Birth	Birth is August 12 1959"
	4.0	Get vitals	
	4.1	Record Last Known Well	Stroke alert
	4.3	Record Blood Pressure	
	4.4	Record Heart Rate	
	4.5	Record Respiratory Rate	"BP is 180 over 90, Heart rate is 110, Respiratory rate is 25 per minute, pulse
	4.6	Record Blood Oxygen Saturation	ox is 98 percent, Glucose is 104, and RACE is 6"
	4.7	Record Point of Care Glucose	KACE IS U
	4.8	Record RACE score under Chief Complaint	

	Task		
Start	No	Task	Knowledge Requirements
Once you enter the ambulance do 1 and 2	1	Log on to REACH	User ID: ems@email.com and Password: 1234
	2	Enter Patient Case	

Neurologist Tasks			
Start	Task No	Task	Knowledge Requirement
You get a call from the hospital for a consult, do steps 1 and 2	1	Log on to REACH system	User ID: neuro@email.com and Password: 1234
	2	Select Patient Case	
Once connected with paramedic do 3, then 4	3	Record patient history	"After the patient woke up this morning at 8 her husband noticed she was slurring her speech, she said she had a headache and couldn't see well" "The patient is retired, has a history of high blood pressure, and a family history of diabetes, Nonsmoker and drinker, and no drugs"
	4	Conduct NIHSS Exam	
	4.1	Level of Consciousness	
	4.2	Eye movement	
	4.3	Visual Field	patient can answer age but not the current month and needed to be asked multiple times
	4.4	Facial Palsy	patient has trouble opening eyes and gripping, gaze is direct, shows partial
	4.5	Arm Movement	minor paralysis on the left side, right arm and leg responds normally to movement and sensation the left arm and leg drift and are
	4.6	Leg Movement	only partially responsive to stimuli, limb ataxia present on the left
	4.7	Ataxia	patient slurs speech and shows indicators of mild aphasia, commands needed to be repeated and loudly to be understood
	4.8	Sensation	
	4.9	Dysarthria and Language	

# Appendix I

		Nurs	ce Tasks (1)	
Task No	Task	Violation Type	Description	Severity
		Visibility of System Status	No update on what (ID/password) was entered wrong	4
		Error Prevention	No indication of format for email or password	б
1.0	Log on to REACH system	Help users recognize, diagnose, and recover from errors	No error message	4
		Help users recognize, diagnose, and recover from errors	System does not show password to check the entered information	3
		Visibility of System Status	No feedback on what was being entered incorrectly	4
2.0	Create patient	Aesthetic and Minimalist Design	The heading and the data entry box are not aligned. Also, there is unnecessary space between them which confuses where to enter the data correctly	1
		Help users recognize, diagnose, and recover from errors	No error message on what is entered incorrectly	4
		Visibility of System Status	No indication of what kind of inputs	1
	Enter medic unit	Visibility of System Status	Medic unit is not clear	ю
2.1	as last name	Aesthetic and Minimalist Design	Inconsistent use of gray highlights	1
		Help users recognize, diagnose, and recover from errors	No error messages of incorrect name formats	2
		Visibility of System Status	No indication of what kind of inputs	1
2.2	Enter current date as first name	Help users recognize, diagnose, and recover from errors	No error messages of incorrect name formats	5

### **Heuristic Violations**

		Nurse 1	Pasks (2)	
Task No	Task	Violation Type	Description	Severity
r C	Fnter Gender	Consistency and Standards	Label for gender input is far separated from the input	2
ì		Aesthetic and Minimalist Design	Label is not in line	1
		Visibility of System Status	Distance of label makes it unclear what input is needed	1
		Visibility of System Status	Date requirements not known in system until after pressing submit	3
		Match Between System and Real World	The system does not have a calendar drop down as seen often with other systems	1
		Match Between System and Real World	Certain date formatting is not allowed	ю
		User Control and Freedom	Certain date formatting is not allowed	3
2.4 E	inter DOB greater than current date	Consistency and Standards	In most of the places, the date is entered in DD/MM/YYYY format. Either this should follow the same standards or accept multiple formats	ε
		Error Prevention	The space between the words "Date of birth" and the data entry column is too much and is not aligned. This creates the doubt on whether that's the correct space to enter the information	7
		Error Prevention	The date entry format cannot be seen. Ideally, it should have a previous entry of format that is required (like MM/DD/YYYY or DD/MM/YYYY) or have a calendar input	2.5
		Recognition Rather than Recall	It would be easier if there was a drop down list for dates	7

			Nurse Tasks (3)	
Task Nc	o Task	Violation Type	Description	Severity
		Visibility of System Status	Distance of label makes it unclear what input is needed	1
v (	Enter EMS as	Match Between System and Real World	EMS is not a room or bed; label should be more inclusive	1
0.7	Room/Bed	Error Prevention	No set up for input to tell user what formats are available	б
		Aesthetic and Minimalist Design	Distance from input	1
	0.01.000 <b>f</b> (0.0000)	Visibility of System Status	If the gray background changing to white is to highlight that option, a better color contrast could be used since most of the other parts of the website has a white background	2
2.6	Urgent" in bottom selection box	Consistency and Standards	There is no label for check boxes	Т
		Recognition Rather than Recall	There is no label for check boxes, which means that because the check boxes are attached to the reason for consult, they assume that is what it is	2
2.7	Enter Reason for Consult as "Stroke	Visibility of System Status	Reason for consult box does not change with the change of consult type	3
	Alert"	Error Prevention	No framework for input	1
		Visibility of System Status	Hard to gain access to the next part of the system- had multiple issues submitting	4
		Consistency and Standards	Bolding and type of the demographics bar are not consistent	1
3.0	Demographics	Consistency and Standards	Non-consistent use of blocks	1
		Consistency and Standards	The patient name is highlighted in the same way as other headings like start/stop camera. However, they are not the same. If the name needs to be highlighted, use different color combination	7
		Help and Documentation	Support number is small and hard to read	2

		Nurse	: Tasks (4)	
Task No	Task	Violation Type	Description	Severity
		Visibility of System Status	Edit button does not make it clear that you edit all demographics with that button	2
		Visibility of System Status	With so much pink on the page, the edit button may not be visible	ŝ
		Consistency and Standards	Format for filling out demographics is different than patient creation	7
		Consistency and Standards	The patient name has a pink background and white letters while the rest 4 has white background and black letter while the last column (room/bed) has white background and pink letter. All these inconsistencies are not required	0
3.1	Update Patient Name	Aesthetic and Minimalist Design	Unnecessary use of multiple (6) edit buttons on the top when everything leads to the same place	5
		Visibility of System Status	Edit button does not make it clear that you edit all demographics with that button	7
3.2	Update Patient Date of Birth	Consistency and Standards	Format for filling out demographics is different than patient creation	7

	Severity	3	7	7	ς	3	7	ŝ	7	1	7	2	7	2
e Tasks (5)	Description	Not clear that there is a box to input this until you scroll over it	Vitals heading is on the right; however, the Last Known Well information is not under the vital section (it is on the right)	Headings are not provided to the section on the left and the ones at the bottom (Medications and allergies) have headings, but are not consistent with the others	The data entry box could have information in it to show the format in which the data needs to be entered. For example, if it is time the information can be HH:MM AM/PM or something similar in a lighter $t_{max}$	There should be feedback given that you entered valid information	Vitals heading is on the right; however, the Last Known Well information is not under the vital section (it is on the right)	The data entry box could have information in it to show the format in which the data needs to be entered. For example, if it is time the information can be HH:MM AM/PM or something similar in a lighter	Multiple areas for data entry which may make things confusing	Abbreviations are used, not full names	Updated vital signs are denoted with an icon that must be clicked additionally	No formatting to detail which measurement are required like in demographics	No set format for measurement	Should be able to scroll over each and have info about data to be entered
Nurse	Violation Type	Visibility of System Status	Consistency and Standards	Consistency and Standards	Consistency and Standards	Error Prevention	Error Prevention	Error Prevention	Flexibility and Efficiency of Use	Match Between System and Real World	Visibility of System Status	Consistency and Standards	Recognition Rather than Recall	Help and Documentation
	Task							Get vitals						
	Task No							0.						

	Severity	7				s ge 4	s ge 4	
Tasks (6)	Description	No format for date and time	Having a "/" between mm and Hg would be an accelerator			The data entry box for blood oxygen is not clearly marked or is using a different term which is not in line with the real world usa, of the term	The data entry box for blood oxygen is not clearly marked or is using a different term which is not in line with the real world usa, of the term	
Nurse	Violation Type	Visibility of System Status	Flexibility and Efficiency of Use			Visibility of System Status	Match Between System and Real World	
	Task	Record Last Known Well	Record Blood Pressure	Record Heart Rate	Record Respiratory Rate		Record Blood Oxygen Saturation	Record Point of Care Glucose
	Task No	4.1	4.3	4.4	4.5		4.6	4.7

	Severity	4	ς	б	7	Э	m	4
Pasks (7)	Description	RACE score option was not visible. (This option was noticed later when I was going through the neuro page)	EMS saying, "RACE is 6" and the title of the data entry box "chief complaint" is not matching	No label for RACE score or indication to put in under complaint in interface	Could not use return button to enter information	No set location for RACE score	EMS saying, "RACE is 6" and the title of the data entry box "chief complaint" is not matching	RACE score option was not visible. (This option was noticed later when I was going through the neuro page)
Nurse 7	Violation Type	Visibility of System Status	Match Between System and Real World	Consistency and Standards	User Control and Freedom	Error Prevention	Recognition Rather than Recall	Flexibility and Efficiency of Use
	Task				Record RACE score under Chief Complaint			
	Task No				4.8			

Task				
No	Task	Violation Type	Description	Severity
		Visibility of System Status	No update on what (ID/password) was entered wrong	4
1		Error Prevention	No indication of format for email or password	б
1 Lo	gon to REACH system	Help users recognize, diagnose, and recover from errors	No error message	4
		Help users recognize, diagnose, and recover from errors	System does not show password to check the entered information	3
		Visibility of System Status	"Pending consults" and "my active consults" should be differentiated more (for example, by highlighting them)	n
		Visibility of System Status	Not clear you can select until scroll over	б
		Match Between System and Real World	Patient name should be in stroke alert, but could be written as something else	1
		User Control and Freedom	Assigned patients should be separated from unassigned patients	7
		Consistency and Standards	The tab is for active consults. However, the pending consults are in the same tab too. Ideally, when the neuro logs in, it should just show the pending tabs which would save time and avoid errors	n
2 Selé	sct Patient Case	Error Prevention	"Pending consults" and "my active consults" should be differentiated more (for example, by highlighting them)	ŝ
		Error Prevention	The tab is for active consults. However, the pending consults are in the same tab too. Ideally, when the neuro logs in, it should just show the pending tabs which would save time and avoid errors	ŝ
		Recognition Rather than Recall	There is no indication that this patient is assigned to me, other than its under pending consults, I would have to remember the name of the patient to identify	7
		Flexibility and Efficiency of Use	Medic information in stroke alert could be its own field rather than name to indicate the correct session	7
		Aesthetic and Minimalist Design	Only relevant information about the patient to identify should be shown, the reason for the consult may not be necessary if you have the consult type already	7

		Neurologist Tasks (2)	
Task No Task	Violation Type	Description	Severity
	Visibility of System Status	"Review of systems" did not look clickable. It took some time and luck to figure that out	3
	Match Between System and Real World	The heading "history of present illness" is kind of confusing since it is a history of present! Better words can be recommended	0
	Consistency and Standards	First task is not on the first page	7
	Consistency and Standards	The use of radio button is inconsistent. Alcohol use and substance use should also have radio buttons since they are single answer questions than multiple choice	2.5
	Consistency and Standards	"Review of systems" did not look clickable. It took some time and luck to figure that out	3
	Consistency and Standards	Not clear how to actually enter information in other than having to click outside the box	ŝ
	Consistency and Standards	More than one location to enter text for history	7
	Error Prevention	None reported selection does not prevent other items from being selected or deselects none reported	ŝ
3 record partern	Error Prevention	Should be an input button to confirm input	б
	Error Prevention	"Review of systems" did not look clickable. It took some time and luck to figure that out. Initially, I entered those information manually under the "history of present illness" option	3
	Recognition Rather than Recall	No format for completing patient history except check boxes	1
	Recognition Rather than Recall	"ROS notes" is not clear. I had to go to the top and figure out it meant review of systems. It had me recalling the information than recognizing it	1
	Flexibility and Efficiency of Use Flexibility and Efficiency of Use	Need to click on the historv tab to complete the first task Multiple scroll bars can get complicated and confusing to use	5 7
	Aesthetic and Minimalist Design	More than one text box with history labels	2
	Aesthetic and Minimalist Design	Unnecessary use of abbreviations (PMH, PSH, FH) which can take additional time or confuse. The information architecture of the page could be improved by removing such unnecessary information.	1

Task No     Task     Violation Type     Description       Task No     Visibility of System Status     "Repeat NIHSS" did not look clicicable       Yisibility of System Status     Visibility of System Status     "Repeat NIHSS" did not look clicicable       Yisibility of System Status     Visibility of System Status     Task No     "Repeat NIHSS" did not look clicicable       Visibility of System Status     Task No     Answers cannot be seen properly for all the options     Percention direction       Natch Berween System and Real     Add colors indicating different levels of severity     Month Berween System and Real     Add colors indicating different levels of severity       User Control and Freedom     No option to undo the radio button in the page (bedide and consulting physician)     Visibility in the options for NIHSS     Some items have multiple selections that could be a 3 on the scale       Visibility in the options for NIHSS Exam     User Control and Freedom     No option to undo the radio button in the page (bedide and consulting physician)       Add Colors indicating different have the same highest level     Consistency and Standards     Some items have nultiple selections that could be a 3 on the scale       Add Colors indicating different have the same highest level     Consistency and Standards     Not all items have or bot ciciclable       Add Colors indit Add C				Veurologist Tasks (3)	
Visibility of System Status   "Repeat NHISS" did nol look clickable     Visibility of System Status   World     Visibility of System Status   Amswers cannot be seen properly for all the options     Visibility of System Status   Marcolous indicating different levels of severity     Match Beuveen System and Real   Add colous indicating different levels of severity     World   User Control and Freedom   No option to undo the radio button in the page (bediade and consulting physician)     User Control and Freedom   No option to undo the radio button in the page (bediade and consulting physician)     User Control and Freedom   No option to undo the radio button in the page (bediade and consulting physician)     Consistemy and Standards   Some items have multiple selections that could be a 3 on the scale     Consistemy and Standards   Not all items have the sume highest level     Consistemy and Standards   Not all items have the sume highest level     Consistemy and Standards   Not all items have the sume highest level     Consistemy and Standards   Not all items have the sume highest level     Consistemy and Standards   Not all items have the sume highest level     Consistemy and Standards   Not all items have the sume highest level     Consistemy and Standards   Not all items have dedid nor look clickable <t< th=""><th>Task No</th><th>Task</th><th>Violation Type</th><th>Description</th><th>Severity</th></t<>	Task No	Task	Violation Type	Description	Severity
Visibility of System Status   Each drop down box should specifically have text indicating the menu drops down     Visibility of System Status   Answers cannot be seen properly for all the options     Match Between System and Real   Add colors indicating different levels of severity world     User Control and Freedom   No option to undo the radio button in the page (bedside and consulting physician)     User Control and Freedom   No option to undo the radio button in the page (bedside and consulting physician)     User Control and Freedom   No option to undo the radio button in the page (bedside and consulting physician)     User Control and Freedom   No option to undo the radio button in the page (bedside and consulting physician)     Conduct NIHSS Exam   User Standards   Some items have multiple selections that could be a 3 on the seale     Conduct NIHSS Exam   Consistency and Standards   Some items have the same highest level     Consistency and Standards   Net all items have the same highest level     Consistency and Standards   Net all items have the same highest level     Consistency and Standards   Should be a final confirmation button that forces you to review all of the inputs     Error Prevention   The confirmation button that forces you to review all of the inputs     Fireo Prevention   The confirmation button that forces you to review all of the inputs     <			Visibility of System Status	"Repeat NIHSS" did not look clickable	ξ
Visibility of System Status   Answers cannot be seen properly for all the options     Match Between System and Real   Add colors indicating different levels of severity     World   User Control and Freedom   Add colors indicating different levels of severity     User Control and Freedom   No option to undo the radio button in the page (bedside and consulting physician)     User Control and Freedom   No option to undo the radio button in the page (bedside and consulting physician)     User Control and Freedom   No option to undo the radio button in the page (bedside and consulting physician)     Ordicat NIHSS Exam   No entimes have multiple selections that could be a 3 on the scale     Conduct NIHSS Exam   Consistency and Standards   Some items have multiple selections that could be a 3 on the scale     Consistency and Standards   None items have the same highest level   Consistency and Standards     Consistency and Standards   Not all items have the same highest level     Consistency and Standards   Not all items have the same highest level     Consistency and Standards   Not all items have the same highest level     Consistency and Standards   Not all items have the same highest level     Consistency and Standards   Not all items have the same highest level     Consistency and Standards   Notal controit option items part area in system <td></td> <td></td> <td>Visibility of System Status</td> <td>Each drop down box should specifically have text indicating the menu drops down</td> <td>б</td>			Visibility of System Status	Each drop down box should specifically have text indicating the menu drops down	б
Match Between System and Real World Add colors indicating different levels of severity User Control and Freedom Add colors indicating different levels of severity   User Control and Freedom User Control and Freedom Very little flexibility in the options for NIHSS   User Control and Freedom No option to undo the radio button in the page (bedside and consulting physician)   User Control and Freedom No option to undo the radio button in the page (bedside and consulting physician)   User Control and Freedom No option to undo the radio button in the page (bedside and consulting physician)   Conduct NIHSS Exam Consistency and Standards Some items have multiple selections that could be a 3 on the scale   Conduct NIHSS Exam Consistency and Standards Not all items have the same highest level   Consistency and Standards Not all items have the same highest level   Consistency and Standards Not all items have the same highest level   Consistency and Standards Not all items have the same highest level   Consistency and Standards Not all items have the same highest level   Error Prevention Should be a final confirmation button that forces you to review all of the inputs   Error Prevention Should be a final confirmation button that forces you to review all of the inputs   Finery Frevention Should be a final confirmation button that forces you to review all of the inputs   Finery Frevention Should be a final co			Visibility of System Status	Answers cannot be seen properly for all the options	б
4 User Control and Freedom Very little flexibility in the options for NIHSS   4 User Control and Freedom No option to undo the radio button in the page (bedside and consulting physician)   6 Conduct NIHSS Exam Consistency and Standards Some items have multiple selections that could be a 3 on the scale   7 Consistency and Standards Some items have multiple selections that could be a 3 on the scale   7 Consistency and Standards Some items have the same highest level   7 Consistency and Standards Not all items have the same highest level   8 Conduct NIHSS <sup>T</sup> did not look clickable Unnecessary use of bold letter throughout   8 Should be a final confirmation button that forces you to review all of the inputs   9 Theoremion Should be a final confirmation button that forces you to review all of the inputs   9 Theoremion Should be a final confirmation button that forces you to review all of the inputs   9 Theoremion Should be a final confirmation button that forces you to review all of the inputs   9 Theoremion Should be a final confirmation button that forces you to review all of the inputs   9 Theoremion Should be a final confirmation button that forces you to review all of the inputs   9 Theoremion Should be a final confirmation button that force sout to review all of the inputs   9 <td></td> <td></td> <td>Match Between System and Real World</td> <td>Add colors indicating different levels of severity</td> <td>7</td>			Match Between System and Real World	Add colors indicating different levels of severity	7
4 User Control and Freedom No option to undo the radio button in the page (bedside and consulting physician)   4 Conduct NIHSS Exam Consistency and Standards Some items have multiple selections that could be a 3 on the scale   Consistency and Standards Consistency and Standards Not all items have the same highest level   Consistency and Standards Not all items have the same highest level   Consistency and Standards Not all items have the same highest level   Consistency and Standards Not all items have the same highest level   Consistency and Standards Unnecessary use of bold letter throughout   Error Prevention Should be a final confirmation button that forces you to review all of the inputs before moving to next area in system   Error Prevention "Repeat NIHSS" did not look clickable   Fistibility and Efficiency of Use With the options being limited and notes for each selection   Fistibility and Efficiency of Use With the options being limited and confising to use   Fistibility and Efficiency of Use Multiple scall bars can get complicated and confusing to use			User Control and Freedom	Very little flexibility in the options for NIHSS	7
4 Consistency and Standards Some items have multiple selections that could be a 3 on the scale   7 Consistency and Standards "Repeat NIHSS" did not look clickable   7 Consistency and Standards Not all items have the same highest level   7 Consistency and Standards Not all items have the same highest level   7 Consistency and Standards Not all items have the same highest level   7 Consistency and Standards Not all items have the same highest level   8 Not all items have the same highest level Immercessary use of bold letter throughout   8 Should be a final confirmation button that forces you to review all of the inputs   9 Teror Prevention Should be a final confirmation button that forces you to review all of the inputs   9 Teror Prevention Should be a final confirmation button that forces you to review all of the inputs   9 Teror Prevention Should be a final confirmation button that forces you to review all of the inputs   9 Teror Prevention Should be a final confirmation button that forces you to review all of the inputs   9 Teror Prevention Should be a final confirmation button that forces you to review all of the inputs   9 Teror Prevention Should be a final confirmation button that forces you to review all of the inputs   9 Teror Prevention Teror Prevention			User Control and Freedom	No option to undo the radio button in the page (bedside and consulting physician)	ŝ
Consistency and Standards"Repeat NIHSS" did not look clickableConsistency and StandardsNot all items have the same highest levelConsistency and StandardsUnnecessary use of bold letter throughoutError PreventionShould be a final confirmation button that forces you to review all of the inputs before moving to next area in system Error PreventionFlexibility and Efficiency of UseWith the options being limited and notes may be needed, there's only an overall not section rather than individual notes for each selectionFlexibility and Efficiency of UseMultiple scroll bars can get complicated and confusing to use	4 Conduct 1	NIHSS Exam	Consistency and Standards	Some items have multiple selections that could be a 3 on the scale	-
Consistency and StandardsNot all items have the same highest levelConsistency and StandardsUnnecessary use of bold letter throughoutError PreventionShould be a final confirmation button that forces you to review all of the inputs before moving to next area in systemError PreventionNotal be a final confirmation button that forces you to review all of the inputs before moving to next area in systemFlexibility and Efficiency of UseWith the options being limited and notes may be needed, there's only an overall not section rather than individual notes for each selectionFlexibility and Efficiency of UseMultiple scroll bars can get complicated and confusing to use			Consistency and Standards	"Repeat NIHSS" did not look clickable	ŝ
Consistency and StandardsUnnecessary use of bold letter throughoutError PreventionShould be a final confirmation button that forces you to review all of the inputs before moving to next area in systemError PreventionShould be a final confirmation button that forces you to review all of the inputs before moving to next area in systemFror PreventionRould be a final confirmation button that forces you to review all of the inputs before moving to next area in systemFror PreventionRould be a final confirmation button that forces you to review all of the inputs before moving to next area in systemFlexibility and Efficiency of UseWith the options being limited and notes may be needed, there's only an overall not section rather than individual notes for each selectionFlexibility and Efficiency of UseMultiple scroll bars can get complicated and confusing to use			Consistency and Standards	Not all items have the same highest level	7
Error PreventionShould be a final confirmation button that forces you to review all of the inputs before moving to next area in systemError Prevention"Repeat NIHSS" did not look clickableFlexibility and Efficiency of UseWith the options being limited and notes may be needed, there's only an overall not section rather than individual notes for each selectionFlexibility and Efficiency of UseMultiple scroll bars can get complicated and confusing to use			Consistency and Standards	Unnecessary use of bold letter throughout	-
Error Prevention"Repeat NIHSS" did not look clickableFlexibility and Efficiency of UseWith the options being limited and notes may be needed, there's only an overall not section rather than individual notes for each selectionFlexibility and Efficiency of UseMultiple scroll bars can get complicated and confusing to use			Error Prevention	Should be a final confirmation button that forces you to review all of the inputs before moving to next area in system	б
Flexibility and Efficiency of Use With the options being limited and notes may be needed, there's only an overall not section rather than individual notes for each selection Flexibility and Efficiency of Use Multiple scroll bars can get complicated and confusing to use			Error Prevention	"Repeat NIHSS" did not look clickable	ŝ
Flexibility and Efficiency of Use Multiple scroll bars can get complicated and confusing to use			Flexibility and Efficiency of Use	With the options being limited and notes may be needed, there's only an overall note section rather than individual notes for each selection	7
			Flexibility and Efficiency of Use	Multiple scroll bars can get complicated and confusing to use	7

		Non-	ist Tooles (A)	
		Incuroing	(+) CASES 1 101	
Task No	Task	Violation Type	Description	Severity
4.1	Level of Consciousness	Match Between System and Real World	Unclear as to why the items are labelled 1a-c, do the items add up to a score for 1 item?	5
4.2	Eye movement			
4.3	Visual Field			
4.4	Facial Palsy			
4.5	Arm Movement	Visibility of System Status	Unclear as to how the unstable selection contributes to overall score	2
4.6	Leg Movement			
4.7	Ataxia			
4.8	Sensation			
4.9	Dysarthria and Language			

# Appendix J

#### **SHERPA** Table

Task No.	Task	Error Mode	Error Description	Consequence	Recovery (	Criticality Prc	bability	Remediation
0	Assess Possible Stroke Patient							
-	Log on to REACH							
1.1	Enter UserID	A7, 8, 9	Incorrect or incomplete User ID	Cannot enter system	Immediately	Г	Μ	Highlight required field, saved personal info, role specific log in
1.2	Enter Password	A7, 8, 9	Incorrect or incomplete Password	Cannot enter system	Immediately	Г	М	Highlight required field, saved personal info, role specific log in
1.3	Click Log In button	A8	Button not clicked	Cannot enter system	Immediately	Г	Г	Enlarge button
5	Create Patient							
2.1	Enter Medic unit as last name	A6, 8	Medic unit not correctly recorded in last name or placed elsewhere	Cannot create patient	Immediately	M	Г	Highlight required field, suggestions for how to fill
2.2	Enter todays date as first name	A6, 8	Date not correctly recorded in first name or placed elsewhere	Cannot create patient	Immediately	Г	L	Highlight required field, suggestions for how to fill
2.3	Enter DOB greater than current date	A6, 8	DOB not correctly formatted or omitted	Cannot create patient	3.2.2	Ц	М	Create rigid formatting, highlight required field
2.4	Enter Gender	A6	Gender incorrectly selected	Will need to change later	3.2.2	Γ	Г	
2.5	Enter EMS as Room/Bed	A8	EMS not written as Room	Cannot create patient	3.2.2	W	Г	Highlight required field, suggestions for how to fill

Task No.	Task	Error Mode	Error Description	Consequence	Recovery	Criticality	Probability	Remediation
2.6	Select Stroke - Urgent in bottom selection box	A6, 8	Reason for consult not selected or stroke not selected	Cannot create patient when omitted, will not be allocated to the right people if incorrectly selected	3.2.2	W	Г	Highlight required field, suggestions for how to fill
2.7	Enter Reason for Consult as Stroke Alert	A8	No reason for consult written	Cannot create patient	3.2.2	Г	Г	Highlight required field, suggestions for how to fill
2.8	Click Submit Button	A8	Button not clicked	Cannot create patient	Immediately	Г	Γ	Enlarge button
с	Update Demographics							
3.1	Gather Demographics							
3.1.1	Ask Patient for Name	11,3	Patient did not hear paramedic or vice versa	Cannot properly update patient case	Immediately	Г	М	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
3.1.2	Ask Patient for Birthday	11,3	Patient did not hear paramedic or vice versa	Cannot properly update patient case	Immediately	Ц	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers

Remediation		nlarge button or make it constantly visible	lighlight required field, suggestions for how to fill	Create rigid formatting, highlight required field	Enlarge button				sual aids in environment displaying proper placement or alignment instructions for sleeve	button label to read Start BP, visual aid	Bright simplified screen
Probability		Ц	L L	Ц	Г				L	Γ	Г
Criticality		Г	Γ	Γ	Γ				Μ	Μ	Г
Recovery		Immediately	Immediately	5.1.2	Immediately				Immediately	Immediately	4.1.2
Consequence		Cannot properly update patient case	Cannot properly update patient case	Cannot properly update patient case	Cannot properly update patient case				Cannot assess blood pressure or heart rate	Cannot assess blood pressure or heart rate	Cannot assess blood pressure
Error Description		Button not clicked	Name not correctly recorded in correct sections or placed	DOB not correctly formatted or omitted	Button not clicked				Step omitted, cuff incorrectly placed or not secured properly J on arm	Step omitted, wrong button pressed	Cannot read output or output not generated by machine
Error Mode		A8	A6, 8	A6, 8	A8				A6,8,9	A6, 8	R1
Task	Record Demographics	Click Edit button as it appears in the header next to name	Change Name	Change DOB	Click Submit	Update Vitals	Gather Vitals	Get BP	Affix blood pressure cuff to patient upper arm	Press NIBP button	Read output
Task No.	3.2	3.2.1	3.2.2	3.2.3	3.2.4	4	4.1	4.1.1	4.1.1.1	4.1.1.2	4.1.1.3

ask Io.	Task	Error Mode	Error Description	Consequence	Recovery	Criticality	Probability	Remediation
0	Read HR output from blood pressure cuff	R1	Cannot read output or output not generated by machine	Cannot assess heart rate	Immediately	Г	Γ	Bright simplified screen
3	Count breaths taken for 30 seconds to get RR	A1,9	Miscount, count breaths for more or less than a minute	Cannot assess respiratory rate	Immediately	Μ	Μ	Provide clicker or counter with timer to measure for time
4	Get SpO2							
÷	Affix pulse ox reader to patient fingertip on opposite hand from BP cuff	A6,8,9	Step omitted, reader incorrectly placed or not secured properly on finger opposite from BP cuff	Cannot assess blood oxygenation	Immediately	Μ	Г	Visual aid for placement
t.2	Read SpO2(%) on reader	R1	Cannot read output or output not generated by machine	Cannot assess blood oxygenation	Immediately	Г	Γ	Bright simplified screen
0	Record Vitals							
-	Locate Nurses Lane	R3	Cannot find nurses lane	Cannot update vitals for all caregivers	Immediately	Ц	Г	Highlight nurses lane for nurse log on, simplify system such that only task relevant screens are available
7	Report BP next to BP	A7,8	Cannot find correct input, report in the wrong format	Cannot update vitals for all caregivers	5.15	Г	Г	Highlight required vitals, ordered fill out form, obvious input boxes
ŝ	Report HR next to HR	A7,8	Cannot find correct input, report in the wrong format	Cannot update vitals for all caregivers	Immediately	Г	Г	Highlight required vitals, ordered fill out form, obvious input boxes

	Task	Error Mode	Error Description Cannot find correct	Consequence	Recovery	Criticality P	robability	Remediation
-	rt RR next to RR	A7,8	input, report in the wrong format	Cannot update vitals for all caregivers	Immediately	Г	Г	Highlight required vitals, ordered fill out form, obvious input boxes
ă	ort SpO2 next to SpO2	A7,8	Cannot find correct input, report in the wrong format	Cannot update vitals for all caregivers	Immediately	Г	Г	Highlight required vitals, ordered fill out form, obvious input boxes
	tt POC Gluc next o POC Gluc	A7,8	Cannot find correct input, report in the wrong format	Cannot update vitals for all caregivers	Immediately	L	Г	Highlight required vitals, ordered fill out form, obvious input boxes
Ch	t RACE Score in ief Complaint	A7,8	Cannot find correct input, report in the wrong format	Cannot update information for all caregivers	Immediately	L	L	Highlight required vitals, ordered fill out form, obvious input boxes
dat	e Patient History							
the	rr Patient History							
/1g	ate to History tab	S1,2	Cannot locate or incorrectly select a different tab	Cannot form a written patient history for chart	Immediately	Ц	Г	Simplified screen for user relevant tasks, training
Ş	ad Patient Age	R1	Cannot locate	Cannot form a concept of patient history	Immediately	Г	Г	Automated highlight of important information, placement of important information on history tab
ad	Patient Gender	R1	Cannot locate	Cannot form a concept of patient history	Immediately	Г	L	Automated highlight of important information, placement of important information on history tab

Remediation	Automated highlight of important information, placement of important information on history tab	Automated highlight of important information, placement of important information on history tab	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers		Highlight or require response	Natural language check for necessary completion, training of necessary items in history
Probability	Г	Г	Μ	Μ	W		Ц	Г
Criticality	Μ	W	M	Μ	Μ		Ц	Μ
Recovery	Immediately	Immediately	Immediately	Immediately	Immediately		Immediately	Immediately
Consequence	Cannot form a concept of patient history	Cannot form a concept of patient history	Cannot form a concept of patient history	Cannot form a concept of patient history	Cannot form a concept of patient history		Patient chart will be incomplete and patient care could suffer	Patient chart will be incomplete and patient care could suffer
Error Description	Cannot locate	Cannot locate	Paramedic cannot hear request, cannot hear response	Paramedic cannot hear request, cannot hear response	Paramedic cannot hear request, cannot hear response		Does not select correct text box	Incomplete history
Error Mode	R1	R1	11	11	II		S2	A9
Task	Read Last Known Well	Read BP	Ask paramedic for situation update	Ask paramedic for medical history	Ask paramedic for Medications	Record Patient History	Select text box under History of Present Illness	Write out patient information
Task No.	5.1.4	5.1.5	5.1.6	5.1.7	5.1.8	5.2	5.2.1	5.2.2

Fask levant check	Error Mode	Error Description Incorrect selections	Consequence Patient chart will be incomplete and	Recovery	Criticality	Probability T	Remediation Require response for categories,
े म		selections made	patient care could suffer	6	-	1	processing
70							
	S1,2	Cannot locate or incorrectly select a different tab	Cannot record assessment for patient chart	Immediately	Г	Г	Simplified screen for user relevant tasks, training
les	S						
cent	Ξ	Patient cannot hear request, cannot hear response	Cannot form an assessment of how alert the patient is	Immediately	Μ	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
.enl	Ξ	Patient cannot hear request, cannot hear response	Cannot form an assessment of how alert the patient is	Immediately	М	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
anc	II p	Patient cannot hear request	Cannot form an assessment of how alert the patient is	6.1.5.4	Μ	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
anc	II I	Patient cannot hear request	Cannot form an assessment of how alert the patient is	Immediately	Μ	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers

Task No.	Task	Error Mode	Error Description	Consequence	Recovery	Criticality	Probability	Remediation
6.1.3	Eye movement							
6.1.3.1	Set screen to patient face	A3,4	Overshoot or under correct camera zoom or position	Cannot see patient movements with clarity to form assessment	Immediately	Γ	Г	Automatic camera adjustments
6.1.3.2	Ask patient to open eyes wide	Ш	Patient cannot hear request	Cannot form an assessment of patient eye movement	Immediately	M	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.3.3	Ask patient to move eyes to the right, keeping their head straight	Ш	Patient cannot hear request	Cannot form an assessment of patient eye movement	Immediately	W	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.3.4	Ask patient to move eyes to the left, keeping their head straight	11	Patient cannot hear request	Cannot form an assessment of patient eye movement	Immediately	W	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.4	Visual Field							
6.1.4.1	Ask paramedic to move next to the patient	П	Paramedic cannot hear request	Incorrect placement, faulty assessment	Immediately	Μ	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.4.2	Set screen to see patient and paramedic	A3,4	Overshoot or under correct camera zoom or position	Cannot see patient movements with clarity to form assessment	Immediately	Г	L	Automatic camera adjustments

Remediation	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers		Automatic camera adjustments	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
Probability	W	M	M	M	Μ		Г	Μ
Criticality	W	M	M	M	Μ		Г	Μ
Recovery	Immediately	Immediately	Immediately	Immediately	Immediately		Immediately	Immediately
Consequence	Cannot form an assessment of patient visual field		Cannot see patient movements with clarity to form assessment	Cannot form an assessment of severity of patient facial palsy				
Error Description	Patient cannot hear request	Paramedic cannot hear request	Patient cannot hear request	Paramedic cannot hear request	Patient cannot hear request		Overshoot or under correct camera zoom or position	Patient cannot hear request
Error Mode	II	II	II	II	II		A3,4	II
Task	Ask patient to look at paramedic face	Ask paramedic to hold up one or two fingers in the patient peripheral on the right side	Ask patient to say out loud how many finger they see	Ask paramedic to hold up one or two fingers in the patient peripheral on the left side	Ask patient to say out loud how many finger they see	Facial Palsy	Set screen to patient face	Ask patient to show teeth
Task No.	6.1.4.3	6.1.4.4	6.1.4.5	6.1.4.6	6.1.4.7	6.1.5	6.1.5.1	6.1.5.2

Remediation	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers		Automatic camera adjustments	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers		Automatic camera adjustments	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
Probability	M	Μ		Г	Μ	Μ	Μ		Γ	Μ
Criticality	М	Μ		Γ	Μ	W	М		Γ	Μ
Recovery	Immediately	Immediately		Immediately	Immediately	Immediately	Immediately		Immediately	Immediately
Consequence	Cannot form an assessment of severity of patient facial palsy	Cannot form an assessment of severity of patient facial palsy		Cannot see patient movements with clarity to form assessment	Cannot form an assessment of severity of stroke effect to arm movement	Cannot form an assessment of severity of stroke effect to arm movement	Cannot form an assessment of severity of stroke effect to arm movement		Cannot see patient movements with clarity to form assessment	Cannot form an assessment of severity of stroke effect to leg movement
Error Description	Patient cannot hear request	Patient cannot hear request		Overshoot or under correct camera zoom or position	Patient cannot hear request	Patient cannot hear request	Patient cannot hear request		Overshoot or under correct camera zoom or position	Patient cannot hear request
Error Mode	11	П		A3,4	11	11	11		A3,4	11
Task	Ask patient to raise eyebrows	Ask patient to close eyes tight	Arm Movement	Set screen to full body	Ask patient to move right arm up	Ask patient to move left arm up	Ask patient to turn palms up	Leg Movement	Set screen to full body	Ask patient to lift right leg
Task No.	6.1.5.3	6.1.5.4	6.1.6	6.1.6.1	6.1.6.2	6.1.6.3	6.1.6.4	6.1.7	6.1.7.1	6.1.7.2

Task No.	Task	Error Mode	Error Description	Consequence	Recovery	Criticality	Probability	Remediation
6.1.7.3	Ask patient to lift left leg	II	Patient cannot hear request	Cannot form an assessment of severity of stroke effect to leg movement	Immediately	Μ	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.8	Ataxia							
6.1.8.1	Set screen to full body	A3,4	Overshoot or under correct camera zoom or position	Cannot see patient movements with clarity to form assessment	Immediately	Г	Ц	Automatic camera adjustments
6.1.8.2	Ask patient to touch their finger to their nose with right hand	П	Patient cannot hear request	Cannot form an assessment of severity of ataxia	Immediately	W	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.8.3	Ask patient to touch their finger to their nose with left hand	П	Patient cannot hear request	Cannot form an assessment of severity of ataxia	Immediately	Μ	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.8.4	Ask patient to touch their right heel to left shin	Π	Patient cannot hear request	Cannot form an assessment of severity of ataxia	Immediately	Μ	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.8.5	Ask patient to touch their left heel to right shin	Π	Patient cannot hear request	Cannot form an assessment of severity of ataxia	Immediately	Μ	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.9	Sensation							
6.1.9.1	Set screen to full body	A3,4	Overshoot or under correct camera zoom or position	Cannot see patient movements with clarity to form assessment	Immediately	Г	Ц	Automatic camera adjustments
6.1.9.2	Ask paramedic to pin prick right cheek	II	Paramedic cannot hear request	Cannot form an assessment of patient limb sensation	Immediately	Μ	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers

Remediation	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers		Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	
Probability	W	M	M	М	W		M	
Criticality	Μ	М	Ж	М	W		L	
Recovery	Immediately	Immediately	Immediately	Immediately	Immediately		Immediately	
Consequence	Cannot form an assessment of patient limb sensation		Cannot use NIHSS cards to assess patient cognition and speech					
Error Description	Paramedic cannot hear request		Paramedic cannot hear request					
Error Mode	11	11	11	11	11		11	
Task	Ask paramedic to pin prick left cheek	Ask paramedic to pin prick right arm	Ask paramedic to pin prick left arm	Ask paramedic to pin prick right leg	Ask paramedic to pin prick left leg	NIH Stroke Scale Cards	Ask paramedic to use NIHSS cards	Object Description
Task No.	6.1.9.3	6.1.9.4	6.1.9.5	6.1.9.6	6.1.9.7	6.1.10	6.1.10.1	6.1.10.2

Remediation	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers, training to use cards to confirm with neurologist rather than verbal		Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers		Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
Probability	W	W	X		Μ	М		Μ
Criticality	Γ	Μ	W		Г	М		Г
Recovery	Immediately	Immediately	Immediately		Immediately	Immediately		Immediately
Consequence	Cannot use NIHSS cards to assess patient cognition and speech	Cannot assess patient object recognition	Cannot assess patient object recognition		Cannot use NIHSS cards to assess patient cognition and speech	Cannot assess patient cognition		Cannot use NIHSS cards to assess patient cognition and speech
Error Description	Paramedic cannot hear request	Patient cannot hear request	Paramedic cannot hear request, cannot hear response		Paramedic cannot hear request	Patient cannot hear request		Paramedic cannot hear request
Error Mode	II	Ш	П		Π	Ш		П
Task	Ask paramedic to show the patient the objects card	Ask patient to identify an object in the card	Confirm object with paramedic	Scene Description	Ask paramedic to show the patient the scene card	Ask patient to describe scene on the card	Sentences	Ask paramedic to show the patient the sentences card
Task No.	6.1.10.2.1	6.1.10.2.2	6.1.10.2.3	6.1.10.3	6.1.10.3.1	6.1.10.3.2	6.1.10.4	6.1.10.4.1

Task No.	Task	Error Mode	Error Description	Consequence	Recovery	Criticality	Probability	Remediation
6.1.10.4.2	Ask patient to read a sentence on the card	П	Patient cannot hear request	Cannot assess patient speech	Immediately	Μ	М	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.10.4.3	Confirm object with paramedic	Π	Paramedic cannot hear request, cannot hear response	Cannot assess patient speech	Immediately	M	M	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers, training to use cards to confirm with neurologist rather than verbal
6.1.10.5	Dysarthria							
6.1.10.5.1	Ask paramedic to show the patient the words card	11	Paramedic cannot hear request	Cannot use NIHSS cards to assess patient cognition and speech	Immediately	Г	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.10.5.2	Ask patient to read a word or phrase on the card	11	Patient cannot hear request	Cannot assess dysarthria	Immediately	Μ	Μ	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers
6.1.10.5.3	Confirm object with paramedic	11	Paramedic cannot hear request, cannot hear response	Cannot assess dysarthria	Immediately	W	M	Enhance audio quality of transmission, closed captioning, visual aid of normal requests made by caregivers, training to use cards to confirm with neurologist rather than verbal
6.2	Record Assessment							
6.2.1	Find assessment item	R2	Cannot locate correct assessment item	Cannot record assessment for patient chart	Immediately	Г	Г	Training to complete NIHSS in order

Remediation	Training to complete NIHSS in order	Visual aid of selection options or examples of behavior for each selection	Prompts for extra notes when making certain selections or training to use notes with assessment items
Probability	Г	Г	Г
Criticality	L	Μ	M
Recovery	Immediately	Immediately	Immediately
Consequence	Inaccurate assessment on patient chart	Inaccurate assessment on patient chart	Incomplete assessment to lower level of patient care
Error Description	Select the wrong assessment item	Incorrect assessment level selection	Insufficient notes on assessment
Error Mode	S2	S2	<b>A9</b>
Task	Select drop down menu	Select level of patient in assessment item	Write unusual responses or patient responses to NIHSS cards in Notes section
Task No.	6.2.2	6.2.3	6.2.4

#### Appendix K

#### Sample Interview Guide: Nurse and Neurologist

Thank you again for agreeing to participate in the study. The goal of the study is to understand the influence of factors such as provider characteristics, physical environment and layout, worker roles, staff workload, stress, job satisfaction and communication flows on the caregiving process. We would like to audio-record the interview to help us capture your responses. May we record the interview?

• If subject has agreed to audio-recording:

I have set up the tape recorder here in front of us. Please speak clearly during the interview so that the tape will record your voice accurately. I may ask you to repeat a response to make sure that it is recorded.

• If subject has not agreed to audio recording and a note taker is not available: I will take notes during our conversation today. I may ask you to slow down or pause for a moment so that I can record what you say accurately.

• If subject has not agreed to audio recording and a note taker is available: My colleague [NAME] will take notes during our conversation today. He/she may ask you to slow down or pause for a moment so that he/she can record what you say accurately.

During the interview, please use only your first name if you refer to yourself. This will help us keep your responses private. Your answers will not be individually reported to your care team members here at the clinic. If we do share information from the interview with clinic staff, we will only report it at the aggregate level, so that it is not obvious who said what. The interview will take about 60 minutes to complete. If you need to take a break during the interview to use the restroom or get a drink, please let me know and we will pause the interview.

If any of my questions aren't clear or you don't understand a word that I use, please let me know and I will rephrase the question for you. Most of these questions refer to the differences between your work before and after using the telemedicine system, here we are talking about the REACH program.

Please remember that you are not required to answer any specific question. You may also leave the interview at any time.

Do you have any questions before we start the interview?

### 1. Changes with regard to the (physical) environment
- What changes did telemedicine implementation create for your work environment? PROMPTS IF NEEDED:
  - Less paper to deal with?
  - Spending more time with computers? Less on the phone?
  - Changes to computer work stations?
- How does the space available to you support your ability to use the telemedicine system for providing stroke care?

**Interactions:** Did changes in the physical environment cause changes in the way your work is organized? **\*\*** specific to physical space

**Interactions:** Did changes in the physical environment cause changes in the way you interact with other people (colleagues and patients)?

## 2. Changes with regard to the person(s)

• Did you receive (extensive) training in the telemedicine?

### 3. Changes with regard to tasks you perform

- What activities do you do now (with telemedicine) that you did not do before? PROMPTS IF NEEDED:
  - Spending more on certain tasks than before telemedicine implementation?
    - More or less time examining the patient? Talking with the patient? Talking to the remote caregiving team?
    - Talking about different things with patients (e.g., histories vs. current problems)?

**Interactions:** What effects have these changes in tasks had your workflow (e.g. what steps you do in what order)?

### 4. Changes with regard to tools and technology

- How did telemedicine implementation in the ambulance for stroke care change the way you use *tools and technology, such as the telephone, and computers*? PROMPTS IF NEEDED:
  - Using certain tools less or more?
    - Spending more/less time looking for patient information?
    - Spending more/less time passing information back and forth with others, or waiting for someone else to finish with an activity?

### 5. Changes with regard to the organization

This section is asking about your group, your organization (MUSC telehealth, Georgetown EMS, Tidelands Hospital)

- How has telemedicine affected the existing *processes*? Do you do other things and you spend your time differently?
- After telemedicine implementation, do processes seem more or less efficient?

## 6. Use of telemedicine in daily caregiving

- What does your workday look like? How do you use telemedicine during your workday?
- How does telemedicine affect your workload (physical, mental, effort, frustration)?
- How does telemedicine affect communication?
  - with caregivers?
  - with patients?
- How aware are you of your surroundings in your care environment with the telemedicine system?
- Does the system make you more accessible to the team of caregivers and the patient?
- Do you feel informed of the past, current, and planned tasks during a patient care scenario?
- Think about your caregiving environment, what are you able to see and hear during a care scenario?
- Are you aware of any trigger factors in this system, any actions made by a caregiver that signals others to complete a task or set of tasks without a set plan (cookie assembly example)?

# 7. Use of teamwork in telemedicine

Your team being the telestroke doctor, the paramedic, and the nurse

- Do you feel that you are able to communicate and coordinate information effectively with team members with the telemedicine?
- Does a shared understanding of both the task and the methods to accomplish the task develop during the session with the telemedicine?
- Does a shared understanding of both the task and the methods to accomplish the task develop during the session with the telemedicine?
- Does everyone on the team fully understand their team role and responsibilities during the session with the telemedicine?
- Are you able to develop awareness of what your other team members are doing during the session with the telemedicine?
- In what ways does telemedicine limit the ability to conduct teamwork?
- Likewise, are there ways telemedicine helps to achieve teamwork?

# 8. Usefulness and usability of telemedicine

- What do you think about the *usefulness* of telemedicine and potential benefits of telemedicine integrated ambulance-based setting? Does ambulance-based telemedicine have benefits? PROMPTS IF NEEDED:
  - Useful for you individually
  - Useful for patients
  - Useful to redesign the system for improved efficiency or effectiveness
- What do you think of *usability* of telemedicine? PROMPTS IF NEEDED:
  - Easy for yourself
  - Easy for patients
- What part(s) of telemedicine integrated ambulance-based system do you like best?
- What part(s) of telemedicine integrated ambulance-based system could be improved?

### 9. Telemedicine implementation and quality and safety of patient care

- How do you think that telemedicine affects quality of care?
- How does telemedicine affect care coordination?
- How do you think that telemedicine affects patient safety or reducing medical errors?

#### **10.** Security and privacy

• How much of an issue is security, privacy, and confidentiality for you? For your patients?

### 11. Telemedicine implementation and patient satisfaction

• Do you think that patients appreciate the use of telemedicine?

### 12. Barriers

- What are the main barriers against using telemedicine to do your work? PROMPTS IF NEEDED:
  - Privacy, security
  - Privacy and security concerns of patients
  - Skills of you and your colleagues
  - Computer skills of patients
  - Workflow adjustments
  - Training
  - Increase in workload
  - Lack of computer support in your workplace
  - Legal risks
  - Loss of face-to-face contact with patients

- Start-up costs
- Maintenance costs
- Others

# 13. Facilitators

- Does telemedicine make your life easier?
- Do you think telemedicine makes life easier for patients?
- Others

#### 14. Final questions

- How does a telemedicine-integrated ambulance-based setting affect you (personally)
- Overall, how satisfied are you with telemedicine?

### Appendix L

#### Sample Interview Guide: Paramedic

Thank you again for agreeing to participate in the study. The goal of the study is to understand the influence of factors such as provider characteristics, physical environment and layout, worker roles, staff workload, stress, job satisfaction and communication flows on the caregiving process. We would like to audio-record the interview to help us capture your responses. May we record the interview?

• If subject has agreed to audio-recording:

I have set up the tape recorder here in front of us. Please speak clearly during the interview so that the tape will record your voice accurately. I may ask you to repeat a response to make sure that it is recorded.

• *If subject has not agreed to audio recording and a note taker is not available:* I will take notes during our conversation today. I may ask you to slow down or pause for a moment so that I can record what you say accurately.

• If subject has not agreed to audio recording and a note taker is available: My colleague [NAME] will take notes during our conversation today. He/she may ask you to slow down or pause for a moment so that he/she can record what you say accurately.

During the interview, please use only your first name if you refer to yourself. This will help us keep your responses private. Your answers will not be individually reported to your care team members here at the clinic. If we do share information from the interview with clinic staff, we will only report it at the aggregate level, so that it is not obvious who said what. The interview will take about 60 minutes to complete. If you need to take a break during the interview to use the restroom or get a drink, please let me know and we will pause the interview.

If any of my questions aren't clear or you don't understand a word that I use, please let me know and I will rephrase the question for you. Most of these questions refer to the differences between your work before and after using the telemedicine system, here we are talking about the REACH program.

Please remember that you are not required to answer any specific question. You may also leave the interview at any time.

Do you have any questions before we start the interview?

# 1. Changes with regard to the (physical) environment

- What changes did telemedicine implementation create for your work environment? PROMPTS IF NEEDED:
  - Less paper to deal with?
  - Spending more time with computers? Less on the phone?
  - Changes to computer work stations?
- How does the space available to you support your ability to use the telemedicine system for providing stroke care?

Follow up if needed:

- Which areas in the ambulance seem most constrained for the tasks you need to perform?
- What are some existing features in your work area that either enhance or impede your ability to perform tasks or interact with team members while using telemedicine to provide stroke care?
  - What is the best positioning of the telemedicine workstation with respect to the patient, ambulance door etc?
  - What is the optimum positioning of screens to support the use of telemedicine for stroke care?
  - Does the lighting in ambulance affect your ability to use telemedicine while performing tasks associated with stroke care?
- What are some common problems you have encountered related to obstructed visibility to people, equipment and tools while using the telemedicine system for providing stroke care?

**Interactions:** Did changes in the physical environment cause changes in the way your work is organized? **\*\*** specific to physical space

**Interactions:** Did changes in the physical environment cause changes in the way you interact with other people (colleagues and patients)?

### 2. Changes with regard to the person(s)

• Did you receive (extensive) training in the telemedicine?

### 3. Changes with regard to tasks you perform

- What activities do you do now (with telemedicine) that you did not do before? PROMPTS IF NEEDED:
  - $\circ$   $\,$  Spending more on certain tasks than before telemedicine implementation?
    - More or less time examining the patient? Talking with the patient? Talking to the remote caregiving team?
    - Talking about different things with patients (e.g., histories vs. current problems)?

**Interactions:** What effects have these changes in tasks had your workflow (e.g. what steps you do in what order)?

## 4. Changes with regard to tools and technology

- How did telemedicine implementation in the ambulance for stroke care change the way you use *tools and technology, such as the telephone, and computers*? PROMPTS IF NEEDED:
  - Using certain tools less or more?
    - Spending more/less time looking for patient information?
    - Spending more/less time passing information back and forth with others, or waiting for someone else to finish with an activity?
- Do you think that "automation" of certain processes in the existing telemedicine integrated system allows you to enhance the caregiving process? How do you adapt when the system is down and you cannot use telemedicine?

**Interactions:** Has the way you changed your use of tools and technology impacted the tasks that you do and the way the work is organized?

### 5. Changes with regard to the organization

This section is asking about your group, your organization (MUSC telehealth, Georgetown EMS, Tidelands Hospital)

- How has telemedicine affected the existing *processes*? Do you do other things and you spend your time differently?
- After telemedicine implementation, do processes seem more or less efficient?

### 6. Use of telemedicine in daily caregiving

- What does your workday look like? How do you use telemedicine during your workday?
- How does telemedicine affect your workload (physical, mental, effort, frustration)?
- How does telemedicine affect communication?
  - with caregivers?
  - with patients?
- How aware are you of your surroundings in your care environment with the telemedicine system?
- Does the system make you more accessible to the team of caregivers and the patient?
- Do you feel informed of the past, current, and planned tasks during a patient care scenario?
- Think about your caregiving environment, what are you able to see and hear during a care scenario?
- Are you aware of any trigger factors in this system, any actions made by a caregiver that signals others to complete a task or set of tasks without a set plan (cookie assembly example)?

# 7. Use of teamwork in telemedicine

Your team being the telestroke doctor, the paramedic, and the nurse

- Do you feel that you are able to communicate and coordinate information effectively with team members with the telemedicine?
- Does a shared understanding of both the task and the methods to accomplish the task develop during the session with the telemedicine?
- Does everyone on the team fully understand their team role and responsibilities during the session with the telemedicine?
- Are you able to develop awareness of what your other team members are doing during the session with the telemedicine?
- In what ways does telemedicine limit the ability to conduct teamwork?
- Likewise, are there ways telemedicine helps to achieve teamwork?

### 8. Usefulness and usability of telemedicine

- What do you think about the *usefulness* of telemedicine and potential benefits of telemedicine integrated ambulance-based setting? Does ambulance-based telemedicine have benefits? PROMPTS IF NEEDED:
  - Useful for you individually
  - Useful for patients
  - Useful to redesign the system for improved efficiency or effectiveness
  - What do you think of *usability* of telemedicine? PROMPTS IF NEEDED:
    - Easy for yourself
    - Easy for patients
- What part(s) of telemedicine integrated ambulance-based system do you like best?
- What part(s) of telemedicine integrated ambulance-based system could be improved?

### 9. Telemedicine implementation and quality and safety of patient care

- How do you think that telemedicine affects quality of care?
- How does telemedicine affect care coordination?
- How do you think that telemedicine affects patient safety or reducing medical errors?

### **10.** Security and privacy

• How much of an issue is security, privacy, and confidentiality for you? For your patients?

### 11. Telemedicine implementation and patient satisfaction

• Do you think that patients appreciate the use of telemedicine?

#### 12. Barriers

- What are the main barriers against using telemedicine to do your work? PROMPTS IF NEEDED:
  - Privacy, security
  - Privacy and security concerns of patients
  - Skills of you and your colleagues
  - Computer skills of patients
  - Workflow adjustments
  - Training
  - Increase in workload
  - Lack of computer support in your workplace
  - Legal risks
  - Loss of face-to-face contact with patients
  - Start-up costs
  - Maintenance costs
  - $\circ$  Others

### 13. Facilitators

- Does telemedicine make your life easier?
- Do you think telemedicine makes life easier for patients?
- Others

### 14. Final questions

- How does a telemedicine-integrated ambulance-based setting affect you (personally)?
- Overall, how satisfied are you with telemedicine?

#### Appendix M

#### **Study Protocol**

- 1. Send email to set up a time and date to complete study, send follow up email the day of with the study materials and instructions on how to download the simulator
- 2. At the study time, call the participant to explain the study and lead them through the session.
  - a. "Hello, thank you for taking the time to complete this study. Your participation will provide valuable data to improve telemedicine stroke care. Please open the first survey link to read through and agree to the consent form provided. If you do not wish to participate you can select 'No, I do not wish to be a part of this study' and we will end this session. If you do wish to continue, please select the 'Yes, I have read the above information and agree to be a participant in this study' and complete the following demographics form"
  - b. "You will be completing two stroke evaluations in two different telemedicine systems using the National Institutes of Health Stroke Scale. You will be connected to a simulated patient and paramedic in an ambulance, they will provide you with all the information you need to complete the assessment. A nurse will have connected before you to collect patient demographics and vitals, it is up to you to determine the patient history, medications, allergies and conduct a NIHSS evaluation. During the sessions we will pause the simulations to ask you questions about the simulation, we will also ask you to evaluate your workload, usability of the telemedicine system, and perceived teamwork after each session. Please do not worry, we are trying to evaluate the telemedicine system, not your performance."
  - c. "When you have completed one of the telemedicine sessions please open the second survey link to evaluate the telemedicine system and your workload during the session."
  - d. "When you have completed that survey you will need to open the simulator again and complete the same evaluation with the other telemedicine system, then complete that second survey again, this time evaluating the second system and your workload and perception of teamwork in that session only."
  - e. "When you have completed both sessions and surveys I will ask you to connect to a zoom call to complete a short interview about your experience in the system"
  - f. "At the conclusion of the interview you will receive your incentive for participation in this study"
  - g. "Again thank you so much for taking the time to complete this study it should take no longer than 90 minutes to complete"

- 3. Once they have confirmed they have completed the consent/demographic survey, have them open the simulator and select the telemedicine condition they will complete first
  - a. "Now that you have completed the consent and demographics you will need to open the simulation program, when that opens please select telemedicine system (A/B). Again, You will be connected to a simulated patient and paramedic in an ambulance, they will provide you with all the information you need to complete the assessment. A nurse will have connected before you to collect patient demographics and vitals, it is up to you to determine the patient history, medications, allergies and conduct a NIHSS evaluation. During the sessions we will also pause the simulations to ask you questions about the simulation. Are you ready to begin?"
- 4. They will complete the first simulation, the researcher will select the correct video response as they listen to the commands of the neurologists. Once they have finished the simulation, direct them to complete the second survey
  - a. "Now that you have completed the simulation please open the second survey, the first survey will ask you to read the definitions of 6 subscales of workload, mental demand, physical demand, temporal demand, performance, effort, and frustration. Please read through those definitions such that you understand each of them before continuing. The next section will ask you to select which of a pair of these subscales is most important in contributing to your workload, for example: when you are completing a task is it more stressful when that task is physically demanding or frustrating? Please complete a selection for all pairs, then you will be asked to rate your experience in the session for each of these subscales on a scale of 0 to 100, please note that performance is rated from low performance rated as 100 and high performance as 0 as low performance typically represents higher workload."
  - b. "the second survey will be an evaluation of how easy the telemedicine system was to use, and the final survey will be an evaluation of how well you felt you and the paramedic worked together as a team during the evaluation"
- 5. When they have completed this survey direct them to start the second session
  - a. "Now that you have completed the evaluation of the first telemedicine system you will need to conduct a session in the second system. You will need to open the simulation program again, when that opens please select telemedicine system (A/B). As before, please complete the history, medications, and allergy information and then the NIHSS to complete the evaluation."
- 6. They will complete the second simulation, the researcher will select the correct video response as they listen to the commands of the neurologists. Once they have finished the simulation, direct them to complete the second survey
  - a. "Please conduct the surveys just as before, only evaluating the second telemedicine system and session experience."

- 7. Once they have completed the survey guide them to the zoom call link to complete the survey
  - a. "Now that you have completed the telemedicine sessions we will complete a short interview through a zoom call. Please select the conference call link to connect to that session such that the interview can be audio recorded to create verbatim transcription of your responses. This should take no more than 30 minutes to complete and you will be given your inceptive for participation at the conclusion of the interview"
- 8. Once they have connected ask them for their permission to audio record the session
- 9. If they agree to recording, read through the interview question protocol
- 10. Once the interview is completed, send them their incentive and thank them for their time.

#### Appendix N

#### SAGAT Questions

Current System First Freeze (10s after clicking on NIHSS tab)

Level 1

- 1. Is your patient taking Eliquis?
  - a. Yes
  - b. No
- 2. What is your patient's current systolic blood pressure?
  - a. <95 mmHg
  - b. 95–120 mmHg
  - c. 120-170 mmHg
  - d. >170 mmHg
- 3. Does your patient have a history of atrial fibrillation?
  - a. Yes
  - b. No
- 4. What is your patient's glucose level?
  - a. <130
  - b. >130

Level 2

- 1. Is your patient on a blood thinner or blood clot prevention medication?
  - a. Yes
  - b. No
- 2. Does the patient have stroke indicating history?
  - a. Yes
  - b. No
- Level 3
  - 1. Does the patient have any history that would prevent them from being a tPA candidate?
    - a. Yes
    - b. No

Current System Second Freeze (end of simulation)

Level 1

- 1. What is the patient's NIHSS score?
  - a. 0
  - b. 1-4
  - c. 5-15
  - d. 16-20
  - e. 21-42
- 2. What is your patient's current heart rate?
  - a. <75 bpm
  - b. 75-95 bpm

- c. >95 bpm
- 3. State your assessment of the patient's speech
  - a. Clear, smooth
  - b. Slurring present
- 4. What is your patient's diastolic blood pressure?
  - a. < 60 mmHg
  - b. 60-80 mmHg
  - c. 80-100 mmHg
  - d. >100 mmHg

#### Level 2

- 1. State your diagnosis of the patient
  - a. Stroke
  - b. Other condition

Level 3

- 1. Will this patient be a candidate for tPA?
  - a. Yes
  - b. No
- 2. If the patient is having a stroke, do you think that this is a large vessel stroke or small vessel stroke?
  - a. Large vessel
  - b. Small vessel
  - c. Not a stroke

New System First Freeze (10s after clicking on NIHSS tab)

Level 1

- 1. Is your patient taking Metformin?
  - a. Yes
  - b. No
- 2. What is your patient's diastolic blood pressure?
  - a. < 60 mmHg
  - b. 60-80 mmHg
  - c. 80-100 mmHg
  - d. >100 mmHg
- 3. Does your patient have a history of atrial fibrillation?
  - a. Yes
  - b. No
- 4. What is your patient's glucose level?
  - a. <130
  - b. >130

#### Level 2

- 1. Is your patient on an ACE inhibitor?
  - a. Yes
  - b. No
- 2. Does the patient have stroke indicating history?

- a. Yes
- b. No

Level 3

1. Does the patient have any history that would prevent them from being a tPA candidate?

L

- a. Yes
- b. No

New System Second Freeze (end of simulation)

evel 1

- 1. What is the patient's NIHSS score?
  - a. 0
  - b. 1-4
  - c. 5-15
  - d. 16-20
  - e. 21-42
- 2. What is your patient's current heart rate?
  - a. <75 bpm
  - b. 75-95 bpm
  - c. >95 bpm
- 3. State your assessment of the patient's Limb Ataxia
  - a. No ataxia present
  - b. Ataxia present
- 4. What is your patient's current systolic blood pressure?
  - a. <95 mmHg
  - b. 95–120 mmHg
  - c. 120-170 mmHg
  - d. >170 mmHg

Level 2

- 1. State your diagnosis of the patient
  - a. Stroke
    - b. Other condition
- Level 3
  - 1. Will this patient be a candidate for tPA?
    - a. Yes
    - b. No
  - 2. If the patient is having a stroke, do you think that this is a large vessel stroke or small vessel stroke?
    - a. Large vessel
    - b. Small vessel
    - c. Not a stroke

### Appendix O

#### **Interview Protocol**

We would like to audio-record the interview to help us capture your responses. May we record the interview?

• If subject has agreed to audio-recording:

I have set up the tape recorder here in front of us. Please speak clearly during the interview so that the tape will record your voice accurately. I may ask you to repeat a response to make sure that it is recorded.

• If subject has not agreed to audio recording and a note taker is not available: I will take notes during our conversation today. I may ask you to slow down or pause for a moment so that I can record what you say accurately.

During the interview, please use only your first name if you refer to yourself. This will help us keep your responses private. If we do share information from the interview with anyone, we will only report it at the aggregate level, so that it is not obvious who said what. The interview will take about 30 minutes to complete. If you need to take a break during the interview to use the restroom or get a drink, please let me know and we will pause the interview.

If any of my questions aren't clear or you don't understand a word that I use, please let me know and I will rephrase the question for you. Please remember that you are not required to answer any specific question. You may also leave the interview at any time.

Do you have any questions before we start the interview?

- 1. Have you used a telemedicine system for your work before, can you describe that experience?
  - a. If yes, how did these systems compare?
- 2. Can you tell me what you thought of the first system you experienced?
  - a. Layout of the information
  - b. Detail of information provided
  - c. General ease of use
  - d. Awareness of the patient status
- 3. Can you tell me what you thought of the second system you experienced?
  - a. Layout of the information
  - b. Detail of information provided
  - c. General ease of use
  - d. Awareness of the patient status

- 4. How did the interface support information movement and transformation? Ex. how did you gather information for your diagnosis, how did the two systems differ in this process?
- 5. Was there a difference in the detail of information provided by the interface? How rich was the information communicated to you by the patient, paramedic, and the display?
- 6. Did you experience any trigger factors in either interface, e.g. an action in the system that triggered you to complete a task?
- 7. How did each system perform in creating scaffolding (a reminder of where you are in the task)?
- 8. Were there ways in either system that supported your awareness of how near or far you were from some goal (the tPA window, normal patient vitals, end of the NIHSS assessment)?
- 9. What do you think are the biggest differences between the two systems?
- 10. What do you think are the potential benefits of using either of these systems to complete a stroke evaluation?
- 11. Did you have any problems or issues in either systems?
- 12. Was one of the diagnoses you experienced more difficult than the other? How so?
- 13. What was the most difficult process?
- 14. Which system do you prefer?
- 15. What is your current computer set up (e.g. mouse vs trackpad, desktop, laptop)?